

This work is protected by copyright and other intellectual property rights and duplication or sale of all or part is not permitted, except that material may be duplicated by you for research, private study, criticism/review or educational purposes. Electronic or print copies are for your own personal, noncommercial use and shall not be passed to any other individual. No quotation may be published without proper acknowledgement. For any other use, or to quote extensively from the work, permission must be obtained from the copyright holder/s.

Understanding and managing the impacts of transience in student-led university food gardens



Rebecca Laycock Pedersen

A thesis submitted for the degree of Doctor of Philosophy in Environment and Sustainability

December 2019



Notes

On language

This thesis is written in British English, however there are quotes throughout that used American and Canadian spellings. For consistency, these spellings have been adapted to British English in this thesis.

In order to enhance readability, abbreviations are not used, except to describe organisations better known by their abbreviation than their full name (such as UNESCO). Unnecessary capitalisations have also been avoided.

On photographs

All photographs in this thesis have been taken by the author unless otherwise specified. Photos at the beginning of each chapter are not research material. Rather, they intended as a visual accompaniment to the written text.

On simultaneous publications

The following publications were written in parallel with this my doctoral work. Sections of these have been included in this thesis.

- Laycock Pedersen, R., Robinson, Z. P., & Surman, E. (2019). Understanding participation dynamics in university student-led food gardens with transient participants. *Sustainability*, 11(10), 2788. doi: 10.3390/su11102788
- Laycock Pedersen, R., & Robinson, Z. P. (2018, September 17-21). Can boundary objects be used for collaboration across timescales? Hope for university food-growing projects. In: *Third International Conference on Agriculture and Food in an Urbanizing Society – Healthy food, socio-biodiversity, and sustainable agrifood systems: innovations from consumption to production*. Paper presented at Third International Conference on Agriculture and Food in an Urbanizing Society, The Federal University of Rio Grande do Sul, Porto Alegre, Brazil. ISBN: 978-85-66094-45-9
- Laycock Pedersen, R., & Robinson, Z. P. (2018). Reviewing University Community Gardens for Sustainability: taking stock, comparisons with urban community gardens and mapping research opportunities. *Local Environment*, 23(6), 652-671. doi: 10.1080/13549839.2018.1463210
- Laycock, R. (2016, June 3-8). Challenges in student-led food growing initiatives and their solutions: The role of boundary objects in managing participant transience. In: *International Student Conference on Environment and Sustainability 2016 Conference Proceedings*. Paper presented at the International Student Conference on Environment and Sustainability 2016, Tongji University, Shanghai, China.

Laycock, R., Robinson, Z., & MacGregor, S. (2015, September 14-17). What can university community gardens offer to an 'alternative' food system?. In: Second International Conference on Agriculture in an Urbanizing Society - Reconnecting Agriculture and Food Chains to Societal Needs. Paper presented at the Second International Conference on Agriculture in an Urbanizing Society, Roma Tre University, Rome, Italy. ISBN: 978-88-908960-3-3.

Acknowledgements

Six years later, this is it! I am done! Though I've been greatly indebted from the day I'd begun.

This is the space where I've a few short words For a grateful thanks to those that deserve.

But to most, I am thankful in such a way That can't be expressed in words, but hey...

It's words that I have right here and right now. I'll continue my thanks in person, anyhow.

First thanks go to Zoe, who's been through it all Through laughter and tears and my stubborn gall.

Not all are so lucky to have a supervisor like you. You supported me no matter what I'd do.

Your curiosity in my work and your trust in me Helped to make me the best I could be.

And Sherilyn, it was you who helped me Understand my epistemology.

"A skin, not a sweater" was what I learned. That phrase, in my mind – it was more or less burned.

Emee and Emma, you joined me late in the game. And thanks to you both for these things I will name:

Emee, your input into my research design The evaluative part, I think you will find

Is where you have had your biggest impact For making practical all things abstract.

To Emma, thank you for your contributions – Thinking through ideas, finding solutions.

Thanks also for bringing me to the food network – Where I traded in my PhD for some practical work.

Thanks to the NUS sustainability team For letting me research the Student Eats scheme.

I want to also say thanks for believing in youth

Students are change-makers. That is the truth.

And students who took part in my research, Without you I'd've been left in a lurch!

You were insightful, brave and captivating; Thanks for being part of this research-creating.

All the people I've worked with are great, But especially those with a pragmatic trait:

Thank you to Keele's Action Research Network, The folks making change though their fieldwork.

Knowing there were others working like me – It was comforting during my inquiry.

It's Mary Brydon-Miller who deserves the credit For connecting me to those who led it.

Mary, you also grew my curiosity In research that is participatory.

To Pip, your friendship has carried me through. Thanks for the pick-ups when I was feeling blue.

Katie, you and I were the best officemates; through Will Smith our laughter permeates.

And Sophie, you're the one that that inspires me With your precision and ESD competency.

Friends at my new job, thanks, too, go to you: Jayne, Giles, Varvara, Rachael, Matilda, too.

Jesko, Yin, others, too many to name. Thanks for bearing with me while my thesis, I tamed.

And my thesis was improved by opportunities to present At conferences abroad, funded with money I spent.

To the KPA, thanks for your kind support For conference fees, hotels, and for transport.

Santander sent me to Rome to present my Literature review on: "Student Gardens – Why?"

The Institute of Environment for Sustainable Development, Thanks for contributing to my professional development. And Edith and Göran, thanks for understanding In your new recruit with a thesis outstanding.

You'll be happy to see that it is now complete. Finishing part-time – not a feat I'd repeat.

Mum and Dad, I can't thank you enough For supporting me through this academic stuff.

You might not get it, and that's okay. So, let's get cream tea on graduation day.

To my amazing Sara, my best friend in the world, You kept me afloat as my thesis unfurled.

Thanks for moving with me, I know it was tough. And for staying for five years – until you'd had enough!

Your love, your kindness, unconditional support, I am so glad you're the one that I did consort.

This cheeky rhyme of acknowledgement, Possibly to your astonishment

Is what I have left you with here As, into this book, you disappear.

So, thanks to the reader, for taking the time To make your way through these couplets of rhyme.

And thanks even more to those reading the rest, For your interest and time, I feel blessed.

N.B.

Although I wasn't able to express this fully in verse, I am grateful for the financial support from Keele's Postgraduate Association for conference attendance and fieldwork on many occasions over the course of my doctoral studies. I would also like to thank Santander and the UNEP-Tongji Institute of Environment for Sustainable Development for travel bursaries to present my research at international conferences.

Bekki

August 25, 2019

Abstract

University student-led food gardens are increasingly used to facilitate learning fostering prosustainability attitude and behaviour change. However, they are led by a transient student population, which impacts how they operate and the benefits they provide. This study undertakes the first explicit and empirical inquiry into how students' transience impacts student-led food gardens, and how these impacts might be addressed to maximise the gardens' benefits. I investigated this through an action research study, using a quantitative systematic literature review to assess the benefits of 'sustainable university community gardens' (a proxy for 'student-led food gardens') and take stock of what is already known about the impacts of students' transience on these gardens. I used a case study of the United Kingdom's National Union of Students' Student Eats student-led food growing scheme to understand how transience impacts the participation dynamics of the student-led food gardens, and the ways in which they are vulnerable to students' transience, drawing on interviews, participatory workshops, photovoice, a fishbowl discussion, and a research diary. This data was used to create a causal loop diagram and assess the vulnerability of the gardens using Biggs et al.'s (2012; 2015c) resilience principles. I found the student-led food gardens operated in a vulnerable state because of the impacts of students' transience. Paradoxically, transience both increased and decreased problematic participation, while exacerbating friction and power asymmetries between students and staff, underrepresentation of longterm stakeholders, lack of monitoring slow variables, and a lack of tolerance to ambiguity and uncertainty. In spite of this vulnerability, evidence suggested the gardens provided opportunities for building sustainability competencies. Actions were taken to address the negative impacts of students' transience and build resilience into the gardens. These fed into the recommendation that a portfolio of strategies is needed to address the immediate challenges of problematic participation, and build social-ecological memory in the gardens.

Keywords

action research, causal loop diagram, community gardens, higher education, participation, resilience principles, social-ecological memory, social-ecological resilience, sustainability education, temporary organisations, turnover

Table of Contents: Brief

Part 1 – The Background

Chapter 1 – Introduction

Chapter 2 – Reviewing sustainable food and sustainability in higher education

Chapter 3 – Framing through social-ecological resilience

Chapter 4 – Participatory action research in student-led food gardens

Part 2 – The Problem

Chapter 5 – Systematically reviewing sustainable university community gardens

Chapter 6 – Understanding the dynamics of problematic participation

Chapter 7 – Assessing the vulnerability of student-led food gardens

Part 3 – The Future

Chapter 8 – Managing the impacts of transience and building resilience

Chapter 9 – Methodological reflections and future directions

Chapter 10 – Conclusion

Table of Contents: Detailed

Notes	iii
On language	iii
On photographs	iii
On simultaneous publications	iii
Acknowledgements	V
Abstract	ix
Keywords	ix
Table of Contents: Key Definitions	xix
Education concepts	xix
Methodological concepts	xix
Participation, transience & volunteering concepts	xix
Resilience & systems theory concepts	xx
Sustainability concepts	xxi

CHAPTER 1: INTRODUCTION

1.1 Introduction	5
1.2 Overview	8
1.2.1 The Background	8
1.2.2 The Problem	10
1.2.3 The Future	11
1.3 A note on the 'voice' in this thesis	11

CHAPTER 2: REVIEWING SUSTAINABLE FOOD AND SUSTAINABILITY IN HIGHER EDUCATION

2.1 Introduction	. 17
2.2 An unsustainable food system and the scholarship to address it	. 17
2.3 Sustainability in higher education	. 20
2.3.1 Living Labs	28
2.3.2 Student Volunteering	30
2.4 Sustainable food and food gardens in universities	. 33
2.4.1 The transience of volunteers in student-led food gardens in universities	36
2.5 Temporary organisations and transience	. 39

CHAPTER 3: FRAMING THROUGH SOCIAL-ECOLOGICAL RESILIENCE

3.1 Introduction	45
3.2 Ontological and epistemological considerations	45
3.3 Rooting in systems thinking	
3.4 Social-ecological resilience	52
3.4.1 Principles for building resilience	
3.4.1.1 Resilience Principle 1: Maintain diversity and redundancy	60
3.4.1.2 Resilience Principle 2: Manage connectivity	62
3.4.1.3 Resilience Principle 3: Manage slow variables and feedbacks	
3.4.1.4 Resilience Principle 4: Foster complex adaptive systems thinking	65
3.4.1.5 Resilience Principle 5: Encourage learning and experimentation	67
3.4.1.6 Resilience Principle 6: Broaden participation	68
3.4.1.7 Resilience Principle 7: Promote polycentric governance	
3.5 Conclusion	

CHAPTER 4: PARTICIPATORY ACTION RESEARCH IN STUDENT-LED FOOD GARDENS

4.1 Introduction	75
4.2 Action research methodology	76
4.2.1 Pragmatic action research	77
4.2.2 Critical action research4.2.2.1 Influences from critical pedagogy4.2.2.2 Influences from the theory of communicative action	80
4.2.2.2 initialities from the theory of communicative action	
4.3 Overall study structure	
4.3.1 Systematic literature review4.3.1.1 Inclusion criteria4.3.1.2 Method	94
4.3.2 Case Study Approach 4.3.2.1 The case study	
4.4 A student gardener studying student gardens as a researcher where she works	111
4.5 Research stages and main methods	114
4.5.1 Sampling and recruitment of research participants	
4.5.2 Recording and transcription	
4.5.3 A messy start: Constructing	
4.5.4 Planning and taking action	
4.5.5 Evaluating	
4.5.6 Bringing it all together	
4.6 Quality measures in mixed methods action research	

4.6.1 Quality in the quantitative systematic literature review	
4.6.2 Quality in the case study	
4.7 Ethical considerations	
4.8 Conclusion	

CHAPTER 5: SYSTEMATICALLY REVIEWING SUSTAINABLE UNIVERSITY COMMUNITY GARDENS

5.1 Introduction	153
5.2 Sustainable university community gardens as compared to urban community gardens	154
5.2.1 Characteristics of articles and gardens researched	154
5.2.2 Benefits of sustainable university community gardens and urban community gardens	157
5.3 The dominating benefits	169
5.4 The absence of students' transience	172
5.5 Conclusion	173

CHAPTER 6: UNDERSTANDING THE DYNAMICS OF PROBLEMATIC PARTICIPATION

6.1 Introduction	177
6.2 Causes of problematic participation	177
6.2.1 Barriers to 'getting people through the garden gate'	178
6.2.2 Barriers to continued engagement	179
6.3 Effects of problematic participation	188
6.4 A causal loop diagram to understand transience and problematic participation	196
6.5 Strategic points for intervention	201
6.6 The role of long-term stakeholder support2	202
6.7 Conclusion	203

CHAPTER 7: ASSESSING THE VULNERABILITY OF STUDENT-LED FOOD GARDENS

7.1 Introduction	207
7.2 Assessing the vulnerability of student-led food gardens	207
7.2.1 Maintenance of diversity and redundancy	. 208
7.2.2 Management of connectivity	.212
7.2.3 Monitoring of slow variables and feedbacks	.215
7.2.4 Fostering complex adaptive systems thinking	.218
7.2.5 Encouraging learning	. 220
7.2.6 Broadening participation	. 223

7.2.7 Promoting polycentric governance	226
7.2.8 Summary of the vulnerability of student-led food gardens	227
7.3 Reflections on the resilience principles	231
7.3.1 Usability of the resilience principles	232
7.3.2 Maximising or managing diversity	233
7.3.3 The need for ontological humility in fostering complex adaptive systems thinking	234
7.3.4 Learning, resilience, and communities of practice 7.3.4.1 Resilient learners or learning for resilience?	
7.4.4.2 Using communities of practice to enhance learning for resilience	
7.3.5 Broadening participation	239
7.3.6 Applying the resilience principles beyond ecosystem services	240
7.4 Conclusion	246

CHAPTER 8: MANAGING THE IMPACTS OF TRANSIENCE AND BUILDING RESILIENCE

8.1 Introduction	253
8.2 Strategies rooted in analysis from Chapters 62	253
8.2.1 The role of university and students' union staff2	254
8.3 Strategies rooted in the analysis from Chapter 72	256
8.4 Strategies suggested by participants2	259
8.4.1 Have an appropriate and well-managed space2	260
8.4.2 Run events and activities2	264
8.4.3 Get support2	265
8.4.4 Nurture interest and creativity2	267
8.4.5 Increase visibility2	268
8.4.6 Create organisational infrastructure2	274
8.4.7 Recruit	276
8.4.8 Document and plan2	277
8.4.9 Embed in wider organisational infrastructure2	282
8.4.10 Maintain healthy relationships and social networks2	284
8.4.11 Connect	285
8.4.12 Create vibrancy and focus on value2	286
8.4.13 Appeal to extrinsic motivation & resource adequately2	287
8.5 Problems without solutions2	288
8.6 Increasing participation through 'action competence'2	289
8.7 Retaining social-ecological memory2	<u>293</u>
8.8 Summary of advice for different stakeholders2	297
8.9 Conclusion	308

CHAPTER 9: METHODOLOGICAL REFLECTIONS AND FUTURE DIRECTIONS

9.1 Introduction	13
9.2 Methodological reflections	13
9.2.1 The changing research focus – and the value of a research diary	14
9.2.2 Gaining privileged access to the separate worlds of students and staff	17
9.2.3 Action research with and about transient partners319.2.3.1 Transient participants lack insight into their own transience329.2.3.2 Difficulty providing 'value' to research participants329.2.3.3 Difficulties implementing 'actions'329.2.3.4 Lacking capacity to engage in research329.2.3.5 The researcher as 'intervention'32	19 22 24 25
9.2.4 Using an embedded case study approach while conducting action research	29
9.2.5 Implications of early career researcher naiveté	31
9.2.6 Problem-focus discomfort	34
9.2.7 Missing stakeholder involvement	37
9.2.8 On the <i>ex-post</i> causal-loop mapping and use of the resilience principles	38
9.3 Future research directions	39
9.3.1 Long-term implications of attitude and behaviour change initiatives in tertiary education34	40
9.3.2 What and how are students learning?	44
9.3.3 Pedagogy in the critical research agenda for urban agriculture	48
9.3.4 Failure	49
9.3.5 Intersectional interrogations of student-led food gardens	51
9.3.6 Student-staff relationships in university sustainability initiatives and living labs	53
9.3.7 Resilience at odds with other aims	54
9.3.8 Understanding transience	54
9.4 Conclusion	55

CHAPTER 10: CONCLUSION

10.1 Introduction	359
10.2 The benefits of sustainable university community gardens	359
10.3 Existing understandings of the impacts of students' transience	360
10.4 Problematic participation: causes, effects, and feedbacks	361
10.5 Student-led food gardens' vulnerabilities in the face of students' transience	363
10.6 Applying the resilience principles beyond ecosystem services	365
10.7 Strategies to address the impacts of students' transience and build resilience	366
10.8 Methodological reflections	370
10.9 Future research directions	372

10.10 Conclusion .	 373

References	377
Appendix A: Wiek et al.'s (2011) key competencies in sustainability	405
Appendix B: Student Eats funding	407
Appendix C: Reflective diary prompts	409
Appendix D: Ethical approval letters	411
Appendix E. Summary of findings from the 'constructing' phase	417
Appendix F: Action plans	425
Appendix G: Photovoice information sheet	439
Appendix H. Fishbowl discussion guidelines	441
Appendix I: Visualisations of the causal loop mapping using TheBrain software	443

Table of Contents: Key Definitions

Education concepts

Action competence	26
Conscientisation	81
Embodied learning	27
Hidden curriculum	
Informal learning	28
Living labs	29
Non-formal learning	28
Sustainability education	22

Methodological concepts

Action research	76
Data saturation	117
Descriptive coding	125
Eclectic coding	
Go-along interview	115
Holistic coding	125
In vivo coding	
Photovoice	130
Protocol coding	134
Simultaneous coding	125

Participation, transience & volunteering concepts

Irregular participation	36
Low participation	36
Problematic participation	36
Short-term participation	36
Temporary organisation	39
Transience	37
Turnover	37

Resilience & systems theory concepts

Boundary interaction	222
Community resilience	53
Ecological resilience	52
Ecosystem services	58
Feedback loop	49
Mental model	65
Organisational memory	
Organisational resilience	53
Social memory	
Social resilience	52
Social-ecological memory	
Social-ecological resilience	54
Social-ecological system	49
Transformation	55
Vulnerability	54
Wicked problem	50

Sustainability concepts

Sustainability	18
Sustainability science	18

Part I: The background

Introduction

Chapter 1

...what happens to a species that loses touch with its habitat? And where will all the conservationists come from when kids no longer have a patch of ground that they can truly call my space?

Robert Michael Pyle, Pulling the Plug, n.d.

Let me get it right. What if we got it wrong? What if we weakened ourselves getting strong? What if the message carried in the wind was saying something? From butterfly wings to the hurricane It's the small things that make great change

Lemn Sissay, Gold from the Stone: New and Selected Poems, 2017, p. 179



Courgette seedlings Spring 2014

1.1 Introduction

There is little more essential to human life on Earth than food. It is central to sustaining our existence and shapes our day-to-day lives profoundly. However, producing food has impacted our planet dramatically and detrimentally. Agriculture takes up 38% of the Earth's land and contributes to approximately 12% of human-caused greenhouse gas emissions (Foley et al., 2011). In spite of grand efforts to feed the world, over 800 million people are still undernourished (World Bank, n.d.). These figures represent only glimpse of our global food system's problems.

Perhaps one of the most concerning problems underlying these issues is a growing disconnection from nature (Kesebir & Kesebir, 2017) and the food systems on which we all depend. Disconnection from nature is certainly an issue that has occupied the minds of educators (Liefänder et al., 2012; Liu & Lin, 2014; Lankenau, 2016), and rightly so, given that education can be an important tool for shifting "our most deeply held, unconscious set of assumptions and values; the things we take for granted; and that which determines our expectations, frames the questions we ask, and structures our approach to what we do" (Zohar, 1997, p. 22; Olson & Raffanti, 2006). These deep-seated shifts within ourselves can give us the capacity to leverage substantial change in the world we live in and the systems we are a part of (Meadows, 1999).

It is with this in mind that universities have turned to their campuses to facilitate learning that supports pro-sustainability attitude and behaviour change. One type of these initiatives are food gardens, which support students in developing their understanding of where their food comes from. Student-led food gardens are places where groups of university students manage and operate spaces on university campuses where edible plants are grown. There are sometimes other participants, such as university staff or local community members,

but students are the driving force behind these initiatives. Student-led food gardens are appearing in many universities across United Kingdom, in part thanks to funding from the National Union of Students. This study focuses on these student-led food gardens funded by the National Union of Students through a scheme called 'Student Eats.' Eighteen gardens were initially funded between 2011 and 2013, but now sixty-five universities have gardens affiliated with the Student Eats network (NUS, n.d.b). Apart from providing learning opportunities that foster pro-sustainability behaviour change in students, these gardens are thought to have a range of other beneficial outcomes, from providing fresh produce and contributing to healthy diets on campus, to reducing carbon emissions (NUS, n.d.c). However, there has yet to be a systematic assessment of the benefits of these gardens reported in peer reviewed literature. Therefore, the first set of questions this thesis will address is:

- 1. According to academic literature about sustainable university community gardens¹,
 - a) What benefits (both discussed and demonstrated) do these gardens reportedly provide?
 - b) How do these benefits compare to the benefits provided by community gardens more generally?
 - c) What is already known about the impacts of students' transience on sustainable university community gardens?

If a review of the literature provides evidence that sustainable university community gardens provide mostly positive outcomes, then we can assume that they could be a contributor to a sustainable future. However, in spite of potential contributions to sustainability, these initiatives face a longevity issue: the transience of students. Students' transience, or to pass through with only a brief stay (Merriam Webster Online, n.d.), is caused

¹ Given that the literature about student-led food gardens is so sparse, this question was broadened to include all sustainable university community gardens. This is explained further in section 1.1.2 and in Chapter 4.

by the time-limited nature of academic programmes and periodic holidays. Its main impacts are therefore short-term participation (when people are involved for limited periods of time) and irregular participation (when involvement is inconsistent or non-regular). These can also lead to low participation, which is when limited volunteer hours are invested overall. Ultimately, these modes of participation are problematic for student-led food gardens because they cause problems, such as inadequate maintenance of gardens and a lack of capacity to run activities (McKinne & Halfacre, 2008). In this thesis, the terms short-term, irregular, and low participation will collectively be called 'problematic participation.'

Providing that the literature review brings forth evidence that sustainable university community gardens provide mostly positive outcomes, it is then important to understand how transience impacts the participation dynamics of these gardens. However, to understand how transience impacts them, we first must understand participation dynamics in student-led food gardens. Therefore, the second set of questions this thesis will address is:

 a) In addition to students' transience, what causes problematic participation in student-led food gardens?

b) What effects does problematic participation have in student-led food gardens?c) What feedbacks between problematic participation, its causes, and effects in student-led food gardens exist (if any)?

Once the causes and effects of short-term, irregular, and low participation, and their feedbacks are known, the next step of this thesis will be to assess how vulnerable student-led food gardens are. The next research question is therefore:

3. In what ways are student-led food gardens vulnerable to students' transience in the context of the participation dynamics outlined above?

Finally, after outlining exploring challenges student-led food gardens face, the final research question is solution-orientated, asking:

4. In what ways can student-led food gardens address the impacts of students'

transience and build resilience into their initiatives such that they continue to persist?

1.2 Overview

I have broken this thesis into three parts. The first part (which begins with this introduction chapter) is intended to provide the necessary *background* information for this study. Beyond this introduction, it includes a narrative-style literature review, and chapters outlining my theoretical framework and methodology. The second part presents empirical findings related to transience and participation *problems*. This includes a quantitative, systematic literature review, causal loop mapping of the participation dynamics in student-led food gardens, and an assessment of the vulnerability of the student-led food gardens. The third part is *future*-orientated. It begins with an empirical chapter outlining strategies to address the problems described in the previous section. Following on, I reflect on my methodology and present some suggestions for future research. The final chapter draws together and summarises the main contributions I make in this thesis. I will now explain what each of these sections contain in greater detail, chapter by chapter.

1.2.1 The Background

After this introduction, I begin the second chapter with a narrative-style literature review. In this review, I explain how our current dominant food system in unsustainable and suggest that sustainability science is the field of scholarship that is most appropriate to address it. I then introduce the rationale for and manifestations of sustainability in higher education, the concept of 'living labs' on university campuses, and the role of co-curricular activity as strategies to move to society towards sustainability and sustainable food systems. Finally, I conclude the chapter with a summary of the current state of sustainable food activity

and food gardens in universities, suggesting that it appears that the transience of students may negatively impact the longevity of student-led food gardens and it appears that this is not yet well understood.

Given that it seems that the longevity of student-led food gardens may be under threat due to students' transience, in the third chapter I explain how I will use resilience as a theoretical framework. I chose to work with resilience because it focuses on the "capacity of a system to absorb disturbance and reorganise while undergoing change so as to still retain essentially the same [...] identity" (Folke, 2016, p. 4), which is the challenge that these gardens are struggling with. The chapter begins with a brief outline of my ontological and epistemological stance, and then goes on to position my research as being rooted in systems theory, particularly complex adaptive systems theory. Then I go on to explain that resilience is a useful way of thinking about how to maintain the long-term success of a complex adaptive social-ecological system, like a student-led food garden, because it accounts for the uncertainty of complexity. Finally, I outline some key concepts in resilience and a set of principles for building resilience proposed by Biggs et al. (2012, 2015c), and explain how I will be using them in my thesis.

The fourth chapter focuses on the methodology. The first part of the chapter uses pragmatism, critical pedagogy, and the theory of communicative action to make the case for using an action research methodology. Action research is a multi-stage methodology which combines a more traditional research agenda with an action-orientated agenda, focusing on addressing a problem or issue. I then outline my interpretation of action research as a methodology and how I use it. After that, I dig down into some of the more procedural parts of the methodology, beginning by explaining the systematic literature review and the case study approaches used. I used the United Kingdom's National Union of Students' Student Eats campus food growing scheme as a case study with a sample of three gardens as subunits of

analysis (Yin, 2014). The methods used focused on developing an understanding of students' transience as a 'problem', and then creating and assessing strategies to address it. Within the case study approach, I used a range of methods including interviews, workshops, photovoice, and a fishbowl dialogue.

1.2.2 The Problem

The first steps I undertake in the second section of this thesis are to (1) assess the main documented benefits of student-led food gardens compared to community gardens in general, and (2) assess the extent to which transience impacts sustainable university community gardens according to the literature. I review the twenty-two articles that met these criteria using quantitative systematic literature review. These articles are assessed according to their overall characteristics and the benefits they reported the gardens provided. These results are then compared with the results from a similar review of urban community gardens (Guitart et al., 2012).

Having assessed the main benefits of sustainable university community gardens and found that there appeared to be a lack of understanding of how transience impacts them, I then seek to better understand the dynamics of problematic participation in Chapter 6. I outline the causes these forms of problematic participation, the effects they have, and if there are any feedbacks. All of this is depicted in a causal loop diagram.

Given the new understanding of students' transience's influence on participation dynamics in the student-led food gardens, in Chapter 7 I use the resilience principles (outlined in Chapter 3) to assess the vulnerability of student-led food gardens to the transience of their own participant base. In addition to this assessment, I also offer an evaluation of my use of the resilience principles.

1.2.3 The Future

Having spent three chapters exploring what is known about students' transience in student-led food gardens, developing an understanding of the dynamics of problematic participation, and assessing the gardens' vulnerability, the final part of this thesis shifts attention from what is currently happening to what could or should happen. Chapter 8 is the most prescriptive chapter of this thesis, presenting strategies to enhance the resilience of student-led food gardens in the face of transience by drawing on the analyses from Chapters 6 and 7, as well as suggestions from the participants of this study.

The focus of the first half of Chapter 9 will then be to take a step back and consider some of the methodological decisions I made in this study, and the resulting challenges, assets, and complications of my approach. I will also offer some advice for researchers adopting similar methodologies, especially action researchers engaging with transient participants or partners. The second half of the chapter will offer some future research directions off the back of this study, for scholars working with sustainability in higher education, urban agriculture, resilience, and sustainability science more broadly. The final chapter of this thesis, Chapter 10, will summarise this study's contributions to research and practice, and make some concluding remarks. Before diving into the literature review, however, I will make a brief note on my stylistic writing choices.

1.3 A note on the 'voice' in this thesis

Here, I will briefly explain my rationale behind the style of prose I have chosen to adopt in this thesis. My ontological and epistemological assumptions will be outlined at the beginning of Chapter 3 and my methodology in Chapter 4, and these will help to clarify why I

have made these choices. However, to dissuade any reader from assuming I have been sloppy or unconsidered I take this opportunity to explain the rationale for my choices early on.

The reason I am compelled to write this section is because the 'academic voice' is traditionally one characterized by objectivity and dispassion. This is assumed to reduce bias and allow the reader to make up their minds themselves based on objective facts. However, this assumes that there are 'objective facts' to be ascertained. This is an assumption about which there is much debate (Jerneck et al., 2011; Moon & Blackman, 2014).

Even as first-person voice is increasingly accepted in academic writing, there are still widespread assumptions that researchers should use a writing voice that is formal and maintains some distance from the subject. However, I maintain that the use of the 'objective voice' is only a 'safer' way of writing - it is not necessary more 'neutral' in all cases. However, researchers who have objected to the 'objectivity' of 'neutrality' have been subject to criticisms that they let their emotions distort their data (Mitchell, 2017), and those that undertake self-examination are "navel gazing" (Leggatt-Cook, 2011, p. 404) and narcissistic (Finlay, 2002). However, I think this backlash against making the author visible is illustrative of a quote from Ahmed (2017, p. 37): "when you expose a problem, you pose a problem." This quote refers to the exposing of a problem associated with misogyny or racial inequality, but also holds true for scholars adopting a different kind of 'voice' in their scholarship. That is, when scholars have identified problems with adopting an 'objective' voice in academic writing, scholarship that does not do so has been targeted and subject to intense criticism. This being said, I don't think making the author visible is unproblematic. Actually, the opposite: there is very much a need to consider the authenticity in the way first-person writing is used, which I do both in my methodology in Chapter 4, and in my methodological reflections in Chapter 9.

For my doctoral research I used an action research approach, and this has been the main influence on my choice to make myself and my experiences visible within the narrative of this thesis. I will briefly explain action research here, but a more detailed explanation and my rationale for using it can be found in Chapter 4. Action research is characterized cycles where the researcher, in collaboration with relevant stakeholders, make sense of a problem, plan actions to address it, take these said actions, and then evaluate the action. This is a *collective* cycle of action and reflection. Within these cycles the researcher also goes through their own personal cycles of action and reflection. Coghlan and Brannick (2014) call these meta-cycles because they are action and reflection in relation to the collective cycles. The purpose of these *personal* cycles of action and reflection are to enable a heightened level of reflexivity and critical thinking. As a result of these cycles, the researcher (in this case, myself) and researcher's experience is drawn into the inquiry. I used a research diary throughout the research process, and this diary was analysed along with the other data I gathered (such as interview data). I also revisited other relevant documents that I accumulated, like proposals I wrote at different stages of my research process and abstracts for conference presentations in order to build the narrative of this study. As such, I have a relatively strong and verifiable personal narrative that accompanies the data that came from interviews, workshops, and other sources. My personal research narrative was used to help explain and verify the evolution of my thinking throughout the thesis process, and to illustrate points made in the discussion. This is done mainly in the Chapter 4.

Another reason I have chosen to make myself so visible within this thesis is because of my 'insider status.' I unpack my insider-outsider positionality in section 4.4, but in short, I have been a student gardener as both an undergraduate and Master's student, and have been highly active in one of the case study gardens examined in this thesis. Therefore, to
conceal my insider status by omitting myself from the narrative of this thesis would provide an incomplete picture.

Situating my writing from a personal perspective has helped me to be as transparent and truthful as possible in my writing. To write dispassionately, or as if I am somehow disconnected from the research I have undertaken would be untruthful and obscure the reality of the study. Instead I have included information to situate myself and my own experiences to help, you, the reader, to understand where my reasoning comes from so that you can use that to assess the reliability of the assertions I make in my thesis.

In summary, I have been shaped by my experiences, and, in turn, I have shaped the direction of this thesis. Therefore, I have made deliberate efforts to be transparent, authentic and rigorous in my use of voice, my methodology, and my interpretation of the data. In this way, I will be clear about my role in this research throughout this thesis and won't make any attempts to conceal my influence.

Reviewing sustainable food and sustainability in higher education

Chapter 2

Education must be conceived as a continuing reconstruction of experience: [...] the process and goal of education are one and the same thing.

John Dewey, 1897, p. 79



Construction of raised beds Winter 2013

2.1 Introduction

Over the course of this chapter, I introduce the areas of scholarship this thesis engages with. My first efforts focus on outlining the unsustainability of the global food system and its local-level consequences, which will be followed by a brief introduction to the field of sustainability science. I explain why I see my research primarily as a contribution to sustainability science, although my research engages with scholarship from a range of disciplines. I then outline some developments, theories and concepts for sustainability education in universities pertinent to this thesis. After this, I come to the central topic of my thesis, which is student-led food gardens in universities. In addition to explaining what I mean by 'student-led food gardens,' I outline some of the positive contributions these gardens may make to sustainability and sustainability education, and the main challenges they face. By the end of this chapter, you should have a good understanding of the potential student-led food gardens have to make to the education of their participants and how this could contribute to a sustainable future.

2.2 An unsustainable food system and the scholarship to address it

The global food system has led to widespread sustainability problems of both environmental and social natures in an effort to feed the world (Foley et al., 2011). While food production has increased in recent decades, there are also major problems connected to achieving equitable distribution of food as a result of complex factors related to the global market and corporate influence in the food industry (Kneen, 1999; Kasa, 2008; Miller & Harkins, 2010; Marktanner & Noiset, 2013; Rocchi et al., 2013). On top of this, our food system causes serious environmental problems, such as soil erosion, water pollution, and loss of biodiversity (IPCC, 2019).

Alongside the rise of these modern sustainability problems, it is thought that people are becoming increasingly disconnected with nature (Louv, 2008; Kesebir & Kesebir, 2017). There is evidence of this in the language we use, for example, references to nature in dictionaries, books, songs and films feature less nowadays than they did in the past (Wolff et al., 1999; Kesebir & Kesebir, 2017). This is worrisome, not only because nature connectedness can contribute to happiness (Capaldi et al., 2014), but also because having a connection to nature makes people more likely to be concerned about the environment and take steps to act on that concern (Nisbet et al., 2008).

A new area of scholarship is developing to address our profound socio-ecological problems. Sustainability science is an emerging field which, in simplest terms, aims to "understand the fundamental character of interactions between nature and society" in order to meet "fundamental human needs while preserving the life-support systems of planet Earth" (Kates et al., 2001, p. 641). Because the problems it grapples with are complex and farreaching, sustainability science is united by the issues it addresses rather than "disciplines it employs," much like agricultural or health sciences (Clark, 2007).

Before explaining sustainability science further, I will briefly outline what I mean by the term 'sustainability.' 'Sustainability' and 'sustainable development' are contested and fraught terms, however most definitions are united by a main driving principle of intergenerational equity. The most widely used definition that captures this is from the Bruntland Commission, which says that sustainable development is development that "meets the needs of the present without compromising the ability of future generations to meet their own needs" (United Nations, 1987). This definition has been criticised for not going far enough (Reed, 2007), as well as for being too vague to operationalise (Broman & Robèrt, 2017), however it has been a useful definition for propelling action.

I would also like to clarify the difference between 'sustainability' and 'sustainable development.' 'Sustainability' often thought to be the 'end goal,' and 'sustainable development' is the path we use to get there (Hector et al., 2014). However, the term 'development,' carries with it a range of connotations, including the ambition of uncontrolled economic growth and "cosmetic environmentalism on the part of both governments and businesses" (Robinson, 2004, p. 374). Or, rather than constituting an entirely new concept, sustainable development is seen as painting development 'green,' which many (myself included) consider to be insufficient for the scale of the problems we face today (Robinson, 2004). I do acknowledge that when others use the term 'sustainable development,' often they are not thinking of it in this negative way. However, so that I am not misinterpreted, I chose to adopt the term 'sustainability' in this thesis. In terms of defining sustainability, I consider the Brundtland definition of sustainable development to be a sufficient definition of sustainabile development to be a sufficient definition of sustainabile development to be a sufficient definition of sustainability for the purpose of this thesis, in spite of its flaws.

Now that I have explained what I mean by sustainability, I will move on to describing sustainability science. Sustainability science aims to develop "a comprehensive, holistic approach to identification of problems and perspectives" relating to sustainability (Komiyama & Takeuchi, 2006, p. 3). Sustainability scientists advocate for 'transdisciplinary' research to deal with sustainability problems (Hirsch Hadorn et al., 2006; Max-Neef, 2005). Transdisciplinarity, as it is usually understood in sustainability science, involves the cooperation between multiple disciplines driven by different guiding questions (such as 'what exists?' in disciplines like mathematics, ecology and sociology, 'what are we capable of doing?' in disciplines like architecture, agriculture and medicine, 'what do we want to do?' in disciplines like planning, politics, and environmental design, and 'what should we do?' in disciplines like ethics, philosophy, and theology) (Max-Neef, 2005). Sustainability scientists also advocate for the use of participatory processes to produce scientific knowledge, while

also taking action to address sustainability problems (Jerneck et al., 2011; Komiyama & Takeuchi, 2006; Kates et al., 2001). I consider this thesis to be a contribution to the field of sustainability science.

2.3 Sustainability in higher education

Sustainability education is a crucial part of movement towards a more sustainable future. Through sustainability education, students can be equipped with the tools to become effective change agents in the pursuit of sustainability (Jucker, 2011). The role of education has long been a concern for environmentalists. In contemporary environmentalism, it was highlighted internationally as early as the 1992 Earth Summit in Rio de Janeiro (UNCED, 1992; Wals, 2012). During the United Nation's Educational, Scientific and Cultural Organisation's (UNESCO) Decade of Education for Sustainable Development (from 2005 to 2014) sustainability education made substantial advancements (Wals, 2012). This has been followed up with the UNESCO's Global Action Programme on Education for Sustainable Development, which focuses on "reorienting education and learning so that everyone has the opportunity to acquire the knowledge, skills, values and attitudes that empower them to contribute to a sustainable future" and "strengthening education and learning in all agendas, programmes and activities that promote sustainable development" (UNESCO, 2014, p. 14). Further evidence that sustainability education is a high priority globally is that sustainability education is included in the United Nation's 4th Sustainable Development Goal, which is: "by 2030 all learners [should have acquired] knowledge and skills needed to promote sustainable development, including among others through education for sustainable development and sustainable lifestyles, human rights, gender equality, promotion of a culture of peace and non-violence, global citizenship, and appreciation of cultural diversity and of culture's contribution to sustainable development" (GAML, 2017).

Sustainability education in higher education is of particular importance because university graduates are often the people who go on to take on leadership roles, and therefore have the potential to make large-scale change. Furthermore, university students are typically at a unique point in their lives when they have just moved from home and may be developing habits that stay with them through their lives (Verplanken et al., 2008; Thompson et al., 2011; van Niekerk & Barnard, 2011). These particular windows of time in a person's life when they can be able to make changes to otherwise inflexible habits have been called "life-course transitions" (Elder, 1998; Verplanken et al., 2008; Schäfer et al., 2012). The idea that life course transitions might be ideal moments for behaviour change interventions is called the 'habit discontinuity hypothesis' (Verkplanken et al., 2008). For example, househunting is thought to be a 'window of opportunity' in which students acquire new travelrelated behaviours (Haggar, 2019). Another example is when young people move from their parental home and begin shopping for and cooking food independently for the first time. If the habit discontinuity hypothesis holds, the students' skills, knowledge, and perceptions on their ability to contribute to positive changes that are developed throughout their university experience may have longer-lasting effects than experiences at other points in their lives (Thompson et al. 2011). It is therefore important to maximise their sustainability learning through their university education. The habit discontinuity hypothesis has been used to justify many of the National Union of Students' sustainability schemes, including Student Eats. Other schemes that have been underpinned by the habit discontinuity hypothesis are Green Impact, an initiative which provides a structured framework for taking pro-sustainability actions on university/college campuses and in students' unions (NUS, n.d.a); Student Switch-Off, an initiative promoting behaviour change around energy reduction (NUS, n.d.f); and Students for Trees, an initiative to "protect, enhance and celebrate woods and trees" on college and university campuses (NUS, n.d.d).

There are many terms that overlap with 'sustainability education', including Education for Sustainable Development (ESD), education for sustainability, and environmental education. Like Sterling (2010) and Hill (2013), I understand 'sustainability education' to incorporate all of these traditions. To formulate my own definition that captures how I understand 'sustainability education', I have drawn on definitions from the United Kingdom's Quality Assurance Agency in Higher Education (Longhurst et al., 2014) and Sterling (2010). In this thesis, sustainability education will be understood as:

the process of equipping students with an operating ecological or participatory worldview, and the competencies needed to work and live in a way that safeguards

environmental and social wellbeing, both in the present and for future generations.

I have deliberately left the 'economic pillar' of sustainability out from this definition. The 'economic pillar' is a contested component of the definition of sustainability. The 'three pillar' approach is typically seen as a means to elevate the social and environmental priorities to on to the agenda of businesses by aligning it with the more traditional priority of profit (Norman & MacDonald, 2004). As such, it has been criticised for valuing an instrumental aim (profit) at the same level as intrinsically valuable social and environmental aims. As such, some scholars who believe that environmental and social wellbeing have *a piori* intrinsic value have argued that environmental and social components should therefore be at the centre of the definition, while the economy should not (Robért et al., 2002; Missimer et al., 2010). As a way of reconciling this some suggest that, rather than 'three pillars,' what we need are, rather, three concentric circles in which the economy is embedded in society, and society is embedded in the environment (Giddings et al., 2002).

However, other scholars continue to highlight the importance of the 'economic pillar' in sustainability because of the unsustainability of present economic systems and the need for alternative, more sustainable economic systems. In other words, *because* economies are

so unsustainable, they need to be considered as a vital component of sustainability. Spencer et al. (2018) have offered some propositions of how economic systems might become more sustainable, such as though better valuing care work, biological reproduction, problematising wealth and power concentration, and welcoming diversity and pluralism. There is also increasingly better recognition of alternative, possibly more sustainable, economic practices, such as sharing economies and collaborative consumption (Hamari et al., 2015), while new economic systems, such as degrowth, are being articulated and envisioned (Martínez-Alier et al., 2010).

In the context of sustainability in higher education, neoliberalisation has had considerable impact on the sustainability agenda of universities in the United Kingdom through the marketisation of higher education (Bessant & Robinson, 2017, 2019). In this new system, "financial control, efficiency, [and] value for money" are prioritised (Bessant & Robinson, 2017, p. 421). Furthermore, competition for students has driven universities to place greater emphasis on the 'student experience' and recruiting international students (Bessant & Robinson, 2017). From this, it is clear that economics has considerable influence on and potential to contribute to sustainability and sustainability education. However, I chose not to include it in my definition of what sustainability education is, because I still see the economy as a means to an end, albeit an important means.

I have also combined both an instrumental and an intrinsic view of sustainability education in the definition. The instrumental (sometimes called 'pragmatic,' 'behaviourist,' or 'ESD type 1') tradition of sustainability education focuses on education *for* sustainability – that is, education "as a means to an end" (Vare & Scott, 2008; Sterling, 2010, p. 513). This tradition tends to be more outcomes-orientated, and is "motivated by a sense of urgency and a passion to increase levels of what is commonly referred to as 'sustainability literacy'" (Sterling, 2010, p. 513). Conversely, the intrinsic (sometimes called 'emancipatory,' 'critical,'

or 'ESD type 2') tradition tends to be process-orientated – that is, education *as* sustainability. It therefore "involves the development of learners' abilities to make sound choices in the face of uncertainty and complexity of the future" (Vare & Scott, 2008, p. 3). Like Vare and Scott (2008), I believe drawing on both these traditions is necessary, and moreover, in practice these two approaches to sustainability education rarely, if ever, exist in isolation. Sometimes learning *for* sustainability will result in learning *as* sustainability, and vice versa.

In my definition of sustainability education, I have also replaced the terms "knowledge and understanding, skills and attributes" from Longhurst et al. (2014, p. 5) with 'competencies.' Education is typically thought to have three dimensions: cognitive, psychomotor, and affective. Together, these are sometimes called "heads, hands, and heart" (Sipos et al., 2008). The cognitive dimension deals "with the recall or recognition of knowledge and the development of intellectual abilities and skills," or the 'head' (Bloom et al., 1956, p. 7). The psychomotor dimension, the 'hands,' deals with the ability to physically manipulate a tool or object, that is, to use motor skills (Bloom et al., 1956, p. 7). Finally, the affective domain, or the 'heart,' has to do with the development of "values, attitudes and behaviours and involves the learner emotionally" (Shephard, 2008, p. 88). Competence is a term that brings together all of these domains, which is useful because all three of the domains are thought to be important for sustainability. There is considerable terminological ambiguity with regards "skills, abilities, capabilities, capacities, qualifications and other concepts" related to sustainability competence in general (Wiek et al., 2004, p. 204). However, the term 'competence' can act as a catch-all for all these terms, as it refers to a "functionally linked complex of knowledge, skills, and attitudes that enable successful task performance and problem solving" (Wiek et al., 2004, p. 204). In the sustainability education community internationally, the term 'competency' is used much more widely and tends to be much more inclusive, capturing not only cognitive learning outcomes, but also affective and

psychomotor outcomes as well (de Haan, 2006; Barth et al., 2007; Brundiers et al., 2010; Mogensen & Schnack, 2010; Wiek et al., 2011; Hedefalk et al., 2014). I have chosen to use the term 'competency' in order for the content of this thesis to have wider relevance given the term's more inclusive nature.

One competency framework that has been particularly influential in the sustainability education community in recent years is Wiek et al.'s (2011) key competencies for sustainability. Reviewing, analysing, and synthesising seminal works in the literature about sustainability competencies and learning outcomes, Wiek et al. (2011) proposed a framework integrating five key competencies for sustainability: systems-thinking competence, anticipatory competence, normative competence, strategic competence, and interpersonal competence. Systems-thinking competence is "the is the ability to collectively analyse complex systems across different domains (society, environment, economy, etc.) and across different scales (local to global), thereby considering cascading effects, inertia, feedback loops and other systemic features related to sustainability issues and sustainability problem-solving frameworks" (Wiek et al., 2011, p. 207). Anticipatory competence is "the ability to collectively analyse, evaluate, and craft rich 'pictures' of the future related to sustainability issues and sustainability problem-solving frameworks" (Wiek et al., 2011, p. 208-209). Normative competence, then, is "the ability to collectively map, specify, apply, reconcile, and negotiate sustainability values, principles, goals, and targets" (Wiek et al., 2011, p. 209). Strategic competence is "the ability to collectively design and implement interventions, transitions, and transformative governance strategies toward sustainability" (Wiek et al., 2011, p. 210). Finally, interpersonal competence is "the ability to motivate, enable, and facilitate collaborative and participatory sustainability research and problem solving" Wiek et al., 2010, p. 211). While Wiek et al.'s (2011) work contains evidence of both the instrumental and emancipatory traditions, overall the framework leans towards the instrumental tradition as

the framework focuses mainly on education for the purpose of problem-solving. More detail on Wiek et al.'s (2011) competencies can be found in Appendix A for interested readers.

A competing, or as I see it, a complementary body of literature focuses around the development of 'action competence' (Bruun Jensen & Schnack, 1997; Breiting & Mogensen, 1999; Mogensen & Schnack, 2010; Hedefalk et al., 2014). Action competence tends to defy definition in that it is an educational ideal rather than a knowledge or skill *per se* (Morgensen & Schnack, 2010), and is more aligned with the emancipatory educational tradition. As such, it is different in essence from other educational competency frameworks with more instrumental aims, such as the one outlined by Wiek et al. (2011)². Action competence is closely linked the German concept of '*Bildung*,' which is associated with self-cultivation, rooted in a critical approach to pedagogy (Morgensen & Schnack, 2010). It has been defined as the "ability, motivation and desire to play an active role in finding democratic solutions to problems and issues connected to sustainable development" which "cannot be reduced to mere education in the sense of cultivation, normalisation, or traditional socialisation" (Morgensen & Schnack, 2010, p. 61). Ultimately, action competence is less of something that someone *has*, but more something that someone *does*.

As I said, I see these as complementary ways of thinking of sustainability competencies. Wiek et al.'s (2011, p. 207, 211) framework offers a more concrete way of delivering sustainability education through an "explicit and commonly shared reference framework" with more concrete concepts, methodologies, and "peer reviewed 'classics.'" However, while this framework is comprehensive, specific, and thorough, it begs some

² Wiek et al. (2011) do suggest that action-orientated competence is embedded within their 'strategic competence.' However, this term is used in a different way and has more to do with the instrumental ability to "collectively design and implement interventions, transitions, and transformative governance strategies toward sustainability" (Wiek et al., 2011, p. 2010), rather than abilities, *motivations* or *desires* to play an active role in sustainable development, as described by Morgensen and Schnack (2010).

readers like Shephard et al. (2019) to wonder, where does the learner's independence and freedom factor into the framework? Are democracy and honesty not equally as important as sustainability? For me, this is where action competence comes in, focusing on process, democracy, and the development of the individual as a good in and of itself, rather than for an instrumental purpose. Attempting to combine these two approaches to sustainability competencies is not unproblematic as there are embedded contradictions. Allowing for freedom and democracy is, to some extent, at odds with the desire to produce a specific outcome. However, in this study I have the benefit of having engaged in a more pragmatic line of inquiry: examining the impacts of students' transience in student-led food gardens and how to manage them. Reconciling the internal contradictions in the field of sustainability education that have existed for decades is far beyond the scope of this study. Instead, I recognise these traditions and draw on these two understandings of sustainability competencies (Morgensen & Schnack, 2010; Wiek et al., 2011) as relevant when broaching the topic of sustainability education in this thesis.

It should be noted that the cognitive dimension of learning is privileged in formal education, including sustainability programmes. However, researchers have highlighted the importance of affective outcomes for sustainability education, with the concern that we are only developing 'heads' and not 'hearts' (Ojala, 2007; Shephard, 2008). An important way of thinking about cognitive, affective and psychomotor development in sustainability education is through experiential learning (Shephard, 2008; Sipos, 2008; Brundiers et al., 2010). Experiential learning is "the process whereby knowledge is created through the transformation of experience. Knowledge results from the combination of grasping and transforming experience" (Kolb, 1984, p. 41).

Another relevant concept which has been influential in sustainability education is embodied learning. Embodied learning is "the active process through which changes and

shifts are experienced in, through, with, and because of the body. It is the mindful attention to, and retention of, this aforementioned process that determines the continuous emergence of self and that facilitates learning and cognition" (Munro, 2018, p. 5). Embodied learning deliberately draws on both the body and the mind in the learning process. Sipos et al. (2008) say that embodied learning engages the "head, hands and heart" (p. 68). The classroom can be a venue for embodied learning, however embodied learning that takes place beyond the classroom can have much more profound impacts than many educators in higher education give it credit for. One strategy to promote students' experiential and embodied learning beyond-the-classroom is through the use of the university campus as a 'living laboratory.' This will be the focus of this next section.

2.3.1 Living Labs

Sustainability education in the higher education context can take place through the formal, informal, and non-formal curriculum, as well as the campus environment, sometimes called the 'hidden curriculum' (Snyder, 1970). The 'hidden curriculum' is made up of "the implicit messages a university sends about sustainability through the institutional environment and values" (Winter & Cotton, 2012, p. 783). This might include message sent though the physical campus environment (How green is it? How accessible?), operations (What is the policy on pesticides? What procedures exist for reporting harassment?), or the work culture ('We compost food waste in our department'). Informal learning refers to learning that takes place in day-to-day life from educational influences in the learner's environment, however non-formal learning refers to organised educational activities that are outside of the formal educational system of schooling and university/college education (Coombs et al., 1973). University food gardens can be included as part of any four of these

curricula, however, in the UK they mainly operate as part of the informal and hidden curricula.

The use of universities as 'living laboratories,' which are often called 'living labs,' is an idea that has been proliferating widely within the sustainability education community, with the recognition that physical university campuses can contribute to and be used as a tool for sustainability learning, and universities as a place for sustainability experimentation. The Environmental Association of Universities and Colleges in the United Kingdom defines living labs within the tertiary education sector as "partnerships or programmes which connect academic activities of the institution (i.e., learning and teaching, and academic research) with non-academic partners" (EAUC, 2018). These non-academic partners can be external to the university, or they can be professional staff within the university, such as estates teams. Many definitions of living labs focus more on their use as a research tool (i.e., Robles et al., 2015; Colding & Barthel, 2017), however, there is a recognition that they can contribute to education (Zen, 2017) and, in the context of sustainability, developing a connection to nature (Colding & Barthel, 2017). The term 'living lab' is sometimes used to refer to a process or methodology, and sometimes to a site. Like Bergvall-Kåreborn and Stålbröst (2009), I see these understandings as complementary – a living lab methodology can be used on the site of a living lab. The EAUC definition is loose enough in this respect to be applicable to either.

Universities have used campus food growing projects as living labs around the world, though not all institutions may label them as such. A well-known example is the University of British Columbia's work developing its Centre for Sustainable Food System's on-campus farm (UBC, n.d.). However, food growing on campuses is also being used in living labs in the UK. For example, the University of Cambridge mentioned their campus community garden in their Living Laboratory Annual Report as early as 2013 (Cambridge Green Challenge, n.d.).

The notion of 'living labs' is relatively new, however the use of on campus food growing has a long history through agricultural training and research (McDowell, 2003). Land grant institutions in the United States have been champions of this. Unfortunately, in recent years there has been an erosion of public funding for these institutions (National Resource Council, 1995), which has resulted in attempts by colleges/universities to recoup funds through increasing tuition fees (Weaver & Diamantides, 1993) and private investments (National Resource Council, 1987). In spite of this, land-grant universities are pioneering the development of sustainable agriculture programmes (Jacobsen et al., 2012; Galt et al., 2013). As such, there is much that can be learned from the land-grant institution model about public engagement in sustainable food growing, as well as producing graduates prepared for and research applied to practical sustainability problems in the 'real world' (McDowell, 2003). However, living labs need not be integrated into the curriculum to contribute to student learning. Learning can take place through this 'living lab' model through student volunteering.

2.3.2 Student Volunteering

Student volunteering in the higher education context is often used or seen as a form of experiential or embodied learning (as in Mooney & Edwards, 2001; Moyer et al., 2014). Student volunteering has been defined as when students volunteer "their time in their local communities through programmes organised at/by their students' union or institution" (Student Volunteering England, 2004; as cited in Darwen & Rannard, 2011). I would clarify this definition to say that 'local communities' include the university community ('inward-facing volunteering') in addition to the surrounding geographic community ('outward-facing volunteering') (Darwen & Rannard, 2011, p. 178). However, 'community' could also refer to communities of interest, practice, or other type of non-geographic community. A concept

related to student volunteering³ is service learning, which is sometimes used synonymously, but more often refers to learning through 'volunteering' undertaken for academic credit (Mooney & Edwards, 2001).

The sorts of benefits student volunteering provide are hard to summarise because they are so context dependent. Squirrell et al. (2009) have provided a good overview of benefits to students as individuals, (including their employability), as well as benefits to the university, and the local community organisations in their report on student volunteering for the National Co-ordinating Centre for Public Engagement (Darwen & Rannard, 2011). Some of the sustainability-relevant benefits were students "doing something worthwhile," having the "opportunity to put theory into practice," broadening students' perspectives, supporting communities in providing volunteer-run services, and building a more cohesive community between the university and surrounding area (Squirrell, 2009, p. 20).

Student volunteering has been taking place for well over a century, providing social services, running clubs and camps, contributing to the war effort during the First and Second World Wars, undertaking raising and giving initiatives, and engaging in campaigning movements (such as anti-apartheid, tenant's rights, gay rights, and environmentalism) (Brewis, 2014). More recently there has been significant support across political parties at a policy-level for student volunteering. For example, in 2001, the Higher Education Active Community Fund was established to promote volunteering and distributed tens of millions of pounds of funding (HEFCE, 2001). Much of student volunteering in recent times has been "organised and supported within universities, rather than by students under the auspices of student unions," and the reason for this can be attributed to the governmental funding of the initiatives (Holdsworth & Quinn, 2010, p. 116). However, according to Holdsworth and Quinn

³ There are a range of other similar terms for these and similar types of activities, such as 'learning-linked volunteering' and 'community-engaged learning' (Darwen & Rannard, 2011).

(2010) "relatively little has been documented on student voluntary work, despite the political rhetoric in support of these activities. Furthermore, evidence of the outcomes of student volunteering is piecemeal and fragmented, reflecting the status of student volunteering within higher education institutions" (p. 114).

There are different types of voluntary activity that students can get involved with: mutual aid or self-help (where "people with shared problems work together to [...] address them" (Rochester et al. 2010, p. 24-25)), philanthropy and service to others (where an organisation "recruits volunteers to provide a service" to others (Rochester et al. 2010, p. 25)), participation ("the involvement on a voluntary basis in the political or decision-making process" (Rochester et al. 2010, p. 25)), and advocacy or campaigning ("collective action aimed at securing or preventing change" (Rochester et al. 2010, p. 25)). Student-led food gardens could fall under any of these types of activities depending on the aims of the initiative and the activities it undertakes.

Because of their transient nature, students often engage in volunteering episodically. Episodic volunteering is on the rise, and it is "made up of separate, especially loosely connected episodes; of or limited in duration or significance to a particular episode, that is, temporary; occurring, appearing, or changing at usual irregular intervals, that is, occasionally" (MacDuff, 2005, p. 52). Of those that take part in episodic volunteering, some are temporary volunteers who take part for a limited duration of time (usually a period of hours). Interim volunteers take part "on a regular basis for less than six months" (MacDuff, 2005, p. 52). Occasional episodic volunteers take part in the voluntary activity "at regular intervals for short periods of time" (MacDuff, 2005, p. 53). Student volunteers engage in all three of these types of episodic volunteering. Much of the research on episodic volunteering focuses on how to retain them for longer (i.e., Hyde et al., 2016). In other words, research has been undertaken to understand how to make episodic volunteers into long-term volunteers. However, in some

cases, as in student-led food gardens, volunteers are inevitably transient. It is therefore crucial that we have a better understanding of how to manage issues stemming from a transient volunteer base given that it is an inevitable reality for some initiatives and organisations, including student-led initiatives.

2.4 Sustainable food and food gardens in universities

Many universities are striving to encourage sustainability through their campus environment and food system (such as those documented by Sharp (2002), Baldwin and Chung (2007), Rojas et al. (2007), and Barlett (2011)). Others are taking advantage of the opportunities to link these endeavours to the formal curriculum (such as those documented by Rojas et al. (2007), Babich and Smith (2010), Cohen (2010), and Bacon et al. (2011)). For example, Sipos et al. (2008) found that a service-learning project, in a course called 'Edibility and awareness: sustainable food systems,' contributed to transformative learning about sustainable food through reflective practice and engagement of head, hands, and heart. Other activities that can contribute to these transformations in university settings are encouraging sustainable food habits though promotional material and demonstrating a commitment to socially and environmentally ethical procurement. Institutional commitments to sustainable procurement can make a sizable impact on the "conventional food chain" as well, since universities are such large consumers (Bartlett, 2011, p. 101, 111). Sustainability leadership in higher education therefore has the ability to contribute to change in the food system through fostering cultural transformations and "[reshaping] relations between food and place" (Barlett 2011, p. 101). This said, universities are a heterogeneous group of institutions with varied funding streams (i.e., public, private), foci (i.e., research, education, extension/outreach), socioeconomic statuses of students, and sizes. These factors affect how sustainable food initiatives manifest at different institutions.

Part of the more recent growth in university-based food growing initiatives can be explained by a budding interest in enhancing the sustainability of universities and the potential of these garden projects to influence the sustainability attitudes/behaviours of their students, society's future leaders (McKinne and Halfacre, 2008; Johnston et al., 2012). My research focuses on student-led food gardens in universities. I will explain my rationale for this in the remainder of this and the following section, however, first I will briefly describe what I mean by 'student-led food gardens.' Student-led food gardens are places where groups of university students manage and operate spaces on university campuses where edible plants are grown. The students can be at any level of education, and there can also be non-student participants as long as they aren't 'leading' it *per se*. When I use the term 'garden,' I refer to both the physical space where the growing takes place, as well as to the social and organisational structures that facilitate the management of the space.

University food gardens, including student-led ones, are often referred to as 'community gardens' or 'allotment gardens' in both academic literature (e.g., Mycock et al., 2010; Somerset et al., 2010) and by students and other stakeholders involved with them. Community gardens are "open spaces managed and operated by members of the local community in which food and/or flowers are cultivated," (Guitart et al., 2012, p. 364). Allotment gardens, on the other hand, tend to be made up of plots rented or used by individual people or families. The foci of this study are communally run gardens, although one of the gardens studied was called an allotment by the students involved in spite of being run communally.

Some universities are also host to larger-scale food growing initiatives and initiatives that are run by paid staff. Student-led food gardens differ from these models, typically used by agricultural universities and land grant institutions, in that they don't have the level of institutional support and integration into university activities. Initiatives run by non-students

will not be the focus of this study given that my research question centres around managing the impacts of transience, and students tend to be more transient than, for example, university staff.

Community gardens have been linked to building community, making fresh food accessible, improving health, providing economic benefits to their users, and contributing to self-efficacy (the belief in one's capabilities to take action on a given problem) (Pierce & Seals, 2006; Teig et al., 2009; Evans et al., 2012; Guitart et al., 2012). Some of these impacts have direct sustainability implications (like growing food sustainably) and other sustainability impacts are more indirect (like education and community networking) (Evers & Hodgson, 2011). In any case, community gardens are multifunctional in that the benefits they provide are manifold (Valley & Wittman, 2018). There is some evidence university student-led food gardens may provide similar benefits, however the literature is limited and there has yet to be any kind of systematic assessment of these benefits.

In addition to providing these potential benefits, student-led food gardens could be a 'sandbox' where students can explore their role in activism, and how food growing might be a part of their life in future. Importantly, university is also the first time many students have control over their diet, food shopping, and other food practices. Engaging with discussions around sustainable food and developing sustainable habits at this malleable stage could impact food behaviours for the rest of their lives (Verplanken et al. 2008, Thompson et al. 2011, van Niekerk & Barnard 2011). However, this subject is yet to be well-explored empirically and specifically. If these gardens are to provide the benefit outlined here, then it is necessary to understand how to sustain the gardens over the long-term. It is therefore crucial to understand how to manage issues caused by students' transience.

2.4.1 The transience of volunteers in student-led food gardens in universities

Gardens run by university students are faced with a transient membership, resulting in short term participation (when people are involved for limited periods of time) and irregular participation (when involvement is inconsistent or non-regular). Short-term and irregular participation can also result in low participation, which is when there are limited volunteer hours invested overall (Figure 1). In this thesis, I will refer to short-term, irregular, and low participation collectively as 'problematic participation' because they are forms of participation that result in problems (although are not problems in and of themselves). However, it is likely that there are also other causes of problematic participation, in addition

Problematic participation	
Low participati	ion
Irregular participation	
Short-term participation	\square_+

Figure 1. The three forms of problematic participation in student-led food gardens: irregular, short-term, and low participation. The causal relationships are depicted using arrows to indicate the direction of the relationship, and pluses (+) to indicate that an increase in the first variable will lead to an increase in the second variable.



Figure 2. The relationship between problematic participation, transience, its other causes and effects. The causal relationships are depicted using arrows to indicate the direction of the relationship, and pluses (+) to indicate that an increase in the first variable will lead to an increase in the second variable. The greyed section of the diagram represents what are thought to be included in the impacts of students' transience.

to transience. These relationships are illustrated in Figure 2 and will be built upon in Chapter

6.

Throughout this thesis I will also be using two related terms: transience and turnover. Transience refers to movement in and out of the university (over the course of degrees and during the academic calendar because of holiday times), and turnover refers to movement in and out of the student-led food gardens, which can be accelerated by short-term participation. The terms transience and turnover are often used synonymously in common language, but I draw this distinction between the two to talk about these two different scales of movement (see Figure 3 for an illustration of this). Students' transience in and out of the university community causes high turnover of participants of student-led food gardens. As such, each 'generation' of students within the garden is short-lived compared to volunteerrun gardens based in less transient communities. The set-up of student-led food gardens is also counter-intuitive because students are absent during a long vacation period during the summer and therefore the gardens risk being neglected at the time when they are in most need of care. Furthermore, the summer is the time where students would get greatest benefit from accessing the food produced. Variability in student volunteering during the academic year due to uneven academic course loads, other holidays, and vocational placement requirements is also common.



Figure 3. A depiction of how transience refers to movement *in and out of the university*, and turnover refers to movement *in and out of the garden*.

Most students could be characterised as episodic volunteers. High rates of volunteer turnover can cause issues with "organisational effectiveness and efficiency" (Starnes & Wymer, 2001, p. 98), and this is compounded when volunteers need special skills, intensive training, to be committed over the long term, and/or there are not enough qualified volunteers (Fischer & Schaffer, 1993). Students' turnover in university gardens means there is a constant need to generate and maintain interest in the garden to ensure there is "adequate caretaker presence" (McKinne & Halfacre, 2007, p. 153). Adrangi (2013) also found that students can also unintentionally neglect work by other local groups on similar projects (such as by initiating new projects (like community gardens, food bank donations) without realising such projects have already been lobbied for or run by the university or within the local community). Such duplication of efforts thereby can create tensions between non-students and students. Furthermore, the neglect of existing work also deprives students and their projects of "a pool of knowledge and mentors, exposure to new ways of doing things and experiences, pooled human and material resources, and staying power that often exceeds" their own (Adrangi, 2013). Much of the research on short-term volunteering focuses on how to retain volunteers for longer (i.e., work by Hyde et al. (2016)). However, the transient nature of student life means that such comings and goings are inevitable. Therefore, understanding how short-term involvement can be best managed within the context of the student led garden is important. However, the interplay between forms of problematic participation, its causes and effects, and transience have yet to be investigated in any kind of systemic way in student-led food gardens, or even sustainable food organisations more broadly. This is particularly important to understand given that, in society at large, transience is increasing (MacDuff, 2005) and self-organising sustainability movements are strengthening (e.g., Extinction Rebellion, 2019; Wahlström et al., 2019). Furthermore, given student-led food gardens' capacity to facilitate learning about, and behaviour and attitude change for,

sustainability it is important to understand how to enhance the resilience of student-led food gardens in the face of transience.

2.5 Temporary organisations and transience

While student-led food gardens are not temporary organisations because they are typically intended to last over the long-term, organisations with transient participants share much in common with them because of their short-term membership. There is much discussion about what temporary organisations are, but the general consensus is that they are organisations that are time-bounded (Bakker, 2010). While some temporary organisations may be truly ephemeral, and disperse as soon their task is complete, often temporary organisations are connected to a more permanent organisation operating over a longer period of time (Bakker, 2010). Furthermore, even if a temporary organisation is not embedded within a more permanent organisation in some capacity, they are still embedded in "enduring personal networks, epistemic communities, and industries in which their participants embedded" (Bakker, 2010, p. 480). In other words, regardless of whether a temporary organisation is embedded within a more formal and long-standing organisation, they are usually still embedded within wider socio-cultural milieu. Three ways in which temporary organisations might share something in common with organisations with transient participants (like student-led food gardens) are in terms of their approach to knowledge, risk, and time (see Table 1). Temporary organisations tend to have linear understandings of time and problems with knowledge retention, but are more comfortable with risk. Permanent organisations tend to have cyclical understandings of time and better-established knowledge management practices, but are more risk adverse. It is not yet clear what approaches organisations with transient participants have to knowledge, risk, or time. With a better understanding of these

approaches, their strengths, and weaknesses, stakeholders of transient organisations could

self-organise more strategically.

Brookes et al. (2017).		
	Temporary organisations	Permanent organisations
Approach to knowledge	 "Excellent domain for context-specific knowledge-creation" (Brookes et al., 2017, p. 1217) Knowledge retention is a problem The orientation towards knowledge is exploratory 	 Reduced reflexivity More established knowledge management practices The orientation towards knowledge is exploitative
Approach to risk	 Changes that result in risk are viewed positively as they present opportunities Undertakes high-risk activity Lacks temporal consistency and therefore is riskier 	 Changes that result in risk are viewed negatively as they are a threat to the longevity of the organisation Undertakes low-risk activity Has temporal consistency and therefore has less risk
Approach to time	• Linear, where tasks are orientated to "fulfil a one-off mission" (Ibert, 2004, p. 1530)	• Cyclical, where "routines are established to deal with constantly (seasonally, monthly, daily, etc.) reoccurring tasks") (Ibert, 2004, p. 1530)

Table 1. The differences between how temporary organisations and permanent organisations approach knowledge and risk. Information included comes from Ibert (2004), Bakker (2010), and Brookes et al. (2017).

2.6 Conclusion

Student-led food gardens have the potential to contribute to a range of benefits, not least to sustainability education, and attitude and behaviour change. However, this potential may not be fulfilled if these initiatives are unable to persist in the face of a transient volunteer base. There is, therefore, a need to better understand how student transience impacts student-led initiatives in universities and how transience and its impacts can be managed to ensure the longevity of these projects and the quality of their outcomes.

These questions are of particular note given the emerging interest in the UK, as well as abroad, about how to use university campuses as living laboratories for developing sustainability practices and learning. This study of student-led food gardens will contribute to this emerging body of research. In particular, it can contribute to understanding how studentled activities can be embedded in living laboratories in a way that genuinely benefits all stakeholders. The examination of transience in student-led food gardens also has implications beyond the university garden plots. Studying transience within student-led food gardens may contribute to understanding how to manage of other self-organising groups with transient participants, regardless of whether their transience is inevitable or preventable. Having described the literature this thesis is engaging with and will be contributing to, I will now move on to outlining my theoretical framework.

Framing through social-ecological resilience

We can't control systems or figure them out. But we can dance with them! Donella Meadows, Dancing with Systems, 2001



Vegetables on display at a food festival Spring 2015

3.1 Introduction

My theoretical framework is rooted in a systems understanding of social and environmental phenomena. Layered on top of this, I draw on scholarship in social-ecological resilience to explain and analyse both how student-led food gardens fit into the wider landscape of change towards a sustainable food system, and the capacity of student-led food gardens to withstand the ongoing changes in their volunteer base and how to increase that capacity. This chapter will present these theoretical perspectives and their key tenets, but before doing so, I will position myself ontologically and epistemologically.

3.2 Ontological and epistemological considerations

My ontological and epistemological orientations are rooted in critical realism. Some social researchers conflate ontology and epistemology (Crotty, 1998), however, critical realists draw an important distinction between ontology, theories of what exists and the nature of existence, and epistemology, theories of knowledge or how we know what we know (Nastar et al., 2018). Epistemological considerations have been at the forefront of my research throughout as I have adopted a research approach with a high degree of reflexivity, which will be outlined in greater detail in Chapter 4.

Unlike interpretivist ontologies, where reality is understood to be socially constructed, a critical realist ontology is similar to positivist ontology in that there is understood to be a reality that exists independent from human understanding of it (Coghlan & Brannick, 2005). However, unlike positivists, critical realists understand there to be different 'domains' of reality (which Bhaskar (2008) calls 'empirical', 'actual', and 'real' domains), and that it is only the 'empirical' domain which can be observed through scientific inquiry. The 'actual' and 'real' domains exist prior to human cognition. The 'real' domain is made up of "objects, their

structures or natures and their causal powers and liabilities," and this includes not only natural things (i.e., gravity) but social things (i.e., human agency) as well (Fairclough et al., 2002). The 'actual' domain is made up of "what happens when [the] powers and liabilities [from the 'real domain] are activated and produce change" (Fairclough et al., 2002). A visualisation of this can be seen in Figure 4. In short, critical realists believe the world is made of very real, however, unobservable things and events, and these can cause events that are observable by scientists (or non-scientists, for that matter). However, just because the events are observable, it does not mean that they are actually observed (scientifically or otherwise).

Empirical
Observable through scientific inquiry
(i.e., experiences and perceptions)
Actual
Events that result from the activities in the 'real' domain
(i.e., events and actions)
Real
Things that have causal powers
(i.e., structures, mechanisms, tendencies, powers, rules, institutions, conventions)
Figure 4. A stratified ontology of critical realism (based on Fleetwood, 2014).

Epistemologically critical realists share more in common with interpretivist traditions. Positivists believe that, through observation and the use of the scientific method we can understand natural and social phenomena. However, like interpretivists, critical realists believe "we 'produce' our experience in the form of our narratives and concepts" (Bhaskar, 1986, summarised in Winter & Munn-Giddings, 2001, p. 259). In this way, critical realists are similar to interpretivists in that they do not think there can be a "one-to-one correlation between knowledge claims and reality" (Koutsouris, 2010, p. 255).

Another way in which critical realism departs from positivism epistemologically is that the means to understand social and natural phenomena are different. Unlike natural phenomena, which are much less relative, society and culture are dynamic and constantly in flux, and as such, critical realists do not strive for universal 'laws' or 'rules' that govern social systems. Instead, understanding of the context in which social phenomena take place is paramount.

The benefit of an ontological and epistemological stance rooted in critical realism is that it can marry natural and social sciences (Nastar et al., 2018), and qualitative and quantitative approaches without resorting to "paradigm 'switching'" (McEvoy & Richards, 2006, p. 66). This is particularly relevant in sustainability science, where the natural world (typically observed through quantitative research) affects and is affected by people (often observed through qualitative research), regardless of human understandings of it. For example, it is clear that our food system is contributing to climate change in a substantial way, regardless if we know it or not. Through an interpretivist lens, it is more challenging to make a consistent argument that this can be taken as a fact, given the espoused socially constructed nature of reality. However, a critical realist perspective on social phenomena means that, like in interpretivism, understandings of these are still understood to be negotiated and relative.

Research operating under a critical realist ontology and epistemology typically does not focus on producing predictive knowledge, either through induction (building theory from observation) or deduction (testing theory through observation). Rather, it employs retroduction, that is, beginning with an observable phenomenon and then generating possible explanations (Belfrage & Hauf, 2017). Critical realism embodies many concepts that are compatible with systems thinking, such as emergence, open systems, stratification and holistic causality (Mingers, 2015). Building on the critical realist ontology and epistemology outlined here, I will go on to root my research in systems theory in this next section.

3.3 Rooting in systems thinking

Sustainability science is firmly rooted in systems theory, and therefore I have drawn on this area of scholarship and on theories that share these roots for my theoretical framework. Systems theory is based on the premise that the world is made up of interacting systems (including inorganic, organic and socio-cultural systems) with emergent properties⁴. In other words, a system is "a perceived whole whose elements are 'interconnected'" (Ison, 2008, p. 140). Systems theorists contend that systems go through cycles of degeneration and renewal and that these cycles operate on different hierarchical levels which are also interconnected. As a result, actions that are taken can have counterintuitive effects and unintended consequences.

For systems theorists,

"The world is not a neat stratigraphic map beginning with inorganic matter, passing to organic matter, and then being transcended by socio-cultural forces. Rather, the world is a complex, interacting array of systems and system processes, bumping into each other in a variety of ways. Social relationships and processes are impacted by the physical world as the physical world is transformed by social activity."

(Greenwood & Levin, 2008, p. 58)

Sustainability science holds that the relationship between the human and natural world are dynamic and "operate within material boundaries and are capable of transforming material living conditions" (Folke et al., 2002; Greenwood & Levin, 2008, p. 59).

There are different schools of thought within systems theory (Ison, 2008). The starting point of systems theory was challenging reductionist thinking in diverse disciplines, like

⁴ Emergence is a "property of a complex system that emerges from interactions of subunits and cannot be understood or predicted by studying individual subunits" (Chapin et al., 2009, p. 345).

biology, mathematics, and philosophy, postulating that it is not enough to study elements of a system in isolation to understand the whole. Cybernetics is a stream of this theory which brought forward the ideas of cyclicity and feedback loops – or, the idea that the output of a system is reused as input to the same system. There are two types of feedback loops: positive or reinforcing feedback loops amplify the changes in a system (the changes amplified can been good or bad changes), and negative or balancing feedback loops dampen changes in, or stabilise, the system (Senge, 2006). These feedback processes can also have delays embedded within them, so consequences can be unexpected or counterintuitive (Senge, 2006). In first-order cybernetics, the interest was not just on the dynamics of these loops, but also the ability to control systems through them. Second-order cybernetics brought a useful epistemological perspective for sustainability science because it postulates that "knowledge is not something we have but arises in social relations such that all knowing is doing" (Ison, 2008, p. 151). This means that, studying complex systems is not objectively collecting knowledge bits, but coming to know a system through social learning.

Complexity sciences, another stream of systems theory, are diverse and not always consistent, but have also brought useful concepts to the table. The main driving idea of complexity is that parts of systems are linked up and interact in multiple, and often unpredictable ways. Complexity sciences have been influential in sustainability science through the conceptualisation of social and ecological systems as linked complex adaptive systems (social-ecological systems) (Folke et al., 2002) Complex adaptive systems are systems "of interconnected components characterized by emergent behaviour, self-organization, adaptation, and substantial uncertainties about system behaviour" (Biggs et al., 2012, p. 425). Although the study of social-ecological systems is a relatively new area of scholarship in systems theory, it has been influential within and beyond the sustainability community (Colding & Barthel, 2019). Social-ecological systems are ecological systems "intricately linked
with and affected by one or more social systems. An ecological system can loosely be defined as an interdependent system of organisms or biological units. 'Social' simply means 'tending to form cooperative and interdependent relationships with others of one's kinds'" (Anderies et al., 2004). These social and ecological systems "contain units that interact interdependently and each may contain interactive subsystems as well" (Anderies et al., 2004).

The idea of 'wicked problems' from the complexity sciences have also been influential in social-ecological systems research (Rogers et al., 2013), sustainability research (Frame, 2008) and sustainability education (Ramaley, 2014). 'Wicked' problems are problems that are difficult to define and there isn't consensus about how to move forward from them because they are characterised by uncertainty and contradiction (Frame, 2008). The global sustainability challenge itself has been called a 'wicked problem' (Blok et al., 2016), as have many sustainability challenges that are more niche, such as wildfire management (Chapin et al., 2008), disappearing coral reefs (Hughes et al., 2013), and food insecurity (Hamann et al., 2011).

Another part of systems theory scholarship that has relevance for my thesis is the idea that there are different places to "intervene in a system" to leverage change to a greater or lesser extent (Meadows, 1999, p. 2). In her seminal work, Meadows (1999) outlined twelve 'leverage points,' starting with shallow leverage points (ones that are easy to leverage but have less of an effect on the system) and moving towards deep leverage points (places in the system that are much harder to change but have a much more substantial effect on the system). The shallow leverage points include things changing system 'parameters' (i.e., changing the amount of land set aside for conservation, minimum wage, taxes) and changing "the size of buffers and other stabilising stocks" (i.e., the number of substitute teachers oncall, the amount of water in a dam). Deep leverage points focus on changing "the

underpinning values, goals, and worldviews of actors that shape the emergent direction to which a system is orientated" (Abson et al., 2017, p. 32). Much of sustainability research and policy has focused on shallow leverage points, and Abson et al. (2017, p. 32) have argued that there needs to be more focus on leveraging change at a deeper level. Sustainability education is a deep leverage point that has a crucial role in changing "the mindset of paradigm out of which the system arises" in order to leverage that change.

In summary, thinking of my research as being underpinned by systems theory means that I understand the issues I have studied as complex, and with social and ecological drivers "under conditions of uncertainty and plurality of values and perspectives" (Popa et al., 2015, p. 2). I consider my research to be transdisciplinary as I (1) collaborated with research participants in setting the research agenda and taking action (Brandt et al., 2013), and (2) drew on "perspectives shared by a wide array of scientists and social reformers of diverse backgrounds and divergent political ideas," rather than being rooted in a "single discipline anchored in [a particular academic department]" (Greenwood & Levin, 2008, p. 57). Understanding student-led food gardens through the lens of systems theory challenges the "radical individualism" that dominates understandings of social behaviour, purporting that the complexities of social and natural systems require a more holistic, open, and democratic theoretical underpinning (Greenwood & Levin, 2008, p. 59). For this reason, I examine each student-led food garden as not only a collection of individual students, but also as a group with synergistic properties, embedded within the wider ecology of a university and society at large.

My research also unfolded in unexpected ways and therefore I found myself adapting my focus and methodology as I went along. Systems theory was a useful theory in this respect because it encourages flexible methods for redefining 'the problem' throughout the research project. Systems theory provided some crucial underpinnings for my research, however a

'resilience lens' helped to operationalise some of these ideas for my data analysis. I will outline what I mean by resilience in the following section.

3.4 Social-ecological resilience

Resilience is a term used in a variety of different fields, such as ecology, organisational studies, psychology, and urban studies, to name a few. The concept has gained significant traction in the field of sustainability, and this school of thought has emerged out of the original work by Holling and Gunderson (Holling, 1973; Gunderson, 2000; Gunderson & Holling, 2002). This early work focused on ecological resilience, or "the capacity for ecosystems to absorb change" (Folke, 2016, p. 4). Since then, the theory has increasingly been understood in much broader terms and has been drawn on in social science and interdisciplinary arenas within the field, with the development of new concepts like 'social resilience' (Adger, 2000), 'social-ecological resilience' (Adger, 2005; Folke, 2006; Gunderson, 2010), 'community resilience' (Berkes & Ross, 2013), 'economic resilience' (Simmie & Martin, 2010), 'urban resilience' (Meerow et al., 2016), and organisational resilience (Seville et al., 2008, Burnard & Bhamra, 2011; Butler, 2018). Many of these terms are not mutually exclusive, and the literature about each of the concepts is quite heterogeneous. I will not outline definitions used in different contexts and disciplines as this has already been done by Quinlan et al. (2016), Meerow et al. (2016), and Brand and Jax (2007), but rather I will explain several alternatives that could have been appropriate to this study, and why I have chosen to work with social-ecological resilience.

Social resilience is "the ability of communities to withstand external shocks, mitigate and recover from hazards" (Adger, 2000; Keck & Sakdapolrak, 2013; Quinlan et al., 2016, p. 125). Some parts of this scholarship tend to focus on resilience at a large scale, and therefore suggested indicators for social resilience include those related to demographic change and income distribution, for example. The scale at which resilience is being applied in this study is smaller and more niche (that is, to student-led food gardens), rendering such strategies (Adger, 2000) for observing social resilience unusable. There is also yet to be coherence within the literature with regards to how one might try to increase the social resilience of a group or system in a more pragmatic sense, though there are some more pragmatic contributions, like Marshall and Marshall (2007), and Obrist et al. (2010).

Organisational resilience "resides in both the individual and organisational responses to turbulence and discontinuities" involving "both the ability to withstand systematic discontinuities as well as the capability to adapt to new risk environments" (Burnard & Bhamra, 2011, p. 5583). The literature on organisational resilience tends to focus on how organisations cope with an acute crisis event (Seville et al., 2008, Stephenson et al., 2010; Burnard & Bhamra, 2011) such as a natural disasters, although some attempts have been made to explore organisational resilience during "normal operational activities" (Butler, 2018, p. 103). However, most established framings of organisational resilience are built around resilience as a response to an acute crisis event (e.g., Burnard & Bhamra, 2011). As such, organisational resilience was a less useful framing for this study.

Community resilience is defined as "a process of adaptation in a community following a disruption, distinguished by factors such as social capital and community competencies" (Cretney, 2014, p.629). The concept suffers from similar weaknesses as social resilience in that there is a considerable lack of coherence within the literature about how to apply the concept prescriptively. Recent interpretations of community resilience have drawn on both the literature on psychological resilience and social-ecological resilience, in an effort to integrate the two (Berkes & Ross, 2013). This would be the most appropriate approach for this study, given that student-led food gardens (which could be considered 'communities') are primarily social entities that interface with ecological systems rather than resource-

dependent "communities of place" (Berkes & Ross, 2013, p. 16). However, this work is still underdeveloped, particularly for use in the prescriptive sense that would be necessary for this study.

In this study, I draw on the scholarship of social-ecological resilience given that the student-led food gardens in this study are underpinned by social aims, such as increasing prosustainability changes in food attitudes and behaviours, as well as environmental aims, such as to produce food sustainably⁵. Social-ecological resilience is the only school of thought within the scholarship of resilience which has a coherent explanation of the relationship between social and ecological systems (unlike social, community, or organisational resilience). However, the scholarship on social-ecological resilience draws on work from these other schools of thought. Previous research on community food growing has also seen the merit of, and therefore adopted, social-ecological framing of resilience (Tidball & Krasny, 2007; Krasny & Tidball, 2009; Barthel et al., 2010). Social-ecological resilience is defined as "the capacity of a system to absorb disturbance and reorganise while undergoing change so as to still retain essentially the same function, structure and feedbacks and therefore identity, that is, the capacity to change in order to maintain the same identity" (Folke, 2016, p. 8). Vulnerability is the antonym of resilience: the lack of a capacity of a system to withstand change. In recent years, there have been influential attempts to synthesise existing research into some generic principles for building social-ecological resilience in the context of sustaining ecosystem services as the concept of social-ecological resilience has matured (Chapin et al., 2009; Biggs et al., 2012, 2015c).

A key concept in resilience is the idea of 'basins of attraction' (Walker et al., 2004). A basin of attraction is an equilibrium state (sometimes called an 'attractor') where a system

⁵ The aims of the gardens studied will be outlined in more detail in Chapter 4 when introducing the case study.

tends to remain (illustrated in Figure 5 as the darker area towards the bottom of the basin). Walker et al. (2004) state that "all real-world [socio-ecological systems] are[...] continuously buffeted by disturbances, stochasticity, and decisions of actors that tend to move the system off the attractor. Therefore, we think of [socio-ecological systems] as moving about within a particular basin of attraction, rather than tending directly toward an attractor" (p. 5). In this way, the black dot in the basin (Figure 5) could be visualised as a ball rolling around in the basin. A system rolling up to the rim of a basin would be a situation in which the resilience of the system is being tested, and the ball sitting at the bottom of the basin would be a system at its equilibrium state. A system with a deep basin would be considered a more resilient system because the walls of the basin are steep, making it hard for the system to move out of the basin of attraction. However, if the ball reaches the rim of the basin because it had a shallower basin that experienced a moderate perturbation (a less resilient system), or it had a deeper basin which experienced a bigger perturbation (a more resilient system), it may tip into a new basin of attraction. This 'crossing of a threshold' causes what resilience scholars call a 'transformation,' which is when a system becomes "a fundamentally new system when ecological, economic, or social structures make the existing system untenable" (Walker et al., 2004, p. 5).



Figure 5. Basins of attraction (Walker et al., 2004).

As can be seen in the above section, resilience can be quite an abstract theory, and as a result has been applied in a range of contexts and interpreted in different ways. For this reason, I adopt a critical approach to socio-ecological resilience. Clarifying a critical approach to socio-ecological resilience is necessary given the wide use of the concept which has been used to ends as diverse as "perpetuating hegemonic values and discourses" and spearheading "a more countercultural form of activism" (Cretney, 2014, p. 631). Given the ecological roots of resilience, it's initial use in the social sciences was predicated on the false assumption that ecological and social systems operate on similar principles (Cote & Nightingale, 2012; Cretney, 2014). Closely related to this is the lack of consideration within resilience scholarship of important social constructs, like power, politics and culture. The absence of these constructs from the theory is what, some have argued, has contributed to the perpetuating of hegemonic values and discourses (Jerneck & Olsson, 2008; MacKinnon & Derickson, 2013). I actively take these aspects into account throughout my study.

In writing about *resilience* in the context of *sustainability*, I also want to be clear about what the differences between these two concepts are and how they can be used. This should not only enhance the clarity in what I mean when I use these concepts, but it should also help to address the questions of what I am suggesting should be resilient, and for whom the resilience should serve (Cote & Nightengale, 2012). Sustainability is a normative term that focuses on equity of access to resources and opportunity, at the present and for future generations (see section 2.2). In short, it is a desirable future vision. Whether resilience is normative or non-normative is more contested (Brand & Jax, 2007; Cote & Nightengale, 2012; Keessen et al., 2013). I choose to treat resilience as a "property of a complex system" that is inherently non-normative as much of the scholarship in social-ecological resilience does (Cretney, 2014; Elmqvist, 2017, p. 352). As such, a system can be resilient in a non-desirable condition. For example, dictatorships can be "resilient across generations" (Elmqvist, 2017, p.

352). Furthermore, resilience and sustainability aims can be at odds with one another. For example, highly efficient energy systems can be more sustainable because less energy is wasted, however, they may be less resilient to changes, such as changes to energy use patterns or unexpected increases in energy use.

3.4.1 Principles for building resilience

Resilience scholars have proposed seven principles for building resilience for sustaining ecosystem services in social-ecological systems (Biggs et al., 2012; Biggs et al., 2015c). The principles are to:

- 1. Maintain diversity and redundancy,
- 2. Manage connectivity,
- 3. Manage slow variables and feedbacks,
- 4. Foster an understanding of social-ecological systems as complex adaptive systems,
- 5. Encourage learning and experimentation,
- 6. Broaden participation, and
- 7. Promote polycentric governance systems.

Although the principles are new, many of the ideas within them represent already existing suggestions from various resilience scholars (Chapin et al., 2009; Cilliers et al., 2013). These principles have already been influential in sustainability and resilience scholarship (Carpenter et al., 2012; Folke et al., 2016; Quinlan et al., 2016), including in research that is considered to be seminal (e.g., Steffen et al., 2015). The principles are also being taught in a Massive Open Online Course run through the Sustainable Development Goals Academy, and has been integrated into 'Wayfinder,' a "resilience guide for navigating towards sustainable futures" (Enfors-Kautsky et al., 2018; Sustainable Development Solutions Network, 2018).

Before I go on to explain the seven principles, I will briefly explain what ecosystem services are, and why I still choose to apply the principles to the resilience of student-led food gardens (which are not ecosystem services). Ecosystem services are the "ecosystem goods (such as food) and services (such as waste assimilation) [that] represent the benefits human populations derive, directly or indirectly, from ecosystem functions" (Costanza et al., 1997, p. 253). Or, put more simply, they are the "benefits that people obtain through their interaction with nature" (Biggs et al., 2015b, p. 13). While student-led food gardens could be said to provide ecosystem services, such as food production, recreation, pollination, or cultural services, the focus of this study is not on how to sustain these, but how to sustain a studentled organisation itself. In this way, I take a normative view of student-led food gardens in which I assume that these initiatives provide important sustainability outcomes which have been outlined in Chapter 2 and will be explored systematically in Chapter 5.

Biggs et al. (2015b) limit the resilience principles to sustaining ecosystem services because, firstly, "the features of society that promote resilience [...] may differ from those that promote resilience of ecosystem services," and secondly, to make the scope of their book "manageable" (Biggs et al., 2015b, p. 17). Biggs et al. (2015b) say that ecosystem services are outcomes not only of ecosystems, but also of social-ecological interactions, citing crop production as an example. Crops require both the ecological conditions, like temperature and rainfall, as well as the social conditions, like skills and technologies, for their provision. However, Biggs et al. (2015b, p. 17) said that they "fully acknowledge that there are other crucial social-ecological systems outcomes of importance to society, such as women's rights or education." If the interactions between social systems and ecological systems through social-ecological systems are as in Figure 6, the focus that the resilience principles have when focusing on ecosystem services have mainly to do with the ecological and socialecological feedbacks, and much less to do with the social feedbacks. It was reasonable to limit



Figure 6. A model of how social and ecological systems interact within a social-ecological system. As Biggs et al. (2015b, p. 8) said, "in the resilience approach, SES are not simply seen as social plus ecological systems. Rather they are viewed as systems centred on the feedbacks between ecological (grey) and social (white) system components, which lie at the interface of social and ecological systems."

the scope of the principles within Biggs et al.'s (2012, 2015c) original work, however, I believe

continuing to limit the principles to ecosystem services is doing them a disservice because

they can have much broader application.

As such, I intend to apply these principles to assess the resilience of student-led food gardens as social-ecological systems. This differs from the original intended use of the principles in two ways. Firstly, the focus will be on the social-ecological feedbacks and the social feedbacks, shifting the emphasis to the social side of the system. Secondly, I will be applying the principles to the resilience of student-led gardens as organisations in a socialecological system, rather than ecosystem services within a social-ecological system. It is therefore my intention to use the principles both to help answer my research questions, as well as to problematise the use of the resilience principles outside of their original intended context.

This is not a far stretch given that social-ecological resilience has been applied to social phenomena outside of the ecosystems services lens, albeit this literature is still underdeveloped compared to the more ecologically-orientated literature about resilience. For example, each year between 2007 and 2017, Stockholm Resilience Centre, one of the most prolific scholarly centres working with resilience in the context of sustainability, has produced more natural science publications than social science and interdisciplinary publications combined (Stockholm Resilience Centre, n.d.). Indeed, resilience has been criticised for attempting to take a concept originally from the natural sciences and applying it in a social science context without sufficient reflection on how the application might be different in a social context from an ecological one (Arora-Jonsson, 2016). However, Biggs et al. (2015c) have used examples of feedbacks of social, ecological, and social-ecological natures to illustrate their points, demonstrating that these principles may well apply to social phenomena as well, or at least some. By applying concepts from social-ecological resilience, and specifically the resilience principles, in the context of a socially-focused study I will extend this body of research and consider further how the concepts and principles may need adapting for social systems.

Three of these principles have to do with the general properties of socio-ecological systems (SES) that should be managed, and four are attributes of governance systems of social-ecological systems that need to be managed. The first three principles are quite abstract, and therefore I will be presenting examples of what they might look like in a school.

3.4.1.1 Resilience Principle 1: Maintain diversity and redundancy

The first principle is to maintain *diversity* and *redundancy*. Diversity is made up of three concepts: "variety (how many different elements), balance (how many of each element), and disparity (how different the elements are from one another)" (Stirling, 2007; as cited in Biggs, 2012, p. 245). Diversity is important to build the resilience of systems because

it "provides options for responding to change and disturbance" (Kotschy et al., 2015, p. 50). The principle to maintain diversity applies to both social and ecological system components. In the case of a school, this might include a diversity of staff competencies for teaching different subjects, or diversity in the landscape of the schoolyard, for example. In the case of diversity in the landscape of the schoolyard, the variety would be the number of different types of landscape features (e.g., playground, football pitch, treed area). The balance could then be the balance between, for example, the number of teaching vs support staff. The disparity would then be the how different the competencies are, for example, are all the competencies related to science and mathematics (such as biology, chemistry, and physics), or is humanities competence also present (such as languages and history)?

Redundancy is the "replication of elements or pathways in a system" (Kotschy et al., 2015, p. 53) and is important for the resilience of a system because, in the event of a loss or failure of a system component, another can compensate. However, it is natural for redundancy to be minimised because it reduces efficiency of the system. In the case of a school, redundancy might include having several teachers with overlapping science-based competencies so that if one teacher falls sick, the someone else can cover for them. Another example would be having two football pitches to enhance the flexibility of the system, so that if a group of younger students was already using one pitch, older children could still use the other one rather than having to wait.

With too little diversity and redundancy, social-ecological systems can become 'brittle' (rather than flexible or adaptive), and when exposed to stressors are less able to 'bounce back,' and can result in the system undergoing a transformation into a new state. In the school example, if there is only one art teacher, and there are no other teachers with artistic competencies this may make the system brittle. If the teacher becomes sick with a long-term

illness and there is no one to run the art classes, the school may need to transform their programme offering to account for the fact that they can no longer teach art.

3.4.1.2 Resilience Principle 2: Manage connectivity

The second principle is to manage connectivity within the social-ecological system. Connectivity means "the way and degree to which resources, species, or social actors disperse, migrate, or interact across ecological and social landscapes" (Biggs et al., 2012, p. 247). How connectivity manifests within social-ecological systems is affected by the "intensity with which components are connected," or the *strength* of the connectivity, and the "presence or absence of links between components and how links are distributed" (Biggs et al., 2012, p. 427-428). Connectivity is important because it "facilitates the exchange of material or information necessary for the functioning of ecological and social processes" in a social-ecological system (Biggs et al., 2012, p. 428). A system that is not well connected will not be able to facilitate these exchanges, resulting in a less resilient system. For example, in a school, if few of the school staff engage with each other, information-sharing may be limited which could reduce the effectiveness of the school's internal governance. These connections can be strengthened or weakened depending, for example, on trust or the degree of reciprocity (Dakos et al., 2015). However, if a system is too interconnected, a disturbance in one part of the system can travel to other parts of the system rapidly. For example, if there is an interpersonal conflict between staff members at a school with high interpersonal connectivity, the conflict could affect the rest of the school more quickly or dramatically than if there were weaker social connections between the staff.

Two key characteristics of social-ecological systems related to connectivity are *modularity*, or the "the extent to which there are subsets of densely connected nodes that are loosely connected to other subsets of nodes," and *nestedness*, or "the degree to which specialist nodes (nodes with few links) interact with subsets of generalist nodes (nodes with a

lot of links)" (Biggs et al., 2012, 428). The difference between modularity and nestedness has to do with scale. Modular connections happen at the same level of the system, for example, the schools within a city could be well-connected modules with loose connections holding the schools together in a wider network. Nested connections happen between different levels of a system. An example would be how a board of education operating at the scale of a city is connected to many individual schools operating at a neighbourhood level. According to Biggs et al. (2012, p. 429), "in highly modular or nested systems, resilience may be jeopardized if some components become overly important compared to others." For example, if there is one teacher in a school who leads many of the school's extracurricular activities and that teacher finds a job elsewhere, many students would be affected.

3.4.1.3 Resilience Principle 3: Manage slow variables and feedbacks

The third principle is to manage slow variables and feedbacks. These slow variables are the background on which other variables depend. A change in the slow variables and feedbacks can cause rapid and unintended consequences to a system if the system crosses a tipping point where the system will be drawn into a new basin of attraction. In the case of community gardens, fast variables might be participation levels and the crops grown, whereas corresponding slow variables might be the land tenure arrangements for the space. When there is a change in a variable, a feedback can occur. Reinforcing feedbacks (often called positive feedbacks) reinforce "subsequent changes of the same type," and balancing feedbacks (often called negative feedbacks) "dampen" them (Biggs et al., 2012, p. 430). An example of a reinforcing feedback loop is a student thinking that they are not smart, and therefore is sloppy with their homework and receives low grade, which then reinforces their belief that they are not smart. An example of a balancing feedback loop is how a school's weekly litter-pick keeps the school grounds tidiness in check. Changes in slow variables and feedbacks can result in nonlinear changes and/or large, permanent changes to the system

such that the system has a different structure and different dynamics - in other words, the defining properties of the system are changed so much that it becomes a different system altogether. Usually these sorts of changes result from "gradual changes in slow variables" combined with an acute "shock" to the system (Biggs et al., 2015a, p. 11).

An example of a situation like this could be when the area around a school becomes increasingly gentrified, therefore driving up land value (a slow variable). After an election (an acute shock), the newly formed local government, incentivised by the high selling price of the land, decide to sell the lot the school's community garden occupies for development. Identifying and monitoring slow variables and feedbacks means that a decline in resilience is more likely to be identified early on. In this case, this might mean checking local real estate websites to monitor the changing housing prices. By monitoring slow variables, work can be done to strengthen stabilizing feedbacks, if possible, and to build resilience into the system in the event of shocks. In this case, this might include establishing a formal land-tenure agreement for the community garden lot with the local government prior to an election.

It may be desirable to increase or decrease the resilience of a system depending on what the desired system-state is. Increasing resilience would be done by strengthening feedbacks that maintain desired regimes, such as the establishment of the land-tenure agreement in the above example. However, if land on the site was contaminated and school wanted to move to another location, they may attempt to weaken or break the feedbacks that keep the garden in their location by approaching the local government and leveraging the rising land prices and development opportunities to convince the government to give them access to a different plot of land. This would be an example of an attempt to decrease the resilience of the system in order to shift into a new system state. In these cases, monitoring of gentrification and land prices (slow variables) can help the school to identify

how they could increase or decrease resilience to move into or maintain a desirable system state.

3.4.1.4 Resilience Principle 4: Foster complex adaptive systems thinking

The fourth principle, the first of the governance-orientated principles, is to foster an understanding of socio-ecological systems as complex, adaptive systems. Given that a complex adaptive system is "a system of interconnected components characterized by emergent behaviour, self-organization, adaptation, and substantial uncertainties about system behaviour" (Biggs et al., 2012, p. 425), then complex adaptive systems thinking is "a mental model for interpreting the world that recognises [complex adaptive system] properties" (Bohensky et al., 2015, p. 143). Shared mental models are "knowledge structures" held by members of a team that enable them to form accurate explanations and expectations for the task, and, in turn, coordinate their actions and adapt their behaviour to demands of the task and other team members" (Cannon-Bowers et al., 1993, p. 221). Bohensky et al. (2015, p. 145) argue that a shared mental model based on complex adaptive systems could offer "insight not only into how managers understand social-ecological systems but also how a manager might act and how he or she perceived the responsiveness of the social ecological system to such management actions." The crucial part of this complex adaptive systems 'cognitive framework' is the non-linear and holistic understanding of social-ecological systems. Through this lens "uncertainty, disturbance, and surprise" are understood to be navigated or managed rather than eliminated (Biggs et al., 2012, p. 432). Furthermore, the recognition of multi-scalar interactions can help foster understandings of why an apparently minor shock to a system can trigger a dramatic change. This sort of thinking also provides rationale for the need to follow other resilience principles, such as managing slow variables and feedbacks, and encouraging learning. A lack of complex adaptive systems thinking can result in management that erodes a system's resilience, and conversely, the adoption of

complex systems thinking can lead to management that maintains or increases resilience in a system.

Bohensky et al. (2015) offer some guidelines for how this fourth principle can be practically used. First, they suggest that an "uncertainty-tolerant culture" should be fostered so that stakeholders don't become paralysed and unable to act without a full understanding of a problem or situation. Secondly, they suggest using a systems framework to bring together different concepts and draw system boundaries, like the Millenium Ecosystem Assessment (MEA, 2005) or Ostrom's (2007) social-ecological system diagnosis framework. Other such frameworks that weren't already mentioned by Bohensky et al. (2015) could include the Sustainable Development Goals (UN, 2019) or the framework for strategic sustainable development (Broman & Robért, 2017), however Bohensky et al. (2015) say that grassroots systems models could also work. Their third suggestion is to "acknowledge epistemological pluralism as a source of complexity" (Bohensky et al., 2015, p. 162). In other words, different ways that stakeholders have of knowing the world may provide different, and possibly competing knowledge claims. These ways of knowing, and their resulting knowledge claims are one of the ways that complexity may manifest, and as such, they need to be acknowledged and incorporated when applying the other principles. The fourth suggestion they make is to "investigate critical thresholds and non-linearities" (Bohensky et al., 2015, p. 162) in order to understand what aspects of the system behave in non-linear ways. The fifth suggestion they make is to change institutional arrangements and functioning to better reflect the reality of complex adaptive system functioning, such as being flexible and adaptable to change (Bohensky et al., 2015). Finally, they also highlight that it needs to be recognised that there are "many barriers to cognitive change" (Bohensky et al., 2015, p. 164), and fostering complex adaptive systems thinking may generate resistance.

3.4.1.5 Resilience Principle 5: Encourage learning and experimentation

The fifth principle is to encourage learning and experimentation. Learning is "the process of modifying existing or acquiring new knowledge, behaviours, skills, values, or preferences" (Biggs et al., 2012, p. 434). As knowledge in complex adaptive systems is assumed to always be incomplete and the landscapes in which knowledge is situated are constantly changing, learning is understood to be crucial for dealing with challenges within them. Drawing on literature about loop learning (Flood & Romm, 1996) and social learning (Reed et al., 2010), Biggs et al. (2012) understand learning to take place at an individual level, as well as "situated within wider groups, organizations, or communities of practice" (p. 434). Single loop learning is learning which asks if things are being done right to enable an "organization to carry on its present policies or achieve its objectives" (Argyris, 1977, p. 116). Double loop learning is a "more comprehensive inquiry" which interrogates into whether the right thing is being done (Argyris, 1977, p. 116). Triple loop learning "involves a more deepseated questioning of values and norms that underlie institutions and actions by asking, 'how do we know what the right thing to do is?'" (Flood & Romm, 1996; Cundill et al., 2015, p. 178). The other learning theory this principle is built on is social learning, which refers to learning that takes place between individuals, existing in "wider social units or communities of practice" (Reed et al., 2010). Such learning develops though deliberate or spontaneous interactions between individual and shared experimentation and reflection (Cundill et al., 2015).

Learning does not always lead to positive outcomes. It can be "maladaptive or dysfunctional," causing conflict when it is based on misinformation (Cundill et al., 2015, p. 186). Biggs et al. (2012) also warn that unequal power relationships can affect learning by determining who learns and what is learned, and these power relationships are crucial to manage to allow genuine and equitable learning and experimentation. Furthermore, power

asymmetries can contribute to partial learning by devaluing, for example, traditional knowledge systems.

Experimentation and monitoring can also be used as tools to facilitate learning, and Cundill et al. (2015) suggest that this needs to be resourced adequately. This can be done in a formal way (i.e., through research) or informal way (i.e., through traditional knowledge management practices). Cundill et al. (2015) say that monitoring should take place at the appropriate scale and over the long-term. Participation in experimenting and monitoring is also a crucial means to encourage learning by helping actors learn about other actors' viewpoints and enable conflict resolution (Biggs et al., 2012). Participation in experimenting and monitoring should both focus on being diverse and representative, as well as be through prolonged engagement (Cundill et al., 2015). Linking to relevant actors and activities in order to embed learning within other learning networks or communities of practice (Lave & Wenger, 1991) can also be effective for sustaining learningful practices (Cundill et al., 2015).

3.4.1.6 Resilience Principle 6: Broaden participation

The sixth principle is to broaden participation, or the "active engagement of relevant stakeholders in the management and governance process" (Biggs et al., 2012, p. 201). Participation can happen to different degrees, and at different stages of a change management process (i.e., both in times of stability and flux). When facilitated effectively, participation can "increase cooperation between actors," increase transparency, and facilitate diverse input into decision-making (Biggs et al., 2012, p. 436). In turn, these can "improve legitimacy [of the governance], expand the depth and diversity of knowledge, and detect and interpret perturbations" (Leitch et al., 2015, p. 204). Participation from diverse stakeholders can also mutually reinforce the fourth and fifth resilience principles by fostering the development of complex adaptive systems thinking (Danielsen et al., 2005) and learning in more stakeholders (Van Rijsoort & Jinfeng, 2005).

There are two types of stakeholders that are relevant for adaptive co-management of ecosystem services (Leitch et al., 2015). While ecosystem services are not the focus of this study, the principles from the two typologies of stakeholders to engage are still relevant. There are (1) those that engage with the issues on the ground and (2) actors working at different "scales and levels of decision making" (Leitch et al., 2015, p. 203). Participation from different levels means that different perspectives, and knowledge can be drawn on to enable capacity building at different levels.

In order for broadening participation to be successful, it should involve diverse actors, including both typologies above, and those with non-scientific knowledge, and ensure that actors are focusing on long-term visions rather than short-term aims during their participation (Biggs et al., 2012). There should also be a shared understanding of governance process in terms of "goals, roles and expectations" (Leitch et al., 2015, p. 215). Effective leadership and "impartial, open, and approachable facilitators" are also important for broadening participation (Leitch et al., 2015, p. 215). Capacity building through learning and adequate resourcing can also support high quality leadership and facilitation, and therefore broaden participation. However, Leitch et al. (2015, p. 203) caution that the ways in which "conflict and power inequity" complicate participatory processes are often "[glossed] over." They can affect the diversity and inclusion of relevant stakeholders, as well as the quality of participation. Therefore, Leitch et al. (2015) suggest that attention to and management of power differentials can help to broaden participation.

3.4.1.7 Resilience Principle 7: Promote polycentric governance

The seventh principle is to promote polycentric governance, which is a governance system, or "the exercise of deliberation and decision making among groups of people in the act of self-ordering their relationships" (Schoon et al., 2015, p. 228), that has "multiple governing authorities at differing scales" (Biggs et al., 2012, p. 437). According to Biggs et al. (2012), "in polycentric systems, each governance unit has independence within a specified geographic area and domain of authority, and each unit may link with others horizontally on common issues and be nested within broader governance units vertically" (p. 437). This governance system reflects the understanding of systems as existing within nested systems, in which disturbances in a smaller system can destabilise a larger system, and conversely, a larger system can help a smaller system 'remember' and retain its properties and functions after being destabilised (Gunderson & Holling, 2002). Perhaps the most crucial feature of a polycentric governance system is that the level at which a problem is governed should reflect the size of the problem (Schoon et al., 2015). By using this system, efforts can be coordinated at a higher level, while devolved governance can allow for autonomy and integration of local-level traditional and non-scientific knowledge.

More practically, polycentric governance provides opportunities for safer experimentation and learning by building redundancies into the system. In this way, if one part of the governance system experiments with a new policy, for example, and fails other parts of the governance system can continue to fulfil its function, thereby preventing the failure from affecting the whole system. Polycentric governance can also broaden participation by involving more diverse stakeholders though different governance systems at different scales. Good polycentric governance both enables and requires "active engagement of individuals in the problems that directly affect them" in order to allow them to have a part in the governance of systems that affect them, while contributing their own expertise and knowledge to enhance the governance process (Schoon et al., 2015, p. 227). Similarly, polycentric governance can facilitate modularity, and "nested cross-scale linkages" between modules (Schoon et al., 2015, p. 234).

Polycentric governance, however, also comes with challenges. One is that there are divergent and unresolved understandings of what polycentricity is and how it should be

assessed (Schoon et al., 2015). All governance systems can be understood to be polycentric to some extent, therefore it is crucial to understand what scales, what activities, and which actors are to be polycentric. This is likely dependent on the governance context. A balance also needs to be struck between the redundancy and efficiency. Too much redundancy can make the governance slow and uncoordinated, whereas if the system is too efficient it can limit experimentation and therefore learning. There may also be trade-offs that need to be managed, and governance that is devolved may not be able identify and consider the implications of the trade-offs that affect a different part of the system. Furthermore, stakeholders pushing an agenda may go "scale-shopping' for a more favourable political venue in which to frame a specific issue" (Schoon et al., 2015, p. 237).

3.5 Conclusion

In this chapter, I have outlined the critical realist ontology and epistemology underpinning this research, and framed my research in the context of systems theory, specifically social-ecological systems theory. I introduced social-ecological resilience as a theoretical lens which will be applied to analyse the findings of this study. Specifically, I introduced Biggs et al.'s (2012, 2015) resilience principles which I will use to assess the vulnerability of student-led food gardens and generate strategies for how to best to support student-led food gardens manage the impacts of students' transience. In the next chapter I will go on to explain how I used action research to these ends.

Participatory action research in student-led food gardens

Chapter 4

If you want truly to understand something, try to change it. Attributed to Kurt Lewin by Henderikus J. Stam, 1996, p. 31

I have often encountered that funny feeling of finding an anomaly [...] It is a little irritating feeling, kind of a pre-sneeze sensation – and it is also very exciting. Learning to trust this message is the toughest lesson I have to teach my students – no less than myself. Susan Leigh Star, 2010, p. 605



Cones marking a newly-built gate Summer 2013

4.1 Introduction

This thesis is based on an action research study of Student Eats student-led food gardens. This chapter will begin by outlining what action research is and the theoretical underpinnings of the methodology, drawing on pragmatism, critical pedagogy and the theory of communicative action. The more practical process of employing the 'spirals' that characterise action research methodology will be described next. These include the four main stages the study was divided into: constructing the problem, planning action, taking action, and evaluating action (Coghlan & Brannick, 2019).

After outlining the action research spirals, I will take a moment to outline my positionality in relation to the study and reflect on how it impacted the research in terms of quality and ethics. This then leads into a description of my overall research design, where I begin by positioning the research as a multiphase mixed methods approach, and then go on to explain the methods I use to address each research question. First, I explain the methods used to conduct a quantitative systematic literature review of sustainable university community gardens, and then I outline how I used a case study approach within the action research methodology. I used a single case study of Student Eats, investigating three gardens as 'subunits of analysis' (Yin, 2003).

The next section will go into the specifics of the methods employed in each of the research stages of the action research spiral. I will outline how interviews, workshops, photovoice, and a fishbowl discussion were used to collect data. I also explain what sampling strategies were employed, and how data was handled and analysed. The chapter concludes with a reflection on quality and ethical considerations.

4.2 Action research methodology

Action research is:

"a participatory, democratic process concerned with developing practical knowing in the pursuit of worthwhile human purposes, grounded in a participatory worldview which we believe is emerging at this historical moment. It seeks to bring together action and reflection, theory and practice, in participation with others, in the pursuit of practical solutions to issues of pressing concern to people, and more generally the flourishing of individual persons and their communities" (Reason & Bradbury, 2001, p. 1).

In its simplest terms, action research combines a research agenda and an action agenda in a participatory way. Action research studies tend to have features akin to longitudinal research designs in that data is gathered over longer periods of time and change is observed (Bryman, 2012). The main difference between a typical longitudinal design and an action research design is that action research focuses on planning and delivering an intervention. Action research is also a solutions-orientated methodology, and as such tends to be focused around addressing a problem or issue in some capacity. Action research is often advocated as a methodology for sustainability science because of its focus on solving urgent problems in collaboration with stakeholders outside of academia (Wittmayer & Schäpke, 2014; Egmose, 2015). Action research and sustainability science also share an advocation for values in research and have the ambition to contribute to a more equitable future.

Different schools of thought on action research have emerged over the years (Nielsen & Nielsen, 2006; Johansson & Lindhult, 2008). The two main alternatives today are pragmatic and critical action research, which are summarised in Table 2. Oftentimes, critical orientations and pragmatic orientations to action research are painted as two distinct traditions in tension

with one another (Johansson & Lindhult, 2008), however within this study I attempt to draw

on the best of both traditions. In the following section I will outline some of the main tenets

of the traditions and their theoretical underpinnings. While doing so, I will also explain how I

draw on each of these traditions in the application of my methodology.

2000, p. 102).		
	Pragmatic action research	Critical action research
Purpose	Improvement in workability of human praxis	Emancipation
Action focus	Experimental, cooperation	Resistance, liberation
Orientation to power	Power as the ability to do, collaborative relation, practical agreement is sought for	Dominant interests, coercive, conflict is acknowledged
Role of researcher/ related knowledge	Closeness, practical knowledge	Distance, episteme, reflective knowledge
Research focus	Action, dialogue	Reflection
Development focus	Experiential learning, learning by doing	Consciousness-raising
Type of dialogue	Cooperative, experience based, action-orientated	Promote openness to the other
Situation	Fragmentation, compartmentalisation	Asymmetrical power relations

Table 2. A comparison of the pragmatic and critical action research traditions (Johansson & Lindhult, 2008, p. 102).

4.2.1 Pragmatic action research

Pragmatic action research focuses on creating useful and actionable knowledge through dialogue. Pragmatism, as a philosophical and theoretical tradition, contains relatively diverse sets of ideas, but what brings them together is a "rejection of foundationalist epistemology" (Parker, 1996, p. 22), which is the idea that knowledge must be secured in some basic, non-self-referential beliefs. For pragmatists, creating knowledge happens only through transformation of "ourselves and the world" (Parker, 1996, p. 23). That is, it is only through interaction between people and the world they are situated in that knowledge is produced. That is, the only way to know anything is to know it in relation to other things, including the self. This is what has been called an 'extended epistemology,' because it goes beyond other theories of knowledge, requiring that knowledge is acted upon in order to be tested (Reason & Bradbury, 2001). This is because pragmatists believe that through action is how knowledge is validated (Reason & Bradbury, 2001).

This extended epistemology includes ways of knowing beyond the "theoretical, propositional knowledge" that dominates academia (Heron & Reason, 2001, p. 183). It incorporates four ways of knowing: experiential knowing, presentational knowing, propositional knowing, and practical knowing. The most grounded way of knowing is experiential knowing, which is an immediate and affective form of knowing had though direct contact with whatever is being known (Heron & Reason, 2001). The next stage, presentational knowing, emerges out of experiential knowing and is an intuitive and creative type of knowing (Heron & Reason, 2001). This sort of knowing is often pre-language, instead being expressed through, for example, the visual arts, music, or dance. Presentational knowing can therefore be thought of as a means of coming to understand and interpret experiential knowing through creative practices. Propositional knowing, the most accepted form of knowing within the academy, is "intellectual knowing of ideas and theories" and is associated with cognitive development, the use of language and abstraction of ideas (Heron & Reason, 2008, p. 367). Practical knowing is associated with psychomotor development, and refers to "knowing how to do something. Its product is a skill, knack or competence – interpersonal, manual, political, technical, transpersonal, and more" (Heron & Reason, 2008, p. 367).

These ways of knowing interweave and can be experienced simultaneously. All of these forms of knowing are mutually reinforcing; that is, a complete understanding will incorporate all four forms of knowing. As Heron and Reason (1997, p. 282) put it, "it is equally important that action not only consummates the prior forms of knowing but is grounded in them. It is in this congruence of the four aspects of the extended epistemology that lie claims to validity" (see Figure 7).



Figure 7. An adaptation of the four aspects of Heron & Reason's (1997) extended epistemology and their relationship.

Pragmatism can also be used as an approach for "transdisciplinarity in sustainability research" to clarify the role of reflexivity (Popa et al., 2015, p. 1). Since sustainability science is such a new area, there is some ambiguity in the understandings of some concepts, including 'reflexivity'. Popa et al. (2015) contend that incorporating a pragmatic understanding of reflexivity will extend it from being used only to build social legitimacy (more akin to a socio-technical approach to action research (Masters, 1995)) to encourage more open-ended approaches to reflexivity where the research design must be flexible enough to allow the problem or research question to be redefined, not only the methodology. Practically, this means that, in addition to considering epistemological and normative questions in reflexive work (such as the "critical consideration of assumptions, values, and socio-institutional" bias and focusing on "convergence on values and normative commitments"), researchers should also engage with more pragmatic concerns (such as "reflexivity on values and understandings in concrete problem-solving and social experimentation processes") (Popa et al., 2015, p. 52-53). As such, action research is a highly appropriate mode of research for adopting this pragmatic approach to reflexivity.

4.2.2 Critical action research

Critical action research is focused more on dialogue and action that promotes emancipation and disrupts power structures (Johansson & Lindhult, 2008). There have been a wide range of thinkers and doers who have influenced this tradition, but the ones that have had the most profound impact on my approach to critical action research are Freire (2000) and Habermas (1984, 1987).

4.2.2.1 Influences from critical pedagogy

Freire's most influential concept is likely 'education as a practice of freedom.' This idea is based on his assumption that all people have an "ontological vocation to become more fully human" (Freire, 2000, p. 74) or that, by their very nature, people are "driven to be more fully human" (Smidt, 2014, p. 22). The phrase 'to become more fully human' implies a process, rather than a static state, and this was reflected in how Freire saw education – as a practice, rather than an end. In Freire's (2000, p. 84) own words, people need to be affirmed "as beings in the in the process of *becoming* – as unfinished, uncompleted beings in and with a likewise unfinished reality." According to Freire, the process of 'becoming human' is negotiated through dialogue, and people who are not 'becoming more fully human' are not free because they are not striving towards their ontological vocation.

The 'banking model of education' (a term coined by Freire (2000)), where teachers are knowers who deposit information into students, does not foster the process of becoming more fully human. This is because, in this model, students are considered merely receptacles of knowledge rather than critical thinkers with the ability to generate and synthesize information and ideas. To Freire, this type of education dehumanises both parties because

the student is objectified and therefore dehumanised, and the teacher's objectification of the student is also a violent and dehumanising act which dehumanises the teacher. Therefore, both parties are not free because they are both trapped in the role of oppressor and oppressed. The opposite of the 'banking model of education' is 'education as the practice of freedom,' and this latter mode of education is also a means to dismantle the former (Freire, 2000). To Freire (2000, p. 80), education as a practice of freedom is education through dialogue, which transforms the "teacher-of-the-students and the students-of-the-teachers" to "teacher-student with students-teachers." This happens though what Freire calls 'conscientisation' (*conscientização* in Portuguese), which is the process of becoming conscious of consciousness, or developing 'critical consciousness'.

Freire's (2000) ideas outlined above emerged from his work with developing literacy in poor communities in Brazil and as such, one might feel his work may not be applicable to education within formal institutions, least of all in a developed country. While not all aspects of his work are relevant in this study, there are a number of concepts that have been valuable. Moving from the educational context to the research context, I found it useful to think of Freire's banking model of education as a banking model of knowledge. In this model, we can see the relationship between the student and the teacher parallel the relationship between the researcher and the researched: both teachers and researchers are the *knowers* and students and researched are *receptacles of knowledge* or *objects of study*. Thinking of my research outside of a 'banking model' repositions my role as a researcher as more than just an extractor of information. When thinking about my researcher role through this lens, the knowledge creation process is more democratic and involves research participants as coinquirers where they help shape the direction of the research. The role of the researcher becomes more reflective and as a facilitator, the research has a responsibility to help make visible restrictive and invisible social structures. In addition to helping situate my role as a

researcher and define the researcher-participant relationship, thinking through the lens of critical pedagogy has also encouraged analysis situated in the lived experiences of the research participants that can elucidate patterns and issues within societal systems. Like critical pedagogy, the theory of communicative action also has emancipatory aims, which will be discussed next.

4.2.2.2 Influences from the theory of communicative action

Like Freire's work developing critical pedagogy, Habermas' theory of communicative action has been influential in action research (Kemmis, 2008). Communicative action is "action orientated towards intersubjective agreement, mutual understanding, and unforced consensus about what to do" (Kemmis, 2008, p. 127). The goal of this type of action is to go beyond instrumental aims (through what Habermas (1984, 1987) calls 'strategic action') and to understand (through communication) what overall purpose the given action is meant to achieve.

According to Habermas, to achieve this type of communication, there needs to be an "open communicative space" (Kemmis, 2008, p. 127). The people engaged in communication in this space need to be committed to the idea of communicative action – that is to say, they are prepared to think beyond their instrumental aims and engage in "meta-level" discussions (Kemmis, 2008, p. 127). Some other important preconditions for communicative action are (1) that every person who could make a contribution needs to be included, (2) there are equal opportunities to speak, (3) there is a lack deception or illusion, and (4) that participants are free from coercion (Kemmis, 2008). Furthermore, within this space, people should speak comprehensively, truthfully, authentically, and appropriately (McNiff & Whitehead, 2001). These preconditions create an 'ideal speech situation,' which is an 'intersubjective space', or a space where people relate to one another and are able to engage in rationale dialogue.

This intersubjective space that Habermas (1984) talks about is a linguistic space. While everyone has access and makes use of language, it is not owned by any single person. Furthermore, it is shaped by ongoing subjective interpretations and use. Habermas (1984) believed that through communication we can come to intersubjective agreements that actions can be based upon. However, these intersubjective agreements are time-bounded because no conversation can be fully 'completed,' and circumstances are always changing. Freire (2000) and Habermas (1987) share this understanding of dialogue as an ongoing practice of learning and meaning-making.

Ultimately, what prevents us from coming to intersubjective agreements through communication is that there are bureaucratic systems (like government, organisations, economic systems) that are 'colonising' our 'lifeworlds', which are spaces of intersubjectivity, where truth and morality exist through our interactions with one another. Individuals occupy multiple lifeworlds, and these lifeworlds are plentiful and can "overlap and interweave" (Kemmis, 2008, p. 129).

To explain what it means for a system to colonise the lifeworld, Habermas (1984) outlines two types of action: strategic and communicative. Strategic actions are instrumental and are useful in complex societies to improve efficiency and expedite processes. However, strategic action can create systems that begin to operate relatively autonomously from the lifeworlds, where genuine, valid truths exist. What this means is that systems become 'uncoupled' from the lifeworld, and in effect, subordinates the lifeworld to the system. Another way to describe this is that the system is 'colonising' the lifeworld with "rationalized models of right action that are inappropriate for relationships between people wherever these should properly be based on valid knowledge, solidarity, and personal capacities" (Kemmis, 2008, p. 130). This might mean that people behave in ways such that their actions contract their values. The problem that results from this is an inability to continue to

construct and reconstruct lifeworlds unburdened by the bureaucratic conditions of the social system. The systems then become "increasingly difficult to manage, since the lifeworld anchoring necessary for system operation is no longer secure" (Kemmis, 2008, p. 130).

Universities, and the higher education sector at large, tend to be highly bureaucratic, particularly with recent neoliberal developments (Lorenz, 2012). Wider systems that influence student led-food gardens, such as the university as an institution, their students' union, and the National Union of Students, are systems that may have colonised the gardens' 'lifeworld,' and may be preventing the students and other stakeholders from engaging with one another in a non-instrumental way.

In this study, I have been guided by Habermas' preconditions for creating ideal speech situations when constructing and engaging in interviews and workshops in order to promote open, honest dialogue and enable people to develop a common 'intersubjective space.' The idea of a genuinely ideal speech situation might be seen as somewhat naïve, and I don't claim to meet all the criteria fully. However, I used the conditions for ideal speech situations as a guide to enhance the 'idealness' of the speech situation.

Critical pedagogy and the theory of communicative action, while emerging from quite different intellectual roots, are highly complementary as they are both political and counterhegemonic (or in opposition to the status quo). They come together in their struggle for democracy, equality, and freedom through participation. They complement each other in that Freire believes that language is never neutral, and Habermas suggests that, although language is socially constructed, perhaps there are situations (i.e., ideal speech situations) in which language can be used to create shared understandings by negotiating what we mean through rational discussion. Both theories are on the more pragmatic side of critical theory, and therefore provide a critical complement to the more pragmatic approaches outlined earlier.

4.2.2.3 Action research spirals

Lewin (1946), often considered to be the father of Action Research, introduced the concept of research spirals that characterise the methodology (see Figure 8). Each spiral is undertaken as a "collaborative venture" between the researcher and participants to examine the present situation, prepare for and undertake some type of action to address the problems as they are then perceived, and finally reflect on the understanding of the situation and the action (Coghlan & Brannick, 2005, p. 23). With each spiral, a clearer understanding of the problem is uncovered, and the actions undertaken become better suited to address the situation. This process is rooted in the pragmatist extended epistemology, where only through testing knowledge in real-life situations can it be validated (Argyris & Schön, 1974; Heron & Reason, 1997).

In each action research cycle, there are four different phases: constructing, planning action, taking action, and evaluating action (Coghlan & Brannick, 2005; French & Bell, 1999; Stringer, 2007). Some models, such as Stringer's (2007), only have three stages, however I will be using the four-stage model to explain my research because it better reflects how my



Figure 8. The continuous cycles of action research as modelled by Coghlan and Brannick (2005, p. 10; 2019).
research manifested. The constructing phase is a process of defining the problem to illustrate the context and prepare for planning action. Planning action involves examining the problem construction, and on that basis coming up with "a first step, or a series of first steps" to address the problem (Coghlan & Brannick, 2005, p. 23). These step(s) are then undertaken as the next part of the cycle, taking action, often referred to as an "intervention" (Coghlan & Brannick, 2005, p. 23). Finally, an evaluation takes place to identify the outcomes and to see if the construction of the problem still seems accurate, how successfully the actions addressed the problem, and what information can be taken forward into the next cycle, should there be one.

It is important that the action research spiral is not merely procedural, and rather remains "a series of commitments to observe and problematize through practicing a series of principles for conducting social enquiry" (McTaggart, 1994, p. 249). I draw on Elliot's (1991) understanding of the action research cycle in which "the general idea should be allowed to shift," to prevent reflection from becoming "merely fact finding" rather than analysis (p. 70). In practice, this means not only evaluating and amending the plan of action, but also revising the general idea underlying the project (Elliot, 1991; McNiff, 2013).

The 'core' research cycle depicted in Figure 8 is also complemented by a more reflective process undertaken by the researcher, which Zuber-Skerritt and Perry (2002) call the 'thesis' research cycle (see Figure 9). This reflection on top of the "core" research cycle is in aid of developing a deeper understanding than what would be achieved through "ordinary problem solving" about what the researcher is doing, why they are doing it, and what the implications of it are (Coghlan & Brannick, 2005, p. 25). Through the process of this 'thesis' cycle, the researcher undergoes a process of "learning about learning," or meta-learning (Coghlan & Brannick, 2005, p. 25). In this 'thesis' cycle, Coghlan and Brannick (2005, p. 25) suggest that three forms of reflection should be undertaken, upon: content (which has to do

with the matters under consideration and what is taking place), process (which is related to the methods and strategies), and premise (which includes reflections on underlying assumptions.)

The 'core' and 'thesis' cycles operate concurrently. Coghlan & Brannick (2014) liken this to a clock in which the hour, minute, and second hands are all cycling around the same fixture. The hour hand takes twelve hours to make a full rotation, and the minute hand only takes sixty seconds. So, in the case of this study, a full 'core' cycle took 2.5 years, however, there were many 'thesis' cycles that were cycled through much more rapidly and intuitively. These two cycles also help to propel the project forward. This is akin to the cogs in Figure 9, where a movement forward in the thesis cycle advances the core cycle, and vice versa. The thesis cog is smaller, and therefore cycled more rapidly than the core cog, which is reflective



Figure 9. A depiction of the relationship between the 'core' and 'thesis' research cycles of action research.

of the timelines in practice. In this thesis, the core cog had more defined phases, while the thesis cycle was more amorphous in its manifestation.

4.3 Overall study structure

My study incorporates both qualitative and quantitative methods using an approach called multiphase mixed methods (Cresswell, 2014). This is when qualitative and quantitative components are separate components in a study. I began with a quantitative systematic literature review, and then in the next research phases I used qualitative methodologies (see Tables 3, 4, 5 and 6). This separation of the qualitative and quantitative components of a mixed methods study enables a less complex analysis process but with the benefits of drawing on different approaches that offer complementary perspectives (Cresswell, 2014).

It is well-recognised that the reality of conducting action research is not as straightforward as models, like the one in section 4.2.2.3, suggest. However, Cook (2009, p. 7) says that the 'messiness' of action research is actually how Reason and Bradbury's (1997) four types of knowing from their extended epistemology (see Figure 7) "come together and jostle with each other" in the process of knowledge co-construction. However, this poses a challenge in communicating the narrative of the research, including the methodology, the timeline, the evolution of ideas, and the findings. As such, the write-ups of action research studies tend to have less conventional formats (Stringer, 2007). In writing up this study, I have used Habermas' (1987) principles for rationale dialogue as a guide for what information to include or omit, as it would be impossible to include everything that happened during the study. As explained earlier, I have therefore tried to be as comprehensible, truthful, authentic and appropriate in my writing as possible (McNiff & Whitehead, 2011). As such, I will begin by briefly explaining how I arrived at what are now the research questions addressed in this

Table 3. The research timeline.

		2013 2014					2015							2016								20	017																						
	J	J A	S	0	N [) 1		F N	ΛA	A M	J	J	А	S	0	Ν	D	J	F	Μ	А	N	J	J	A	S	5 C)	N D	J	F	Μ	А	Μ	J	J	А	S	0) N	D	J	F	Μ	А
	Systematic Literature Review																																												
Articles																																													
<u>Note</u> :	Note:																																												
• Dui																																													
ado	ded d	uring	this pe	erio	d.																																								
	National Union of Students																																												
Interviews									\bot						2	1																					1		\bot	\bot	\bot	\bot			
Embedded unit: Keele University																																													
Workshops									\perp									1						_					2		1			1						┶					1
Interviews																		1									1	L						1						1			1		
Notes:																																													
		work				,																																							
		works	•																													-					-								
		ints ar													-								s no	t aı	udio	rec	cord	ed į	giver	n the	e lim	itec	nui	mb	er of	f pa	rtici	pan	its m	near	nt th	nat I	didı	n't	
		a dise																																											
		kshop					int	ende	ed to	o be	an	evalu	uatic	on w	ork	sho	p, ł	now	eve	er it v	Nas	рос	orly	atte	ende	ed b	by th	IOSE	e wh	o ha	d no	ot vo	olun	tee	red	to d	deliv	er a	ın 'a	ictic	on.' I	t wa	as		
		e not																																											
• The	e solic	itatio	n of p	hote	ovoi	ce ir	nag	ges to	ok	plac	e b	etwe	en N	Лау													eiveo	d be	etwe	en A	ugu	st 2	016	i an	d Fe	ebru	lary	201	.7.						
			<u>г г</u>		-		-	-		-	-	-	1		En	nbe	dd	ed u	init	: Un	iver	sity	of V	Nar	wic	k	-	-		-			1	-		-	-	-					-	1	
Workshops					_	_			+		_							_	1		-	_		_	_	_	_		1		1				1	+			+	+	_	—		_	
Interviews					_				+		_	_						_	-							_	2	2			-			_		┢	1		—	+	_		1		
Visits																			1								1	L			1				1		1								
Notes:	c• •										10																										c								
		work							to r	myse	elt a	and m	iy re	sea	rch	whe	ere	I IN	vite	d th	e ga	arde	ner	s to	o pai	rticij	pate	e. It	was	not	aud	io r	ecoi	rde	d to	allo	ow to	orn	nore	e nat	tura	lint	erac	tior	IS
		myse																C . I																											
• The	e wor	kshop	in Fel	orua	ary 2	016	0100	cused	d bc	oth o	n c	heck	ing t	he a													nnin	g.																	
	Embedded Unit: Sheffield University																																												
Interviews	+ +		+			_	+		+		_								_	-	_	_	_	_		_		_	2		+	_		1	_	+	_	_	+	+	+	+			+
Visits													<u> </u>			L				1									1		1	1	<u> </u>	1									<u> </u>		
			<u> </u>		1	1	-		—		1		Na	atio	nal	Uni	on (of St	tud	ents	St	ude	nt E	ats	Cor	ntere	ence	э	-	-						_			—	—	_	_			-
Workshop																																1	1										1		

Table 4. Legend for Table 3.

Actively constructing
Passively constructing
Planning action
Evaluating

Table 5. Summary of the number of interviews, people interviewed, and workshops undertaken.

	NUS Staff	Keele University	University of Warwick	Sheffield University	Student Eats Conference
Number of interviews	4	6	4	3	0
Number of people interviewed	3	6	3	3	0
Workshops	0	5 (3 recorded)	4 (3 recorded)	0	1

Table 6. Research questions and corresponding methodologies, data sources, and data collection and analysis.

Research question	Methodology	Data source(s)	Data collection method	Analysis method
 According to academic literature about sustainable university community gardens, What benefits (both discussed and demonstrated) do these gardens reportedly provide? How do these benefits compare to the benefits provided by community gardens more generally? What is already known about the impacts of students' transience on sustainable university community gardens? 	Action research	 Peer reviewed literature 	Quantitative systematic literature review	 Protocol coding according to thematic areas from Guitart et al.'s (2012) literature review
 In addition to students' transience, what causes problematic participation in student-led food gardens? What effects does problematic participation have in student-led food gardens? What feedbacks between problematic participation, its causes, and effects in student-led food gardens exist (if any)? 	Action researchCase study	 Three student-led food gardens Student Eats conference 	 Semi-structured interviews Workshops Photovoice Fishbowl dialogue Reflective research diary 	 Eclectic coding, drawing mainly on holistic coding and descriptive coding Causal loop mapping using The Brain software
• In what ways are student-led food gardens vulnerable to students' transience in the context of the participation dynamics outlined above?				 Protocol coding according to Biggs et al.'s resilience principles (2012, 2015c)
• In what ways can student-led food gardens address the impacts of students' transience and build resilience into their initiatives such that they continue to persist?				 Eclectic coding, drawing mainly on holistic coding and descriptive coding Cross-referencing the codes with those from the causal loop diagram

study. A discussion and reflection on the methodological issues will be presented in Chapter

9.

4.3.1 Systematic literature review

In the early stages of my study, I knew I wanted to adopt an action research approach. This was inspired by my pragmatic orientation to research and wanting to 'make a difference,' as well as by the sustainability science literature that calls for sustainability research to engage with practice (Kates et al., 2001; Clark & Dickson, 2003; Lang et al., 2012). From my Master's thesis (Laycock, 2013) I had a rough idea of the landscape of literature about community gardens and the problems they face, however I felt less confident in the realm of community gardening in a university context. Inspired by Guitart et al.'s (2012) quantitative systematic literature review of urban community gardens, I decided to draw on the study's methodology to undertake a quantitative systematic literature review of sustainable university community gardens before approaching any particular garden to work with.

The systematic literature review addressed the first research question, which is: *What is the state of the literature about sustainability-driven community gardens in universities?* My aim was to produce a synthesis of the literature about sustainable university community gardens in order to map current patterns and direct future research efforts (Petticrew & Roberts, 2006). I will explain why I use the term 'sustainable university community gardens' rather than 'student-led food gardens' in section 4.3.1.1, but first it is necessary to explain what a quantitative systematic review is and why it was appropriate for the 'constructing' phase of my thesis.

Quantitative systematic reviews are an approach to reviewing literature that identifies, appraises and synthesises all primary literature on a subject by defining an area of study, systematically and exhaustively identifying literature in the area, and developing a

database with information summarising the current status of the literature (Petticrew & Roberts, 2006). They have been used widely in the social sciences, and increasingly in environmental studies and sustainability science (de Medeiros et al., 2014; Terstappen et al., 2013; Partelow et al., 2018; Nikulina et al., 2019). This style of review is beneficial for mapping trends in the literature and highlighting reliable findings and uncovering biases, through providing "reproducible, reliable [assessment] of the current status of a field of research" (Petticrew & Roberts, 2006; Guitart et al., 2012, p. 365). By appraising the literature using a transparent and clearly articulated methodology, there is reduced potential for authorial bias compared to narrative-style reviews (Guitart et al., 2012). Being a new scholar, my knowledge of the literature was limited and therefore a systematic approach meant that I could appraise the literature and be confident that I was not missing anything (Pickering & Byrne, 2014).

This said, a review was certainly not sufficient to 'construct' the problem to be addressed by the research, least not in a participatory way. The construction of the problem was also shaped and defined through qualitative methods, both in conversation with research participants and through my own experience with student-led food gardens. The rest of the constructing phase will be outlined in section 4.5.3.

Some of the limitations of the systematic literature review method are that the reliability of the review is dependent on the quality and scope of articles analysed, and there is a certain amount of subjectivity in developing inclusion criteria for studies included in the review and identifying themes in diverse literature (Garg et al., 2008). However, this style of review was appropriate for my thesis for several reasons. Given that literature about sustainable university community gardens is newly emerging, a systematic literature review was an appropriate method to take stock of what has been written to date to guide future research efforts. Because Guitart et al. (2012) had published a similar review about urban

community gardens, I was able to use the robust set of themes they had already developed from in the literature. This helped to increase the robustness of the themes, and it also meant I could identify the ways in which sustainable university community gardens were similar and different to urban community gardens.

4.3.1.1 Inclusion criteria

When creating the inclusion criteria I used for the review, I tried to create criteria that would include articles about gardens most similar to Student Eats student-led food gardens. The criteria I eventually arrived at were that the gardens in the articles had to be 'sustainable university community gardens.' The use of the term 'community garden' originated from my initial research focus. Early in my doctoral studies, I focused my research questions around 'university community gardens.' I was planning on working with some of the National Union of Students' Student Eats gardens, and number of them called themselves 'community gardens.' As such, it seemed to be the most appropriate term. In section 4.2.2.3, I wrote that action research is meant to encourage a shift in thinking and that the general idea underlying an action research project should be allowed to change as the issues at stake are better understood (Elliot, 1991; McNiff, 2013). This evolution in the terms I used to refer to the subject of this study is evidence of this shift and should be seen as part of the narrative of how this study developed over time. It was only though the initial inquiry into sustainable university community gardens that I came to see that the term 'student-led food garden' that I use elsewhere in this thesis most accurately represented the Student Eats gardens, as explained in section 2.4.

The three criteria for inclusion in the review were that the articles must have written about garden(s) that: (1) were affiliated to a higher education institution, (2) were community gardens, in terms of being operated and managed collectively by the local community (including any combination of university students, staff, alumni and residents, as well as people living in the area surrounding the university or garden), and (3) had a commitment to

sustainability through practice and/or ethos. Table 7 expands on some of the details of these

criteria.

Including gardens affiliated with higher education institutions was a natural criterion

given this research's focus on food gardens in universities. The second criterion was intended

to exclude farms exclusively run by the institution and allotment-style gardens, as the Student

Eats gardens were all run collectively by a group (rather than by an individual) or at the very

least had a collectively-run part. This literature was both voluminous and shared little in

Table 7. Descriptions of the inclusion criteria for articles included in the quantitative systematic literature review.

Criteria	Description
The garden(s) in the articles must be Affiliated with a higher education institution	 Garden must have had a sustained relationship with a higher education institution Garden was on university property, created by the university, or the university had sustained engagement with the garden and planned to continue to do so Could include participants not affiliated to a university Example: Gardens affiliated with Cornell's Mosaics Garden programme were excluded, because in spite of the fact that many people from the university played an important role, the gardens were off-campus and run independently from the university (Doyle & Krasny, 2003) Example: The Siyakhana Project of the University of Witwatersrand (Wills et al., 2010) was included because, in spite of being off-campus, the garden was created and directly affiliated with the University through various departments' applied research projects
The garden(s) in the articles must be A community garden	 Garden must have been an open space "managed and operated by members of the local community in which food, flowers" and/or other plants are cultivated (Holland, 2004; Pudup, 2008; Kingsley et al., 2009; Guitart et al., 2012, p. 364); the community could have included university students, staff, alumni and residents, but could also have included people living in the surrounding area Included community supported agriculture, farms, botanical gardens and a wildlife habitat garden, in addition to more the conventional smaller-scale community gardens. This mirrors the literature reviewed by Guitart et al. (2012) which included some larger-scale projects (Wade 1987; Karaan & Mohamed, 1998), projects involving restoration ecology/native ecosystem regeneration (Stocker & Barnett, 1998) and those run for commercial purposes (Flynn, 2001) Must have had a strong agenda or practice for involving people from the university or the area around the garden and/or university (gardens excluded by this criterion include Delate (2006), Ferguson et al., (2006), Schroeder et al. (2006)) Could not be private gardens (Nell et al. (2000) was excluded by this criteria)
The garden(s) in the articles must have A clear commitment to sustainability	 The commitment must have gone beyond just food growing (given that not all food growing is "sustainable"), for example, a specific commitment to organic food production (e.g., Kobayashi et al., 2010), facilitating research into and practicing ecological restoration (e.g., Barton et al., 2010; Hockenberry Meyer et al. 2010), and horticultural therapy (e.g., Hockenberry Meyer et al. 2010) Must have included either sustainability practice and/or ethos – the actual contribution to sustainability or the intent to do so (gardens excluded by this criterion include Byers (1999), Hamilton (1999), Olsen et al. (1999), Stimart (1999), VanDerZanden and Cook (1999), Haynes and Trexler (2003), and Stephens et al. (2006))

common with the university community gardens I intended to research. Therefore, I used the

qualifier 'sustainable' to both omit much of the less-relevant literature about university farms,

as well as to explore university community gardens' potential contributions to a sustainable

future.

Articles, such as reports or other accounts, which were not peer reviewed (such as

Lund & Orth, 2010) were not included in the review. The types of articles that were included

were conference proceedings, feature articles, technical papers, perspective pieces and

research articles. The differences and similarities between the term 'sustainable university

community gardens' and the term 'student-led food gardens' that is used in the rest of this

thesis can be seen in Table 8.

Table 8. The differences and similarities between the inclusion criteria for the review about sustainable university community gardens, Guitart et al.'s (2012) inclusion criteria for their review of urban community gardens, and the definition of student-led food gardens in universities used in this thesis.

Criteria	The term used in the quantitative systematic literature review: Sustainable university community gardens	The term used in the rest of this thesis: Student-led food gardens in universities	The term used by Guitart et al. (2012): Urban community gardens
Self-defined "community garden"	Can be self-defined 'community garden,' but does not need to be	Can be self-defined 'community garden,' but does not need to be	Must be a self-defined 'community garden'
Location	In urban or rural areas	In urban or rural areas	Only in urban areas
Relationship with an HEI	Must have a relationship with a higher education institution	Must have a relationship with a higher education institution	Can have a relationship with a higher education institution, but does not need to
Student leadership	Can be led by students, but does not need to be	Must be led by students	Can be led by students, but does not need to be
Sustainability practice(s) or ethos	Must have sustainability practice(s) and/or ethos	Has a sustainability practice and/or ethos given the mandate of Student Eats funding	Can have sustainability practice(s) and/or ethos, but does not need to
Food growing	Can grow food, but does not need to	Must grow food	Can grow food, but does not need to

4.3.1.2 Method

Articles were found searching the following electronic databases between June 2013

and August 2016: EBSCO, ISI Web of Knowledge and Google Scholar. The keywords used to

find the articles were 'university,' 'community garden,' 'food growing,' 'campus,' 'higher

education,' and 'sustainability.' Further articles were found by exploring the references of the initial papers and other literature reviews, as well as those papers which had cited Guitart et al.'s (2012) review of urban community gardens. Most articles were found and analysed before August 2014, after which the review was 'topped-up' as new articles were published until August 2016.

Each article was coded using protocol coding (Saldaña, 2013) according to its authors' affiliation(s), journal's discipline area, year of publication, location of the garden(s) studied, research methods used, garden characteristics and benefits of the gardens, drawing on the thematic areas used by Guitart et al. (2012). The garden characteristics were then broken down into types of gardens (e.g. food gardens, botanical gardens), what plants were grown (e.g. edible, ornamental), whether or not produce was sold and who it was sold to, who participated in the garden (e.g. students, faculty), and if the garden was used in the formal curriculum, and if so, for what subject area.

The benefits (both those that were discussed and those that were demonstrated) that the garden provided were categorised using the themes identified by Guitart et al. (2012). Discussed benefits were "only mentioned as the authors' opinion and/or findings in previous literature", while demonstrated benefits "were confirmed as a result of the research" (Guitart et al., 2012, p. 365). Determining if a benefit was discussed or demonstrated posed a challenge, particularly for articles that did not clearly outline methodology/methods because it was difficult to determine what they were asserting their research contribution was. Furthermore, this categorisation also had the potential to be used as a judgement of the reliability of a paper's findings (i.e., saying that there is not sufficient support for this finding, therefore classing it as a 'discussed' benefit). In order to mitigate the challenges this posed, like Guitart et al. (2012, p. 365), I didn't attempt to assess the rigour of the papers given (1) the interdisciplinary nature of the topic of the review and resulting diversity of criteria for

rigour, and (2) that "each paper assessed had already been evaluated within its discipline as of a suitable standard for publication in the academic literature."

The types of benefits I categorised my results into were: 'social benefits' (which were benefits to a group of people, such as networking and community development); 'access to fresh foods;' 'economic benefits;' 'health benefits;' 'reduced crime/increased safety;' 'educational benefits;' 'environmental sustainability benefits' (such as the use of sustainable/organic farming/gardening methods); 'cultural heritage/development;' 'life satisfaction⁶;' 'environmental equity' (a state when a particular group/community does not bear disproportionate negative environmental effects/impacts) and 'increased biodiversity⁷.' For this review, I developed three new categories of benefits that did not fit under any of the themes identified by Guitart et al. (2012). These were: 'engagement, attitudes and behaviours for sustainability;' 'additional benefits to individual participants⁸;' and 'benefits to the institution.'

The results from this review are presented in Chapter 5 and helped to 'construct' my understandings of sustainable university community gardens in order to better understand what is already known about student-led food gardens and similar initiatives. I will now go on to explain how I continued to 'construct' my understandings of the problems in student-led food gardens through a case study approach.

⁶ Guitart et al. (2012) didn't define what they meant by 'life satisfaction,' so I categorised benefits that enhanced participants' sense of well-being (such as a 'slower pace of life' and improved self-confidence/esteem and/or empowerment) under this category.

⁷ Although it could be argued that Guitart et al.'s (2012) 'increased biodiversity' theme fit under 'environmental sustainability,' it is retained as a separate category to better align with Guitart et al.'s review.

⁸ While a number of benefits identified by Guitart et al. (2012) could fall under the 'additional benefits to individual participants' category (such as education, life satisfaction), I retained Guitart et al.'s existing categories, and added a new category called 'additional benefits to individual participants' (such as employability benefits and decreased shopping trips) for the benefits that didn't fit under any other of Guitart et al.'s categories to enable clarity in comparing the findings.

4.3.2 Case Study Approach

After the bulk of my quantitative systematic review was complete, I adopted a case study approach to focus on student-led food growing in universities in the United Kingdom. A case study is "an empirical inquiry that investigates a contemporary phenomenon within its real-life context; when the boundaries between phenomenon and context are not clearly evident; and in which multiple sources of evidence are used" (Yin, 1984, p. 23). Action research studies are usually case studies because they both focus on an "intensive analysis of a single case" or series of cases (Bryman, 2012, p. 709; Dick, 2014). Action research partners well with a case study approach because they both take place in the field, tend to be holistic, and acknowledge the importance of context (Dick, 2014). Using case studies can enable the development of a rich, in-depth understanding of the phenomena by retaining their "holistic and meaningful characteristics" (Bryman, 2012; Yin, 2003, p. 2).

Case study design is not as prescriptive as many other study designs (Yin, 2014), however there are four main types of case study designs, depending on whether there are one or multiple cases, and whether they are holistic or have multiple units of analysis. Table 9 shows these four different types of designs and gives an example of each. I used a single case study approach with three embedded units of analysis (see Figure 10). A single case design was chosen because the research question allowed for the examination of the impacts of transience in student-led food gardens in England as a whole unit. Three gardens were used as embedded units of analysis. Four gardens were approached, however only three ended up being part of the study. The reasons for this will be explained in the following section.

Embedded units of analysis were used was because there were too may gardens funded through Student Eats to be able to conduct action research in all gardens in the depth that such a research approach requires. The gardens were chosen using literal replication,

Table 9. Basic types of designs for case studies, including examples of each (Yin, 2014, p. 50). A single embedded case study is used in this thesis.

	Single case designs	Multiple case designs
Holistic (single unit	One case embedded within a context	Multiple cases within multiple contexts
of	A simplified example:	A simplified example:
analysis)	Case: student-led food gardening in English	Cases: student-led food gardening in English,
	higher education institutions	Canadian and Swedish higher education
	Context: English higher education	institutions
	environment	Contexts: English, Canadian and Swedish
		higher education environment
Embedded (multiple	One case with multiple units of analysis within a context	Multiple cases with multiple units of analysis within multiple contexts
units of		
analysis)	A simplified example:	A simplified example:
	Case: student-led food gardening in English	Cases: student-led food gardening in English,
	higher education institutions	Canadian and Swedish higher education
	Embedded units of analysis: student-led food	institutions
	gardens in English higher education	Embedded units of analysis: student-led food
	institutions	gardens in English, Canadian and Swedish
	Context: English higher education	higher education institutions
	environment	Contexts: English, Canadian and Swedish
		higher education environment



Figure 10. A visual depiction of the relationship between the 'case,' it's context, and the embedded units of analysis. The 'case' is embedded within a wider context, and sometimes embedded units of analyses are employed when a specific phenomenon within a case is to be examined in greater detail (Yin, 2014).

where units are chosen on the basis of their similarities in order to corroborate each other

and enhance the credibility of the findings (Yin, 2003). This means the gardens were not

chosen for comparative purposes, and therefore the results will be presented with limited

comparison between the gardens. The 'core' action research cycles operated separately at

each garden, however, in the 'thesis' cycle, the data from all the cases were pooled and they

were used to examine Student Eats as a phenomenon. The case and the embedded units of

analysis within their wider context will be explained in the next section though the example of

a single, embedded case design from Table 9 is a rough illustration.

4.3.2.1 The case study

For my case study, I focused on student-led food growing in universities in England funded by the National Union of Students' Student Eats programme. The reason for choosing to examine only those affiliated with Student Eats was because it meant there was coherence to the case: they all received similar funding and had similar ambitions for their gardens. It also meant that the gardens were easy to identify and get in contact with. The National Union of Students is a confederation of students' unions in the United Kingdom and had a sustainability team that focuses on social and environmental responsibility⁹. The sustainability team ran a range of different programmes focusing on, for example, pro-environmental behaviour change, embedding sustainability in the curriculum, and reducing the environmental footprint of universities and their students' unions. One of these programmes is Student Eats, which supports "institutions across the United Kingdom in cultivating their own student-led growing sites for fruit and vegetables" (NUS, n.d.c, emphasis mine). Prior to Student Eats, the National Union of Students ran a programme called Freshers' Freshers supporting seven student-led food growing initiatives in universities. Freshers' Freshers acted as an unintentional pilot for Student Eats, as the National Union of Students subsequently received funding of £315 000 from the Big Lottery's Local Food Fund for Student Eats. Some of this funding was distributed to universities and their students' unions by the National Union of Students in 2012 through their own funding call of approximately £10 000 (see Appendix B). Eighteen sites were funded, all of which were in England, fourteen of which were at universities (higher education institutions) and four of which were at colleges (further education institutions) (NUS, n.d.b). There were institutions with food-growing sites for

⁹ At the time of publication, the sustainability team had split from the rest of the National Union of Students to for its own charity, called Students Organising for Sustainability. They have since continued to deliver the same portfolio of activities as they had done previously.

students prior to the funding (such as Staffordshire University, n.d.), however the funding fed the growing interest in student-led food gardens in universities. Subsequent funding initiatives from the National Union of Students also helped spur on engagement, with a £5 million Students' Green Fund supporting several campus food growing initiatives (NUS, n.d.e), and the £998 000 Student Eats Food Enterprise funding, which had already supported the setup of approximately seventy student-led food enterprises throughout the United Kingdom at the time of publication.

Based on personal communications with staff at the National Union of Students, and their own publication (NUS, n.d.c), the ambition of the initial Student Eats funding from 2012 were to:

- Build physical infrastructure where food is grown,
- Sell produce or value-added products to sustain the initiative financially,
- Increase availability of fresh, local food on campus,
- Strengthen links with the local community,
- Facilitate student learning about sustainable food,
- Provide access to 'ethnic and exotic' crops for international students and students from different backgrounds,
- Engage hard-to-reach students,
- Create a network of growing sites,
- Create pro-sustainability changes in students' food attitudes and behaviours,
- Increase students' employability,
- Produce food in an environmentally sustainable way, and
- Connect students to nature.

Much of the focus of the initiatives was on students learning about growing healthy, environmentally-friendly, and ethical food (NUS, n.d.c). Importantly, the initiatives were meant to be student-led since they were funded by the National Union of Students. However, in practice, some gardens were coordinated or even run mainly by university or students' union staff.

The small number of potential study sites (subunits of analysis) for a qualitative study like this means that a purposive approach to selecting the study sites was necessary (Cresswell, 2014). The three student-led food gardens were chosen as embedded units of analysis in consultation with a staff member at the National Union of Students. The two criteria used to decide which gardens to choose were the extent to which the gardens were successful and student-led. A staff member from the National Union of Students advised on this matter. Student-led gardens were chosen because of the emphasis on students' transience in the research questions. I included 'success' as a criterion for my study given that I wanted to work with gardens that had the capacity to engage with a research project, and suspected that if the initiatives were struggling, they may not have had the capacity¹⁰. The fact that the gardens used as embedded units of analysis were the most student-led initiatives meant that they were also exemplars of the issues associated with student transience¹¹. Four gardens were recommended, three of which became the focus of this study. Initially all three agreed to participate, however, I ended up not pursuing one of the cases because after initial contact was made, communication dwindled and, through social media, I discovered the garden was not as student-led as had been thought. All the gardens were at universities, likely because the higher level of staff support for the gardens at the

¹⁰ The focus on 'successful' gardens is a methodological issue I will take up in Chapter 9.

¹¹ In section 9.2.3.5 I will problematise this idea of 'exemplifying issues associated with student transience,' and discuss the extent to which these case studies genuinely did this.

further education institutions which meant they didn't meet the student-led criterion. One of the gardens was my university, Keele, and since I was the main contact on behalf of my university, I did not need further permission to study it. I contacted the other gardens with contact information provided by the gatekeeper at the National Union of Students to invite them to participate (see Table 10).

Table 10. Description of the gardens studied.

	Keele University	University of Warwick	University of Sheffield
Size of university	Approx. 10 000 students	Approx. 25 000 students	Approx. 30 000 students
Location of university	Rural	Rural	Urban
Extent to which the garden was student-led	 Mainly undergraduate participants Supported by a PhD student (myself) Some support from grounds staff (e.g., mowing lawns) 	 Mainly undergraduate participants Supported by a PhD student Some support from grounds staff (e.g., provision of compost) 	 Mainly undergraduate participants Limited postgraduate participation Some support from grounds staff (e.g., provision of compost)
Garden characteristics	 Eleven raised beds of approx. 10 m², two of which run by students, the other 9 rented out to staff and campus residents Located within a historical walled garden with a locked gate Students had access to a large greenhouse run by the biology department 	 Roughly 18 raised beds (approx. 2m²), a polytunnel (approx. 15 m²), several of small plots and borders dug directly into the ground, and a shed Located behind student residences in a low-traffic area on campus 	 Six raised beds (approx. 3-4 m² each), several beds dug directly into the ground (approx. 12 m² total), a shed, a small glass greenhouse, and a pond Located in between other university buildings and residences, next to an open green space

4.3.2.1.1 Embedded unit of analysis 1: Keele University

Keele University is a small (approximately 10 000 students), rural, post-war institution with a broad-base of studies, including health, natural science, social science, and the humanities, on a 600-acre campus. The institution considers itself to be research intensive with a strong educational profile. Most students and many staff live on the campus, meaning there is a residential community near the garden. The garden was set up through the Estates and Development Directorate in a disused historical walled garden of a family estate the university had been built on (see Figures 11a and 11b). The raised beds took up a small



Figure 11a. A Google maps satellite image of Keele University's walled garden.



Figure 11b. A Google maps satellite image of the raised beds within Keele University's walled garden.

proportion of the whole walled garden space. Within the garden, much of the space was used by the estates team as a space for storage (for items such as unused bricks, wood chips, equipment, etc.), and for a greenhouse operated by the biology department. A large, closed off area contained disused and decrepit bothies (basic housing for farm labourers) and Victorian greenhouses.

Ten 1m x 10m raised beds were set up initially to be let out to staff, university residents and groups (such as the international office and the chapel) before the Student Eats funding was awarded. It initially had a traditional allotment style set-up with several communal beds let out to groups. When the Student Eats funding was awarded to the institution, an eleventh bed was built, and the international office's bed was given to the newly formed Student Eats society.

The Student Eats funding had been applied for by a university staff member, so substantial effort was made to recruit students to the project and set up a student society for the garden at the students' union. For the first few years the Student Eats garden ran as an independent society, and then after several years students decided to merge it with 'Think:Green,' the student sustainability society, as one of a portfolio of different initiatives they ran.

Over the years, the space was developed further by students, with the creation of a social space with foraged tables and benches, a composting system using wooden pallets, and a small (1.5m x 1.5m) plot that was dug directly into the soil. Students and other bed-users initially had access to the greenhouse, however after several years the space in the greenhouse for students diminished and was used for storing plants for the estates team.

The university grounds team mowed the lawns several times a year, would occasionally engage in dialogue with student gardeners, and would often let students use some of their unused bricks or pallets, and take their grass clippings or leaves to compost.

They occasionally had conversations with students about how to use and develop the space but provided limited support to students for that.

The students' union supported the garden in the same way they supported other societies. They did this through reminding them to have elections, providing avenues for their society to get more funding, providing a webpage for the society through the students' union website, and running 'freshers'' and 'refreshers'' fairs to recruit new students to their society. On several occasions students' union staff attended the gardening sessions.

I began working as a Sustainability Project Officer at the university while the final raised bed was built in the winter of 2013. Part of my role was to help support the emerging Student Eats society in establishing the initiative. I was due to start my PhD later that year, so my initial involvement was as a staff member of the university (though at the time I was still a student completing my Master's thesis at another institution).

4.3.2.1.2 Embedded unit of analysis 2: University of Warwick

The second garden was at the University of Warwick, which is also a rural university but more than double the size of Keele (in terms of student numbers) with approximately 25 000 students, a large number of which live in on-campus accommodation (University of Warwick, 2019a, 2019b). Warwick has a similar academic profile to Keele, including teaching and research in medicine, science, social science and the arts (University of Warwick, 2019a). Established in 1964, the university also considers itself to be research intensive with a strong educational profile. The garden (which students called 'the allotment') was small, fenced-off, and in a secluded area behind student accommodation. It was managed entirely collectively (see Figure 12). There were roughly 18 raised beds (1m x 2m), a polytunnel (5m x 3m), and several of small plots and borders dug directly into the ground. The Student Eats funding was used to set up the garden's physical infrastructure (which did not exist prior to the funding). The garden was entirely student-run with mostly undergraduate participants, however, there

were some postgraduate students and university staff involved as well. Notably, a doctoral student had an informal leadership role within the garden throughout the duration of my research engagement. The university's grounds team also supported the initiative by supplying compost.



Figure 12. A Google maps satellite image of the University of Warwick's allotment.

4.3.2.1.3 Embedded unit of analysis 3: Sheffield University

The third garden was at the University of Sheffield, a city-based, multi-campus, research-intensive university established around the turn of the 19th Century. With approximately 30 000 students, the fields of study at the institution mirror the other two cases, with the addition of engineering. The garden was established using the Student Eats funding, and consisted of six raised beds (approx. 1m x 4m), four tiered beds on a hill (approx. 1m x 4m), beds dug directly into the soil on the south side of the garden, a shed, a small glass greenhouse (approx. 1.5m x 1.5m), a pond, and a strawberry patch. Throughout my research engagement with the garden, it was entirely run by undergraduate students and therefore best exemplified the trait of 'students' transience,' though the university's grounds team also supported the initiative by supplying compost. The garden itself sat in between other

university buildings and residences, next to an open green space.



Figure 13. A Google maps satellite image of the University of Sheffield's allotment.

4.3.2.1.4 Stakeholders of student-led food gardens

There were a variety of participants and stakeholders involved with the gardens. The main focus of this thesis is on the student participants, which were mainly undergraduate students. These tended to be the most transient of the participants. Within this group there were also exchange students who were especially transient and would typically leave the university to continue their studies after one or two semesters. There was also a very high proportion of international students involved with two of the case study gardens, some of which stayed in the United Kingdom over summer holidays. A handful of postgraduate students were involved in all three of the gardens, though most were involved in passive or irregular capacities. The exception was at Warwick where there was a doctoral student who took on an informal leadership role, similar to the role I took on at Keele.

University and students' union staff were involved in different capacities in the

different gardens, as outlined in the above sections. Gardeners from two of the gardens also

had encouraged university and students' union staff to take part outside of their formal

responsibilities, but this typically only happened during the summer and in an irregular

capacity.

Broadly, I identified four main types of people taking part in the gardens. These are depicted in Figure 14. These typologies are not expected to be transferable to other projects, or even complete understandings of the typologies in the three gardens studied. They are based only on my interactions and observations, and therefore only representative of the people who were engaged in my research. As such, these will not be used as an analytical tool

Long-term mentors

The gardeners in this group were doctoral students, like myself, however in some gardens there were also students' union or university staff members who performed similar functions. They were based at the university throughout the year. These people typically had some gardening knowledge and, because of their long-term involvement in the project, knew about the history of the space, the crop rotation, the various gatekeepers, and organisational structure of the university. Other group members would typically rely on these volunteers for both strategic planning of the space, as well as guidance of what to do on a day to day basis.

Student leaders

These gardeners were typically undergraduate students involved in the project quite intensively, often having taken on formal roles within the student society. Some of these students had some knowledge about gardening, and those that did not had an interest or curiosity in it. Given that these students were mainly undergraduates, they tended to have quite specific patterns of involvement as outlined in section 2.4.1.



These gardeners were less involved in the gardens than the student leaders, but more active in taking initiative and determining the direction of the project than passive volunteers. These gardeners may have also been an active part of other student societies or other activities outside of their academic work, which is why they chose not to take on a leadership role in their Student Eats garden. That being said, some students from this typology later went on to fit the 'student leader' typology. These volunteers were typically happy with informal or formal activities.

Passive volunteers

These gardeners typically invested much less of themselves in the project than the other types of volunteers. They may or may not have had gardening experience or an interest in gardening. Their motivations for taking part ranged from socialising, to learning to garden, to getting volunteering hours for an award or their CV. These volunteers typically wanted a more formal volunteering opportunity.

Figure 14. Typologies of student volunteers involved in the student-led food gardens studied. These are to be used to give context to the quotes provided in subsequent chapters.



or presented as a finding; rather they are used solely to give context to statements made by participants whilst preserving a higher degree of anonymity. In future research, it may be of interest to explore the extent to which these typologies hold up under scrutiny with a larger sample of cases and participants.

4.4 A student gardener studying student gardens as a researcher where she works

Reflection on positionality is considered to be an important part of action research (Coghlan, 2019). While I would not want an exercise in reflexivity to become a "self-indulgent focus on the self thus distancing the researcher from the research focus itself" (Kobayashi, 2003, p. 349), I do think it is important for the purpose of transparency to outline some key aspects of my identity that may have had bearing on the study. These include that I am white, female, able-bodied, and from an upper-middle class family. At the time of fieldwork, I was in my late twenties. I hold British and Canadian citizenship but grew up in Canada. As a British citizen, I am not considered to be an 'international student' as I did not need a visa to live, work, and study in the UK. However, I am often perceived to be an 'international student' because of my Canadian accent. I have been involved and set up student gardens during my undergraduate and Master's studies, and see myself as a student gardener (although this has changed over time as discussed below).

The parts of myself that require attention are the multiplicity of roles I embodied within the study and my student gardener identity. The main roles I embodied during the study were 'researcher,' 'staff member,' and 'student gardener.' At times these roles overlapped, conflicted and became muddled. I often felt myself shifting between these roles when in interviews, workshops, and whilst gardening. When I moved to the UK and began my professional role at Keele University as a Sustainability Project Officer in early 2013, part of

my job was to support a student group to manage the student beds in the walled garden. Later that year, after I graduated from my Master's programme, I officially became a doctoral student. However, from the beginning I had a personal interest in the garden and tending to it, which at times, went beyond my normal job role. For example, I tended to the garden much more actively during the summer while students were away, dropped in to weed on lunchtime walks, and occasionally came in on the weekends to garden. At times, I felt myself to be just another one of the students gardening at Keele, but at other times I felt myself, and was perceived to be, a staff member acting on behalf of the university. In my research diary, I reflected:

In my own garden at Keele, I clearly occupy an insider role because I am very involved in it in my Sustainability Project Officer role. However, it's not as simple as that, because it is actually often described as a 'student-led project,' and since I am active in the garden as staff, I am technically not 'in' the garden, so to speak. But I still feel that I am an insider because, of all the people involved in the garden, I probably spend the most time with it (both in terms of actual gardening, as well as in an administrative capacity). Furthermore, since I am engaged with it in a more continuous way (having worked with it already for a year and a half, expecting to work with it for at least three more years, and staying over the summer) I think the best way to describe me as an insider, but with a caveat – an insider with some outsider elements as a result of my employment status.

The relationship-building with gardeners that happened while I was a 'student gardener' at Keele led participants to be franker and more open with me in interviews and workshops. My insider knowledge, having tended to the garden with other gardeners, allowed me to ask more nuanced questions that were rooted in our shared realities.

Furthermore, as a new staff member, I experienced many of the challenges students also faced in engaging with 'the university.' For example, in the beginning I struggled to navigate the university's organisational structure. Who was responsible for what? How should I go about finding something out? However, as time progressed, this 'student gardener' identity started to shift as I embraced my professional role at the university and learned my way around the university structures. It was then that I started to see the gardens and how they were managed in a different way, and as such, asked questions that corresponded more to the perspective that most university staff members would have. For example, in workshops I questioned how realistic some students' proposed activities were given the student lifespan in the garden, which I would have been more supportive of earlier.

My roles at the Universities of Warwick and Sheffield were much simpler and remained static over time. I was an outsider to their projects as a researcher, though as a fellow university gardener I was perceived as a sympathetic outsider. Because of my outsider role, I was not as emotionally tied up in the success of their initiatives. This afforded me some reprieve from the struggle of balancing my researcher, staff member, and student gardener roles during data collection, which allowed me to be more relaxed and attentive in interviews and workshops. Furthermore, when the gardeners from Warwick and Sheffield engaged in my research, they did so out of curiosity or, as I reflected in my research diary, wanting to *contribute to something bigger*, in contrast to the participants at Keele who felt a sense of obligation to take part because they knew me. As such, participants from Warwick had more independence in carrying out the 'actions' planned through the research because they were not doing them for me, but for themselves.

Throughout my doctoral work, as can be seen in this section, I have been operating in a liminal state between being a student and scholar, in which I occupied multiple overlapping roles. This boundary spanning role (Adler et al., 2009) gave me unique insights to student-led

food gardens that would not have been accessible to a more senior researcher, nor to a fulltime doctoral student. This was complemented by my outsider role as a researcher in the gardens at Warwick and Sheffield in which my emotional distance from the initiative allowed for more impartial inquiry. Further reflection on my positionality in section 9.2.2.

4.5 Research stages and main methods

The action research approach guided the overall structure of the study. I attempted a construct-plan action-take action-evaluate process in each garden. These cycles were undertaken in the gardens at Keele and Warwick, however, the participation at Sheffield dropped off so dramatically in the autumn of 2015 that I made no attempt to support the participants in planning or taking action.

Within the action research approach, several methodologies and methods were used to triangulate the findings (Bryman, 2012; Cresswell, 2014). I will first give an overview of the data collection methods used and then explain the research timeline in a more narrative style, explaining context-specific methodological concerns and approaches to analysis.

The principle method I used was the semi-structured interview, a widely-used method in qualitative research (Bryman, 2012). Semi-structured interviews are a method where research participants are asked questions according to a pre-defined set of themes or questions, but "the interviewer usually has some latitude to ask further questions in response to what are significant replies" (Bryman, 2012, p. 716). They are neither as rigid at structured interviews, which means that the interviewee can help shape the interview direction meaning that emergent lines of inquiry can be exploited, nor as unfocused as unstructured interviews, which can produce nebulous, confusing, or off-topic data (Bryman, 2012). The interviews in the study ranged between approximately twenty minutes and two hours in length, and in total there were seventeen conducted between all three case study gardens. Interviews were an appropriate method because they allowed the participants to construct their answers to the questions on their own terms, and to challenge the questions they were being asked, as well as the overall direction of the research. It also allowed for the data collection process to be more of a negotiation of meaning in which I, as the researcher, was required to justify, rationalise and adjust my research direction as appropriate. One example was when one interviewee expressed frustration that early stages of the research were so problem-centred and they were concerned that this would disempower students and discourage participation. Upon discussion, I found that it was mainly in the written material I had sent participants that had prompted this reaction. I therefore made sure all future interactions with research participants were framed in a solutions-orientated manner, including written material. A further discussion of this will take place Chapter 9.

Interviews were useful in this study because it meant that the interviewees (such as those from the gardens at Warwick and Sheffield, and from the NUS) could provide historical information I could not have otherwise uncovered (Cresswell, 2014), and they also provided insight into how participants experienced the impacts of transience and the extent to which they understood its impacts. In one case I used a go-along interview, which is when "fieldworkers accompany individual informants on their natural outings, and – through asking questions, listening and observing – actively explore their subject's stream of experiences and practices" (Kusenbach, 2016, p. 154). This allowed for in-situ inquiry and for me to help the gardener complete their tasks while they took part in my research. This said, some of the weaknesses of the approach was that information was "filtered through the eyes of the interviewees" and may not have represented other realities, and that as "not all people are equally articulate and perceptive," this can affect the quality of the data (Cresswell, 2014, p. 191). The topics covered in the interviews and types of questions that were asked varied

between interviews because of the shifting focus of the research. As such, the topics of the interviews will be discussed below as I discuss each stage of the research in turn.

During the entirety of the case study component of the research, I used a research diary. Research diaries are widely used in action research for a variety for purposes (Reece, 2014). One way I used my research diary was to log factual information, such as descriptions of activities or events that took place, for recall at a later date (Reece, 2014). I also used my research diary as a tool to work through and reflect on the content, process and premise of my research (Coghlan & Brannick, 2005) in order to be reflexive on a practical, epistemological, and ideological level. Prior to beginning my research diary, I constructed a series of questions to respond to, based on journaling exercises from Coghlan and Brannick (2005). These questions can be found in Appendix C. My research diary also served as an outlet for expressing frustration, worries, disappointment, and fears about my research in a safe way. Being able to write about these things allowed me to express them and 'park' them so that I was able to reduce their influence on my facilitation or bias my research decisions. I have always used writing as a means to work through my thoughts and feelings, so this approach felt like a natural and unforced way to engage in a reflexive practice.

4.5.1 Sampling and recruitment of research participants

Because the study took place over a long period of time with a transient population, recruitment activities were ongoing. There were two levels of sampling: sampling of embedded units of analysis and sampling of participants (Savage, 2005). I chose to sample gardens (embedded units of analysis) that exemplified the trait of being 'student-led' and that were successful. This was outlined in section 4.3.2.1. To sample participants I used purposive sampling, which is to choose participants in a strategic rather than random way (Bryman, 2012). The populations I sampled from were gardeners in the three gardens to be studied and

staff working with Student Eats at the National Union of Students. Some participants I had previous relationships with, while others I did not. As a result, I used different purposive sampling strategies. For some, I recruited participants from "particular subgroups of interest" (participants from each of the most student-led food gardens and staff at the National Union of Students) (Bryman, 2012, p. 419). I also used opportunistic sampling (Bryman, 2012) by attending garden sessions at University of Warwick and recruiting participants that happened to be in attendance. At Keele, I used typical case sampling of participants that I knew had dropped out of the initiative to better understand why they had dropped out although they were still at the university (Bryman, 2012). Those interviewed from the garden at Sheffield University were the only potential interviewees available at the time given low participation.

The question of adequate sample size in qualitative research is a subject of debate (Bryman, 2012). Generally speaking, sample sizes should "not be so small as to make it difficult to achieve data saturation" nor "so large that it is difficult to undertake a deep case-orientated analysis" (Onwuegbuzie & Collins, 2007, p. 289). Data saturation is "when no new information is discovered in data analysis, and this redundancy signals to researchers that data collection may cease" (Faulkner & Trotter, 2017). I believe that data saturation is a subjective ideal towards which we strive, and I contest the idea that data saturation can ever be fully achieved, at least in research in which the 'general idea is allowed to shift' (Elliot, 1991). This is because the data collection and analysis could continue forever as the researcher continuously uncovers new nuances that shift the research focus. However, at some point the researcher must choose to stop collecting and analysing data, and begin writing up their findings. I had a high level of data saturation in my study as I was repeatedly receiving the same answers, in spite of having a small number of interviews and leading a small number of workshops. This is somewhat unsurprising as students had short lifespans as participants within the gardens, and therefore lacked insight into the issues studied (a

phenomenon which will be unpacked further in Chapter 6). Therefore, I would expect that increasing the number of students interviewed or the length of time spent collecting data would not have provided new information. Longer-term stakeholders, such as doctoral student gardeners or staff members of the National Union of Students offered considerable insight, however the number of these stakeholders were limited (though some long-term stakeholders were interviewed twice, as relevant).

4.5.2 Recording and transcription

All interviews and workshops were audio recorded and manually transcribed myself because it allowed for a more "thorough examination of what people say" (Bryman, 2012, p. 482). Whilst transcribing, I also made analytic memos (Saldana, 2013) to record any reflections or 'aha' moments as I went along. While some participants were perturbed by the use of a recording device, most were immediately comfortable. One person expressed concern about the use of the recording device and the use of direct quotes. In this case, I assured them that I would not quote them without checking with them first. The interviewee was forthcoming and natural both times I interviewed them, although occasionally corrected themselves and asked for particular sections not to be quoted. Another interviewee's demeanour immediately changed when the recording device was on, becoming uncomfortable. At times during the interview, I switched off the recording device to allow for more natural conversations and interactions. During the interview, another interviewee chose to write information on a piece of paper that they deemed to be too sensitive to be recorded. Unfortunately, the audio recording for one interview was corrupted, however most of the interview was recorded by a second device. This partial recording was transcribed in lieu of the full recording.

The choice to record and transcribe was fairly straightforward because there were many benefits to the quality of the research. Of course, much of the information conveyed in interviews has to do with *what* is said, but also crucially important is *how* participants say what they say (Bryman, 2012). This includes tone, body language, gestures, and eye contact, for example. If the interviewer's focus is on note taking, they may miss these subtle cues that can help the interviewer prompt and probe for further understandings (Bryman, 2012). Transcription inevitably 'flattens' reality, which is a weakness of the approach (Crichton & Childs, 2005). Although not fully remedying this issue, I addressed this by revisiting the audio recordings during data analysis to help to reanimate the transcript in my mind. While I did not re-listen to all recordings, I listened to some of the workshops several times after transcribing because of the complexity and multi-layered nature of the conversation that took place. This also allowed for quality checks of the transcription (Bryman, 2012). Another benefit of transcription is that it "allows the data to be reused in other ways from those intended" (Bryman, 2012, p. 482). This was crucial in this study given my commitment to reflexivity that allows the overall research agenda or question to shift.

4.5.3 A messy start: Constructing

Early stages of my research were more informed by the pragmatic tradition, and as my understanding of the questions and issues the student-led food gardens were grappling with grew, my research shifted towards a more critical orientation. My original orientation was towards providing practical knowledge developed through experiential learning that could be used by student gardeners and the National Union of Students in their support for studentled sustainability initiatives (more akin to the pragmatic tradition). As my engagement in the initiatives deepened, I began to see the heterogeneity within the gardens' volunteer populations, the conflict between long- and short-term stakeholders, and need for

consciousness-raising among the different stakeholder groups. This marked a shift towards a more critical orientation.

Early iterations of my research questions were guided by an apparent need to assess the environmental sustainability of student-led food gardens (as discussed in the results of the systematic literature review in Chapter 5). It was with this agenda that I applied for ethical approval (see section 4.7 for a discussion of ethics in this study and Appendix D for ethics approval letters). The ambition of my research at this stage was to use an action research approach to support the participatory development of indicators for social and environmental sustainability in sustainable university community gardens¹². This being said, I was conscious of how much students' transience impacted these gardens. However, rather than seeing it as a dominating issue that needed examination, I saw it as background issue that the students needed to manage in their gardens, and a methodological issue that I needed to manage in my research.

In October and November of 2014, I interviewed three staff members at the National Union of Students who had been working with Student Eats. These interviews were semistructured and exploratory. They covered the following themes:

- The history of Student Eats
- The aims of Student Eats as a project and the gardens themselves
- How they were measuring and assessing the impact of Student Eats
 - The purpose of the evaluations of the scheme that were being undertaken
 - The extent to which they felt their evaluations were effective and what could be improved

¹² I felt the development of both social and environmental indicator sets would be more useful to create a holistic tool that could be used not only to assess social and environmental sustainability, but also to manage trade-offs.

- How they understood the term 'sustainability'
- Identifying gardens to use as embedded units of analysis

Two of these interviews took place in the National Union of Students' office, and one took place over skype. They ranged between 30 minutes and two hours. Following these interviews, I designed a workshop at Keele to begin developing indicators for environmental sustainability in student-led food gardens. However, students struggled to come up with useful and practical indicators for their vision, mainly because students struggled to see the point of it. This is what led to the most crucial revelation of the workshop, which was that all tasks in the garden were subordinate to one crucial task, which was to maintain sufficient participation levels in the face of a transient student population. This revelation marked a shift in my research agenda.

Over the course of the next few months I reformulated my research questions and amended my ethics application (see Appendix D). I interviewed a student who had taken part in the workshop at Keele to gain further insight into the issues discussed in the workshop. I also found Silva's (2013) recently published report on the different models of engagement for monitoring in urban stewardship, which led me to a data collection toolkit for monitoring the outcomes of urban food growing (e.g., the amount of food produced, environmental, health, and economic outcomes) (Design Trust for Public Space, n.d.). I thought this data collection toolkit could be used to track the vital signs (e.g., levels of participation, food produced, change in attitudes) of the gardens before and after the interventions took place as a way to assess whether the interventions had an impact. I introduced this at Keele progressively during ordinary gardening sessions in early 2015, at the garden at Warwick at an introductory workshop in January 2015, and at Sheffield during one of their gardening sessions in February 2015. This data collection strategy was abandoned in the Autumn of 2015 as the data was
inconsistently recorded, it was labourious, and provided little insight into the garden's vital signs due to the inconsistency of recording.

After this, I came to the conclusion that I had begun the research with too many preconceived ideas of what challenge needed addressing in these initiatives and the level of engagement in my research that the gardens could sustain. As a result, I scaled my approach back and conducted five interviews (one at Keele, two at Warwick, and two at Sheffield) and three workshops to better understand issues associated with transience and participation at Keele¹³ (November 2015, eight participants) and Warwick (December 2015, five participants). The workshops were based on Vaughn and Lohmueller's (1998) group level assessment method. The method is qualitative and participatory group research method where data is collaboratively generated and analyzed by large group of stakeholders (Vaughn & Lohmueller, 1998; Vaughn et al., 2011). The purpose of the method is both to research through producing data, but it also is intended to create participant ownership and bring groups of people together (Vaughn & Lohmueller, 1998; Vaughn et al., 2011). The workshop, as originally conceived, includes seven steps and has more focus on collaborative and participatory analysis. I amended the workshop to fit the needs of my study and the student participants involved. This meant the workshop was shortened from the recommended three hours to two hours, and the focus was more on brainstorming and thinking through of the main issue(s) in the gardens collectively. The workshops were quite flexible in their structure as some students arrived late and others needed to leave early. In the beginning, I introduced myself and my study to the group as there were usually new participants present. There was

¹³ The two workshops conducted at Keele in November 2016 were extensions of the same activity. The first workshop was poorly attended so I ran it again the following week. In the first week participants answered questions written on paper taped to the greenhouse walls, but it was not audio recorded as I chose not to facilitate a discussion given the limited number of participants. The discussion I facilitated on the second week was audio recorded.

then some time spent talking about, reading and filling in the information sheet and consent forms (more on this in section 4.7). After this, the main part of the workshop started, where participants walked around the room responding to questions on the posters. There was approximately 20 minutes spent on this. The questions were designed so there was overlap between them make sure all the main reflections on the topics were captured, with same question being asked several times using different phrasing (see Figure 15a and 15b). After this first part, participants were invited to walk around and silently read and reflect on others' responses for 10 minutes. For the remaining time (approximately an hour and 45 minutes at both gardens) was used for a discussion of the posters and the main issues present in the gardens. At Keele the workshop was held in the greenhouse in the walled garden, while at Warwick the workshop was held in a small seminar room.

Following this, I began the first of two phases of data analysis. I transcribed all audio data and photographed all the posters with written responses. I then compiled the data, including analytic memos, in the Computer Assisted Qualitative Data Analysis Software programme, NVivo. I used thematic analysis as a strategy to make sense of the data, and coded according to the main questions that I was posing to participants at the time of the research, which were:

- How students' transience impacted the gardens
- Causes of problematic participation
- Effects of problematic participation
- Solutions suggested¹⁴

I coded using a number of different methods over the course of two cycles, a process called eclectic coding (Saldaña, 2013). Some of the data were coded as small chunks of

¹⁴ While I did not solicit solutions in interviews or workshops, many participants volunteered them. As such, I chose to code them.

WHAT IS THE BIGGEST CHALLONG THE GARDEN PACES? - CONTINUITY (Marginal the year and between years) // - Regulazity rat so much to do in the winter maybe the much in spain? - The (EVIL) UNIVERSITY MANAGMENT TEAM WANTING TO BUILD A PARKING LOT - Regurancy uduntees, who an -INConsistent effort from Istudents who have many and varying commitments

Figure 15a. Written responses to the question, 'What is the biggest challenge the garden faces?'.

IF THE GARDEN WERE TO FAIL, IT WOULD BE BECAUSE NO ONE TO KUN IT NEXT YEAR The uni wanted to durdep on that land. Lack of appreciation (as mentioned Tabove) sud input our (21st Century) cultural numbers towards natural things llot orough members

Figure 15b. Endings to the sentence, 'If the garden were to fail it would be because...'.

information (called 'splitting') in the first cycle, and then grouped into themes (called 'lumping') in the second cycle (e.g., positive outcomes of student-led food gardens, and the problem of participation). Other groups of data were analysed by 'lumping' and then 'splitting' (e.g., issues related to transience, problems identified). For both types of data, the first cycle of coding was made up of a combination of holistic coding, *in vivo* coding, and simultaneous coding (Saldaña, 2013). Holistic coding is coding larger sections of text by capturing the overarching theme of the text, rather than splitting the codes out line-by-line (Saldaña, 2013), and this was the main method used when 'lumping.' Simultaneous coding, when the same passage of text is coded more than once, was necessary because of the complexity and messiness of the content and conversations. *In vivo* coding is when the research participants own words are used as the code (Saldaña, 2013). In line with the ethos of action research, I aimed to include as much of the participants' voice' as possible, hence the use of *in vivo* coding (Stringer, 2007). For data that was analysed first by 'splitting' and then 'lumping,' I used descriptive coding, which is when "the basic topic of a passage of qualitative data" is summarised in a word or short phrase (Saldaña, 2013, p. 88). I also complemented the descriptive coding with simultaneous and *in vivo* coding where appropriate.

In these early stages of the research, problematic participation and transience were often conflated by both the participants and myself because they were so interlinked. Often during the interview process, I would seek to explore how students' transience impacted the gardens, but this would quickly shift into a conversation about how to increase participation. Untangling 'transience' from 'participation' identifying their causal relationship became a key focus at this stage. Once it was established that transience caused short-term and irregular participation, a more fine-grained analysis of how transience interplayed with the participation dynamics in student-led food gardens was possible. Having coded in a relatively open-ended manner, I used causal loop mapping to make sense of the relatively long list of codes. Causal loop diagrams are analytical tools that can be used to demonstrate how different variables are related. They are appropriate for understanding wicked problems and complex systems where linear causal thinking is insufficient (Hjorth & Bagheri, 2006). The

process of mapping involved identifying the variables in the system (in this case, many of these were the codes or agglomerations of codes) and how these variables increased or decreased the occurrence of other variables. At this stage, I produced a report (seen in Appendix E) summarising the key findings and presenting them as a causal loop diagram (Figure 16). This causal loop diagram, originally constructed by hand on a whiteboard and then later translated to a digital version, was further refined in later stages of analysis. The final version can be seen in Chapter 6.

4.5.4 Planning and taking action

The planning action phase was much briefer than the constructing phase. It was made up of two workshops, one at Keele with thirteen participants and one at Warwick with five participants in February 2016. From the interviews and attempts to reach out to students at Sheffield, I knew that there was very low or possibly non-existent participation and therefore there wasn't capacity to engage with my research as actively at that time. The workshops at Keele and Warwick had two components: reviewing the findings from the constructing phase, and planning actions to address the problem(s) uncovered in the first phase. Prior to the workshops I sent the report to those who took part in the research in the 'constructing' phase and posted it in the gardens' social media groups. I invited feedback, including changes or additions, however no changes or additions were suggested in the feedback I received. For the workshop at Keele, I created a large-scale version of the initial causal loop diagram from Figure 16 (see Figure 17).

This was placed on the floor in the centre of the greenhouse for the duration of the workshop. There was not space for this for the workshop at Warwick, so we worked off paper versions of the report in Appendix E. For the first part of the workshop, I (re-)introduced participants with the information sheets and consent forms. Then I provided participants with



Figure 16. Initial causal loop diagram mapping the relationship between problematic participation, its causes, effects, and transience.



Figure 17. Post-it notes stuck on to a hand-drawn version of the causal loop diagram from Figure 16.

the report and invited them to read the report and provide any feedback, including changes or additions. Reviewing the report also (re)acquainted the participants with the issue the research was intended to address. In both workshops, there was not any dissent in the feedback, and the comments I received were relatively minor. Therefore, I led them to brainstorm ways to address the issues outlined in the report (see Figure 18 and 19). This was done in an informal discussion around a table at Warwick as there were only five participants.

At Keele, there were more participants therefore, the students were asked to brainstorm by independently writing their ideas down on post-it notes, and then everyone presented their top three to the group. In the end, rather than coming up with one or several larger actions to address together, at both gardens, students planned a portfolio of smaller actions to be taken individually. The actions that were actually followed through on will be discussed in Chapter 8, but a summary of all planned actions can be found in Appendix F.



Figure 18. The floor of the greenhouse where workshop participants were arranging post-it notes with suggested solutions to the problems outlined in the causal loop diagram in Figure 16.



Figure 19. The actions that were chosen to take, with the name of the person responsible and a deadline to complete the task. Names are censored to ensure anonymity.

I followed up with students at both Keele and Warwick several times to see how they were

progressing with them and to encourage them to follow through.

Towards the end of this phase I arranged an interview with a gardener from Sheffield University, who was the sole volunteer at the time. The purpose of the interview was to explore the status of the garden, both to develop a better understanding of a garden that had oscillated from being a highly successful student-led initiative to virtually inactive, as well as to explore the possibility to run an action planning workshop. I concluded at this stage it would not be possible to run such a workshop for the garden at Sheffield.

4.5.5 Evaluating

Throughout the evaluating phase, my main focus was on evaluating the actions taken, however I also continued to seek out input to clarify my understanding of the impacts of transience and the dynamics of problematic participation. I, first, attempted an evaluation workshop at Keele in late May 2016, but it was poorly attended and none of those who had signed up to take actions attended. In lieu of a full workshop, I conducted interviews with two students in the following days who were due to graduate in order to evaluate the impact of the actions and assess which had actually been implemented. At Warwick I successfully conducted an evaluation workshop in June 2016. In the workshops and interviews, we went through the list of actions that the students said they would take and (1) discussed whether they were followed through on, (2) if not, why, (3) if so, what the impact was, (4) if the action was effective in addressing issues associated with transience, and (5) what actions they would take in future.

Around the time of the evaluation workshops I began a photovoice component to my study. Photovoice is a participatory methodology that "asks participants to photograph aspects of their lives" so that the knowledge generated genuinely represents the views of the research participants (Bendell & Sylvestre, 2016, p. 357). The purpose of this was to gather data to assess the impact of the actions taken given that, as actions were being taken, it

became apparent that it would be challenging to assess their impact given the transience of students involved in the gardens and the difficulty in attributing changes in the garden to the actions taken.

Typically in photovoice studies, the first step is to train and prepare participants to take part by advising on the mechanics of taking photographs, issues of ethics and power, and to outline the topic of the project itself (see Appendix G). In the next stage, participants take photos responding to a set of prompts or questions and write an accompanying caption. In trying to reach out to a larger number of students, I visited gardening sessions to encourage participants to take part and provided information on social media in lieu of in-person training. However, only four participants took part in total, all of whom I had briefed inperson. Students sent me the photos and captions, and since only eight submissions were received I had them all professionally printed on photo boards. These photo boards were displayed at the Student Eats conference in February 2017, which three of the four photovoice participants attended.

Students were invited to take photos to answer one or more of the following questions:

- What does the gardening session look like in the [garden/allotment/initiative] this week?
- Is the participation (the quality/the amount) better than usual this week? Is it worse?
- What helps improve participation in the [garden/allotment/initiative]?
- What hampers participation in the [garden/allotment/initiative]?
- What do I want to tell other people about participation in the [garden/allotment/initiative]?

I completed the photovoice component of the study with a focus-group style workshop at Keele University. A focus group is like a group interview, but in addition to

eliciting information from participants, it is also allows for interaction between participants to be observed (Bryman, 2016). In the workshop I displayed their photos and we discussed what the photos were of and why the participants took them. We also talked about some of the actions that were taken from the action planning phase and how effective they were at addressing the issues associated with transience. Finally, I invited students to reflect on what the biggest challenges were for the garden and how they would address them.

Several more interviews were arranged to complement the other activities towards the end of the study. In May of 2016, I interviewed two students from Keele who had already graduated (one of whom had already been interviewed before) to explore their understandings of and thoughts on managing the impacts of transience and problematic participation in student-led food gardens post-studies. In August 2016, I arranged an interview with a staff member at the National Union of Students to report back on my initial findings and gather their perspectives again, given the shift in the research agenda.

At the Student Eats Conference in February 2017, where the photovoice photos were displayed, I also conducted a fishbowl discussion with seventeen conference attendees (see Appendix H for the discussion guidelines used). The purpose of this was to broaden the analysis to the scope of Student Eats as whole, rather than just the three gardens used as embedded units of analysis. This method was chosen instead of a questionnaire, for example, because of the limited number of Student Eats gardens at the start of the study (eighteen). At the time of data collection, a fishbowl discussion therefore seemed appropriate to broaden the analysis while allow for more participant-driven framing and discussion that a survey would not allow. Since the beginning of this study the number of Student Eats gardens substantially increased. At the time of publication there were over sixty Student Eats gardens.

A fishbowl discussion is a large-group dialogue method (Garrison & Munday, 2012). Typically, there is a ring of between four and five chairs in the centre of the room, which is

then encircled by larger ring(s) of chairs. Participants sitting in the inner ring are the only participants allowed to speak, but participants are free to move between the inner and outer rings depending on whether they would like to speak or listen. The advantage of the discussion is that a relatively intimate discussion is able to take place, whilst involving a larger number of people than a focus group, for example, would be able to accommodate. The weakness of the method is that not all participants may be willing to speak in front of such a large audience (Garrison & Munday, 2012). However, in this case the group was still quite small and therefore most participants seemed comfortable volunteering contributions. Participants in the discussion were invited to respond to the following questions:

How can we maintain sustainable participation given:

- The typical undergraduate degree is only three years, after which most students leave?
- Most student leave for the summer (and spring/winter) holidays?
- The ebbs and flows of academic workloads, course work and exams?

At the Student Eats Conference, I also interviewed a new student gardener from Warwick's garden to follow up on some of the content they brought up in the fishbowl discussion.

4.5.6 Bringing it all together

Following the evaluating phase, I began to consolidate all the data and findings into a coherent narrative. A second round of transcription and data analysis took place, examining all the data after the constructing phase. The approach taken was very similar to the first round of data analysis and was analysed in the same NVivo project. It built on the same sets of codes from the initial round of coding, but expanding with two new thematic areas:

- Solutions to problems identified
- Changes in the research focus

After this was completed, further analysis was undertaken for each of the research questions.

For the second research question, the initial causal loop diagram was mapped in The Brain software (see Appendix I), and then built upon using the new dataset in order to clarify nuances in the connections, and add new variables and connections. The Brain enabled a more complex analysis than a hand-drawn causal loop diagram would because of features that enabled zooming in and out on particular variables. It also provided different ways of visualising the connections that helped me to think through the relationships more clearly. Once all causal connections had been mapped, I was able to produce condensed versions of the causal loop diagram which are presented in Chapter 6.

For the third research question, I conducted another layer of analysis over the analysis I had already undertaken, using protocol coding. Protocol coding is coding to an already existing frame (Saldaña, 2018), which was, in this case, Biggs et al.'s (2012, 2015c) resilience principles. I created tables in Microsoft Word framed by the seven resilience principles, and recoded previously coded passages of text that demonstrated alignments or misalignments with the resilience principles. Given that earlier iterations of analysis had not been undertaken with the resilience principles in mind, I then returned to the transcripts and my research diary to identify any passages demonstrating further alignments or misalignments. The results of this are presented in Chapter 7.

Finally, for the fourth research question, I returned to the data coded in NVivo to examine the solutions proposed by participants and the actions actually taken. These were condensed into fourteen main strategies and actions, and then were cross referenced with the challenges outlined in the causal loop diagram in Chapter 6. This produced a table indicating which solutions address which problems. These solutions were also complemented with analyses emerging from the discussions in Chapter 6 and 7, to produce a table

summarising all advice for managing the impacts of transience and problematic participation, broken down by stakeholder group. These can be seen in Chapter 8.

4.6 Quality measures in mixed methods action research

Mixed methods studies can be challenging to apply consistent quality measures to as quantitative and qualitative research usually have different quality criteria (Bryman et al., 2008). As such, I will discuss the quality measures of the quantitative systematic literature review separately from the case study component.

4.6.1 Quality in the quantitative systematic literature review

Quality criteria for quantitative social research tends to be much more uniform than for its qualitative counterparts (Bryman et al., 2008). The traditional quality criteria for quantitative research are the extent to which the methodology is valid, reliable, replicable, and generalisable (Bryman et al., 2008). I assert that the methodology I used for the quantitative systematic literature review was of a high quality and outline why in the following paragraphs.

There are several types of validity: measurement validity, internal validity, and external validity. Measurement validity is whether or not a measure "reflects the concept it is supposed to be denoting" (Bryman, 2016, p. 41). Because of the limited number of articles included in the review, it is possible the articles do not accurately represent the benefits of sustainable university community gardens. As such, I temper my findings with this knowledge, and try to focus on the main trends rather than the minutiae. However, because the review was exhaustive, the findings do represent the current state-of-the-art. Internal validity has to do with causality (Bryman, 2016). The purpose of a quantitative systematic literature review is not to demonstrate causality, therefore this criterion is not relevant. External validity deals with the generalisability of the findings (Bryman, 2016). The external validity of this study is very high since the review was an exhaustive and systematic review of all articles about sustainable university community gardens.

Reliability is concerned with whether or not a study is repeatable (Bryman, 2012). That is, if the same protocol were to be undertaken again under the same circumstances, would the results be consistent? Assuming that the databases searched do not change their search algorithms this would be the case. One factor that could affect the reliability of the method is that different institutions have different agreements with publishers and therefore a researcher from a different institution may not have access to all articles reviewed. However, I did not exclude any articles because of lack of access.

Replicability is concerned with whether the study can be replicated, and therefore the methodology needs to be provided in adequate detail (Bryman, 2012). I have been sufficiently specific and detailed in describing the methodology I used such that it could be replicated. Because I used an established protocol (Guitart et al., 2012), I even demonstrated the replicability of the methodology. When there is analytical ambiguity in categorising the data presented in Chapter 5, I explain my choices and rationalise them. As such, it would be possible for other researchers to replicate the study with a high degree of precision.

4.6.2 Quality in the case study

A subject of heated debate in action research and case study research is how to demonstrate and assess research quality (see Bradbury & Reason, 2001; Feldman, 2003; Kyburz-Graber, 2004; Feldman, 2007). Part of this is due to the connotations that certain terms, such as validity and reliability, have (Heikkinen et al., 2007). Some action researchers (Feldman, 2007) and case study researchers (Yin, 2014) still make use of these more

'traditional' terms. Others use different terms, like authenticity (Coghlan & Brannick, 2005), credibility (Tracy, 2010), and workability (Heikkinen, 2007).

Ultimately, at the core of this debate is the question: how can we know if this research is of a good quality and can be trusted? Instead of being caught up in the language, I have identified what I believe are appropriate quality measures for the case study, and I will explain how I demonstrate them. The quality measures I consider to be important for the case study are:

- The appropriateness of the methods to the concepts being studied (sometimes called construct validity) (Yin, 2014);
- 'Workability,' or how useful the knowledge is in practice (Heikkinen, 2007);
- Transparency, or the clarity and depth in explaining the methods and procedures used to arrive at the study's conclusions (Hiles, 2008); and
- Appropriateness in defining the domain of generalisability (Yin, 2014).

To demonstrate the appropriateness of the methods to the concepts being studied, I have taken two main steps. First, I have provided sufficient definition and theoretical engagement with the concepts I use in this study (see Chapters 2 and 3) to ensure that there is clarity in what I am examining. For example, I have defined what I mean by transience, participation and their differences. I have also given a detailed description the resilience principles in Chapter 3 that will form a core part of my analysis. Second, I have explained why the methodology and methods I have chosen to use are appropriate to my research question and these constructs. For example, my research question focuses on understanding and *addressing* a problem, which is why action research is an appropriate methodology. Furthermore, other studies similar to this one have used similar methodological approaches. For example, Saldivar-Tanaka and Krasny (2004) used an action research approach to study community development in community gardens, Cook and Quigley (2013) used photovoice

with higher education students in an environmental education context, and Yap (2018) used a research diary for his doctoral work studying urban community gardens.

The 'workability' of the knowledge generated in this study is demonstrated by the actions taken. Importantly, 'workability' is not only about what works, but it is also about what does not. The workability of the findings was also assured through embedding aspects of all four of Reason and Heron's (2001) ways of knowing in my study. Practically and experientially, I engaged in gardening in the gardens studied, and took part in organising for the garden at Keele. One of the interviews took place while gardening, and several others took place before, during, and after gardening sessions. These provided opportunities to reflect and refer to garden specific features and phenomena. For example, participants being interviewed while gardening or physically in the garden space meant they more easily related abstract ideas like students' transience to the practical and experienced realities in the gardens. For example, in one garden-based interview, an interviewee had recently laid landscape fabric (a fabric placed around plants to prevent weeds) and this became a focal point of the discussion about how to manage the impacts of transience:

I suppose my strategy, as you can see from this fruit patch, is to make it very low maintenance, growing fun stuff that can be eaten, and then hopefully people come down once and think 'that wasn't hard and it was fun.' That, in a nutshell is what I'm trying to do.

I attempted to embed propositional inquiry into my study through the photovoice component of the study and the use of my reflective diary. I also used posters with questions for participants to respond to with brightly-coloured pens and sticky-notes in workshops to encourage embodied and creative thinking that a sit-down discussion might not inspire.

One of the main ways of being transparent is being clear and complete in what has been done (Hiles, 2008). Striving for clarity and completeness is also one of the ways to

demonstrate replicability of a study, though I do not use replicability as a quality criterion for the case study. Yin (2014) does suggest that replicability should be used as quality criterion in case study research, and while this may be appropriate for some case study research, it is not appropriate in my study given that the ambition of action research is to make lasting change (McNiff & Whitehead, 2011). As such, action research is deliberately non-replicable as it is focused on creating change in a naturalistic context. So, while the strategies for ensuring transparency and reliability are similar, I have no ambition for this study to fulfil the latter criterion.

I have sought to be transparent and clear through the use of first-person inquiry to enhance my own reflexivity through the research process and documenting the evolution of my own thinking in this chapter. Another way to demonstrate transparency is to provide a 'thick description,' which is "in-depth illustration that explicates culturally situated meanings and abundant concrete detail" (Geertz, 1973; Bochner, 2000; Tracy, 2010, p. 843). The purpose of a thick description is to provide relevant and meaningful background information to the data that contextualises it. This allows the reader to understand the phenomena described in its full complexity and with more nuance. This includes not only focusing on "who is talking, and what they are talking about, but also who is not talking and what is not said" (Tracy, 2010, p. 843). In this chapter, I provide a thick description of each of the garden contexts, and provide as much context, complexity, and nuance in the following chapters as possible with restricted space and without breaching promises of anonymity.

Generalisability is one of the more highly debated quality criteria for action research and case study research (Heikkinen, 2007). Traditionally, generalisability has to do with statistical generalisation, or the probability that the findings of the study are applicable to the wider population (Yin, 2014). In much of action research and case study research, this type of generalisability is not appropriate (McNiff & Whitehead, 2011; Yin, 2014). However, many

researchers assert that there is scope for a certain amount of generalisability from action research and case study research (Dick, 2002; Yin, 2014). This is through transferability and naturalistic generalisation, which are when learnings have relevance in other contexts or situations (Tracy, 2010). Transferability is the applicability of findings from one study to another situation. For example, in this study, findings about student-led food gardens may be applicable to other student-led initiatives or a variety of other organisations with transient participants. Naturalistic generalisation happens when the provision of a 'thick description' allows the reader to "make choices based on their own intuitive understanding of the scene, rather than feeling as though the research report is instructing them what to do" (Tracy, 2010, p. 845). While I believe that language cannot be used in a neutral way, and therefore any kind of 'thick description' will always 'lead' the reader to some conclusions, providing a 'thick description' makes it more possible for the reader to come to their own conclusions and genuinely assess how generalisable the findings are.

There are also other quality criteria that are appropriate to this study, however have less to do with the methodological quality and more to do with the overall quality of the study, including the study of a worthy topic, the significance of the research contribution, and the meaningful coherence between the study's purpose, methods, and literature (Tracy, 2010). While these are important considerations, I will not discuss them further here, because they have been and will continue to be demonstrated throughout this thesis.

Earlier case studies of sustainability in higher education have struggled to balance the need for context specificity with rich, thick descriptions (Cresswell, 2014), and the extent to which the findings can be widely generalisable (Corcoran et al., 2004). Because of this, Corcoran proposed some guidelines for enhancing rigour in case studies of sustainability in higher education. I explain how this study meets these quality criteria in Table 11.

Table 11. How this study meets Corcoran et al.'s (2004) criteria for case studies of sustainability in
higher education. Table adapted from Bessant (2017).

Quality criteria for case studies of sustainability in higher education		How this study meets these criteria
Clarity in the purpose of the case study		I have outlined research questions in Chapter 1 and in this chapter (see Table 6) I explain which questions are addressed through case study methodology.
Inclusion of data collection methods, including:	Explicit mention of the role of the author/s in the conduct of the study	My positionality is discussed in section 4.4. As I am the sole author of this thesis, all data was collected and analysed by myself.
	Establishment of a clear purpose for the case	I outline the aims of Student Eats in section 4.3.2.1 ad my research questions in Chapter 1.
	All the people involved in the phenomenon were included in the case	As this study uses a single, embedded case with three subunits of analysis, this means that not all Student Eats gardens were included. Furthermore, it was nearly impossible to engage all people in the case given the transience of the students involved. This would have meant that I would have needed to attend all gardening sessions. Like Bessant (2017, p. 152), I believe that case studies are not only an "illustrative examination of a bounded and confined system or entity, but a methodological research design approach embedded within theoretical questioning and reflection." As such, this "renders the design of case study research flexible and contingent upon the researcher and research aims in question" (Bessant, 2017, p. 152). The use of case studies with embedded units of analysis is an established practice when multiple layers of the case in question are being studied (Yin, 2003, 2008). This study considers Student Eats as the case, but it also includes three gardens as embedded units. This enables a fuller understanding of the initiative at an English-scale and at the scale of the gardens.
	The case's potential to contribute to an improvement in the field of sustainability in higher education	Given that this is an action research study, this is embedded in the research design. Corcoran et al. (2004) say action research is methodology well-suited to meet this criteria. Actions were already taken in the gardens during the study to address the impacts of transience in student-led food gardens. This study also includes a chapter (Chapter 8) that focuses specifically on solutions to the challenges outlined in Part 2.

4.7 Ethical considerations

The Belmont Report was a landmark publication establishing core principles of ethical

research 'with human subjects' (The National Commission for the Protection of Human

Subjects of Biomedical and Behavioural Research, 1978). These core principles are:

• Respect for persons: "individuals should be treated as autonomous agents" and

"persons with diminished autonomy are entitled to protection" (The National

Commission for the Protection of Human Subjects of Biomedical and Behavioural

Research, 1978, p. 4).

- Beneficence: "do not harm" and "maximise possible benefits and minimise possible harm" (The National Commission for the Protection of Human Subjects of Biomedical and Behavioural Research, 1978, p. 6).
- Justice: "research should not unduly involve persons from groups unlikely to be among the beneficiaries of subsequent applications" (The National Commission for the Protection of Human Subjects of Biomedical and Behavioural Research, 1978, p. 10).

These principles underpin ethical review processes in the UK and many other countries, and are embedded in, for example, the UK's Economic and Social Research Council's Framework for Research Ethics and the UK Research Integrity Office's guidance for ethical research (ESRC, 2015; UKRIO, 2019). The main way these translate into practice is through attention to fairness in inclusion or exclusion when selecting 'subjects,' obtaining informed consent, and assessing the risks and benefits of the study, ensuring the benefits outweigh the risks (Brydon-Miller, 2009).

Overall, the study is relatively low-risk given that I was not intentionally seeking out vulnerable people to study or asking invasive personal questions. Taking part in the study did not pose any major risks to participants, other than taking their time away from their gardening or work tasks. However, the immediate potential benefits for participants of taking part were that actions implemented through the research could help mitigate or manage the impacts of students' transience. The long-term potential were that the findings from the study could help stakeholders of student-led initiatives mitigate or manage the impacts of students' transience in their initiatives.

Informed consent was obtained from all of the study's participants. For planned interviews, the participants were sent an invitation letter, information sheet, and consent form to review in advance of meeting with me. Upon meeting, I provided two paper copies of

the information sheet and consent form, which I gave the interviewee time to read. In some cases, I reviewed the consent form with them verbally. The interviewees were invited to ask questions if they had any. At this point they were invited to choose whether or not they would like to take part in the study. All participants who made it to this point consented to taking part.

For the early workshops, I posted the invitation letter, information sheet, and consent form on the relevant garden's Facebook group, where I invited participants to come take part in the workshop. For others, I created a Facebook event for the workshop, where I posted the invitation letter, information sheet, and consent form. In nearly all cases, these were not reviewed by participants before the workshop, so the first stage of the workshop would involve going through the information sheet and consent form together and inviting questions, as I did with interviewees.

The information sheet first outlined the aims of the study, explained why they (the potential participant) was invited to take part, and that they were free to decide whether they wished to take part or not. They were also informed that were free to withdraw from this study without giving reasons. The information sheet also outlined all the possible research activities the participants could take part in, when and the number of times they would take place, and what they would involve. For each of these activities, I also outlined how the information would be used in the study. For example, information from data collection activities at the beginning of the study was used to shape the overall research direction, whereas, the fishbowl at the conference was used to assess the findings' generalisability more broadly. In the information sheet, potential participants were also told about the possible benefits of taking part. This section of the information sheet was revised and approved by the research ethics committee several times as the research focus shifted.

require a time commitment. The information sheet also said that it would only be myself and my supervisors who would have access to the information provided by participants. It also said that their personal information (name, identifying characteristics) would be anonymised, though it would possible that they could be identified in publications by people who knew them because of their role. They were also provided with information about how the research was funded and who to get in touch with in the event of the problem.

I have also taken measures to ensure participants' anonymity. Digital copies of transcripts were stored on a password protected computer. Paper copies of transcripts and consent forms were stored in a locked office. As stated in the information sheets for participants, quotes from interviews, workshops, and the photovoice focus group that were included in this thesis were sent to participants for approval. Consent was obtained to use quotes without additional approval from participants in the fishbowl discussion at the Student Eats conference because it would have been too difficult to identify the source of the quotes.

In addition to some of these more traditional concerns in research ethics, action researchers have also outlined some other ethical issues that are of particular concern to action researchers. In particular, reciprocity is considered to be especially important (Maiter et al., 2008). Brydon-Miller (2009) and Hilsen (2006) have begun exploring a new way of thinking about ethics in action research. They propose a shift from contractual ethics, in which informed consent is an "explicit agreement [that] is in most cases documented through the use of a written consent form" (Fischman, 2000, p. 35), to covenantal ethics, which is "the unconditional responsibility and the ethical demand to act in the best interest of our fellow human beings" (Hilsen, 2006, p. 27). Contractual ethics emphasises ethics as a "legalistic exchange," whereas covenantal ethics has a focus on ethics as a process of negotiation with a "focus on relationships, respect, and responsibility" (Brydon-Miller, 2009, p. 2, 7). This is an appropriate approach to ethics in action research because research plans usually involve

some ambiguity about what the later stages of the research will look like because of its participatory and flexible nature (Boser, 2006). As such, it is difficult or impossible for participants to truly give prior informed consent.

Therefore, in this thesis my approach to ethics was constructed to both fulfil both the institutional requirements for a contractual ethical agreement, while adopting a covenantal approach to ethics in my interactions with participants. I amended my ethics application twice following its first approval as the focus and methodology of the study shifted in response to participant input. One of the instances that prompted one of the amendments was when a participant did not agree to the wording in the information sheet. In the information sheet I had written that I would observe 'inconsistencies between what you say and what you do' which the participant thought sounded accusatory. While this had not been my intention, I did agree that the phrasing did not express the empathy I had for the research participants. I therefore changed the phasing to 'I will also look to see if there are any differences between plans you make and how they are carried out to find out what aspects of it you may find challenging and what obstacles may affect your ability to follow through with them.' This phrasing, while imperfect, was intended to be more sympathetic and solutions-orientated rather than critical. The participant was more comfortable with this latter phrasing. This revision was formally made in my ethics application and approved by the research ethics committee.

I told all participants that I would provide them with the quotes I intended to use in publications prior to using them. All participants consented to all quotes provided, with the exception of one, who was a key informant. As such, there are several passages of text in this thesis which could benefit from illustrative quotes, but for ethical reasons some quotes were not used. I also asked the key informant for feedback on a passage of text about a particular incident that I planned to include in my methodological reflections. The informant disagreed

with my interpretation of the incident and asked for the passage not to be used. I sought out further clarification from the participant in order to better construct my understanding of the incident to amend the passage to better reflect the events as the informant saw them, however they did not respond to my request. At the time of publication, I still do not have a full understanding of the perspective of the key informant, however I felt that removing the section entirely would omit an important and necessary methodological reflection. As such, I have amended the passage of text, taking into account the limited perspective they did provide and removing their direct quotes. In this way, I have attempted to both adhere to the formal ethics requirements of my study, my personal ethical principles, and my commitment to scholarly rigour through transparency.

Gender-neutral pronouns (they/them/their) are used when describing research participants. This is not a reflection of how participants self-define. Rather, they are intended to provide a higher degree of anonymity given that some interviewees had specialised roles, meaning that specifying their gender would make it possible for them to be identified.

4.8 Conclusion

In this chapter I have sought to explain my approach to studying university student-led food gardens using an action research approach, in which I have drawn on both pragmatic and critical traditions. This thesis is based on a multi-phased project, beginning by constructing the understanding of the impacts of transience in student-led food gardens. To do this, I conducted a quantitative systematic literature review of sustainable university community gardens and began a case study of the United Kingdom's National Union of Students' Student Eats food growing scheme. Three gardens at different universities were used as embedded units of analysis to enable a more in-depth study of the phenomena at play, while drawing on the knowledge and expertise of staff members of the sustainability team at the National Union of Students. I employed interviews, participatory workshops, photovoice, a fishbowl dialogue, and a research diary. Constructing the understanding of the impacts of transience took over two years of inquiry, which led to and overlapped with the next stages of the research, planning and taking action to address the impacts of transience in student-led food gardens. Two of the gardens took part in this phase and undertook a portfolio of actions (which will be described in Chapter 8). These were then evaluated, again, employing participatory workshops, interviews, photovoice, and a research diary. A fishbowl discussion held at the Student Eats Conference in 2017 complemented the more in-depth inquiry at the three gardens used as embedded units of analysis. At the end of the chapter, I outlined the quality criteria against which this study should be held, and how I have adopted both contractual and covenantal approaches to ethics in my study in order to both fulfill institutional requirements and stay true to my own ethical principles. This chapter concludes the first part of this thesis, providing the background to the remaining parts. I will now go on to explore the 'problem' as it has been constructed, beginning with presenting the findings of my quantitative systematic literature review.

Part 2: The problem

Systematically reviewing sustainable university community gardens

Chapter 5

This mutual skill sharing and the informal learning environment it creates allowed participants who had previously had negative educational experiences to enjoy learning with the Garden Project: 'This is like a real learning experience, it's not like anyone's twisting my arm to go there.'

Erika Mundel & Gwen Chapman, 2010, p. 169

We posit this learning as holistic, taking into account the five human senses, the mind, the body, and heart: gardeners touch, feel taste, smell, listen, see, and learn bodily and emotionally.

Hongxia Shan & Pierre Walter, 2015, p. 23

In order to maintain the garden and conduct community outreach activities, a substantial number of volunteers were needed, and recruiting these volunteers was the most significant challenge faced... Since the garden needs care, regardless of how minimal it is, there needs to continually be a caretaker available to perform those tasks.

Kristian McKinne & Angela Halfacre, 2007, p. 150, 152



Potting herb cuttings Winter 2013

5.1 Introduction

This chapter will offer an overarching assessment of the benefits provided by sustainable university community gardens and compare these benefits with those reported in Guitart et al.'s (2012) review of urban community gardens. I then will go on to explore what is known about students' transience impacts on sustainable university community gardens. Before doing this, I will first present the authors' affiliation(s), journal's discipline area, year of publication, location of the garden(s) studied, research methods used, and garden characteristics reported in the articles about sustainable university community gardens in order to contextualise the findings. These article and garden characteristics will also be compared to those of urban community gardens as reported by Guitart et al. (2012).

Before beginning, I would also like to remind the reader that sustainable university community gardens were reviewed because the pool of articles about student-led food gardens was too small to provide a meaningful review. As such, I cast the net somewhat wider to capture not only student-led food gardens, but also their associated counterparts. There are two main differences between sustainable university community gardens and student-led food gardens. Sustainable university community gardens can be led by students, but do not need to be, unlike student-led food gardens. Also, crucially, sustainable university community gardens can grow food, but they do not need to by definition. On the other hand, student-led food gardens grow food, by definition. I will begin by presenting and comparing the characteristics of the articles about sustainable university community gardens and urban community gardens and the gardens researched.

5.2 Sustainable university community gardens as compared to urban community gardens

5.2.1 Characteristics of articles and gardens researched

I reviewed twenty-two articles covering twenty-three gardens (one article mentioned two gardens) (see Tables 12 and 13). The majority (77%) of the articles about sustainable university community gardens were written by authors affiliated with institutions in the United States, much like those about urban community gardens (61%). The articles about sustainable university community gardens were published in journals about horticulture (36%), education (36%), health (14%), non-profit/social marketing (9%), and sustainability (5%). The 36% of the articles in journals about horticulture were published in a single journal, several of which were published in a special issue. The disciplines of journals publishing articles about urban community gardens, on the other hand, were "geography (28%), environment and planning (24%), society and culture (23%), health (12%), education (9%), economics (3%) and conservation biology (1%)" (Guitart et al., 2012, p. 366). All articles about sustainable university community gardens were written since 1999, with most written between 2005 and 2012, while articles about urban community gardens reviewed by Guitart et al. (2012) were published between 1985 and 2011, with most being published since 2000. Only eight (36%) of the articles about sustainable university community gardens reported methods they used, seven of which used qualitative methodologies/methods, including case studies, participatory action research, narrative evaluation, autoethnography, interviews, participant observation, and document reviews. The only other study that described its methods was a mixed-methods experimental study using questionnaires. Most articles about urban community gardens used a qualitative or mixed methods approach (using interviews, surveys, text analysis, focus groups, case studies and participant observation), and the

remaining five studies used quantitative approaches (Guitart et al., 2012). In their review on American community gardens, Draper and Freedman (2010, p. 458) found 40% of the articles they reviewed used quantitative methods, however they challenged the rigour of many of these articles for failing to include a "control group or taking pre- and post-test measures." The need for more rigorous quantitative studies about community gardens has also been highlighted in Robinson-O'Brien et al.'s (2009) literature review of garden-based youth nutrition intervention programmes.

Most (78%) sustainable university community gardens were located in the United States and nearly half were described as food gardens (48%). Others were described as farms (13%), community supported agriculture (an agricultural model in which consumers share the economic risks with the producers by paying for their share of produce at the beginning of the season; sometimes consumers volunteer with the garden/farming tasks) (13%), botanical gardens (13%), a wildlife habitat garden, native plant garden, and allotment. This mirrors the

Key characteristics of articles		Sustainable university community gardens	Urban community gardens
		(Articles, n=22; Authors, n=52)	(Articles, n=87; Authors, n=195)
Author affiliation	United States	77%	61%
	Canada	10%	9%
	Australia	8%	13%
	UK	4%	9%
	South Africa	2%	2%
	Other	0%	6%
Journal discipline	Horticulture	36%	0%
	Education	36%	9%
	Health	14%	12%
	Non-profit/social marketing	9%	0%
	Sustainability	5%	0%
	Geography	0%	28%
	Environment and planning	0%	24%
	Society and culture	0%	23%
	Economics	0%	3%
	Conservation biology	0%	1%
Methods used	Qualitative	32%	58%
	Quantitative	0%	6%
	Mixed methods	4%	36%
	Not mentioned	64%	0%

Table 12. Key characteristics of articles about sustainable university community gardens.

	Key characteristics of gardens	Sustainable university community gardens (Gardens, n=23)	Urban community gardens (Gardens, n=89)
Garden	United States	78%	57%
Location	Canada	13%	6%
	Australia	4%	13%
	South Africa	4%	2%
	UK	0%	9%
	Other	0%	13%
Garden type	Food garden	48%	100%
	Farm	13%	No data
	Community supported agriculture	13%	No data
	Botanical garden	13%	No data
	Wildlife habitat garden	4%	No data
	Native plant garden	4%	No data
	Allotment	4%	No data
Participants	Students	87%	No data
	Faculty/academic staff	57%	No data
	Paid interns/garden managers	35%	No data
	Community members	26%	No data
	Organisations	22%	No data
	Other	57%	No data
Plants grown	Edible plants	74%	No data
	Ornamental plants	22%	No data
	Native/endangered species	17%	No data
	Wildlife-habitat-enhancing plants	9%	No data
	Medicinal plants	9%	No data
	Didn't say	9%	No data

Table 13. Key characteristics of the gardens written about in the literature about sustainable university community gardens.

literature reviewed by Guitart et al. (2012) which included some larger-scale projects (Wade, 1987; Karaan & Mohamed, 1998), projects involving restoration ecology/native ecosystem regeneration (Stocker & Barnett, 1998), and those run for commercial purposes (Flynn, 2001).

In the literature about sustainable university community gardens, students were the most commonly reported participants (reported by 87% of gardens), followed by faculty (academic staff) (57%), paid interns/garden managers (35%), community members (26%) and organisations (22%). Most (74%) sustainable university community gardens grew edible plants, but ornamental plants (22%), native/endangered species (17%), wildlife-habitat-enhancing plants (9%), and medicinal plants (9%) were also grown. Over a third (39%) of the sustainable university community gardens were reported to sell produce, which was most

often sold to the campus community or to campus food outlets. Nearly two-thirds (65%) of the sustainable university community gardens were used in the formal curriculum through a wide variety of disciplines, with the most common ones being agriculture/horticulture, applied studies/projects, education, and environmental subjects/geography (Figure 20). Types of participants, plants grown, and the number of gardens selling produce and embedded in educational curricula in the literature about urban community gardens were not reported by Guitart et al. (2012).





5.2.2 Benefits of sustainable university community gardens and urban community gardens

In order to understand whether sustainable university community gardens might contribute to a more sustainable future, I compiled the reported benefits of these gardens. I then compared these benefits to those of the urban community gardens reviewed by Guitart et al. (2012) to assess how sustainable university community gardens perform compared to community gardens more widely. All the benefits of urban community gardens identified by Guitart et al. (2012) were discussed and/or demonstrated in the literature about sustainable university community gardens (Figure 21a and 21b). The most frequently mentioned benefits of sustainable university community gardens were educational and social benefits, benefits to the university, and additional benefits to individual participants. In the articles reviewed by
Guitart et al. (2012), social benefits were most frequently mentioned. However, the other most frequently mentioned benefits of urban community gardens differed. These highlyreported benefits were access to fresh foods, health benefits, and economic benefits. The rest of this section expands on the benefits of sustainable university community gardens¹⁵. Social benefits were demonstrated in fourteen (68%) of the articles about sustainable university community gardens. The proportion of these articles with demonstrated social benefits was nearly consistent with the number of articles it was discussed in (73%). This stands in contrast with 60% of the articles about urban community gardens discussing social benefits, but only 38% demonstrating them. Social benefits of community gardens and urban food growing have also been widely reported in other reviews (Draper & Freedman, 2010; Zasada, 2011; Orsini et al., 2013; Mok et al., 2014; Taylor & Taylor Lovell, 2014). The interactions between different groups of people were one of the main social benefits reported to emerge from sustainable university community gardens. Social interactions reportedly took place between students of different disciplines, faculty and students, growers and consumers, and across cultures. Likely resulting from these interactions, sharing of ideas and experiences, the development of trust and caring among gardeners, and the establishment of a supportive 'community' around specific issues (for example, sustainable food or health promotion) were reported. Some of the social benefits extended beyond the garden. For example, Mundel and Chapman (2010) reported a sense of civic engagement being fostered in participants. Others reported that participants supported each other in cases of illness and death in the family (Wills et al., 2009) (Table 14).

Access to fresh food was discussed in ten (46%) and demonstrated in six (27%) articles

¹⁵ Benefits to health, life satisfaction, environmental equity, biodiversity, and associated with reduced crime/increased safety are not discussed further because their coverage is very similar in articles about both sustainable university community gardens and urban community gardens, and apart from health benefits they were only reported by discussed and/or demonstrated by one or two articles each.



Figure 21a. The percentage of discussed and demonstrated benefits in articles about sustainable university community gardens (n=22). The benefits below the line were three new categories of benefits that were developed for this review as they did not fit under any of the themes identified by Guitart et al. (2012).

*Environmental sustainability benefits exclude benefits that may have been caused by dietary changes (e.g., the displacement of unsustainable food products in participants' or other benefactors' diets by food from the garden). This was discussed in several of the articles; however, I felt that while it could be considered both an environmental sustainability benefit and a benefit in terms of providing access to fresh foods, the category that best captured this outcome was 'access to fresh foods'.



Figure 21b. The percentage of discussed and demonstrated benefits in articles about urban community gardens as assessed by Guitart et al. (2012) (n=87). The benefits below the line were three new categories of benefits that were developed for this review as they did not fit under any of the themes identified by Guitart et al. (2012).

Table 14. The types of social benefits discussed and/or demonstrated in the literature about sustainable university community gardens and the articles reporting them.

Social benefits	Articles citing the benefits
Increased interdisciplinary and faculty- student interaction	Hoffman et al. (2007); Roubanis & Landis (2007); Barton et al. (2010); Kobayashi et al. (2010); Apul & Philpott (2011)
Establishing a supportive 'community' around a specific issue (e.g., sustainable food, health promotion)	Kobayashi et al. (2010); Mundel & Chapman (2010)
Connecting food consumers with growers	Lewis & Affolter (1999); Kobayashi et al. (2010)
Sharing ideas and experiences	Wagner & Fones (1999)
Cross-cultural communication	Hoffman et al. (2007); Shan & Walter (2015); Datta (2016)
Fostering a sense of civic engagement	Mundel & Chapman (2010); Johnston et al. (2012); Datta (2016)
Social interaction in the garden spurred action beyond the garden	Hoffman et al. (2007); Wills et al. (2009); Mundel & Chapman (2010); Datta (2016)
Building trust and caring	Datta (2016)

about sustainable university community gardens. Several articles mentioned that the gardens collaborated with university food services to serve the garden's produce (Ross, 2005; Biernbaum et al., 2006; Wharton & Harmon, 2009; Kobayashi et al., 2010). One article mentioned that their community supported agriculture scheme contributed to "emergency food system efforts in communities and to government-sponsored nutrition assistance programs" (Wharton & Harmon, 2009, p. 114). In Mundel and Chapman's (2010, p. 172) article about a garden in Vancouver, Canada, the emphasis was less on access to fresh food, and more on growing and gathering traditional food in an effort to "[promote] health through decolonizing Indigenous diets" by "[returning] to more traditional diets and food practices." While these findings indicate that access to fresh food is a notable benefit of sustainable university community gardens, the extent to which urban agriculture and indeed, community gardens, can contribute to self-sufficiency in cities is currently hotly debated and needs to be better understood (Mok et al., 2014). Guitart et al. (2012) did not look for access to fresh food as a discussed benefit in their review, but it was demonstrated as a benefit in 43% of the articles about urban community gardens. The discrepancy between the number of sustainable university community gardens and urban community gardens demonstrating

access to fresh food may be because a number of the articles about sustainable university community gardens implied (but did not explicitly state) that access to fresh foods was a benefit, and I chose not to count these articles as demonstrating access to fresh foods. On the other hand, Guitart et al. (2012, p. 386) wrote that they "assumed that [access to fresh foods] was a demonstrated benefit," indicating that articles that were not explicit about this benefit may have still been counted, accounting for this discrepancy in the access to fresh food demonstrated between the reviews.

Economic benefits were discussed in a greater proportion of articles about urban community gardens (42%) than in those about sustainable university community gardens (27%). However economic benefits were demonstrated in a similar proportion of articles in both bodies of literature (17% in urban community gardens and 14% in sustainable university community gardens). Kobayashi et al. (2010) said that garden they studied sought to be economically sustainable, and Wharton and Harmon (2009) reported that their community supported agriculture scheme distributed economic risk between the farmers and consumers. These economic benefits differed from the ones mentioned in Guitart et al.'s (2012) article, where the economic benefits mentioned were more participant-centred, such as individuals saving or making money. Such participant-centred economic benefits were not mentioned in the articles about sustainable university community gardens.

Educational benefits were found to be a key outcome of sustainable university community gardens (Table 15). They were discussed in all articles, while seventeen articles (77%) demonstrated them. This is in contrast to only 34% of articles about urban community gardens that discussed educational benefits and 13% that demonstrated them (Guitart et al., 2012). Similarly, in their literature review of community gardens in the United States, Draper and Freedman (2010), found that nearly a third of the articles were about youth gardening programmes, many of which reported educational benefits as an outcome. Walter's (2013)

assertion that the pedagogical potential of community gardens for adult learning is

substantial and under-researched is supported by the discrepancy between discussed and

demonstrated educational benefits in the literature about urban community gardens, and

Draper and Freedman's (2010) findings that community gardens facilitate youth education.

Educational benefits	Articles citing the benefits
Development of skills	Markhart (2006); Wills et al. (2009); Barton et al. (2010); Mundel & Chapman (2010); Somerset et al. (2010); Apul & Philpott (2011); Datta (2016)
Inter-/multi-disciplinary education	Apul & Philpott (2011); Johnston et al. (2012)
Science, sustainability and environmental education	Apul & Philpott (2011); Johnston et al. (2012); Datta (2016)
Improved outcomes of academic work	Hoffman et al. (2007)
Changing participants' attitudes towards 'education'	Mundel & Chapman (2010)
Increasing participants' agency in their own education through using more informal methods	Markhart (2006); Wharton & Harmon (2009); Johnston et al. (2012); Datta (2016)
Enabling participants to take a more active role in their education	Kobayashi et al (2010); Apul & Philpott (2011); Datta (2016)
'Learning how to learn'	Biernbaum et al. (2006); Apul & Philpott (2011)
Enabling learning that fosters "knowing, connecting, and hybrid knowledge production across cultures"	Shan & Walter (2015)
Providing insight into "challenges facing colonized people"	Mundel & Chapman (2010)
Linking formal and informal learning	Datta (2016)

Table 15. The types of educational benefits discussed and/or demonstrated in the literature about sustainable university community gardens and the articles reporting them.

The reviewed articles suggested that community gardens could facilitate the

development of skills/knowledge outcomes, including learning about science, sustainability and/or the environment. They also described different strategies used to enable learning, including multi- or interdisciplinary approaches, and linking formal and informal learning. The informal learning in the garden (through, for example, learning from experience or using the garden as an outdoor classroom) also changed participants' attitudes towards 'education' by allowing "participants who had previously had negative educational experiences to enjoy learning" (Mundel & Chapman, 2010, p. 169) and increasing individuals' agency in their own education. Apul and Philpott (2011, p. 74) also found that students were 'learning how to learn' in the garden they studied by "[identifying] the knowledge base and technical skills they needed to solve" the problems they faced, and then finding strategies to "develop their competencies."

Several articles about sustainable university community gardens also mentioned cultural aspects of education. For example, Shan and Walter (2015, p. 19) said that one of the gardens they studied fostered "knowing, connecting, and hybrid knowledge production across cultures." In their case, these activities took place between 'westerners' and Chinese immigrants. Mundel and Chapman (2010, p. 172) also mentioned how non-aboriginal students participating an aboriginal health promotion garden provided the students with insight into "challenges facing colonised people." The importance of such cultural exchanges and learning will vary between gardens, but these articles highlight the potential of community gardens for facilitating cultural understanding. However, it should also be noted that gardens are not *de facto* intercultural educational spaces – cultural and racial tensions from other aspects of society can be replicated and reproduced in the garden environment (Mundel & Chapman, 2010).

Environmental sustainability benefits were discussed in twelve (55%) articles about sustainable university community gardens. This is more than the 22% of articles reported to have discussed environmental sustainability in Guitart et al.'s (2012) study of urban community gardens. The proportion of articles demonstrating environmental sustainability in each of these bodies of literature is also consistent with above disparity: 27% of articles about sustainable university community gardens and 2% of articles about urban community gardens demonstrated environmental sustainability (Stocker & Barnett 1998; Holland, 2004). This was an expected result given that the inclusion criteria for this review included sustainability (Table 6 and 7). The most commonly mentioned contribution to environmental sustainability in sustainable university community gardens was the use of organic farming/gardening methods, but other gardens were improving environmental conditions in other ways, such as

through ecological restoration (Hockenberry Meyer et al., 2010), bioremediation (Hockenberry Meyer et al., 2010), and growing endangered plant species (Lewis & Affolter, 1999).

One of the challenges I faced while analysing the articles was determining what actually constitutes a benefit to environmental sustainability in community gardens. I decided that environmental sustainability benefits excluded those caused by reported dietary changes (i.e., displacement of unsustainable food products in participants' or other benefactors' diets by food being produced more sustainably in the garden) because the links between dietary changes and environmental sustainability are too complex to know whether or not they actually have contributed to environmental sustainability without doing a lifecycle analysis. Although there is potential for community gardens to produce environmental sustainability benefits, the literature is limited, particularly in terms of demonstrating these benefits.

The reason why these benefits are not well-demonstrated in the literature may be because environmental impacts and benefits are often linked and have trade-offs, making them challenging to assess. However, they are particularly important to explore because community gardens can also have negative environmental impacts. For example, it has been found that community gardens have "lower levels of arthropod diversity and biocontrol activity by arthropods than undisturbed vacant lots" (Lovell & Taylor Lovell, 2013, p. 291, citing Gardiner et al., 2013 and Yadav et al., 2012), and that an excess of nutrients can also accumulate in garden soils and pollute water sources (Witzling et al., 2011).

Because of the disparity between the claims over and evidence of community gardens' contribution to environmental sustainability, further research is needed to better understand the extent to which they may actually be providing these benefits. This echoes calls from Mok et al. (2014, p. 21) who, when reviewing literature on urban agriculture more broadly, found that a better understanding of the "risks posed by pollutants from agriculture

to urban ecosystems and from urban ecosystems to agriculture" and "the carbon footprint of urban agriculture and use of 'food miles'" is needed. Guitart et al. (2012) suggest that two main questions need to be addressed in order to establish whether food growing projects can be considered 'sustainable'. These are (1) 'What do gardens grow?' to explore agrobiodiversity, and (2) 'How is the food grown?' to explore "the sustainability of gardening practices" (Guitart et al., 2012, p. 369). Recently there have been some attempts to contribute to answering these questions. For example, Guitart et al. (2014, 2015) have assessed the agrobiodiversity in community gardens by exploring the types of food planted using a colour-classification system, and have investigated gardening practices used in community gardens and their ecological integrity. Silva and Krasny (2016) have also suggested that open-access online tools for measuring impacts and outcomes of community gardens should be used (such as Farming Concrete's Data Collection Toolkit (Design Trust for Public Space, n.d.)). They also suggest practitioner-researcher partnerships for more in-depth investigations into the environmental sustainability benefits of gardens (Silva & Krasny, 2016).

However, fostering engagement in research about community gardens from scholars from natural and technical science disciplines will be challenging if it is not regarded as being 'research-worthy' or impactful within those disciplines. Though, there might be fertile ground for a wide range of disciplines to engage with sustainable university community gardens through student research projects, if not through more advanced scholarship. A good example of this is Apul and Philpott's (2011) article about project-based learning in a sustainability engineering graduate programme in a sustainable university community garden. In their article they described how students developed an engineering-based solution for watering the garden, and "estimated the life-cycle cost and environmental impacts of their proposed solutions" (Apul & Philpott, 2001, p. 1).

Cultural heritage and development benefits were discussed in six (27%) articles about sustainable university community gardens. They were also demonstrated in five (23%) of the articles. Cultural heritage benefits were discussed in 18% of articles about urban community gardens and demonstrated in 7% of them. Guitart et al. (2012) did not define the term 'cultural heritage,' however, Shan and Walter (2015, p. 2) has said that the term implies "racial and ethnic boundaries [are frozen] in time," whereas, in reality, culture is dynamic and changing. We chose to adapt Guitart et al.'s (2012) category of benefits, calling it cultural heritage and development to account for ongoing changes in culture.

Some of the benefits associated with cultural heritage and development in sustainable university community gardens included using traditional methods and techniques of growing, cooking and eating associated with a particular culture and/or time period (Lewis & Affolter, 1999; Mundel & Chapman, 2010), accessing culturally appropriate food not available in supermarkets (Somerset et al., 2010), and sharing memories and stories of Aboriginal people (Phillips, 2011). For migrants, the gardens offered opportunities to re-establish and develop traditions and identities, such as through developing 'hybrid foods,' like "eating Chinese vegetables using Western salad sauces" (Shan & Walter, 2015, p. 27). Hoffman et al. (2007) and Datta (2016) also reported that the gardens they studied facilitated reductions in ethnic tensions. Guitart et al. (2012, p. 368) found that most of the research about urban community gardens was about gardens in "low income earning areas with different cultural backgrounds in industrial cities in the USA." There was much less written about the class of the participants in the literature about sustainable university community gardens, though seven articles mentioned involving participants of varying cultural backgrounds.

Additional individual benefits to participants was a new category of benefits that came from this study to account for the additional positive outcomes that benefitted an individual and were additional to those included in Guitart et al.'s. (2012) study. These additional

benefits were discussed in eleven (50%) articles about sustainable university community

gardens and demonstrated in nine (41%). The benefits cited included professional

development and mentoring, decreased food shopping trips, the enjoyment of food growing

and contact with nature, academic improvement, and developing a sense of place (Table 16).

Although Guitart et al. (2012) did not include these benefits explicitly, they may have

characterized some similar benefits as those associated with 'life satisfaction'.

Table 16. The types of additional benefits to individual participants discussed and/or demonstrated in the literature about sustainable university community gardens and the articles reporting them. While a number of benefits identified by Guitart et al. (2012) could fall under the 'additional benefits to individual participants' category (such as education, life satisfaction, etc.), I retained Guitart et al.'s existing categories, with other benefits to individual participants being captured in this category to enable clarity in comparing the findings.

Additional benefits to participants	Articles citing the benefits
Professional development and mentoring	Wagner & Fones (1999), Somerset et al. (2010), Markhart (2006), Mundel & Chapman (2010), Apul & Philpott (2011)
Decreased food shopping trips	Wharton & Harmon (2009)
Enjoyment of contact with nature and food growing	Roubanis & Landis (2007), Wills et al. (2009), Somerset et al. (2010)
Academic improvement	Hoffman et al. (2007)
Immigrants developing a sense of place	Somerset et al. (2010); Shan & Walter (2015); Datta (2016)

Thirteen (59%) of the articles about sustainable university community gardens discussed *benefits to the university*, and the same number of articles also demonstrated these benefits. The main benefit to the university was research output, which was mentioned in eleven (50%) of the articles. Other benefits included increasing student engagement or involvement with the university (Apul and Philpott, 2011), outreach (such as inviting people or groups from the community on the campus to see the food growing initiative) (Falk et al., 2005; Biernbaum et al., 2006; Markhart, 2006; McKinne & Halfacre, 2008; Wharton & Harmon, 2009; Somerset et al., 2010), teaching (McKinne & Halfacre, 2008), and improved appearance of the campus (Hoffman et al., 2007). Though Guitart et al.'s. (2012) research involved several university-based projects/projects aligned with universities, they made up a small proportion of the articles reviewed, and therefore they did not use this category of benefits. *Engagement, attitudes and behaviours for sustainability* were discussed in eleven (50%) and demonstrated in six (27%) articles about sustainable university community gardens. Articles mentioned how volunteering in a garden can deepen participants' relationship with food and the environment (Wharton & Harmon, 2009; Wills et al., 2009; Mundel & Chapman, 2010; Phillips 2011), and that gardening can also promote an interest in (Wagner & Fones, 1999), a sense of responsibility for (Johnston et al., 2012), and even a "dedication for fighting for" the natural world (Apul & Philpott, 2011, p.70). These weren't reported by Guitart et al. (2012), possibly because they did not include sustainability explicitly as an inclusion criterion. It is also possible that these benefits may have been written in the articles reviewed by Guitart et al. (2012), but were included within one of the other categories of benefits. This latter suggestion seems likely since two of the sustainable university community garden articles mentioning engagement, attitudes and behaviour towards sustainability (Wills et al., 2009; Mundel & Chapman, 2012) were also reviewed by Guitart et al. (2012).

To summarise, many of the benefits produced by urban community gardens that Guitart et al. (2012) identified align with those identified in this review of sustainable university community gardens. However, there were different emphases on which benefits were important in the two different areas of literature, which educational benefits dominating the literature about sustainable university community gardens. There were also some additional benefits of sustainable university community gardens literature identified: engagement, attitudes, and behaviours for sustainability, additional benefits to individual participants, and benefits to the university. The similarity in the range of benefits identified in both reviews suggests that these might provide a useful framework in reviewing the efficacy of community gardens projects in a range of different settings. This could be supported by the development of tools (such as Farming Concrete's Data Collection Toolkit (Design Trust for

Public Space, n.d.)) to measure and track food production, as well as environmental, social, health, and economic benefits in farms/gardens.

There also seemed to be some gaps in the literature. The vast majority of the research reviewed was about gardens in the United States and used mainly qualitative approaches. This therefore means that literature may not fully represent the breadth of community garden initiatives that exist, and the perspectives different disciplines could provide. As highlighted earlier, the contribution of sustainable university community gardens and urban community gardens to environmental sustainability is not well-understood. Furthermore, the potential of urban community gardens to contribute to informal adult learning about sustainability may be underacknowledged. I will now go on to discuss dominating benefits of sustainable university community gardens in greater detail, and then consider the absence of discussion about the impacts of students' transience.

5.3 The dominating benefits

The multifunctional nature of community gardens is celebrated as they can achieve pro-sustainability aims through a portfolio approach, rather than focusing on maximising a single outcome (Lovell, 2010; van Leeuwen et al., 2011; Taylor & Lovell, 2013). It is important to recognise the multiplicity of the benefits of these initiatives because if they are evaluated on the basis of a single outcome, their benefits may seem insignificant. Furthermore, focusing only on maximising a single benefit may result in a net reduction of benefits, when all outcomes are considered.

That said, education appeared to be the dominating benefit of sustainable university community gardens in a strictly quantitative sense, particularly compared to how much less it was mentioned in the literature on urban community gardens. While only three articles mentioned contributing to science, sustainability and environmental education (Apul &

Philpott, 2011; Johnston et al., 2012; Datta, 2016), many of the reported educational benefits seen in Table 15 could be seen as the development of sustainability competencies (Wiek et al., 2011). The gardens were reported to be home to learning into the "challenges facing colonized people," (Mundel & Chapman, 2010, p. 172), and facilitating "knowing, connecting, and hybrid knowledge production across cultures" (Shan & Walter, 2015, p. 19). This is evidence that participating in sustainable university community gardens can facilitate the development of normative competence, or the "ability to collectively map, specify, apply, reconcile, and negotiate sustainability values, principles, goals, and targets" (Wiek et al., 2011, p. 2019). Participating in the gardens also seemed to facilitate the development of interpersonal competence, "the ability to motivate, enable, and facilitate collaborative and participatory sustainability research and problem solving" (Wiek et al., 2011, p. 211). The intercultural learning highlighted above (Mundel & Chapman, 2010; Shan & Walter, 2015), contributes to interpersonal competence development through "pluralistic and trans-cultural thinking" (Wiek et al., 2011, p. 211). Furthermore, engaging in cross-cultural communication (Hoffman et al., 2007; Shan & Walter, 2015; Datta, 2016), building of trust and caring (Datta, 2016), and establishing a supportive 'community' (Kobayashi et al., 2010; Mundel & Chapman, 2010) demonstrate the use of communication skills and empathy, which are key features of interpersonal competence (Wiek et al., 2011).

There is also evidence of the use and development of action competence, "ability, motivation and desire to play an active role in finding democratic solutions to problems and issues connected to sustainable development" (Morgensen & Schnack, 2010, p. 61). Participants of sustainable university community gardens appeared to develop action competence through exercising a greater sense of agency and confidence in taking action. For example, gardeners were inspired to take a more active role in their education (Kobayashi et al., 2010; Apul & Philpott, 2011; Datta, 2016) and developed a sense of civic engagement

(Johnston et al., 2012; Datta 2016). Furthermore, the social interaction in sustainable university community gardens spurred action beyond them as well (Hoffman et al., 2007; Wills et al., 2009; Mundel & Chapman, 2010; Datta 2016).

The fact that the majority of articles in this review were about gardens in the United States has considerable bearing on the generalisability of these findings to different geographic contexts. Service learning, or learning through 'volunteering' undertaken for academic credit (Mooney & Edwards, 2001), is a well-established tradition in the United States. However, in many other places, like England, service learning is not a common practice. The learning outcomes from activities that are undertaken for academic credit would likely differ from activities undertaken voluntarily in one's free time. Furthermore, the use of service learning models may contribute to more consistent and predictable participation, meaning that the impacts of students' transience may not be as pronounced. This reinforces the need for studies of sustainable university community gardens in other geographic contexts, like this study of student-led food gardens in England.

As highlighted earlier, social benefits were the second most frequently reported set of benefits of sustainable university community gardens. In contrast to educational benefits, however, the proportion of articles about sustainable university community gardens reporting them are much more consistent with the proportion of articles about urban community gardens reporting them. The other outcome that was mentioned most frequently was that the sustainable university community gardens provided benefits to the university. This is a valuable benefit, and perhaps ones that universities might benefit from acknowledging further. However, for student-led food gardens (compared to sustainable university community gardens), many of these benefits may be less notable since many benefits to the university are unlikely to be capitalised on given the limited nature of university staff involvement. For example, if university staff are not involved in student-led

food gardens, they are less likely to use the gardens for teaching, research, or outreach activities.

The fact that sustainable university community garden's main contributions might be educational highlights that they can play an increasingly important role in the sustainability education agenda in universities. Given that education is an international priority for moving towards a sustainable society (UNCED, 1992; Wals, 2012), strategies are needed for embedding sustainability in higher education. Sustainable university community gardens, including student-led food gardens, are one way to do this.

5.4 The absence of students' transience

Knowing what the impacts of students' transience are is key to understanding how sustainable university community gardens function, as highlighted in Chapter 2. However, while students' transience is logically a defining factor of sustainable university community gardens, it is an issue that has not been adequately explored. Only one of the articles mentioned the issue at all, and it only mentioned it as one of several challenges faced by the gardens (McKinne & Halfacre, 2008). In particular, they said that summer caretaker presence and community interest may be difficult to establish and maintain when student volunteers do not know appropriate staff, postgraduate students, or community members to approach (McKinne & Halfacre, 2008). However, the factors that contributed to community interest were not explored, and neither were the knock-on effects that a lack of caretaker presence might have. In summary, little is known about how students' transience impacts student-led food gardens, in spite of the fact that the phenomenon may impact how successfully they are able to run and how it impacts their outcomes. This underscores the need to explore how students' transience impacts student-led food gardens and how these impacts might be managed in order to provide the benefits outlined in this chapter.

5.5 Conclusion

Like the literature about urban community gardens, the literature about sustainable university community gardens is still immature and under-explored. Both bodies of literature are similar in terms of the geographical scope of their authorship and gardens they investigated (both United States-centric), demonstrating the value of a study based in England. Like the literature about urban community gardens, the literature about sustainable university community gardens is from a wide range of disciplines and currently qualitative methods are used much more widely than quantitative methods.

Overall, there was a similar portfolio of benefits provided by sustainable university community gardens compared to urban community gardens, indicating that a better understanding of how students' transience impacts them would be valuable, especially given the notable absence of discussion around the impacts of students' transience on sustainable university community gardens. The benefits reported most frequently for both sustainable university community gardens and urban community gardens include social, educational, economic, and health benefits, as well as benefits associated with accessing fresh food. The dominating benefits of sustainable university community gardens were educational and social benefits, and benefits to the university. It was also clear from the reported educational and social benefits that sustainable university food gardens have the potential to contribute to the development of sustainability competencies. With an understanding that sustainable university community gardens can provide a portfolio of benefits including contributing to sustainability education, I will now move to Chapter 6 where I will begin to paint a picture of how transience impacts the dynamics of problematic participation in student-led food gardens.

Understanding the dynamics of problematic participation

Chapter 6

...sustainability is about passing to future generations. As students, we are, in generations, coming and leaving, and coming and leaving.

Student gardener, Keele University

"All it takes," said Crake, "is the elimination of one generation. One generation of anything. Beetles, trees, microbes, scientists, speakers of French, whatever. Break the link between one generation and the next, and it's game over forever."

Margaret Atwood, Oryx and Crake, 2003, p. 261-262



Garlic afflicted with garlic rust June 2015

6.1 Introduction

Given that the main impact of students' transience on student-led food gardens is problematic participation, it is necessary to understand the dynamics of problematic participation before speculating how to intervene. Therefore, the purpose of this chapter is to explain the causes of problematic participation, its effects, and if any feedbacks exist between these. This is done drawing on the data from the case study component of the research. As outlined in Chapter 2, an *a priori* assumption was made that the causes of problematic participation increased problematic participation, and problematic participation increased their effects (see Figure 2). These uncontentious assumed relationships will now be used as a frame to hang the results of the study on. As such, the first section will outline the causes of problematic participation, and will be followed by an explanation of the effect of problematic participation. Then, I will build on the assumed causal relationships in Figure 2 to map the feedbacks. Before presenting the results, it should first be noted that most participants in the gardens were undergraduate students, and therefore the results presented will pertain to them unless otherwise specified. Quotes will be written in *italics*, with the source of quotes provided according to the typologies from Figure 14 in Chapter 4 in order to give context to the statements made.

6.2 Causes of problematic participation

The causes of problematic participation were categorised into either 'barriers to getting people 'though the garden gate'' or 'barriers to continued engagement.' I will unpack each of these in turn.

6.2.1 Barriers to 'getting people through the garden gate'

To even begin to have an adequate level of participation in a garden, people must make it through the garden gate. However, many participants said that a cultural barrier or background cultural indifference prevented people from getting involved in the projects because students involved in the gardens had difficulty relating to the wider student body (and likely vice versa). The term 'culture' was understood in a very broad sense, referring to norms, beliefs, attitudes, practices, values, etc. shared by a group of people. For example, some said that many students, particularly international students, did not find food growing appealing because of its association with poverty, colonisation, or being a relic of the past. Conversely, students also speculated that some people did not take part because it was seen as part of a subculture that they did not identify with:

...there is that kind of, like, that whole kind of coterie of dreadlocks and hippie kind of aesthetic. That people don't want to be associated with. (Long-term mentor) Participants speculated that another reason more people did not get involved was because they did not like the actual practice of gardening, whether that be its physicality or getting *mud under your fingernails* (Long-term mentor).

One of the most frequently mentioned barriers preventing students (and other potential volunteers) from getting involved in the gardens was that people were busy, lacked time and had competing commitments. When gardeners were busy or lacked time, it resulted in low or irregular participation. Some students said participation in their garden was not a priority compared to other activities and commitments, echoing Hustinx et al.'s (2005) findings. It was also noted that having competing commitments was not something that could be changed. Students were at university for their degrees and therefore academic work

would be the top priority for most, especially since [university] is already quite an intense experience (Passive volunteer).

A lack of external support also resulted in barriers to 'getting people through the garden gate' because it meant less help with recruitment was given. Student volunteering in England is still marginalised in terms of being resourced by universities and therefore many student-led organisations are under-supported, particularly in comparison to institutions in the United States where experiential learning through volunteering is a more established tradition (Brewis & Holdsworth, 2011).

Student's expectations and misconceptions about what gardening entailed and/or the gardens could offer discouraged people from taking part. Quite a few participants mentioned that a lack of confidence, stemming from a lack of knowledge of food growing, prevented students from getting involved in their garden. Other reasons the participants thought others might not want to take part was that they could not see the benefits of it. One gardener said that this might be because young people are used to instant gratification and since the benefits are often not seen immediately, they might become discouraged or bored. Some also said that cost, or perceptions of cost (i.e., for joining the student society, or paying for seeds or tools), could prevent people from wanting to get involved. On top of this, some students did not know that such an initiative existed at their university which prevented them from getting involved. This was compounded by the fact that the gardens themselves could be hard to find. All three gardens were located off the beaten track and, in some cases, there was limited advertising.

6.2.2 Barriers to continued engagement

After participants came through the garden gate, the next challenge was to encourage them to return. The barriers to continued engagement included specific barriers related to

students and student life, and some wider barriers relating to the organisation of the gardens, as well as participants missing an internal drive or competence. In addition to being a barrier to 'getting people through the garden gate,' misconceptions and unrealistic expectations also adversely impacted students' continued engagement.

Possibly the most defining feature of students and student life is transience. However, students' transience meant that eventually nearly all students involved in the gardens at the beginning of this study stopped coming back by the time data collection drew to a close. As such, students' transience increased short-term participation which resulted in a number of issues that will be picked up on in Section 6.3.

Some students said they were unwilling to commit to regular participation or to taking on positions of responsibility for fear that they would not be able to follow through. One student also said that the types of students that their garden attracted tended to be people who were less willing to commit to defined roles or plans because they joined their garden specifically for its easy-going atmosphere. However, students' lack of commitment was often recognised as being a natural thing, because being at university is an opportunity to experiment, learn, and try new things, so naturally, students would realise there are some activities they enjoy more than others.

Unsurprisingly, given students' varying schedules and that student-led food gardens were often one of many academic and social activities on students' plates, students did not always remember to come to gardening sessions even if they wanted to. Poor weather also dampened participation. This was especially problematic given that summer, the most enjoyable season for gardening weather-wise, was when most undergraduate students tended to be away. One garden did not have access to a sheltered space big enough for the group to be inside (such as a greenhouse, shed, or polytunnel) and this was especially challenging during the winter months. Whereas, in the gardens with sheltered space,

gardening sessions could run regardless of weather. A different weather-related issue was that the seasonal participation of students affected students' ability to learn about growing food holistically because they experienced the growing year in a fragmented way because of the way holidays were placed in the academic calendar.

The way the gardens were organised also presented some barriers to continued engagement. Students expressed differences in the degree of organisational structure they preferred. Some student gardeners said they appreciated and/or wanted a high degree of autonomy that a formalised structure would inhibit, whereas others struggled with the *messiness* as they *[came to the garden] expecting someone to tell you what to do* (my research diary). There is an increasing professionalisation of volunteering in the volunteer sector at large (Howlett, 2010), and volunteers want more formalised support than they used to (Rochester, 2009). Most student volunteers in England take part in unstructured volunteering opportunities (Holdsworth, 2010), though students involved in the gardens expressed an interest in both types of opportunities.

The extent of organisational structure played a role determining the extent to which students' 'psychological ownership' (a feeling that something is yours) of the garden was fostered. Psychological ownership made people feel connected to the space, accountable for what happened to it, and encouraged them to keep coming back. Of this, one student said:

...you can make [the garden] yours and you can decorate it. You can make it look nice. But you also have to clean it up. Like, there is some ownership. Some responsibility. [...] I think that that's super vital and that's really powerful... (Student leader)

This is in line with previous research that found that psychological ownership in community gardens is closely entwined with a sense of responsibility and concern for the space (Eizenberg, 2012), and that gardeners' sense of ownership is an important contributor to knowledge- and skill-sharing in community gardens (Laycock, 2013) (these terms will be

picked up on further in due course). Some of the main issues related to ownership over the gardens were a tension between the need for individual or collective ownership over tasks, people claiming ownership without contributing to the garden, and long-term participants (i.e., university/students' union staff or postgraduate students) and/or highly involved participants having too much control compared to short-term or less involved participants.

Students being able to self-determine or have control over the garden and the tasks they we involved in was seen to help build a sense of ownership and responsibility over the space, echoing similar findings about community gardens (Eizenberg, 2012). Universities owned the land the gardens were built on, but this did not seem to have limited students' self-determination of the space in two of the gardens. However, students from the other garden repeatedly expressed that they felt they had been limited by the grounds team of the university because they were not allowed to 'break ground' outside of their raised beds without scanning the ground for high-voltage cables first (see Figure 22). Participants also said that students' union-run growing spaces with concrete plans (e.g., sowing calendars) helped provide the consistency to keep it going, but that plans also prevented development of students' psychological ownership and self-determination over the space.

The desire to self-determine was often tied to students' political views. One student felt that people exercising their agency over a public place was a form of resistance in a world where there is increasing privatisation of the commons:

[The garden is] a powerful form of resistance. I think sometimes people forget that. The tidal forces of the food system are so strong and so toxic. It's just amazing that there's anything that can withstand that. That's important to me. You know, because,



Figure 22. Signage from university grounds staff issuing a blanket ban on digging in the soil. Grounds team staff appealed repeatedly to student gardeners to request the grounds team scan the ground for cables before digging. Students broke ground on numerous occasions without requesting scans, after which this sign was erected (censoring covers the university logo and contact information). you watch something like Tomorrow's Food¹⁶ and I think yeah, all the kind of tidal

forces within the development of the food system are so counter to this [garden]. And

yet, these things still exist. A thorn in the side of people who are triumphantly, kind of,

modernist. (Long-term mentor)

However, this long-term mentor also said that sometimes it could be a substantial

challenge to actually encourage people to exercise their agency over the space. Whether

intentionally or unintentionally, longer-term participants tended to exert more control over

the space that prevented newcomers from feeling a sense of psychological ownership over

their garden. It is also important to note that not all students expressed a desire to exercise

their agency over the space. These students tended to be those seeking more structured

ways to take part.

¹⁶ Tomorrow's Food is a television programme about food innovations for 'the future of food' (BBC, 2019), examining, for example, the future role of vertical farming in food production and robots in food service.

As found in previous studies (Staeheli et al., 2002; Spierings et al., 2018), cohesion within the gardening group was an important determinant of continued engagement, however cliquey-ness within the core gardening group prevented some students from wanting to come back. Conflict and internal politics within the group also prevented continued engagement. Some of the issues that conflict centred around included students having different ambitions for the garden space (e.g., what it should grow, how to grow it), and differences in ideas of how the project should be managed (e.g., how and whether money should be spent or saved). Internal politics and conflict were sometimes related to the group being too 'clique-y' or expecting others involved in the garden to share their politics. For example, one former student gardener spoke about groupthink within the group that made it difficult for people with different opinions to take part:

- Student leader: ...everyone had to be vegetarian. Or vegan [...] I mean, obviously they want outsiders to join, but they don't want outsiders to join at the same time.
- Interviewer: It's almost like they want people to join, but they also want them to take on their values as well.

[Sound of agreement]

There was a strong relationship between internal politics/conflict and ownership issues because conflict often resulted in 'power grabs' and more authoritarian approaches to managing the gardens.

A lack of a shared vision caused conflicts within the membership of gardens because it meant students invested their time (and money) in activities that moved the garden in different, and sometimes incompatible directions. Furthermore, without a shared vision, students felt loss of direction and motivation to continue to take part. One student said:

How do you define this project? You know, like, where is it going? What's the goal and stuff? I don't know. (Student leader)

Some students also speculated that one of the reasons why their garden might not continue in future would be because the university wanted to develop the land, indicating that a shared vision, not only between students, but between all stakeholders was important.

It was also thought that some students were missing an internal drive or competence, and this led them to drop out. A lack of responsibility was one of these missing pieces. There was a relationship with the extent of ownership participants felt over the space and the degree to which they felt responsible, which was also thought to be mutually reinforcing. However, active gardeners found it challenging to distribute the workload among volunteers because it was difficult to maintain a balance between delegating too much and too little responsibility. This was because, when new participants took on too much responsibility, it caused them to become overwhelmed, overburdened, or burned out, and this ultimately reduced participation. This will be discussed in greater detail in Section 6.3 below. It should also be noted that some students deliberately avoided taking on a higher degree of responsibility because they knew they did not have the capacity to be more involved.

Participants also reported that some gardeners lacked initiative or struggled to make decisions independently. This may have been caused by a lack of confidence or inexperience which meant they had difficulty understanding the level of initiative and independent decision-making needed for such initiatives to run effectively. In my research diary, I reflected that part of the reason these gardening projects can struggle with participation/transience is because most students have not been involved in projects that require this much initiative, particularly in a consistent way over a long period of time. This may also have been exacerbated by a lack of support and encouragement from long-term stakeholders for students to take initiative and be confident in their decision-making capacity.

Many students said they lacked motivation to take part more actively and that they did not know what to do to motivate themselves. One student leader called this the black box of motivation. This student said that their own motivation to take part had dwindled because they were leaving soon anyway and therefore did not see what good [it would] do for the garden to continue taking part. This indicates that it was not just the physical impacts of transience that limited participation (i.e., the students were in a different geographic location from the university and the garden), but the psychological impacts of transience did as well (i.e., lack of attachment to and responsibility for the garden). The students were quite hard on themselves and their peers for what they perceived as laziness (Student leader). However, Holdsworth (2010) has suggested that scholars have focused too much on individualised motivations to explain student involvement in volunteering. She asserts that student volunteering is a complex phenomenon and many students volunteer "without a clearly defined purpose" (Holdsworth, 2010, p. 435). A number of the students who took part in this study seemed to have internalised a preoccupation with motivation rather than thinking about taking part in their garden as just something they do incidentally or habitually. This could have been because, in some cases, they were asked about why they (or others) chose to take part. However, in many cases participants brought up motivation on their own accord when being asked about transience and improving participation more generally.

One student leader did go beyond discussing motivation and said their habits prevented them from having more consistent involvement. They explained that their *irregular lifestyle* was one of their *bad habits* that prevented them from being more regularly involved (see Figure 23). It appeared that when students were not participating at the level they wanted to they blamed it on bad habits or a lack of motivation, but did not seem to consider developing more positive habits (such as having a more consistent schedule) to change their participation behaviour.

Summer

The reason why I didn't go to garden last week is because I was still sleeping at 13:00! !! This is due to one of my bad habits, irregular lifestyle. Sometimes I sleep very late at night and even go to bed after sun rise. And then I get up in the afternoon. It sounds very irregular, isn't it?

Figure 23. One student's photovoice submission explaining why they missed so many gardening sessions during the summer.

Student's misconceptions and (unmet) expectations about what the gardens could offer also prevented continued participation. Students said their gardens were not sustainable enough, producing enough, big enough, and/or organised enough. Some also believed that there was not enough freedom for students to decide what to do (though, in some cases, this was actually true, as illustrated in Figure 22) and that they would do a different type of work than they expected (especially with regards to gardening versus administrative work). Students' unrealistic expectations and misconceptions also meant that sometimes their ambitions were bigger than their capacity to actually act on them, resulting in frustration and demotivation. One of the misconceptions that was widespread and caused a great deal of frustration and disappointment was that recruitment activities would yield more new recruits than they actually did.

In addition to all the barriers to participation outlined, a number of participants said they just *don't know* why people did not participate. Of this, one long-term mentor said that they wouldn't even know where to begin in terms of... prioritising what those [barriers] are [...] Because they are quite different for each person that's approaching the garden. The complexity of factors influencing participation and the resulting uncertainty about what the best strategies to improve participation made it challenging and frustrating to try to improve participation. These barriers to participation caused problematic participation, which then resulted in problems in the gardens.

6.3 Effects of problematic participation

There were many problems caused by problematic participation, and some of these amplified problematic participation itself, as well as the barriers to participation. Many of these problems had to do with the temporal impacts of transience. There were, however, also knowledge retention and deficit issues, and some more general knock-on effects of problematic participation.

Because of the speed of turnover in the gardens, the constant need to recruit was the most immediately felt problem. Participants found themselves mostly organising and facilitating *rather than just participating* (Long-term mentor). Some found it hard to appreciate the good things about being part of their garden because they spent so much time focused on recruitment rather than the gardening activities they had hoped they would be doing. The constant need to recruit caused friction within friendship groups as well. One student said they had repeatedly asked friends to take part in their garden and *when [their friends] didn't want to*, it brought *tension to their friendships and negatively impacted [their circle of friends]* (Student leader). The constant need to recruit also resulted in what one self-directed volunteer called an *existential crisis* for their garden. They said that:

...[maintaining participation was gardeners'] number one priority [...] but also something that is also taken for granted. And it's kind of an existential crisis if they actually identify surviving as an aim. (Self-directed volunteer)

When the gardens struggled with participation, their main aim became making sure they did not fold (i.e., the initiative was successful if it continued to exist), and this led students to question the purpose and value of their garden. Why bother recruiting people to take part if their main task will be to recruit more people?

I also observed that the 'feel' of the garden group changed rapidly. I attributed this to the fast-changing and unpredictable social network, and therefore organisational structure, that was caused by the high turnover of students through the garden. The short time that students were involved in the gardens was not enough for them to settle comfortably into a role, with students having 'formal' roles (such as president or secretary) which did not always correspond with the roles they actually played within the group:

So, [one of the other gardeners], who's there all the time, said to us, 'who wants to be exec?' and I was like, 'uhhh... I'll do the president thing.' It's admittedly not a very functional exec. I kind of... yeah. I do a lot that I'm not meant to, technically, in the role. So we just kind of do whatever. (Student leader)

This affected the continuity, retention of knowledge, recruitment of new participants, and running of activities.

In some cases, there was friction between short-term stakeholders (typically students) and long-term stakeholders (such as university or Students' Union staff members). In one garden, the friction stemmed from a lack of trust between students and staff members. This led to some students feeling exploited rather than supported:

It's just not really our space is it? You can't really do what we want with it. It's really limited, so much. When you don't feel like you own the place. Yeah, it's just, you're, like, just extracted for labour basically.

When students were involved in the initiative for a limited period of time, some university staff and other long-term stakeholders found it difficult to trust the students, especially when they felt that past students had not been trustworthy. This problem was thought to be reinforced by stereotypes about students, such as students being lazy.

In temporary organisations, 'swift trust' is reported to develop between participants, rather than more conventional types of trust that take longer to develop (Meyerson et al., 1996). 'Swift trust' is trust that is taken as a given upfront, and then verified or adjusted in time (Meyerson et al., 1996). There was evidence of this between short-term stakeholders (typically students) but not between short-term and long-term stakeholders in the studentled gardens. This may be because long-term stakeholders did not see the gardens as temporary organisations, in spite of the fact that the gardens might have felt that way for students. Trust was also complicated by unequal power relations between long-term and short-term stakeholders.

The gardens also had varying degrees of discontinuity that affected their longevity. Observing these projects over several years, I witnessed substantial turnover in volunteers which resulted in the 'reinvention' of the projects when a new generation of students arrived, sometimes as often as once a year, with a redefinition of aims, activities, planting plans, and so on. This was especially true of the garden run entirely by undergraduate students (without doctoral student input). Discontinuity resulted in a lack of strategic planning and waste of resources (e.g., soil, seedlings), volunteer energy, and potential. Given the high turnover of students, it was hard to ensure the project's longevity. Long-term thinking is especially important for food-growing projects because their success is cumulative. For example,

improving soil health requires work over the course of many years, and mismanagement resulting in, for example, potato blight, can take years to rectify.

Participants also reported not seeing the results of their work given their often seasonal/short-term engagement. This included both gardening work and administrative/strategic work done on the project. For example, some students took action to increase the visibility of the project and never saw whether increased participation resulted from it. Furthermore, because students knew they would not see the results of certain types of strategic work, some did not prioritise it. For example, one student said their group decided not to prioritise planting asparagus because by the time it would be possible to harvest it, the students who planted it would be gone.

Given that students' engagement in the gardens was usually time-bounded, their thinking about them was often limited temporally, though there were exceptions to this, particularly in doctoral and mature students. Students thinking and operating on such short timescales led them to become frustrated with how long it took to make plans a reality. One of the implications of this short-term thinking was that students involved in a short-term capacity lacked insight into how their own transience caused problems in their garden. Some students, for example, saw their garden as more permanent than it actually was and did not express concerns about who would lead it the following year. Others had expectations for the initiative that were not always realistic, for example creating much more elaborate growing plans than was achievable. In other words, students had fragmented understandings of how to cope with transience because they did not grasp the ways in which participants' transience impacted their garden.

The lack of insight into transience as a problem may indicate that long-term and shortterm stakeholders in student-led food gardens have different conceptions of time pertaining to their garden, akin to the different perceptions of time in temporary and permanent

organisations. Ibert (2004) has argued that a linear understanding of time is more applicable to temporary organisations, rather than a cyclical understanding of time. This is because temporary organisations are intended as 'one-off missions,' while an organisation operating over a longer period of time will usually have reoccurring routines (e.g., regular meetings). Therefore, it makes sense that shorter-term participants of student-led food gardens exhibited linear understandings of the time (as in Figure 24a) while longer-term participants exhibited cyclical understandings (as in Figure 24b).



Figure 24a. A linear conceptualisation of time in a student-led food garden that would be typical of participants who have had short-term engagement with the garden.





Understanding that long-term and short-term stakeholders might have different

conceptions of time helps to explain why there were tensions between these stakeholders.

New students [were] coming in every year who [had] no idea who these [long-term

stakeholders were] (Student leader), and they were encountering the challenge of

approaching people which they found intimidating (such as university staff) anew every year.

For staff, however, it was increasingly routine to engage with students asking similar questions year-on-year. The incongruence between the time scales staff and students were working on meant that students and staff had different expectations for how long it takes, for example, to respond to emails and follow through on requests. These different expectations led to ambivalence from staff and frustration for students.

It is also this linear conceptualisation of time that caused problems in retaining knowledge within student-led food gardens, because of the interruption of learning processes (lbert, 2004). In other words, students "had to give up learning in the moment in which they have gained the deepest insight into the task at hand" and this inhibited the development of organisational memory (lbert, 2004, p. 1534). While most students appeared to conceptualise time in their gardens in a linear way, some seemed to develop a more cyclical understanding of time through prolonged engagement. For example, students often entered the gardens with unrealistic expectations and misconceptions of what they could get out of the garden or achieve within it. By the time many left, they had much more realistic ambitions for the garden, though readjustment of expectations did not tend to benefit the gardens because students left the university as their expectations readjusted.

Food growing is a knowledge intensive practice, as is running a student-led organisation, and therefore it was important that gardeners had or could develop their knowledge about food growing and running a voluntary organisation. Participants reported a skill and knowledge deficit in these areas, and this was exacerbated, and in some cases caused, by a lack of knowledge transfer within the garden as an organisation. While there was a skill and knowledge deficit some students did come with prior knowledge of food growing, from having worked at a garden centre or coming from a farming family, for example.

I observed that there were four main types of knowledge and/or skills that were necessary for the gardens be sustained. There was basic horticultural know-how, or a set of
generalisable skills and knowledge about food growing. There were also place-based knowledge and skills specific to gardening, for example, what has been planted where in previous years. Administrative and interpersonal knowledge and skills, on the other hand, were not related to food growing specifically, but rather to do with sustaining the garden as an organisation. There were also generalisable and place-based versions of these types of knowledge and skills (see Table 17). The gardeners who came with existing knowledge and skills brought with them the generalisable varieties (e.g., how far apart to plant kale), but the deficit in place-based knowledge and skills (e.g., who to contact if you need compost delivered) persisted due to the high turnover of gardeners. In the study of temporary organisational amnesia' (Brookes et al., 2017). It should also be noted that gardeners learned about other things (as described in Chapter 5), but these are not discussed here as the focus is on the instrumental knowledge/skills that directly contributed to the gardenes sustaining themselves.

Types of knowledge/skills	Generalisable	Place-based
Horticultural	e.g., how far apart to plant kale, how much sun tomatoes need	e.g., potato blight has recently affected one of the beds and therefore potatoes cannot be planted there, one of the beds drains faster than the other and therefore needs more water
Administrative/Interpersonal	e.g., the most strategic way to recruit new participants, how to manage voluntary workload to avoid burnout	e.g., who to contact if you need compost delivered

Table 17. Types of instrumental knowledge and skills needed by participants of student-led food gardens to sustain their garden.

The lack of knowledge- and skill-sharing between 'generations' of students resulted in inefficiencies, duplication of work, and reduced student confidence/willingness to take initiative. Some said that inadequate participation or participants lacking necessary skills and knowledge prevented them from applying for funding they needed. Furthermore, when students with less know-how did not get support, it discouraged them from taking part in future. Paradoxically, temporary organisations are thought to be excellent places for learning, however, knowledge sharing and retention within temporary organisations tends to be much less successful (Brookes et al., 2017; Bakker et al., 2011). This appears to also be true of organisations led by transient participants, like student-led food gardens.

In addition to the temporal impacts of transience and knowledge retention/deficit issues, there were some more general knock-on effects of short-term, irregular and low participation. The most common one that was brought up by the participants was feeling overburdened, overwhelmed, or just burned out. Those that were regularly involved tended to get burdened with more responsibilities than they'd like. To compound this, those coordinating the gardens were sometimes reluctant to delegate responsibility for fear that they would overburden new recruits and scare them away. However, this often resulted in the coordinators becoming overburdened instead. Furthermore, because longer-term gardeners shielded the new recruits from some of the work, which was often the administrative work rather than the horticultural work, the new recruits did not have an understanding of all that the longer-term participants were doing to keep the garden afloat. For some, the responsibility was too much to take, which would cause highly active participants to abandon the project entirely. As such, burnout accelerated an already high turnover rate. Of this, one student said:

It seems like, in my experience, [active students] get fed up after a year, because it's too much responsibility. And then they never come again, after they're no longer on the committee. Even though they did it all the time before. Sort of, like, overload. It goes both ways. You can get too much, like, involvement, just out of, like necessity, and get just not involvement because people won't come back. (Self-directed volunteer) On top of this, low or irregular participation, compounded by the constant need to recruit, led to inadequate maintenance of the physical gardening space. There was therefore wasted

food-growing potential as the gardens became overgrown and harder to maintain, forcing gardeners to play catch-up with the tasks.

Having lower and more irregular participation also meant there was less social interaction between gardeners—a widely reported benefit of the initiatives, and a reason why people continued to come back. Problematic participation also caused negative emotional responses. The uncertainty of how many people would turn up on a week-to-week basis and how the project would be maintained made participants feel worried. Participants felt annoyed at others for not showing up, a long-term mentor was disappointed that undergraduates did not get to enjoy the joys of summer gardening, while some judged others for not pulling their weight (Long-term mentor), and generally felt fed up (Self-directed volunteer) with trying to keep the project alive without much result. Frustration was the main negative emotional response reported. Some felt frustrated by the gardens' unfulfilled potential and that others were not taking a more active role. One of the staff members from the National Union of Students said that sometimes it took a long time to set up a university garden, and this caused frustration and lost initial momentum from students. On top of feeling negative emotions, some felt they could not share their feelings without alienating others, and therefore had to keep them to themselves:

I don't want to tell them [the other gardeners] that I'm annoyed because then that will make them even less likely to come. (Student leader)

6.4 A causal loop diagram to understand transience and problematic participation

There were numerous causal connections identified by participants, many of which were outlined in the above sections. Therefore, the assumed causal relationships outlined in Figure 2 were elaborated on by introducing new causal links, creating feedbacks. Two additional variables were also added. These new variables and causal connections are depicted in a simplified causal loop diagram in Figure 25, and a more detailed diagram in Figure 26. The new variables that were introduced were that: (1) every year there was a new pool of potential recruits; and that (2) some of these potential new recruits would become new participants. The other notable change from Figure 2, other than the new causal connections, is that students' transience was found to have special role in both directly increasing and indirectly decreasing problematic participation.



Figure 25. A causal loop diagram outlining the relationships between causes and effects of problematic participation. Plus (+) signs on an arrow indicate that an increase in one variable leads to an increase in the other in the direction of the arrow. Minus (-) signs indicate that an increase in one variable leads to a decrease in the other in the direction of the arrow. A1, A2, and A3 mark the three accelerating feedback loops in the system. These are described in the text in greater detail. The greyed section of the diagram represents the impacts of students' transience. Note: (1) Transience is a barrier to participation, however because of its special role in both directly increasing and indirectly decreasing problematic participation, it is depicted as its own entity, and (2) the term 'causes of problematic participation' was substituted for 'barriers to participation', and 'effects of problematic participation' was changed to 'problems' to best reflect the responses given by research participants.

There were three accelerating feedback loops identified. In the first feedback loop

(A1), the effects of problematic participation indirectly increased these modes of participation

by increasing the phenomena that cause them. An example of this (depicted in Figure 27a)



Figure 26. An expanded causal loop diagram outlining the causes and effects of problematic participation, based on Figure 25. Plus (+) signs on an arrow indicate that an increase in one variable leads to an increase in the other in the direction of the arrow. Minus (-) signs indicate that an increase in one variable leads to a decrease in the other in the direction of the arrow. A1, A2, and A3 mark the three accelerating feedback loops in the system.



Figure 27a. An amplifying feedback loop of a vicious cycle in which low participation causes a constant need to recruit new participants which results in unmet expectations, further decreasing participation. Plus (+) signs on an arrow indicate that an increase in one variable leads to an increase in the other in the direction of the arrow. Minus (-) signs indicate that an increase in one variable leads to a decrease in the other in the direction of the arrow.



Figure 27b. An amplifying feedback loop of a vicious cycle in which lack of opportunities for students to self-determine leads to fewer new participants. This then leads to low participation, causing issues between short- and long-term stakeholders, which then reduces the opportunities to self-determine. Plus (+) signs on an arrow indicate that an increase in one variable leads to an increase in the other in the direction of the arrow. Minus (-) signs indicate that an increase in one variable leads to a decrease in the other in the direction of the arrow.



Figure 27c. An amplifying feedback loop of a vicious cycle in which low participation and less social interaction are mutually reinforcing. Plus (+) signs on an arrow indicate that an increase in one variable leads to an increase in the other in the direction of the arrow.

was that when there was low participation in the garden, it resulted in a constant need to recruit to try to increase the number of gardeners. However, this meant that students who joined the garden with the expectation that they would be able to spend most of their time gardening found their expectations unmet. As a result, those students dropped out, decreasing participation levels further.

In the second feedback loop (A2), causes of problematic participation reduced the number of new participants in the garden. Having fewer new participants lowered participation, which resulted in problems. These problems then increased the causes of problematic participation. This is illustrated in the example in Figure 27b: low participation created mistrust of students (short-term stakeholders) by university staff (long-term stakeholders) about their ability to maintain the space. This made the university staff reluctant to give students more control of the garden, which resulted in even fewer opportunities for students to self-determine, disincentivising new participants from continuing to take part. In the third feedback loop (A3), problematic participation and its resulting effects were mutually reinforcing. For example, low participation reduced social interaction in the garden, which then decreased participation further, as in Figure 27c.

There were many more causal relationships that could be drawn from these three feedback loops, as well as more complicated relationships embedded within this simplified causal loop diagram. For example, many of the problems caused by problematic participation also increased negative emotional responses (as seen in the initial prototype of the causal loop diagram in Figure 16. I have chosen to illustrate the dominating causal relationships and feedbacks in order to prevent the analysis from becoming too convoluted.

Unchecked accelerating feedback loops, like the ones outlined above, ultimately destroy the system they are a part of (Meadows, 1999). The check within this system was students' transience. Students' transience provided a buffer to the accelerating and self-

destructive feedbacks loops by providing a regular and predictable inflow of potential new recruits, which increased the number of new participants. This reduced low participation, and in some cases, also short-term and irregular participation. Yet, at the same time, transience also was the most important contributor to increasing short-term participation, thereby often resulting in low participation.

Given the three accelerating feedbacks and the fact that the buffer to these feedbacks (students' transience) increased the very same phenomena it buffered (problematic participation), it is clear that the student-led food gardens operated in perpetually vulnerable states. It is therefore unsurprising that participants expressed that there was no hope in moving away from the vulnerability:

I, personally, find it difficult to see how [the maintenance of the garden] is ever going to be a sustainable situation. (Student leader)

This uncertainty, vulnerability, and, in some cases, hopelessness characterised the internal reality of student-led food gardens. The discussion in the following section, however, will begin on a more optimistic note by outlining where and how the system can be most strategically intervened in to reduce short-term, irregular, and low participation.

6.5 Strategic points for intervention

Here, will explore some strategic points for intervention to address the impacts of students' transience, rather than focusing on reducing transience itself, as has been done previously (Hyde et al., 2016). I will also make some initial reflections on how some of the issues highlighted in the above sections might be address, however this will be considered more substantively in Chapter 8.

In spite of the precarity of the student-led food gardens, knowing what the accelerating feedback loops are means that stakeholders can intervene at strategic points to

slow the feedbacks. Because feedbacks are present, both 'upstream' (focusing on mitigating the problem) and 'downstream' (focusing on adapting to the problem) solutions are relevant. I therefore advocate a holistic approach to addressing the impacts of transience and problematic participation. This includes addressing barriers to 'getting people through the garden gate', barriers to continued engagement, and effects of problematic participation simultaneously. The gardens studied each had different issues that dominated their gardening group. As such, the decision to intervene on a particular issue should be made at the level of the individual garden in question. However, the causal loop diagrams in Figures 25 and 26 could still be used practically by stakeholders of student-led food gardens to plan action to reduce the negative impacts of problematic participation. This could be done by identifying the issues from the causal loop diagram that dominate in their garden and then using this prioritisation to determine the most strategic actions to take. The causal loop diagrams could also be used to assess how holistic the current menu of actions that are being taken to address the impacts of students' transience are. Stakeholders could do this by mapping actions onto the causal loop diagrams to assess if both upstream and downstream interventions are being taken. I do this with the proposed strategies to address the impacts of students' transience in Chapter 8 of this thesis.

6.6 The role of long-term stakeholder support

In Chapter 2, I highlighted that student-led food gardens are not temporary organisations, but I also speculated that they may have features akin temporary organisations because they have such high turnover of participation. In this chapter, I noted that short-term stakeholders of student-led food gardens have conceptions of time and approaches to trustbuilding akin to those seen in temporary organisations. However, long-term stakeholders exhibited conceptions of times and approaches to trust-building that are more akin to permanent organisations. As such, student-led food gardens seem to be not-quite-temporary, yet not-quite-permanent in terms of how trust-building manifests and how their stakeholders conceptualise time. This indicates that student-led food gardens, as organisations with transient participants, might actually have an organisational form unique from those seen in temporary or permanent organisations. This is because stakeholders appear to have dual conceptions of both time and knowledge, which produce challenges with managing these different and, at times, conflicting approaches and conceptions. In section 8.2.1, I will expand on this by drawing on the literature about temporary organisations to suggest that student-led food gardens can capitalise on having both long-term and short-term stakeholders in order to both retain knowledge within the garden as well as maximise student learning.

6.7 Conclusion

The internal realities of student-led organisations at universities have not yet been well-explored in the literature, therefore this chapter has taken an initial step towards filling this gap by examining the causes and effects of problematic participation in student-led food gardens, and the feedbacks between these. In this chapter, I have mapped the causes and effects of problematic participation in student-led food gardens, and the feedbacks between these. There were two main causes of problematic participation, which were barriers to 'getting people through the garden gate' and keeping participants coming back. Students' transience prevented students from coming back because of the time-limited nature of university degrees, and this caused short-term participation. However, students' transience also had an important role in increasing participation overall by refreshing the pool of potential new recruits on an annual basis. Problematic participation naturally caused a range of subsequent problems. I mapped how the causes and effects of problematic participation influenced each other and identified three accelerating feedbacks, which were buffered only

by an annual refreshment of the pool of potential new recruits. This mapping illustrated the vulnerability that the gardens' participants described and felt acutely. Within the diagram we can also find the beginnings of how to intervene to de-accelerate the feedbacks. I recommended that both upstream and downstream solutions should be implemented, and that university and students' union staff could play a crucial and subtle role in supporting students. This could be through creating a hybrid management structure where staff input into specific parts of the garden or at particular times of the year, such as during holidays. Towards the end of the chapter, I also reflected that student-led food gardens have features that are akin to both temporary and permanent organisations. Short-term stakeholders experienced the gardens through linear conceptions of time, while long-term stakeholders experienced them through cyclical conceptions of time. These dual approaches to and conceptions of time indicated that student-led food gardens, as organisations with transient participants, might actually have an organisational form unique from temporary and permanent organisations. These conflicting understandings of time were a source of friction, and because the dominating stakeholders were students, 'organisational amnesia' was also prevalent, which exacerbated the gardens' precarity. I also suggested that it might be appropriate to consider further the extent to which these gardens should be student-led and what role university or students' staff might play in supporting them. Having mapped the impacts of students' transience on the dynamics of problematic participation and offered an initial assessment of the gardens' vulnerability, I will offer a more holistic and overarching assessment of the gardens' vulnerability using Biggs et al.'s (2012, 2015c) resilience principles.

Assessing the vulnerability of student-led food gardens

...the nature of it, is like, inherently non-continuous. Because students, they stay for a year, and then they do something else.

Student gardener, University of Warwick



Carrots and onions in eroding soil June 2013

7.1 Introduction

This chapter focuses on assessing the vulnerability of the student-led food gardens. I do this by drawing on the principles for building resilience (Biggs et al., 2012, 2015c) outlined in Chapter 3 and considering the extent to which they manifest in the gardens. This means I will consider how well diversity and redundancy are maintained, connectivity, and slow variables are managed, complex adaptive systems thinking is fostered, learning and experimentation takes place, participation in governance is fostered, and polycentric governance is present. Because this is the first time these resilience principles have been fully applied outside of the context of sustaining ecosystem services, I will also assess their broader applicability to other contexts. The first section of this chapter will be spent assessing the overall vulnerability of student-led food gardens. The second section will focus on the extent to which the resilience principles have relevance beyond the application to ecosystem services, and if and how they might need adjustment for different contexts. Some of the themes from Chapter 6 will be revisited in this Chapter, such as the quality of relationships between university/students' union staff and students, ownership issues, cliquey-ness and cohesion, and so on. These, however, will be approached through the lens of the resilience principles to offer a different perspective.

7.2 Assessing the vulnerability of student-led food gardens

In this thesis, I use the term vulnerability to refer to the likelihood of a system "to change to a new state in response to a stress or disturbance" (Chapin et al., 2009, p. 17). This is an antonym to resilience, "the capacity of a system to absorb a spectrum of shocks or perturbations and still retain and further develop the same fundamental structure, functioning, and feedbacks" (Chapin et al., 2009, p. 9). As such, if the resilience principles 207

manifest in the student-led food gardens to a high degree, the system would be considered resilient, but if they do not manifest the gardens would be considered vulnerable.

7.2.1 Maintenance of diversity and redundancy

The first resilience principle is to maintain diversity and redundancy, therefore I my initial step was to assess the extent to which diversity and redundancy were maintained in the student-led food gardens. Participants brought up several axes of diversity that appeared to impact the resilience of the gardens: the types of participants (and their length of participation, reasons for taking part, management style preferences, and opinions/worldviews), the diversity of activities run, and the diversity of crops grown. These are summarized in Table 18.

Given the instability and resulting uncertainty in student-led food gardens, building in diversity and redundancy is of crucial importance. I reflected on this in my research diary:

You don't really know who's going to be the next committed individual to take over the project – are they organized? Are they disorganized? Do they know a lot about food growing? Do they not? Do they prefer to experiment and learn about food growing that way? Or do they prefer a more formal learning experience and would benefit from some documentation of the growing practices used at the site? It's all an unknown, so if you want to build some real resilience into the project, you need to be prepared for all eventualities. And not just prepare for the garden to continue in its current state, however it's run. So, this means anticipating that there might be people coming in who want to experiment, and therefore a crop rotation might go out of whack. So, choosing to plant disease resistant crops and perennials can help for this scenario. Or the next person to come in might be interested in running the project in a more formalized way – so they could benefit from documentation and a handover manual. The more

Axes of diversity/ redundancy		Key findings	Overall diversity/redundancy	
Types of participants	Length and regularity of participation	 Undergraduate students (rapid turnover), postgraduate students and staff (lower turnover) were the main types of participants Uneven balance of participants, with mainly undergraduate student participants Few 'leaders' Students took regular holidays where they left university for long periods, whereas postgraduates and staff were around more consistently over holiday periods 	There was some variety in length/regularity of participation, however there was an uneven balance between different lengths of participation. There was low redundancy in participants taking part for longer periods.	
	Reasons for taking part	 So varied and disparate that the only way to strategise was to create different ways participants could engage, including through: horticultural/garden maintenance, administration/recruitment, or socialising Few people interested in the administration/recruitment compared to horticultural/garden maintenance and socialising activities 	Reasons for taking part were varied and highly disparate.	
	Management style preferences	 Some wanted more structured opportunities for taking part, others wanted more freedom to decide how they engaged Overall, very informal management styles used in the three gardens studied, however some participants that took part in the fishbowl discussion at the Student Eats conference reportedly had gardens that offered more structured opportunities (particularly at further education institutions) 	There was variety in management style preferences within each garden, although management styles practiced were relatively homogenous within gardens. However, there was variety in management styles practiced between gardens.	
	Opinions and worldviews	 Often varied viewpoints, usually thought of as a source of conflict rather than an asset A certain extent of groupthink with regards to 'green issues' in some cases Some students who were sustainability advocates had difficulties relating to the wider student population 	Diversity and disparity of viewpoints between gardeners varied between gardens and over time. In some cases, there was limited diversity of viewpoints within the garden, but high disparity between gardener and non-gardener viewpoints.	
Activities run		 Three main activities: horticultural/garden maintenance, administration/recruitment, and socializing Administration/recruitment activities not seen as enjoyable, but sometimes dominated out of necessity 	There tended to be high levels of variety and disparity in activities run, but at times there was an uneven balance between administration/recruitment and other activities.	
Crops grown		 More diverse crops meant that people did not want to take them home as they did not like them or know how to cook them Less crop diversity meant easier management, but this mainly happened in gardens that were less student-led 	Diversity in crops grown varied from year to year, but in general there tended to be high levels of variety and disparity, but balance between the amounts of different crops was uneven. Diversity of crops was sometimes limited to make management easier.	

Table 18. The extent to which diversity and redundancy were maintained in the student-led food gardens.

systems, and the more types of systems, put into place to cope with transience, the more resilient the project will be.

As such, diversity and redundancy can allow for spontaneous and serendipitous opportunities to be exploited.

Gardeners zealously sought to create redundancy in participation in general, because it was thought that the more people involved, the less likely drop-offs in participation would be. There were three main types of participants: undergraduate students, postgraduate students, and university or students' union staff (though campus residents or local community members had limited involvement in some cases). Undergraduate students were overwhelmingly the main participants, which, to some extent, represented their proportion of the university population. Their rapid turnover in and out of the gardens increased the diversity of interests, opinions, and approaches, which complicated relationship building with long-term stakeholders:

...there's usually, kind of one or two of people that will be involved with the project every, sort of three or four years. So every, sort of academic cycle, that maybe three years at university. And I can imagine that the diversity that comes with that can probably be really, really hard to make a concrete relationship, or to even kind of think of a concrete relationship [with long-term stakeholders]. (Student leader)

There were efforts to increase postgraduate and staff participation, as they were usually present during the summer and undergraduates were not. Having different types of participants often resulted in diversity in the length and regularity of participation. However, it was also noted that a balance between staff and student participation needed to be maintained to ensure students still had autonomy over the garden.

As highlighted in section 6.2.2, there was heterogeneity in participants' management style preferences: some wanted the gardens to offer more structured volunteering

opportunities, while others wanted more freedom in how they used the space. The three gardens studied in detail all had fairly informal organisational structures since they were student-led and this put off students who wanted to take part in more formal opportunities. However, some participants that took part in the fishbowl discussion at the Student Eats conference said their gardens offered more structured opportunities for taking part. These opportunities were mainly offered at further education institutions where there was more staff support for student initiatives.

There was also variability in participants' opinions and worldviews. Rather than being perceived as an asset, varied opinions within the garden were typically referred to as a source of conflict, especially the more disparate they were. Furthermore, it was noted that students involved in the gardens had difficulty relating to the wider student body because their views were too divergent (as in section 6.2.1).

Participants took part in the gardens for a variety of reasons. In fact, a long-term mentor said that *it's kind of confounding how many different reasons there might be for someone coming down to a community garden.* The reasons prospective gardeners became involved or chose to keep coming back were so varied and so disparate that it was difficult or possibly futile to try to create a strategy to appeal to all potential new recruits, other than to try to create a space *where people can engage in multiple ways* (Long-term mentor). As such, there was diversity in the types of activities run in the garden, with the main differentiators being whether it was a task focused on horticultural/garden maintenance, administration/recruitment, or socialising:

So what we've done is that we do our activities, our sessions, and do different types of sessions. So we do learn to grow, and then to cook, and then to harvest. And marketing. And all of these things which also includes a broader diversity of students.

And then we do social events as well, which is associated with our project. (Students' union staff member)

Most students took part in gardens for the horticultural/gardening tasks or for socialising, and administration/recruitment activities were not seen as particularly desirable to work with. Furthermore, because administration/recruitment tasks were so crucial (to the point that at times they began to dominate) it was hard to maintain a balance between horticultural/gardening tasks and the administration/recruitment tasks. The distribution of which participants performed different activities was also uneven, with long-term participants typically taking on more administration/recruitment tasks. This meant that shorter-term participants were not always aware of burden that their long-term counterparts were shouldering.

7.2.2 Management of connectivity

Different aspects of connectivity were mentioned by participants, reflecting modular connectivity, "subsets of densely connected nodes that are loosely connected to other subsets of nodes" (Biggs et al., 2012, 428), and nested connectivity, "where specialist nodes interact with subsets of generalist nodes" (Biggs et al., 2012, 428) (Table 19). The connections between the students in the group and between different student societies were modular, while the connections with other parts of the university (both the physical and the human parts) and Students' Union were nested.

Within the gardening group, there was an ongoing struggle to balance the amount of connectivity between participants such that there was a lively, cohesive social group, but that was not too cliquey and therefore open to newcomers:

...it's really hard to find the right balance, I think, because if it gets too closed then it's more like a friendship group that all do it together. It's hard for new people to join in as

Туј	pes of connections	Key findings
Modular	Connections within Student Eats gardens	There was difficulty balancing between being too cliquey and not cohesive enough, when fostering a lively, cohesive social group.
	Connections between Student Eats gardens	Connections to other Student Eats gardens provided inspiration and a feeling like gardeners are a part of something bigger.
	Connections between student societies	Students involved in Student Eats societies often were active in other societies as well.
Nested	Connectivity to the university	The relationships between participants in the gardens and university staff were weak in all gardens. In some cases the relationships were positive or neutral, but in one case they were fraught with tension and mistrust.
	Connectivity to the students' union	The relationships between the gardens and students' unions were not particularly strong, but were regularly upkept by the students' unions through their organisation of fresher's fairs and reminders about elections.
	Connectivity to the National Union of Students' Sustainability team	The relationships between the National Union of Students' sustainability team and the gardens were mainly through the gardens' leaders and were characterised by information sharing and the facilitation of relationship building between gardens.
	Gardens' connection to the rest of the physical campus	All three of the gardens were in low-traffic areas on campus and one was walled-in and only accessible during working hours.

Table 19. The characteristics of the types of connectivity at play in the student-led food gardens.

well. [...] It needs to be cohesive enough to keep people in, but not so cohesive that it

leaves people out. (Self-directed volunteer)

However, the high turnover of students meant that there was a natural disruption to the

formation of cliques.

There were also modular connections between the gardens and other student societies, where students who were active in the gardens were often also active in other student societies. One student leader said that by taking part in the garden, *you kind of get roped into everything*. [...] *One minute you've got loads of time, and then the next you're signed up to, like, 10 billion things*. Students reportedly enjoyed this, but it did mean that much of the sustainability and other extra-curricular activity at the universities was being underpinned by a small, overworked group of individuals.

The two nested connections brought up by the research participants were the connectivity of garden groups to the wider university social structures, and connectivity of the physical garden site to the wider physical campus. The two main social structures that the gardens were nested within were the university and students' union. The connections were relatively weak in the three gardens studied by virtue of the sampling criteria. However, what appeared to be the main determinant of the quality of these connections was their character. In some gardens the relationships between students and the university/students' union were trustful, mutual, and at worst, indifferent. However, in one of the gardens in particular, there were problems with mistrust and power asymmetries which made student-staff relations fraught (as outlined in section 6.2.2 and 6.3). In the gardens where the relationships were positive, the gardens benefitted from, for example, donations of compost. However, in the garden where the relationships were fraught, students were given limited scope to self-determine, which lowered their sense ownership of the space, further escalated the tensions, and lowered participation.

The relationships between the gardens and students' unions were not particularly strong but were generally positive. The relationships were kept alive mainly though the students' unions' organisation of fresher's fairs and their reminders about elections. The connection between the National Union of Students' sustainability team and the gardens were facilitated mainly through the long-term mentors and student leaders. The sustainability team shared information with the garden groups and the facilitated relationship building between gardens, such as through annual conferences. In some cases, the sustainability team also arranged visits to particularly inspirational Student Eats gardens, and students at some gardens also took initiative to arrange visits to other Student Eats gardens.

Finally, the gardens' connectivity with the rest of the campus impacted the likelihood of people stumbling across them and how well-known they were. One garden was *walled-up* with a *padlock on the gate* (Student leader) outside of working hours, which meant the space was not always readily accessible to students. These physical barriers also created a mental

distance, with a *mindset that [the garden] is a secluded and exclusive space for people* (Student leader).

In summary, modular connections within Student Eats gardens and between different student societies were highly variable and changed rapidly. Nested connections between the gardens and the university, and students' union varied between gardens, but tended to be weak (given the gardens' student-led nature). Overall, it was the quality of connections that mattered, with some gardens having good quality relationships with their university and students' union, and one having a more tense relationship. The connections between Student Eats gardens and between gardens and the National Union of Students were weak, but provided positive outcomes overall. The connections between the gardens and the rest of the physical campus were weak, and as such provided limited opportunities for serendipitous engagement.

7.2.3 Monitoring of slow variables and feedbacks

The dominating variables and feedbacks relating to the impacts of transience are outlined in Chapter 6, and in this section I will consider the slow variables and feedbacks in particular (Table 20). Students identified a number of the slow variables, an important precursor to monitoring them (Biggs et al., 2015a). While participants identified causality between these variables, often they didn't discuss them in a way that indicated they saw them as feedbacks. New gardeners typically were not cognisant of these variables, but over time they became aware of them. However, nearly all gardeners left the gardens during the course of this study and therefore there were few who had a grasp of how these variables had changed over time.

There were several slow variables and feedbacks of note. One was the students' indifference or apathy towards gardening as an activity, or sustainability in general. In the

Table 20. The slow variables and feedbacks at play in student-led food gardens

Slow variables	Related feedbacks	Key findings
The 'cultural barrier': Students' indifference/apathy to food growing and sustainability	+ Cultural barrier +	• Indifference/apathy towards food-growing and sustainability could be changed though participation in the gardens, but low participation limited this 'culture change'
Relationship between long- term and short-term stakeholders	Healthy relationship between long-term and short-term stakeholders Problematic participation	 In gardens where the quality of the relationship between long-term stakeholders and short-term stakeholders was good, more autonomy over the garden space was granted to students which increased the quality and amount of participation (creating a virtuous cycle) In the garden where the quality of the relationship was low, this cycle became vicious, where students lacked opportunities to self-determine and participation became more problematic
Physical state of the garden	+ Inadequate maintenance of the garden + +	 At the times where the state of the gardens and their infrastructure were inadequately maintained, people were discouraged from participating because of how much work needed to be done to lift the gardens out of a state of disrepair (a vicious cycle) Conversely, when the gardens were well-maintained there was less problematic participation because people gravitated towards wanting to be part of something successful (a virtuous cycle)

Table 20 (continued). The slow variables and feedbacks at play in student-led food gardens

Slow variables	Related feedbacks	Key findings
Emphasis on recruitment vs other goals	Problematic participation + + + + + +	• An over emphasis on recruiting new participants to take part in the garden resulted in a loss of purpose for participants (described in Chapter 6 as an 'existential crisis) which caused participants to drop out or participate irregularly

previous chapter, one gardener called this a 'cultural barrier,' which I used as an *in vivo* code. This was a variable that was of crucial importance and dramatically impacted the recruitment of new participants. However, it was also one that that little additional action could be taken on. The gardens themselves were set up in part to break down this *cultural barrier*, and many students were already active in trying to make change in this area.

Another slow variable of note was the relationship between long-term and short-term stakeholders. In the gardens where the relationship was a productive one, students had more autonomy over the space and the support of staff. In the garden where the existing relationship was tension-filled, students across generations had their autonomy stripped away before they even began taking part because the foundation of mistrust had already been laid.

Another slow variable included the overall physical state of the garden. This included the state of the raised beds, greenhouses, compost bins, seating areas, and the soils, to name some of the key aspects. The built structures in the gardens impacted how easy the space was to maintain in periods of low participation. For example, participants said raised beds kept the space looking tidy. On the other hand, some of the built structures in the gardens degraded over time: greenhouses windows broke, and a rabbit-proof fence was shredded by a string trimmer. During the period of the study, these were not repaired. A lack of monitoring of what had been planted where year-on-year also caused the spread of potato blight in one of the gardens.

7.2.4 Fostering complex adaptive systems thinking

There was little evidence of complex adaptive systems thinking (see Table 21). Unsurprisingly, there was no systems framework used or acknowledgement of epistemological pluralism as a source of complexity given the type and scale of the initiatives.

Table 21. Key findings about the extent to which complex adaptive systems thinking was embedded in
student-led food gardens.

Characteristics of complex adaptive systems thinking	Key findings
Uncertainty tolerant culture	 Considerable evidence of participants lacking tolerance to uncertainty Some evidence of participants developing tolerance to uncertainty through participation
Use of a systems framework	No evidence
Acknowledgement of epistemological pluralism as a source of complexity	No evidence
Investigation of critical thresholds and non- linearities	No evidence
Change institutional arrangements to reflect complex adaptive systems functioning	• While there were formal roles filled by students (president, secretary, etc), there was flexibility in term of how these were fulfilled to account for the rapid turnover and changing landscape of the garden group

While there was some evidence of participants identifying the importance of slow variables, most participants took part for such a short period such that investigation of critical thresholds and non-linearities could not be expected. However, there was evidence that many struggled to tolerate uncertainty, a key feature of complex adaptive systems thinking (Bohensky et al., 2015). Students expressed discomfort and stress about the uncertainty of how many people would turn up on a week-to-week basis, and the long-term uncertainty about how the project would be maintained:

... the uncertainty that, you know... every session I'm wondering how many people are going to turn up. Did I make a good enough effort remind people that there is a session on today? Is the weather going to stop people from turning up? And then, also, the more general thing that has to do with that is who's going to run it over the summer when the students that are going home are gone? Who's going to run it when the most amount of people... and also who's going to run it next year? Is it going to dissolve? Or is there going to be a group of people in the spring that are like, 'Yeah, we want to take it on!' (Student leader) However, those that were involved over a longer period of time developed more tolerance to the uncertainty of and caused by participation. For example, a long-term mentor said:

It's not something that bothers me. Like, 'God, I need people to turn up this week!' Because that would just be unsustainable. Because then you'd just be constantly fretting. Uh. But yeah, I see it [in] other people. People come down and say, 'Is anyone else coming'?

This indicates that complex adaptive systems thinking might be fostered through taking part in student-led food gardens. However, since there is such a high turnover rate, those that develop complex adaptive systems thinking would likely leave the university before the competence can be exploited in the gardens. There was also some evidence of changes in institutional arrangements to reflect complex adaptive systems functioning. As highlighted in section 6.3, student societies had formal roles, like president, secretary and treasurer, but in practice the functions these roles fulfilled were extremely flexible to account for rapid turnover in participation.

7.2.5 Encouraging learning

Most participants expressed that there was a knowledge deficit in the garden, with many students reporting they did not *feel qualified to participate* (Student leader). However, some students did report learnings that were instrumental for maintaining their garden, such as horticultural and administrative/interpersonal knowledge/skills (see Table 17 in Chapter 6 for examples of these). This type of learning, however, appeared to be limited. This is unsurprising as many participants participated irregularly and peripherally, especially given their seasonal hiatuses. Furthermore, the limited learning that took place often failed to be retained within gardens between generations of students (Table 22).

Table 22. Key findings related to learning in student-led food gardens.

Key findings

- Types of learning taking place:
 - o Knowledge/skills
 - o Tacit knowledge
- A knowledge deficit was present
- There was a lack of knowledge retention within the gardens
- Opportunities for learning through experimentation were present
- Most students viewed learning as a passive exercise
- Maladaptive learning was present
- Diversity of viewpoints is a potential opportunity for learning
- Unrealistic expectations and misconceptions are potential opportunities for learning

In all three of the gardens, learning through experimentation was encouraged by

virtue of the fact that most people knew relatively little about gardening. However,

participants tended to discuss education and learning as an outcome rather than a process,

which made them eager to seek out expertise and reluctant to experiment. This was

compounded by students viewing learning as a passive exercise. For example, one student

leader said that other students were willing to take their advice because the other students

lacked confidence and therefore were eager to seek out *authority*, even though they (the

student leader) had absolutely no idea what [they were] doing.

Some of the learning reported by participants also appeared to be somewhat

maladaptive, describing learning that made them feel stressed or disengage from their garden:

I realized how much I've [...] been aware of the limitations... the kind of stressors that come along with being involved in a project like this. Like, seeing everyone saying it's really stress relieving... it's not really anymore for me... (Long-term mentor)

I can now see [the disenfranchisement in the] third- years that I met in first year. I can now see how they had a similar thing of just not doing anything anymore. And at that point I was like, 'What are you doing? There's so much to do, let's do something!' But now I understand a lot better why you would be in that place. And... yeah, I think a lot of it was just, 'Okay, I've done so much. I think it's okay to not do anything anymore.' That was a big, big part of that. And also, just, like, feeling like it wouldn't... I don't know. Like, what good would it do? (Student leader)

As mentioned in section 7.2.4, other gardeners reported that they learned to be tolerant to the uncertainty they experienced in the gardens. When asked whether this sort of learning could be shared with others, a National Union of Students staff member said that it would not be possible to pass it on. As such, it seemed that the learning that enabled students to be more tolerant of uncertainty was a tacit form of knowledge, meaning it may not be possible to codify or communicate it. However, this National Union of Students staff member suggested that having reassurance from peers or mentors that what they were going through was normal could help alleviate the worry and anxiety caused by uncertainty until they learn how to better cope.

In sections 6.1.2 and 7.2.1, I outlined how the gardens' participants encountered others with viewpoints that were different from their own in the gardens with regards to, for example, organisational preferences or views on sustainability issues. The interactions between those with divergent viewpoints could be considered a 'boundary interaction,' that is, an interaction where "experiences brought in from the outside [for example, of a garden] generate friction as they encounter the socially defined competence of the community]" (Bendt et al., 2013, p. 19). These boundary interactions can enable learning through negotiation of meaning between the individual and community (Bendt et al., 2013). However, if the competences of the individual and community are too convergent, learning is limited. Conversely, competences that are too divergent will mean there is "no common ground upon which to negotiate meaning" (Bendt et al., 2013, p. 20). As such, the clique problem that limited outsider engagement may have not only limited participation but may have limited learning as well.

Likewise, the finding that students had unrealistic expectations and misconceptions about what they could expect from student-led food gardens indicates that there is substantial potential for learning about, for example, what reasonable expectations are in voluntary initiatives, and how to cope with some of the challenges that come with these (such as growing low-maintenance plants to deal with irregular participation).

7.2.6 Broadening participation

Broadening participation was a central focus of the student-led food gardens (and indeed, a central focus of this thesis) because the principal results of students' transience were short-term and low participation. However, Biggs et al.'s (2012, p. 201) resilience principle defines broadening participation as the "active engagement of relevant stakeholders in the management and governance process," not just participation in an activity in general. Therefore, here I will assess the extent to which participation was broadened in the governance process (see Table 23). In the case of student-led food gardens, that which needed governing was how to use the garden space (e.g., what to plant where, what activities to focus on in gardening sessions), recruitment and associated activities (e.g., how and when to recruit, marketing), and decision-making about the garden group as an organisation (e.g., decisions about the society fees and budget, elections).

There were some stakeholders who were involved in governance, some involved in gardening, and some involved in both (Figure 28). The day-to-day governance of the gardens tended to be undertaken overwhelmingly by long-term mentors and student leaders. It was an ongoing challenge to include other gardeners in governance, because they didn't want to put other students off by overburdening them with responsibilities.

Difficulties broadening participation in the student-led food gardens in general naturally led to difficulties broadening participation in their governance. Some gardeners took

Table 23. Key findings about considerations when broadening participation in governance of studentled food gardens.

Considerations when broadening participation in governance	Key findings
Clarity of goals and expectations in governance	• No evidence, appeared that goals and expectations in governance were largely unspoken
Involvement in governance	• The 'right' stakeholders were not always involved in decisions (e.g., the grounds team were not informed of digging directly into the soil, students were not involved in long-term planning of the land the garden occupied)
Facilitation and leadership	 Effective facilitation and leadership between different stakeholders came from the National Union of Students' sustainability team which facilitated conversations between universities' grounds and sustainability teams, students' unions, and student gardeners Long-term mentors and student leaders also played a key role in broadening participation in governance through engaging other students and reaching out to university or students' union staff in some cases
Capacity building for participation in governance	• Students' capacity for participation in governance was built through learning and experimentation, though this capacity typically was not retained within the garden because of students' turnover
Power differentials	• The power differential between university staff and student made it <i>intimidating</i> (Student leader) for students to approach staff
Resourcing	 There were no financial resources for participation in governance given the small scale of student-led food gardens The universities seemed to dedicate limited time and effort towards broadening participation in the governance of the space



Figure 28. A Venn diagram of those participating (or not) in the governance of student-led food gardens.

leadership with governance because they worried that if they did not, then no one else would step up or others would find it too burdensome and leave the project. This was not a baseless worry. A passive volunteer I interviewed who had been involved in one of the gardens said that the reason they had stopped coming was because it had stopped being *pure fun* and they started feeling blamed and guilty for not committing more to the project. Because of attempts to diffuse governance responsibility, they felt the fun had been taken out of the garden and stopped taking part. However, not distributing responsibility also caused longterm and/or highly invested individuals to become overburdened with work and/or burn out. Those who were overburdened were cognisant of this issue, but often found it was very challenging to let go of control because the worry and risk involved for the overall success of their garden. Other reasons they reportedly found it difficult to delegate were because it was sometimes seen as less time efficient or more draining than the person in charge doing it themselves, or because they did not feel there were enough people to take on the tasks.

Typically, this 'control' would manifest in subtle ways, such as the more engaged gardeners holding the historical knowledge of the project and more gardening knowledge. Because knowledge was held by participants with higher or longer-term engagement, an information deficit and inadequate information-sharing were brought on. On this, a student leader said:

I said, 'We need to organize everything that's going to be planted this year.' And he [a long-term mentor] was saying, 'Oh, the way we've done it before is we've just done it whenever. It allows people to do what they want, it's very flexible.' And I was trying to make the point that 'If you fall off a cliff, so does this allotment.' Not that I'm wishing that upon him. But he's very much holding all the knowledge, and he's saying he doesn't want to hold all the knowledge. Um, and the control. But he does. And because he holds that authority, what he says goes.

In one of the gardens, the 'right' stakeholders were not always involved in decisions. For example, in one garden students did not inform the grounds team before they dug directly into the soil (as mentioned in section 6.2.2), but likewise, students were not involved in long-term planning for the land the garden occupied. The most effective facilitation and

leadership in governance came from the National Union of Students' sustainability team, who facilitated conversations between universities' grounds and sustainability teams, students' unions, and student gardeners. Long-term mentors and student leaders also played a key role in broadening participation in governance through engaging other students and reaching out to university or students' union staff, in some cases. However, power differentials were not always acknowledged or addressed (as highlighted in sections 6.2.2 and 6.3). University grounds staff typically did not have time allocated within their work schedules to dedicate to engaging with gardeners or to broaden participation in the governance of the space.

7.2.7 Promoting polycentric governance

There were four main 'governing authorities,' to use the term employed by Schoon et al. (2015). Each of these operated at different, but at times overlapping, vertical scales. At a micro-level, the student group tending to the garden made most of the on-the-ground decisions about, for example, when to hold gardening sessions, what to do in the gardening sessions, and what the overall aims and ambitions for the garden should be.

The students' unions supported students with governing and promoting their student societies. The garden societies were one of dozens of societies supported by the students' unions. The students' unions required the societies to have elections annually, and organised 'fresher's fairs' where students could advertise their societies and recruit new members.

The universities themselves owned the land the gardens operated on, and therefore had the responsibility and the mandate to ensure the space was presentable and safe. Some of the universities also had staff supporting the student societies to varying extents, such as student support or academic staff. The National Union of Students' sustainability team supported gardening initiatives at different universities all over the UK with information about food growing and running a food growing society. Their staff also focused on building

relationships between gardens at different universities through site visits and annual conferences.

The linkages between the different governance units were outlined in section 7.2.2. These linkages can also be seen in Table 24, where they are stratified into horizontal linkages (linkages between governance units working at similar scales) or vertical linkages (linkages between governance units working at different scales). In most cases, the 'domains of authority' appeared to be appropriate, though in the garden with tension between staff and gardeners, gardeners had less control of how use the garden space. Horizontal linkages were not especially notable, and seemed to be providing positive functions overall, but there did not appear to be any that were more important than others. However, the vertical linkages were a little more complicated. Relationships between the gardens and the National Union of Students and the gardens were positive and supportive. Relationships between university and students' union staff and the student-led food gardens were weak (likely due to the nature of the sampling: gardens with limited staff involvement were chosen as embedded units of analysis). University and students' union staff could have sought out gardener participation in governance of the garden space to a greater extent, particularly in the garden where the relationship was fraught.

7.2.8 Summary of the vulnerability of student-led food gardens

Overall, the student-led gardens in this study were very vulnerable to the impacts of students' transience. This was especially true of the garden at Sheffield, because it did not have postgraduate participation. A seemingly successful garden could rapidly shift into a garden at risk of total neglect. I was told by a staff member at the National Union of Students that one entirely student-run garden was one of the most successful gardens, and when I came to meet them the first time in the spring, the gardening session I attended was lively

Governance unit	Domain of authority	Horizontal linkages	Vertical linkages
Student-led food garden group (student society)	 Tending the garden space and running the gardening society While the garden groups had autonomy over their prescribed growing space in all gardens, in one of the gardens students were limited to growing in the raised beds and not allowed to break ground outside of the beds 	 Participants of some gardens visited other Student Eats garden sites throughout the course of the study Participants of the gardens met and learned from participants of other Student Eats gardens at the Student Eats conferences several times during the course of the study Many students participating in the gardens were involved with other student societies within their university 	 The relationships between the gardens and students' unions were not particularly strong, but were regularly upkept by students' unions through their organisation of fresher's fairs and election reminders The relationships between gardeners and university staff (particularly grounds staff) were weak in all gardens. In some cases, the relationships were positive or neutral, but in one case they were fraught with tension and mistrust The relationships between the National Union of Students' sustainability team and the gardens were mainly through the gardens' leaders and were characterised by information sharing and the facilitation of relationship building between Student Eats gardens
Students' union	Supporting all student societies to self- govern and promote their societies	• The National Union of Students' sustainability team was in contact with the students' unions and universities about spending funding, new funding opportunities, and annual renewal of the institutions membership to the Student Eats programme	• The relationships between student-led food gardens and students' unions were not particularly strong, but were regularly upkept by the students' unions through their organisation of fresher's fairs and reminders about elections

Table 24. Key findings about the promotion of polycentric governance in student-led food gardens.

Governance unit	Domain of authority	Horizontal linkages	Vertical linkages
University	 In charge of managing the campus grounds (including the garden sites) and initially with setting-up of the garden infrastructure in some cases (raised beds, sheds, etc) Some universities had staff supporting the student societies to varying extents 	• The National Union of Students' sustainability team was in contact with the students' unions and universities about spending funding, new funding opportunities, and annual renewal of the institutions membership to the Student Eats programme	• The relationships between university staff and participants in student-led food gardens were weak in all gardens, but in some cases the relationships were positive or neutral, but in one case they were fraught with tension and mistrust
National Union of Students sustainability team	 Supporting Student Eats initiatives at different universities all over the UK with information about food growing and running a food growing society Building relationships between gardens at different universities 	• The National Union of Students' sustainability team was in contact with students' unions and universities about spending funding, new funding opportunities, and annual renewal of the institutions membership to the Student Eats programme	• The relationships between the National Union of Students' sustainability team and the gardens were mainly through the gardens' leaders and were characterised by information sharing and the facilitation of relationship building between staff and students, as well as between Student Eats gardens

Table 24 (continued). Key findings about the promotion of polycentric governance in student-led food gardens.
and active. I tried to reach out again in the autumn and struggled to get a response. When I came back again, I was met by only one student who told me they were managing the garden alone. Students' transience, combined with periods of extended holidays, created this vulnerability. However, the gardens at Keele and Warwick each had the engagement a single, long-term mentor, which made these gardens considerably more resilient. The positive relationships between gardeners and other long-term stakeholders in two of the gardens contributed substantially to the gardens' resilience because there were not issues stemming from mistrust or conflict (e.g., reduced opportunities to self-determine, negative emotional responses).

For those actually participating in the gardens, the main issue related to managing connectivity that was struggled with was maintaining the balance between being too cliquey and not cohesive enough. Given the high turnover, this changed rapidly, making it difficult to make an overarching assessment of the extent to which it contributed to the vulnerability or resilience of the gardens. The remote location of the gardens on their university campuses, however, limited serendipitous and spontaneous engagement that came from stumbling across the gardens, reducing redundancy in participation.

Gardeners who had only recently started engaging in the gardens tended to be less tolerant to uncertainty (e.g., whether or not other gardeners would participate week-toweek), however it seemed that taking part enabled gardeners to develop their tolerance to uncertainty, possibly out of necessity. However, because of the high turnover of gardeners, few active gardeners actually demonstrated tolerance to uncertainty, with the exception of long-term mentors, and some student leaders. Furthermore, students' own transience made it extremely difficult for them to monitor slow variables and feedbacks, although these were crucial considerations for the gardens.

While there were good conditions to contribute to learning, including opportunities for learning and experimentation, and interaction with gardeners with diverse viewpoints, gardeners mainly viewed learning as a passive exercise which limited learning. Both instrumental and intrinsic forms of learning took place, however usually the instrumental learning, or the learning contributing to fulfilling the aims of the gardens, was not retained within the gardens due the students' transience. As such, the learning taking place in the gardens appeared to make limited contributions to the gardens' resilience, however the gardens provided space for participants to learn about how to "make sound choices in the face of uncertainty and complexity of the future" (Vare & Scott, 2008, p. 3). As such, the learning in the gardens appeared to do less for the resilience of the student-led gardens, but produced more resilient gardeners.

In summary, student-led food gardens are highly vulnerable to their own students' transience. However, there is evidence of misalignments with the resilience principles that could be realigned through actions taken by different stakeholders of student-led food gardens. Chapter 8 will delve into how, knowing the ways in which student-led food gardens are vulnerable, their resilience might be able to be improved. Before this, however, I will offer some reflections on the resilience principles.

7.3 Reflections on the resilience principles

The resilience principles (Biggs et al., 2012, 2015c) are relatively new, although they do draw on a range of previous scholarship about enhancing the resilience of social ecological systems (Anderies et al., 2006; Walker & Salt, 2006; Walker et al., 2006). Because of how new they are and the fact that I have attempted to apply them outside of their intended context (that is, to sustain ecosystem services), it is fitting to offer some reflections on how their application played out in practice. I have reflections on the phrasing of the first principle (maximising diversity), ontological humility in the application of the fourth principle (fostering complex adaptive systems thinking), making sense of the fifth principle (encourage learning) in the context of intrinsic and instrumental learning, and how literature about communities of practice could be used to develop the fifth principle. Finally, I conclude with an overall assessment of the appropriateness of applying the resilience principles beyond ecosystem services and offer some tentative suggestions of how the principles could be adjusted for other contexts. Before doing this, however, I offer some general reflections on the usability of the principles.

7.3.1 Usability of the resilience principles

The principles were challenging to work with given that they are so broad. I will use the first principle to demonstrate this. In section 7.2.1, I presented six axes of diversity that were relevant to managing participation in student-led food gardens. Variety, balance, disparity, and redundancy were considered for each of these. That means that there were eighteen aspects to consider in order to be exhaustive with this principle. This is only one of seven principles, making a qualitative and holistic assessment of resilience a monumental task.

To make this task less monumental, users of the resilience principles could make *a priori* decisions about which axes of the principles would be most strategic to consider. I did not collect data with the resilience principles in mind, and as such I did not make an effort to prioritise these through my data collection. For example, in the Wayfinder guide (Enfors-Kautsky et al., 2018) users are invited to reflect on the "aspects of diversity and redundancy [that] are most important in [the] system" and how to maintain these.

The difficultly of managing the complexity and scope of the analysis, however, reflects a more inherent issue with studying complex systems. It is difficult to maintain a balance

between accounting for and retaining the complexity of reality whilst creating tools or frameworks simple enough to understand and use. The resilience principles have made an important step in operationalising a relatively theoretical concept. The Wayfinder guide (Enfors-Kautsky et al., 2018) has made further considerable steps in making the resilience principles much more user-friendly through providing a step-by-step process to follow. However, having applied the principles to assess the vulnerability of student-led food gardens, I would suggest that the principles are still very challenging to apply meaningfully without considerable time, resources and engagement. Therefore, further research could experiment with the principles in different settings to explore if it is possible to identify axes of diversity, connectivity, and feedbacks that are common for particular systems to make the approach more widely accessible. For example, are there common axes of diversity that are relevant for maintaining resilience of organisations, and are there a different set of common axes of diversity that are relevant when maintaining the resilience of ecosystem services?

7.3.2 Maximising or managing diversity

An observation of crucial importance to the continued development of the resilience principles was that maximizing diversity did not always contribute to resilience in the studentled food gardens. High disparity in different viewpoints and motivations for taking part in the gardens affected connectivity in the gardens, causing interpersonal conflicts and making it difficult to meet everyone's needs/wants and accommodate different preferences. These axes of diversity negatively impacted participation, threatening the resilience of the gardens. However, it was clear that diversity contributed to resilience in other cases. For example, running a range of different types of activities meant that there were opportunities to cater to the preferences of a wider pool of existing or potential participants, thereby increasing resilience through increasing the redundancy in the number of participants involved. As such,

diversity had different effects on the resilience of the gardens depending on which axes were considered.

This finding is not an anomaly. For example, researchers of super-diverse populations have found that diversity can "undermine local integration and encourage insularity" at a neighbourhood level (Pemberton, 2017, p. 67). This insularity, or lack of connectivity, could result in 'brittleness,' and therefore reduce resilience, within neighbourhoods. Given that diversity can have paradoxical effects on resilience, I suggest that it would be preferable to phrase the principle as 'manage diversity and redundancy' rather than 'maximise diversity and redundancy.' This phrasing echoes phrasing used in the second principle, to 'manage connectivity.' When managing diversity, taking into account the period of time over which changes in diversity occur may also be important. In this study, the turnover of participants in the gardens was much faster than would be expected in other types of community-led gardens or sustainability organisations in general, leading to rapidly changing levels of diversity. This speed of change made it more challenging for to develop a common purpose and foster a sense of continuity. As such, temporality ought to be considered when applying the first resilience principle.

7.3.3 The need for ontological humility in fostering complex adaptive systems thinking

In the Habermasian tradition, shared understandings are crucial for meaningful, noninstrumental action. As such, having a 'shared mental model' as suggested by resilience (Biggs et al., 2012; Bohensky et al., 2015) and sustainability scholars (Broman & Robért, 2017) is crucially important. However, it is also worth issuing a word of caution about how these mental models ought to come about. For Habermas (1984), the construction of the lifeworld is fundamentally a democratic, co-constructive and participatory activity at its core. As such, I question whether a 'top down' approach to establishing shared 'mental models,' even the

complex adaptive systems model, is appropriate in a sustainability context in which democracy and participation are of paramount importance. Changing the way that "people, individually and collectively, think about and make sense of social-ecological system dynamics" (Bohensky et al., 2015, p. 145, italics in original) with a particular direction in mind (in this case, towards thinking though a complex adaptive systems lens) runs counter to democratic dialogic processes advocated by scholars like Habermas. The complex adaptive systems lens has been a particularly useful one for this study, however, like all theoretical perspectives, it likely is incomplete. As such, the application of this principle should be done with ontological and epistemological humility. Advocating a complex adaptive systems view of the world is ultimately an ontological question as it deals with the essence of what exists, the properties of what exists, and the relationships between things that exist. It is also an epistemological question as well, as complex adaptive system thinking means accepting that uncertainty must be tolerated and embraced as it is impossible to fully understand the world (Bohensky et al., 2015). Bohensky et al. (2015) advocate acknowledging epistemological pluralism in order to foster complex adaptive systems thinking, evidencing the recognition of a need for epistemological humility.

However, ontological humility should also be exercised when fostering complex adaptive systems thinking. Complex adaptive systems thinking has been useful to many in the sustainability community (some examples of which can be seen in Bohensky et al. (2015)), however, there are also critics of systems thinking, particularly for its application in social systems (Spronck & Compernolle, 1997). These critics suggest that power (Spronck & Compernolle, 1997) and culture (Arora-Jonsson, 2016) are not adequately considered in system thinking. Furthermore, there are difficulties defining system boundaries and making sense of "equilibria, thresholds, and feedback mechanisms" in the face of competing ontologies (Olsson et al., 2015, p. 4). These views are important to engage with with humility

in the movement to a sustainable, resilient future because they may provide important insights into the limitations and shadow-side of complex adaptive systems thinking. Only with ontological humility can these insights be harvested. Bohensky et al. (2015, p. 157-158) did caution that "attempts to foster complex adaptive systems thinking can compromise resilience when a complex adaptive systems model is deliberately or inadvertently treated as static" or "intended to assist transition to a new management paradigm, which is seen as the end point." Similarly, Cilliers et al. (2013) advocate that 'rules' for dealing with a problem should be acknowledged as provisional and should be continuously revised (as in the fifth resilience principle about learning and experimentation). However, ontological humility has not only to do with how the 'rules' of a system are modelled according to complex adaptive system thinking, but also has to do with the very thinking behind complex adaptive systems, like the assumptions of non-linearity, interconnectedness, and uncertainty in social-ecological systems. In order to be truly reflexive, we must be open to alternative worldviews, including those that do not adopt a complex adaptive systems understanding of the world.

However, ontological humility is also distinct from ontological pluralism (see Table 25). Ontological humility means acknowledging that one's understandings about the world and how it functions could be incomplete or even wrong. Ontological pluralism is the understanding that there are different 'worlds' and ways these 'worlds' function. In the case of addressing sustainability problems, I do not advocate ontological pluralism because it may lead to the bewilderment and paralysis (Cilliers et al., 2013), and therefore inaction. However, ontological humility can be embodied more pragmatically. For example, if one acknowledges that there may be limitations to complex adaptive systems thinking, they can still provisionally use and advocate for it with the acceptance that it could be flawed. However, ontological pluralism is more immobilizing. If one assumes that there are parallel worlds, one which might function according to complex adaptive systems thinking, and one which does not, it is

impossible to make shared decisions about how to act because one cannot know with any

degree of certainty how a system will respond to a given action.

Table 25. A matrix illustrating the differences between ontological and epistemological humility and
ontological and epistemological pluralism.

	Humility Acknowledging that one's understandings could be incomplete or even wrong	Pluralism <i>Coexistence of two or more worlds</i> <i>or ways of knowing</i>
Ontological The nature of the world and how it functions	Acknowledging that one's understandings about the world and how it functions could be incomplete or even wrong	Understanding that there are different 'worlds' and ways these 'worlds' function
Epistemological How we can know the world and how it functions	Acknowledging that one's understandings about how we can know the world and how it functions could be incomplete or even wrong	Understanding that there are different ways of knowing the world and how it functions

The suggestion to have humility with the application with applying the fourth resilience principle (to foster complex adaptive systems thinking) may seem to be a counterintuitive reflection given that in this thesis, I have aligned my work so closely to this particular way of working and thinking. However, as much as systems thinking feels second nature to me, I am open-minded to the idea I might later find it to be incomplete or inadequate. Though, at this moment, I believe this principle was contextually appropriate, because much of the sustainability thinking in university settings in Western European contexts, particularly amongst young, university-educated people, is rooted in complex adaptive system thinking.

7.3.4 Learning, resilience, and communities of practice

7.3.4.1 Resilient learners or learning for resilience?

Biggs et al. (2015b, p. 186) say that "experience has shown that learning can enhance the resilience [...] primarily through its influence on governance and decision-making processes." Based on observations during this study, I would suggest that learning can indeed enhance governance and decision-making processes within student-led food gardens, but crucially, it can also provide gardeners with transferable knowledge and skills that they can take with them beyond the gardening context. This learning can prepare the gardeners to be more effective pro-sustainability change agents in their careers and/or personal lives, not necessarily because they understand more about (un)sustainability, but because they have developed tacit understandings of how social systems function and change happens. This finding echoes a call from Sterling (2010, p. 511) who called for an integrative paradigm for sustainability education "that reconciles instrumental and intrinsic educational traditions, informed and infused by resilience theory and social learning." Biggs et al.'s (2012, 2015c) principle about learning and experimentation encourages learning for instrumental reasons. However, emphasising learning that creates resilient learners (or intrinsic learning) can potentially have important benefits for the gardens, which, perhaps more importantly, can contribute to the sustainability of our society at large.

7.4.4.2 Using communities of practice to enhance learning for resilience

Another more pragmatic reflection, given that learning is already theorised through a social learning lens in the resilience principles, is that it may be useful for resilience scholars and practitioners focusing on learning to promote resilience to refer to Wenger et al.'s (2002) work on how to cultivate communities of practice. Communities of practice are "groups of people who share a concern, set of problems, or a passion about a topic, and who deepen their knowledge and expertise in this area, by interacting on an ongoing basis" (Wenger et al., 2002, p. 4). There are some synergies between Wenger et al.'s (2002) suggestions for cultivating communities of practice and the resilience principles. For example, Wenger et al.'s (2002) first suggestion for cultivating a community of practice is to design for evolution. In other words, flexibility and adaptability are important for creating communities of practice, like they are when fostering resilience in a system. However, Wenger at al.'s (2002) suggestion for cultivating communities of practice could also build on some of the resilience

principles. For example, the sixth resilience principle is to broaden participation in general. However, Wenger et al. (2002) suggests that different levels of participation need to be encouraged to enable 'peripheral participation' (Lave & Wenger, 1991). Peripheral participation can be a pathway for less active participants to become more active in the community of practice and can also provide alternative perspectives which are conducive to learning. As such, it may be advantageous to both broaden participation as well as create opportunities for different levels of participation. Furthermore, a number of Wenger et al.'s (2002, p. 51) suggestions were not reflected at all in the recommendations from the learning resilience principle, such as "create a rhythm for the community" to foster a sense of familiarity, and "combine familiarity and excitement" to create a safe environment for candid conversation while keeping members engaged and interested. As such, Wenger et al.'s (2002) work is highly compatible with the resilience principles in general and, in particular, could enrich the fifth resilience principle.

7.3.5 Broadening participation

The principle of broadening participation focuses on "the active engagement of relevant stakeholders in the management and governance process" (Stringer et al., 2006; Leitch et al., 2015, p. 203). However, the assumed governance conditions in the resilience principles are very different from the governance conditions of student-led food gardens by virtue of the fact that the principles are written for sustaining ecosystem services. Leitch et al.'s (2015) assumption appeared to be that (1) there is interest in participation, limited mainly by powerful stakeholders, and (2) there are large (or at least larger) numbers of stakeholders than can be expected in student-led food gardens (as illustrated by the example of urban water planning in Indonesia which involved over 500 people (Leitch et al., 2015)). Furthermore, because the focus of the principles is on ecosystem services, the authors of the

resilience principles (Biggs et al., 2012; Leitch et al., 2015) drew on the literature about adaptive management, adaptive governance, and adaptive co-governance which are not (as) relevant in other contexts, such as student-led food gardens. Based on this, it would be useful to see this principle applied in different contexts and to draw on literature about participation in governance on a smaller scale (e.g., Rosol, 2010; Silver et al., 2010; Yap, 2019)

7.3.6 Applying the resilience principles beyond ecosystem services

Another important question to ask is who the resilience principles are written for. They are all written in the imperative, but when applied to a self-organising group of people without a leader, it is unclear who the message is directed to. For example, should students' union staff be ensuring complex adaptive systems thinking is fostered in students? Or should students support other students in fostering complex adaptive systems thinking? Furthermore, how does one ensure the application of the resilience principles doesn't become patronizing to other stakeholders?

This 'who' question really came to light because the resilience principles were stratified into two levels: key social ecological system properties to be managed and key attributes of the governance system (Biggs et al., 2015c; see Figure 29). Another way to think of it would be that the first three principles focus on an 'other' that needs to be *governed*, and the remaining four represent an *I* or a *we* that *governs* the 'other'. For example, Kotschy et al. (2015) explained how the first principle (to maintain diversity and redundancy) could be used to enhance the resilience of the system that provides crop pollination (an ecosystem service). They said:

"The dangers of relying too heavily on a single species to provide this important ecosystem service [pollination] have been brought into sharp focus recently in the United States, where honeybee populations have undergone sharp declines due to a

number of interacting factors [...] The decline in honeybees has led to declines in crop production, affecting food security and the livelihoods of farmers [...] Scientists in California have sought to increase the resilience of pollination services by enhancing both the redundancy and diversity associated with pollination [...] Encouraging pollination of crops by native bees, and increasing the abundances of these species on farms, adds redundancy to the ecosystem service of crop pollination by increasing the number of species that contribute to this service [...] Farmers, landowners and scientists in California are working together to enhance access of wild pollinators to crops..."

(Kotschy et al., 2015, p. 68-70)

In this particular example, "farmers, landowners, and scientists" (Kotschy et al., 2015, p. 70), were *governers* who were managing the diversity and redundancy of the pollinators (the *governed*). However, in this study, the participants in student-led food gardens were both the



Figure 29. The resilience principles "grouped into those that relate to general SES properties to be managed (P1-P3) and those that relate to key properties of the SES governance system (P4-P7)" (Biggs et al., 2012; Biggs et al., 2015c, p. 25).

governers *and* the governed. The participants were in charge of the governance of the garden, but the impacts of their own transience was what they were managing. As such, in this context, the the sixth (broadening participation) and seventh (polycentric governance) principles were more like guidance for application than they were principles in their own right.

The student-led food gardens were systems with both social (the student society) and ecological (the garden) components. However, the main perturbation that threatened the student-led food gardens' resilience was a social one, that is, the students' own transience and the resulting impacts. As such, what needed 'governing' was actually 'those who were to govern'. With this shift in emphasis towards the social system, the stratification into 'system properties' and a 'governance system' becomes blurred. This is because the governance principles about broadening participation and polycentricity deepen the analysis of the other principles (especially maintaining diversity/redundancy and managing connectivity) (Schlüter et al., 2015). In other words, when such a strong focus was placed on the social component of the system (as in this study), stratifying the system properties and the governance system did not make sense because the governance system is an inextricable part of social systems. In practice, this resulted in excessive effort stratifying the findings into the principles without enhancing the quality of the analysis.

For potential users that want to apply the resilience principles outside the context of sustaining ecosystem services, the first three principles hold up well. The first two are also clarified by the key considerations from Biggs et al.'s (2015c) work for: diversity and redundancy, variety, balance, and disparity should be considered (Kotschy et al., 2015), and for connectivity, the presence/absence, distribution, intensity, strength, modularity, and nestedness of connections should be considered (Dakos et al., 2015). The principles about

participation and polycentricity can also be used to focus attention in the application of the first two principles towards governance.

However, I suggest that the principles about fostering complex adaptive systems thinking (principle 4) and encouraging learning and experimentation (principle 5) should be merged. This would allow for more humility in the application of the fourth principle. The field of complex adaptive systems thinking is not static, it is constantly evolving. As such, learning from practice and the application of these principles could provide more nuance in how we understand complex adaptive systems to operate. Also, I would suggest that learning and experimentation should be encouraged not only for improved governance, but also to create 'resilient learners' (Sterling, 2010). In other words, learning and experimentation should be encouraged with an end goal in mind (instrumentally motivated), as well as being encouraged without a particular aim or sense of direction to help uncover 'unknown unknowns' and create resilient (intrinsically motivated) learners.

To summarise, in the above sections, I found that:

- the application of the resilience principles was somewhat unwieldy,
- when studying a social ecological system where the focus is on the 'social' component the separation between the system properties and their governance system collapses,
- too much diversity can negatively affect connectivity and learning, thereby reducing resilience,
- the speed of change in diversity can affect the extent to which it increases or decreases resilience,
- there is a need for more ontological and epistemological humility with regards to the recommendation to use a complex adaptive systems lens, and
- both extrinsic and intrinsic orientations towards learning should be embedded in the fifth principle.

Based on these findings, I suggest that the resilience principles could be amended in order to: simplify their use by reducing their number and explicitly outlining the axes that could be considered within each principle; enhance nuance in the wording; offer guidance with more ontological humility; and include an intrinsic perspective on learning. I therefore propose that the resilience principles, when used for sustaining social systems or constructs, could read as the following:

- 1. Manage diversity and redundancy
 - ...with respect to variety, balance, and disparity...
 - a. ...including in participation in governance
 - b. ...including through polycentric governance
- 2. Manage connectivity

...with respect to presence/absence, distribution, intensity, strength, modularity, and nestedness of connections...

- a. ... including in participation in governance
- b. ...including through polycentric governance
- 3. Manage slow variables and feedbacks
- 4. Encourage learning and experimentation

...with respect to the system and its governance, complex adaptive systems, and

unknown unknowns

These suggestions are, of course, just that: suggestions. The resilience principles have been developed in collaboration with many experienced resilience scholars who have worked at the interface of research and practice for years (Biggs et al., 2012, 2015c). These proposed amendments come from a single doctoral study from a lone scholar, and as such they represent limited experience within a limited scope. On the other hand, however, this study is the first to go beyond applying these principles to sustaining ecosystem services, to the best of my knowledge. I have applied these principles to assess the resilience of social systems (student-led food garden groups) that interface with ecological systems (the physical gardens themselves). The use of the principles in this way has demonstrated their applicability beyond their original intended context. However, it has also demonstrated their limitations and incompatibility within contexts where the social component is in focus. This is therefore why I suggest amending the principles as above for their application to social-ecological systems where the emphasis is on the social component of the system. That is, when the object of governance and those governing are one and the same, my proposed framing of the principles may be of better use.

There is substantial examination of the relevance of the resilience principles in ecosystem services governance in Biggs et al.'s (2015c) book, but if the principles are to be useful elsewhere, there also needs to be further examination of the resilience principles in other such contexts. For example, there could be further examination of the applicability of the principles to the resilience of sustaining social systems and constructs, such as governance regimes, policy landscapes, shared mental maps, or women's rights that interface with ecological systems.

Another potential contribution to future research could be considering how different stakeholders could engage with the principles. The principles feel as though they are written towards stakeholders with power and/or resources who have the capacity to make largerscale interventions. However, there are likely ways that less powerful or well-resourced stakeholders could make contributions to improving resilience of social-ecological systems. Future research could consider what sorts of interventions or strategies for improving resilience could be taken by different stakeholders with difference capacities.

7.4 Conclusion

The student-led food gardens were highly vulnerable to their own transience, as this chapter has demonstrated. This vulnerability was made particularly acute in the case where there were few or no long-term participants, like doctoral students (lack variety in and imbalance between different participants' length of participation), and where relationships with long-term stakeholders were riddled with mistrust and conflict (poor quality connections). A cohesive gardening group was also important for resilience, but when the cohesiveness became cliquey, the garden groups became brittle. Gardens lacking connectivity with the rest of the campus also lacked more incidental engagement.

I also found that those who had only recently started engaging in the gardens tended to be less tolerant to uncertainty, however it seemed that taking part enabled participants to develop their tolerance. However, students' transience limited monitoring of crucial slow variables and feedbacks, such as how an overemphasis on recruitment caused forms of problematic participation, and how the quality of the relationship between long-term and short-term stakeholders and the quality of participation was mutually reinforcing. I found that the learning that took place in the gardens made limited contributions to the gardens' resilience, however, it may have enabled the learners themselves to become more resilient.

Having applied the resilience principles to sustaining student-led food gardens (and not ecosystem services), I also offered some reflections about their transferability to other contexts. The principles were useful to assess the vulnerability of student-led food gardens, however I found it challenging to assess all the principles in full because there were so many factors to consider. Furthermore, I also found it difficult to disaggregate the properties of student-led food gardens as systems and their governance given the focus of the study was weighted more heavily on the social component of the gardens. It also became clear that too

much diversity can negatively affect connectivity and learning, thereby reducing resilience. The speed of change of diversity can also impact connectivity. Finally, I suggested that there is a need for more ontological and epistemological humility with regards to the recommendation to use a complex adaptive systems lens, and that both extrinsic and intrinsic orientations towards learning should be encouraged in the resilience principles. The next part of this thesis will be future-orientated, with the next chapter outlining suggestions for how to address the impacts of students' transience and enhance the resilience of student-led food gardens, drawing on the analysis from this chapter and Chapter 6.

Part 3: The future

Addressing the impacts of transience and building resilience

Chapter 8



Students holding kale, spinach and beet tops June 2013

8.1 Introduction

This chapter is the first of the future-orientated part of this thesis. It focuses on how to manage the challenges associated with transience and participation in, and building the resilience of student-led food gardens. Out of the analysis in Chapters 6 and 7 some strategies to dealing with these challenges emerged. I will first present these. Then I will go on to share the strategies and actions suggested and taken by the research participants. Those taken are described and evaluated towards the end of sections 8.4.2, 8.4.5, 8.4.6, 8.4.7, and 8.4.9. A more detailed reflections on the challenges associated planning, taking, and evaluating action will be found in Chapter 9. Towards the end of the chapter, I will also provide advice to stakeholders of student-led food gardens as a practical offering.

In previous chapters, the question of *who* has not been as central as it is in this chapter. This is because the focus has been on the challenges related to students' transience and participation as they have manifested within student-led food gardens. The question of *who* in these cases is less relevant because it is more productive to identify the behaviour or circumstances that has caused a problem than the person or people who have. However, this chapter is solutions-orientated, and different stakeholders in student-led food gardens have different capacities and capabilities to act on the problems outlined in Chapter 6 and 7. As such, when making recommendations in this chapter, I will be using active voice as much as possible to indicate what sorts of actions are best suited to which stakeholder.

8.2 Strategies rooted in analysis from Chapters 6

In Chapter 6, I suggested that both upstream (focusing on the causes of problematic participation) and downstream (focusing on the effects of problematic participation) strategies should be considered. This is because there were feedbacks at play between problematic participation, their causes and effects. As in systems with feedbacks, what is seen as 'upstream' can be easily turned on its head and considered to be downstream depending on how the system is mapped. I also said that the causal loop diagram could be used to help map which parts of the system solutions address. However, I also have another suggestion that could make a substantial impact on managing the impacts of transience, which is to consider the role of university and students' union staff.

8.2.1 The role of university and students' union staff

Because the focus of this study was on participation in student-led food gardens with transient participants, I studied Student Eats' most student-led food gardens as exemplars of transience. But should student food gardens on university campuses actually be led by students? This is a question I was forced to ask myself, in spite of the framing of my thesis. While many students wanted to have control over and autonomy in the decisions made about the space their garden occupied, there were also some students that expressed that they would have liked more guidance from university or students' union staff. This echoes Brewis & Holdsworth's (2011, p., 174) findings that students appreciate support in their volunteering activities, and that "students who are supported by their university to volunteer report better experiences of volunteering than volunteers who are non-supported." Indeed, in Bakker et al.'s (2011, p. 502) study of project knowledge transfer in temporary organisations, they asserted that "there is a clear and unambiguous responsibility of the project owner (the permanent parent organisation) in project knowledge transfer." The responsibility for knowledge transfer was taken on by doctoral students in two of the gardens studied. However, the fact that many students felt that their knowledge/skills were inadequate, and they lacked adequate mentorship indicates that 'organisational amnesia' was still a problem.

This was likely compounded by students' absence over the summer, which meant that they missed out on learning about an important part of the growing calendar.

While many students wanted to self-determine, self-determination does not necessarily preclude support and guidance from university or students' union staff if the support and guidance is on students' terms. However, ensuring staff input into the space is supportive without being overbearing is a fine line to walk. Experienced or skilled staff may be able to manage this intuitively, however others might need clearer models to offer the right level of support. One model of how staff could support in student-led food gardens is to create a 'hybrid' management structure by splitting the garden into two different parts: one with more staff support and structured volunteering opportunities, and another with less staff involvement and a higher degree of student self-determination. Another model could involve staff at key periods, temporally. For example, staff could limit involvement during term-time, and offer hands-on support during holidays and recruitment at the beginning of term.

Having more staff involvement in the space, assuming it does not infringe on students' creativity or self-determination, could provide other positive benefits beyond catering to students' different management preferences. For example, regular interaction between long-term and short-term stakeholders could help staff and students see each other as human beings rather than sources of frustration, developing understanding and trust. Also, staff members being on the garden site more regularly means they would be more likely to notice when the gardens need outside support to restore student engagement or to help manage the space itself. As such, maximising the extent to which the gardens are student-led is likely not desirable. Long-term stakeholders of student-led food gardens could, therefore, have a subtle but crucial role in mitigating the causes and effects of problematic participation, whilst navigating more stable and long-lasting paths for these gardens.

8.3 Strategies rooted in the analysis from Chapter 7

In Chapter 7, I assessed the vulnerability of student-led food gardens. Rooted in this analysis and drawing on my adaptation of Biggs et al.'s (2012, 2015c) resilience principles, there are some natural conclusions that can be drawn. With regards to diversity and redundancy, supporters of student-led food gardens should strive for an appropriate balance between staff and student participation to maximise the benefits to students, while providing the consistency and support from longer-term stakeholders. Ideally, opportunities should be created for both structured volunteering and more free-form ways of taking part. There should also be a variety activities that cater to different people's interests, including both social activities and gardening activities. It is also important that long-term mentors and student leaders are not only saddled with administrative work, but get to take part in the more 'fun' parts, like the actual gardening activities.

In terms of decision-making about the gardens, stakeholders should be involved for the decisions on matters that pertain to them, as advocated for polycentric governance models (Schlüter et al., 2015). As such, students should be involved or responsible for making decisions that pertain to them and the garden as much as possible. University staff, particularly grounds staff who are not usually expected to operate in a student-facing role, should have time specifically allocated to this task. Support for student-led food gardens should come from a diversity of actors. In particular, staff from the National Union of Students can play a facilitative role in supporting student-staff conversations by regularly checking in. University staff working with the university campus operations have a challenging role to play, which is to support gardeners to manage the physical garden space while maintaining the autonomy of the gardeners as much as possible.

One of the most rapidly changing factors in student-led food gardens is the connectivity in the social relationships between gardeners. Supporters of and participants in student-led food gardens need to ensure the gardening group is lively and cohesive, but also open to new recruits in order to prevent becoming clique-y. The connections between the university/students' union staff and student gardeners needs to be characterised by trustful and mutual relationships to mitigate frustration and conflict. University and students' union stakeholders, in particular, need to take on a facilitative and supportive role in governance, particularly to do with retaining knowledge within the garden. Power asymmetries may not be possible to break down, but university and students' union staff should be attentive to these asymmetries to develop strategies to compensate for them. The relationships between student-led food gardens and students' unions should also be trustful, facilitative, supportive and focus on retaining knowledge within the garden. Long-term mentors and student leaders should engage less-active participants in decision-making, and act as a bridge between the less active gardeners and university, students' union, and National Union of Students staff.

The National Union of Students' sustainability team has a special and vital role to play through working with and between the university, students' union, and student-led food gardens. The staff at the National Union of Students did and ought to continue to play a facilitative and mediating role between the university, students' union, and student-led food garden, to strengthen and improve the quality of the relationships of the different stakeholders. The National Union of Students also can and does connect student-led food gardens from different universities to inspire, spread information, and create a sense of being part of something bigger.

Connectivity between the gardens and the rest of the campus should also be maintained. Although it would be inadvisable to move gardens once they have already been established, new gardens that are being built should be placed in a space that is accessible to

participants, both in terms of being in a location that people pass through regularly and in terms of not being behind lock and key.

Slow variables and feedbacks that I identified as being in need of monitoring were students' indifference/apathy to food growing and sustainability, the quality of the relationships between long-term and short-term stakeholders, the physical state of the garden, and the balance between an emphasis on recruitment compared to the 'core business' of the garden. Long-term stakeholders of student-led food gardens are in the best position to monitor these over time because of students' transience. As such, long-term stakeholders should remain attentive to changes in these variables and feedbacks, an intervene early when changes for the worse are observed to prevent amplification of their decline.

In terms of learning and fostering complex adaptive systems thinking, I suggest that all stakeholders involved would benefit from tolerance to ambiguity and uncertainty, and therefore learning that builds tolerance and uncertainty should be fostered. Based on initial observations from this study, it seems that a tolerance to ambiguity and uncertainty can be developed over time in student-led food gardens. As such, an important strategy to build up this tolerance, at least for students, is to keep coming back and taking part.

I also suggest the approach that the gardens take to knowledge should emphasise retention as well as generation, as the knowledge retention in the student-led food gardens was weak. Gardeners should try to think of learning as a process rather than an outcome in order to experiment and engage more actively in learning, and reduce their perceived reliance on outside 'expertise.' Other stakeholders should support gardeners in adopting this process-orientated view of education. University, students' union, and National Union of Students' staff should provide reassurance to gardeners to help alleviate lack of confidence and discomfort with ambiguity and uncertainty. Furthermore, gardeners should remain open

to encountering people with different opinions and viewpoints, and because having a diversity of viewpoints within the gardens can create 'boundary interactions' that can foster learning (Bendt et al., 2013). This is unpacked further in section 8.6. University, students' union, and National Union of Students staff should encourage and facilitate the inclusion of such diversity of viewpoints.

8.4 Strategies suggested by participants

In this section, the strategies presented are ones identified by the research participants. This was an action research study, so some of these actions were actually undertaken by gardeners (see Table 26). I have clustered the suggestions for how to address the impacts of students' transience into fourteen strategies. These strategies are not mutually exclusive as they often served multiple purposes or supported other strategies. These strategies are to:

- Maintain an appropriate and well-managed space
- Run events and activities
- Get support
- Nurture interest and creativity
- Increase visibility
- Create organisational infrastructure
- Recruit
- Document and plan
- Embed within wider organisational infrastructure
- Maintain healthy relationships & social networks
- Connect

- Create vibrancy and focus on value
- Appeal to extrinsic motivation
- Secure funding/resourcing

In Chapter 6, I outlined the barriers to participation and problems caused by problematic participation. These strategies presented above are intended to address these barriers and problems. As such, Table 26 provides an overview of the suggestions to address the impacts of transience, and which problems from the causal loop diagram in Figure 26 they are thought to address. The findings presented here are based on the transcripts of interviews and workshops, and my research diary and analytical memos. I have not taken additional interpretive licence in this table. This means that there may be some relationships between suggested solutions and problems that could be drawn but I have chosen not to do so in order to present the data as was communicated by participants and as I recorded it. I have done this in order to make a clear distinction between the information that comes from the data and the further analysis I will lay over it. There are two exceptions to this, however, because their relationship was so straightforward. First, creating organisational infrastructure addressed problems associated with the extent of organisational infrastructure, which will be unpacked further in section 8.3.6. Second, logically, securing funding and resourcing deals with a lack of funding. It should be noted that there were no suggestions to address students' transience directly as that was recognised to be a natural phenomenon that should not be changed. I will now explain each of the themes in turn.

8.4.1 Have an appropriate and well-managed space

It was widely thought that having an appropriate and well-managed space was important for addressing issues associated with transience and problematic participation. Having an appropriate and well-managed space meant that it was located in an appropriate

Table 26. The suggested strategies to address the impacts of transience mapped on to the problems identified in Chapter 6. Actions that were actually taken by research participants are in bold. See Table 27 for the legend.

27 for the lege	end.					Stra	ateg	gies	sug	gest	ted				
		Maintain an appropriate and well-managed space	Run events and activities	Get support	Nurture interest and creativity	ncrease visibility	Create organisational infrastructure	Recruit	Document and plan	Embed within wider organisational infrastructure	Maintain healthy relationships & social networks	Connect	Create vibrancy and focus on value	Appeal to extrinsic motivation	Secure funding/resourcing
	Problem addressed	Σ	Ŗ	Ō	ž	2	ΰ	ž	ă	ш	Σ	ŭ	Ū	Ę	Se
Getting	Cultural barrier														
through the	Busy, lack of time, competing commitments														
garden gate	Lack of external support														
	Expectations & misconceptions ¹⁷														
	Lack of confidence														
	Do not see the benefits														
	Cost														
People haven't heard of it															
	Physical Space														
Barriers to	Unwilling to commit														
coming back	Students experimenting														
	People forgetting to come														
	Poor weather and seasonal issues														
	Extent of organisational infrastructure														
	Ownership issues														
	Lack of opportunity to self-determine														
	Cliqueyness and cohesion														
	Conflict & internal politics														
	Lack of responsibility														
	Lack of initiative														
	Lack of motivation														
Problematic	Low participation														
participation	Short-term participation														
	Irregular participation														

¹⁷ Expectations and misconceptions were thought to be barriers both to getting through the garden gate and to coming back. For the sake of simplicity, this problem was included only In the former category.

Table 26 (continued). The suggested strategies to address the impacts of transience mapped on to the problems identified in Chapter 6. Actions that were actually taken by research participants are in bold. Strategies suggested



Table 27. Legend for Table 26. The colours represent the participant(s) that said which strategy addressed which problem.

Student(s)	University, students' union, and/or National Union of Students' staff	Myself (research diary/analytical memos)
X		
	X	
X	Х	
		Х
	Х	Х
X	Х	Х

location, well-maintained, and aesthetically appealing. Some participants appreciated the fact

that their garden was secluded because it meant it was a space where they could get away

from it all and would encourage them to keep coming back. Others, however felt that a more

central space would be more accessible, and easier to stumble across and therefore a

physical reminder to attend sessions. This was thought to contribute to reducing low and irregular participation. In addition to being accessible in terms of being in a central location, it was also important that the space to not have other barriers to entry, such as locked gates or only being open during working hours.

Physical infrastructure was also thought to be important to prevent the gardens from becoming overgrown during periods of low participation. This meant gardeners had less of a hurdle to overcome when the garden was kick-started again. This helped prevent gardeners from becoming overburdened, overwhelmed, and just burned out, and the resulting negative emotional responses. As such, creating appropriate physical infrastructure meant that the space could be left unused for a year or two and still bounce back. Raised beds were thought to keep the space looking nice even if they became overgrown, and some gardens used fabric ground covers to reduce the need to weed. Not all gardens had sheltered space (such as greenhouses, sheds, or polytunnels), but it was thought that these were important so that participants would develop a *good memory association* (Student leader) even on days with poor weather. Furthermore, appropriate spaces for social events, such as a seating area, was important for improving social interaction.

Growing appropriate crops, like early/late cropping and low maintenance plants, was also thought to be important so that gardeners would see the benefits of their efforts. In this way, the bulk of the food produced is ready to eat before students leave for the summer, or after they come back, and there would be a reduced need to water in summer months.

It was important participants had licence to choose how to use the space to build their sense of ownership and responsibility towards the space. It was also important there was enough space for all people who attend to have tasks so they feel their work has a purpose. Indeed, having a physical space was thought to be an important feature when comparing to

other student-led initiatives because it meant that the space actually provided a sense of continuity by just existing:

...having a physical place gives [the garden] continuity, like, by itself. Just by being there. (Student leader)

This will be revisited in section 8.6.

8.4.2 Run events and activities

Another strategy which was widely suggested was to run different types of events and activities. Events and activities were thought to break down the 'cultural barrier' (see section 6.2.1) by bringing new people into the garden and raising awareness of the initiative, while increasing participation and social interaction, though some events/activities were more successful at this than others.

It was crucial to have regular sessions on a set day and time on a weekly basis throughout the year. This helped to remind people to attend and was thought to keep participation going throughout the winter in spite of poor weather. Participants also said that there should be varied activities and events, which might include social events, visits to other gardens, cookery classes, 'taster' sessions at the beginning of term, celebrations (like a harvest festival), and formal courses. Events co-run with other societies brought new people into the gardens. Socials were important for helping people see the benefits of taking part, and to foster a sense of social cohesion, while visits to other gardens and 'taster' sessions were thought to generate interest and inspire. Formal courses led by experts and skill workshops were thought to increase the overall knowledge/skill level in the gardens, while applied skills workshops were also thought to contribute to a sense of ownership and contribute to garden maintenance.

8.4.3 Get support

Research participants thought that gardeners should seek out support, and that university and students' unions' staff, National Union of Students' staff, peers, and more experienced and knowledgeable gardeners should provide it to help alleviate the burden of responsibility from falling too heavily on a small group of individuals. Appropriate support was thought to build confidence in people being supported, increase knowledge and skills within the garden group, and improve the maintenance of the space. Such long-term stakeholders had better insight into the challenges that were caused by students' transience, and also have longer-term views on the gardens. As such, they were in a good position to reassure students when they faced challenges they did not yet understand or experienced frustration and disappointment.

Support from students' union and university staff was thought to be valuable because they had complementary knowledge, skillsets, and experience to student gardeners. As such, the support they offered would be unlikely to be replaced through involvement of more student gardeners. It was noted that support from students' unions and universities might look quite different. Students' unions are set up to advocate on behalf of students and support student societies and student life and therefore could offer support with running their student society as outlined in in section 7.2.7, and support knowledge handover. Their support could help to counter the 'cultural barrier,' and raise awareness about the gardens through, for example, the gardens' participation in 'freshers' fairs' run by students' unions. It was also thought that students' unions could help prevent cliquey-ness and increase social interaction within the society through reaching out to potential participants beyond the friend groups that might dominate the gardens.
University staff could offer different support. Academic staff could help gardeners connect the garden to the formal curriculum by embedding visits into their classes or designing assignments or activities that require visiting the garden (an example of which can be found in section 8.4.9). Grounds staff working with the university campus grounds or estate could support students with maintaining and developing the physical space. In some cases, grounds staff already did provide this support, through mowing the lawn around the garden, providing compost and installing water taps. If staff are experienced gardeners or interested in gardening, they could also support the gardens sharing their knowledge/skills or maintaining the garden over the summer. However, it was noted that not all support was good support. A staff member from the National Union of Students said that some staff support could be overbearing and put students off. Appropriate staff support would therefore be intervening only as needed, such as in moments when participation dropped off and needed a 'kick-start'. A kick-start, according to the staff member from the National Union of Students, would involve publicising the garden in order to recruit new members. Students also expressed hesitancy with getting more support from the university or the students' union:

I think it would be sad if it became more under the university's control. Um. So I'm hesitant to get more involved in the [students' union], like a lot of [other Student Eats gardens] are.

Furthermore, gardeners said that the support from university staff was not always timely in terms of responding to emails or requests for support. This echoes the analysis from Chapter 6, which suggests that support from external stakeholder should be done in a way that retains gardeners' self-determinacy.

The National Union of Students' sustainability team felt that they were important support for the gardens by drawing university and students' union staff's attention to the

garden groups periodically through checking in via email, on-site visits, and at semi-regular conferences. This external prompt encouraged students' union or university staff to 'kickstart' gardens that were struggling with low participation or at risk of folding. Receiving support from the National Union of Students was also thought to make gardeners feel as though they were part of something bigger, and reduce a sense of purposelessness caused by the constant need to recruit. Support from peers was also thought to be important for developing a sense of ownership in supporting peers and reducing isolation, decreasing negative emotional responses.

8.4.4 Nurture interest and creativity

It was thought to be important to nurture interest and creativity to break down the 'cultural barrier' (section 6.2.1) and build confidence by meeting people where they are at. Nurturing interest and creativity were also thought to foster a sense of ownership and motivation through creating opportunities for gardeners to self-determine. There were a variety of different ways to do this, one of which was to provide high-quality leadership and facilitation. This might mean providing mentoring to new recruits and reassurance that they are doing a good job. It also means not overburdening new recruits, while at the same time providing them with encouragement to take on a bigger role and to come back again. One participant said it was important for people in a leadership or facilitative role not to tell people what to do, and instead, to let them ask. In-person communication was thought to be important to nurture interest and creativity, indicating that it is important for people in facilitative or leadership roles to have well-developed soft skills. A long-term mentor reflected on their facilitative role in the following passage:

But ultimately, that, it's my responsibility to kind of fashion that dynamic [where people can engage in different ways]. I'm not saying I'm successful in it. But that's

what I see myself doing a lot of the time. I think. What kind of atmosphere that

pertains here?

What can be seen in this passage is a long-term mentor talking about the flexibility required in their facilitative role, where they need to pay close attention and sense into the needs of the group every week in order to try to create an atmosphere where people can engage in ways that suit them.

Other suggestions for how to nurture interest and creativity were to seek out participant input. This might include arranging opportunities for people to discuss what they want to do or finding out what participants want for the garden and from taking part. Participants also said that it was important for gardeners to be allowed to make changes, be creative, and experiment with the garden space. One gardener said:

...what about the... the general dynamic of the university being very, very careful, and very, very cautious with what they do? And what they allow? Maybe? And students being more progressive and wanting to do more edgy things... or more – just try. Because it's a playground, isn't it? A university. You try out things...

As can be seen in this passage, the risk adverse nature of the university was thought to stifle students' creativity and self-determination. This is an important tension for university and students' union stakeholders to be cognisant of.

8.4.5 Increase visibility

Increasing the visibility of the space was thought to be a key way to address the impacts of transience. One way to do this was advertising through the student newspaper, social media, leafletting, emailing, blogging, and creating videos, for example. Mostly, advertising was used to make people aware of the gardens, but social media was also used to remind people to come to gardening sessions and to create a sense of vibrancy in the society

through a lively online community (which will be elaborated on below in section 8.3.12). Participants also said that, to be successful in recruiting new participants, the messaging used in advertising should communicate the benefits of taking part and be tailored to the audience. Including gardening tips in advertising and on social media was thought to increase potential recruits' confidence to take part. Having more people engaging on Facebook was also thought to increase psychological ownership and create a more cohesive social group.

Another way to increase visibility was to focus on the physical garden space. This included putting the garden on the university's official campus map and creating signage. The garden could also be made more visible through ephemeral actions, like sharing and handing out food or plants, walking through campus with food-in-arm, or talking about the garden in social conversations.

Increasing visibility was mainly thought to address barriers to participation, through raising awareness of the initiative so that more people knew about it and were reminded to come, and thereby breaking down the 'cultural barrier' through raising the gardens profile. Ultimately, increasing visibility of the gardens was thought to increase participation. Being on a campus map and being active on social media were thought to contribute to continuity in the project through creating a sense of permanence (through the map) and acting as a catalogue of the historical activities that have taken place (through social media).

Some actions to increase visibility were actually taken by participants. One gardener from the University of Warwick connected with a person tasked with updating their university's campus map, and as a result their garden was included in the most recent version. At Keele University, signage was created to direct people towards a garden and explain what it was, and a twitter account, blog, and Facebook events for gardening sessions were created. It was thought that the sign, being on the campus map, and Facebook events

had positive impacts, but it was hard to assess because the impacts were difficult to perceive in the short-term and/or there were many mediating variables at play.

Although the sign may have had an impact, it is unlikely it is continuing to do so because, since the sign's creation, it has been subject to the wear and tear of outdoor weather, and is therefore no longer readable (see Figures 30a and 30b). The impact of embedding the garden on the campus map (see Figure 31) was difficult to assess on a short timescale, although I suspected it would be possible it could have a more long-lasting impact. I discussed this in the following exchange between myself and a long-term mentor. To give context, in my first statement, I speculated that if the garden at Keele had been embedded in a campus map, it would give it a sense of permanence that would not have existed otherwise.

Interviewer:	Yeah, I think kind of one of the things about being on a campus
	map is they can't bulldoze over it. For a while at least. Because
	the campus map is updated. Because it looks kind of awkward if
	you had a garden and then you bulldoze over it, you know.
	Which I think [] could be a risk, so.

Long-term mentor: I think we're safe here. Everyone keeps saying it's not land they ever want to build on.

Interviewer: Yeah.

Long-term mentor: And they are building everywhere. But, you're right. It has some kind of institutional memory, just kind of like a physical memory.

Although there was no immediate intent to build on the space, having the garden in the institution's official map may make it more visible, especially to newcomers on campus. This could create a sense of permanence that might not have been there otherwise, dissuading university planners from building on the space.



Figure 30b. A sign in the garden pointing to the entrance of the garden (October 2016). The sign was placed so that people walking along the path overlooking the garden would be able to see it.



Figure 30c. The sign after several years of wear and tear such that the text is no longer visible (May 2018).



Figure 31. The University of Warwick's Student Eats garden now features in the university's official campus map (University of Warwick, 2015).

The Twitter account and blog for the garden at Keele University, however, were much less effective. At the time this thesis was published, the blog had not been updated since its inception, and the Twitter following was small and was mainly made up of non-students (see Figure 32 and 33). The impact of the Facebook events was difficult to assess as there were so many mediating variables. After the initial spree of creating the events, however, no new events were created which meant this action only lasted a few weeks. This said, following the workshop where this action was planned, I regularly posted on Keele's Facebook page approximately an hour before every gardening session and it seemed to be an effective way to remind people to come to the garden. Of this, one student leader said:

Every time, actually, sometimes I forget to go to Student Eats. Oh, I am so irresponsible! Every time Bekki will comment, 'One o'clock pm, who will come?' Okay, I come! Because sometimes I forgot and it really helps.



Figure 32. A screenshot of Keele's Student Eats Twitter page. As of August 2019, the account has been inactive for nearly a year (@KeeleStdentEats, n.d).



FOLLOW US ON FACEBOOK



Figure 33. A screenshot of Keele's Student Eats blog. As of August 2019, the account has been inactive for over three years (Student Eats, n.d.).

There was no evidence that social media had any impact on recruitment from outside

the garden, knowledge retention, knowledge handover, or any other expected impacts.

However, having taken this action and seeing the results (or lack thereof) may have taught

gardeners that these actions are not effective for increasing participation, and this knowledge would likely be carried with the gardeners in their future endeavours. However, if this was learned, it is unlikely that this new knowledge was invested back into the garden since most gardeners were involved for a short period of time. In this way, it would be expected that instrumental learning for the resilience of the gardens was not so effective, however the gardens provide sites for learning that gardeners could take forward into their future endeavours.

8.4.6 Create organisational infrastructure

In Chapters 6 and 7, I wrote that some gardeners preferred a high level of organisational infrastructure, while others found a formal set-up to be off-putting. As such, the suggestion to create organisational infrastructure needs to be applied in context-specific ways, using a great deal of flexibility according to the preferences of current participants. The main, and least contentious, way to create organisational infrastructure in the gardens was to establish a society with committee roles, having elections for these roles, and assigning individual responsibility for tasks (as appropriate).

Gardeners and National Union of Students staff thought that gardeners having individual roles and assigned responsibilities would contribute to a feeling of responsibility and psychological ownership, and increase motivation to continue to take part. A staff member at the National Union of Students said that the Student Eats societies with many roles were particularly effective because having more roles meant more people felt obligated to stay involved. Furthermore, having more roles meant that there would be better distribution of tasks, preventing participants from becoming overburdened and improving garden maintenance overall. Having organisational infrastructure was thought to contribute

to maintaining continuity in the gardens and making handovers between generations of volunteers more likely.

In one of the gardens, one of the actions proposed was to set up elections for a new committee. This was done, and a new committee was elected. However, since the garden was a society within the students' union, it is likely this would have happened even if this research had not taken place. The gardeners who were involved in taking this action were unable to attend future research workshops or interviews and therefore I was unable to ask them how they perceived their impact. However, in my analytic memos, I reflected that:

It's not enough to just have roles, they need to be seen as socially legitimate. Many people seem to think that the roles are not democratic enough¹⁸ and are just so we can continue to fit within the student union's structure.

Students also reflected that although there were elections and people took on formal roles, often people did not actually fulfil the roles they had signed on to:

We had beds manager and we had a president going. But they aren't really doing it. (Passive volunteer)

So, [a long-term mentor], who's there all the time, said to us, 'who wants to be exec?' and I was like, 'uhhh... I'll do the president thing.' It's admittedly not a very functional exec. I kind of... yeah. I do a lot that I'm not meant to, technically, in the role. So we just kind of do whatever. So yeah. (Student leader)

However, the roles seemed to be important in a more symbolic sense. When students took on a role, it created a sense of responsibility and obligation towards the garden, even if they did not fulfil their role as it was written. As such, I suggest that it is still advisable to attempt

¹⁸ What I was referring to was that gardeners had often said elections and the roles were not an exercise in democratic governance of the gardens but, rather, only undertaken in order to be considered a legitimate student society within the Students' Union.

to create some organisational infrastructure, even if it does not function in the way it was intended, because it may be providing unintended benefits.

8.4.7 Recruit

Many strategies and actions suggested by participants centred around different ways to recruit new participants, and which potential participants it would be most strategic to recruit. It was felt that the aim should always be to target as many people to recruit as possible, recognising that many people targeted by recruitment would never show up. Many research participants placed a heavy emphasis on the need to build diversity and redundancy in participation. As can be seen in Table 26, students focused most of their suggestions around the barriers to participation and participation itself, with yellow squares featuring less in the rows of the effects of problematic participation. This was also evident in the conversations with students, particularly self-directed and passive volunteers, because the emphasis in the conversations tended to be on strategies to increase participation, with any other impacts of these strategies being thought of as secondary. As one gardener said:

...if they are more, a larger group of members we will... we don't need to worry about [participation] in the summer. (Student leader)

It was also thought to be important to get people involved early in the year, and early on in their degree to maximise the length of time they could be involved in the garden to reduce the need for recruitment and retain a sense of continuity. Use of social media, leafletting, emailing, and other modes of advertising were suggested frequently to raise awareness of the initiative, thereby increasing participation. However, a student leader said that less-active volunteers asking their friends to come was one of the most effective ways to introduce people to the garden and break down the 'cultural barrier'. They said:

...sometimes random people would show up that people just took with them. You know, it was just friends of the core group. And that was really interesting. Because I had a really hard time convincing the people already in the group to come. [...] But it was weird that it was apparently really easy for those people to mobilise other people that I've just not reached. Or I've just not tried in the right way. And so that was just them seeing them in the morning and saying 'hey, do you want to come help out?' and that's it. There's no big invitation, it's just a small thing to do.

Some people brought friends to the garden as one of the planned actions. One student leader said that the person they brought only *lasted thirty-five minutes*. A long-term mentor said that in spite of this, inviting friends *extends the network* and that in their garden they rely heavily on *word-of-mouth* for recruitment because it was so effective.

It was also thought that diversifying recruitment was valuable, especially in recruiting 'freshers' (first year undergraduate students), postgraduates, and other long-term participants, like university staff or campus residents. These potential participants were strategic to recruit because they would be able tend to the garden for longer periods than other potential recruits, and some are more likely to be around over the summer months. It was also thought to be useful to recruit participants who were knowledgeable about gardening to elevate overall competence within the garden.

8.4.8 Document and plan

Documenting the history of the gardens and planning for their future was thought to address upstream and downstream causes of problematic participation, and problematic participation itself. Records could be kept to demonstrate impact, catalogue useful resources (e.g., gardening advice), and provide future gardeners with vital information for future strategic planning. Many participants mentioned the need to know what had been planted

where year-on-year to prevent disease, while some gardeners also felt they would benefit from a to-do list or a how-to guide to aid in the week-to-week maintenance of the garden when more knowledgeable gardeners were not present. Gantt charts or visual calendars with pictures of what needed doing when were suggested formats for these aids. Participants said the tangible products of documentation and planning (such as written or online documents) ought to be part of a handover between generations of gardeners.

The main way documenting and planning in the gardens was thought to address the challenges associated with transience was through addressing the knowledge and skill deficit, and especially the lack of knowledge and skill transfer. Documenting and planning were also thought to address participants' lack of confidence by providing information and direction to support their activities in the gardens. A staff member from the National Union of Students thought that a template for a growing calendar would increase psychological ownership as participants could 'plug' their ideas into the template. Research participants said that documenting activities in the gardens and providing handover documents would improve maintenance of the garden and prevent gardeners from becoming overburdened, overwhelmed and just burned out, presumably through increasing the efficiency operations in the gardens. Having a legacy strategy was thought to reduce the risk of low participation in future. In my research diary, I also noted that keeping records would be a way of providing better continuity between generations of students. It was also thought by some that keeping records (for example, recording quantified impacts/outcomes) and having a vision would help to enable potential participants to better understand the purpose of the gardens and care about them more, thereby decreasing the 'cultural barrier.'

Having a shared vision which gardeners input into was thought to contribute to gardeners' ability to self-determine, and therefore build their sense of ownership over the

space. A vision was also thought to be important to increase continuity and foster longerterm thinking. A student leader reflected on this:

...you need a bit of a long-term vision for a garden because you don't, obviously... the results aren't immediate. Like, you have to wait a long time. Especially with some crops, like you just have to leave them in the ground for a few years before you get a really good crop. Like we were thinking of doing asparagus, but I was like, there's no point because by the time it's ready to eat, we'll all be gone. Just about choosing your crops really. Like, things that give you a bit more of an immediate benefit rather than the ones you have to wait a long time for. But gardens are more of a long-term project.

The vast majority of participants felt that to-do lists, guidance documents, and even just provision of general gardening advice would be valuable tools to address the negative impacts of students' transience. In the fishbowl workshop, a students' union staff member described how their garden had taken this to the extreme. They only grew ten crop types, and each person was assigned responsibility for only one. They said:

So we have a small library. And everyone who has a small plot has a book. That shows how to grow that kind of vegetable. So they are given the book, and taken through this basic process of growing, of sowing seeds, like that.

However, this system took place in a college (a further education institution) and had considerable staff time dedicated to their garden. As such, some people were sceptical that such provision of to-do lists and guidance documents would have much of an impact in their food gardens that were more student-led. In one of the evaluation workshops, this was discussed when an 'action' had not been completed. The 'action' was to create a board with information about what tasks needed doing so that even if the knowledgeable gardeners were not there, others could still participate. The following conversation took place:

Long-term mentor: I mean, would people like a board that said weed bed number 6? And, you know, thin out carrots in bed number 5? Would that be useful to people?

[silence]

Active gardener:	Because if it says thin out carrots in bed number 5, would people	
	know what that meant? Or?	
Passive volunteer:	No. It's my second time coming. So uh. I wouldn't know how to	
	do a lot of things. So, I need someone to tell me.	
Interviewer:	So, you need that actual, kind of, person-to-person contact. Okay.	
	So, in terms of that one, scrap that? As an action?	

Long-term mentor: Yeah.

The sentiment of the long-term mentor was one that I also shared because I felt that students who were looking to engage passively were not likely to engage in gardening that required the level of independence that reading to-do lists and gardening resources, and figuring out what to do would require. Furthermore, lack of confidence was a big barrier. While information provision might increase confidence to some extent, it seems more likely that a more human-centred approach to learning and confidence-building would be appropriate. Otherwise, a much less labour-intensive way of accessing information about growing would be to find gardening information that is widely available online.

In my research diary I also reflected that:

Having a really well-mapped growing calendar or anything like that would probably teach students more about gardening skills and knowledge in a lead-by-example sort of way. But in terms of managing a complex project, and having the opportunity to self-determine and show initiative, a more 'messy' project does provide many more opportunities for learning – in a problem-based-learning sort of way.

Building a narrative for the gardens was thought to be important for participants to have a sense of the overall purpose of the initiative and feel as though their actions are a part of something bigger. Narrative is widely recognised to be important for creating pathways for sustainable futures (Veland et al., 2018) and thought to be a crucial pedagogical device for sustainability education (Tooth & Renshaw, 2009). Documenting and planning were thought to be a crucial strategy for building a narrative for the gardens. A student leader reflected on the role of narrative and how to build it in the following exchange:

Student leader: Scenario A: you're a student invited to a gardening session. You turn up. You've been told, hey, let's turn this soil over. Let's plant some things there. You can take some of it home. We meet again next week. That's it. Scenario 2: You get to this place. You meet somebody. And they begin by asking you about how you are and how you got into this, and then they tell you about this place and how long it's been running. And what the story behind it is. Why people are doing it. Why they are doing it together. Why they are doing it in their free time. What's the, kind of, pay, in terms of vegetables and how they think it's a good idea to do it that way. You see, I think if you create this narrative and this story around it, it makes people... it makes it so much easier for people to understand what it's about. And why it's important that they do it. [...] If you have some more depth to it. Rather than, hey, we are doing this out of practicality. I don't know what the alternative would be. What would you say if you don't have a story? 'We garden because we garden'? I guess you always have some kind of story. I guess it's... the question is not 'do you have a narrative or no narrative?' The question is how much

effort do you put in the narrative. Because you always create some kind of story, right?

Interviewer: Yeah. And so... because I completely agree. I think the whole narrative is really important. But, how do you see a narrative, for example, in the garden being established and taking place when people are turning over. Is it...

Student leader:Records. Keeping records. Maybe a one-pager of the history of the
garden. The have some kind of yearly calendar of the, kind of, bigger
structural things that have changed within the years. Year one we
got two beds. Year two we made the compost. Year three, we did
that. Right? Something to aid that... turnover. That... I don't know.
The progress of the years. To keep record of that. I think that's it.

8.4.9 Embed in wider organisational infrastructure

Another strategy for addressing challenges associated with transience and participation was to embed gardens in existing organisational infrastructure. Embedding in existing organisational infrastructure was most important for times when participation dropped off. Once embedded, university, students' union, or National Union of Students staff could 'kick-start' a struggling group by advertising and recruiting new members.

One way to embed the gardens in wider organisational infrastructure that was already widely done was to affiliate gardens with students' unions through becoming a student society. This meant they would be expected to have a committee, elections, and, in some cases, membership fees. It also meant that they would benefit from the support that students' unions offered societies, such as funding and advertising opportunities. It was thought that an affiliation with the students' union would help with knowledge handover and increase continuity. Embedding the garden into students' union volunteering schemes was also thought to make more students aware of the initiative.

The gardens could also be embedded into the university's organisational infrastructure, such as through adding volunteer sessions to university calendars, into internal communications, and on university webpages. Once embedded, these were thought to contribute to a sense of permanence and legacy, whilst serving as automated reminders of the gardening sessions. These were all strategies that contributed to the visibility of the gardens. It was also thought to be helpful to embed the garden into the operation of the grounds or estates maintenance team, although gardeners were cautious about this for fear that more involvement from the university might cause the garden to become 'institutionalised':

It's not come to fruition yet, but one of the tasks I was set was to kind of embed the garden in... the actual running of the estates more substantially. I mean, it's complicated, and everyone's wary. But it's quite nice to feel that. On the other side too, we don't want to just be co-oped by Estates. But we want more help. And we want to be... kind of... 'institutionalised' as it were. So I sent a few emails. And there are these sustainability champions that work within Estates. And they've been really good at kind of saying, 'we'll be your kind of go-to people, and we'll talk to the head of estates,' who's really hard to get in contact with, 'about regular compost deliveries, cutting back the pathways...' (Long-term mentor)

In order to recruit more people and increase participation, participants also suggested Incorporating the garden into the formal curriculum. One way to do this would be through a service learning inspired model, where students would receive academic credit for their involvement. However, this is not the only way it could be done. For example, at one of the gardens studied, international students in a summer English course visited the garden and

engaged in gardening activities. This resulted in some students coming back. Students also regularly used one of the gardens as the topic of an assignment in which students conducted environmental audits of on-campus sustainability. However, different strategies for embedding gardens in the curriculum may have different impacts on students' experiences in the gardens and the benefits that participation may provide. Providing academic accreditation for participation, for example, may increase participation levels and reduce the impacts of transience. However, it may also dampen the capacity for the gardens to remain student-led, and thereby limit creative development and opportunities for more 'intrinsic' forms of learning. When deciding how to embed gardens in the curriculum, consideration should therefore be given to the types of benefits and educational outcomes that are being aimed for, and what types of activities can maximise such benefits and outcomes.

A staff member from the National Union of Students also said that it was important to embed responsibility for the garden into university or students' union's staff member's role description. That way, if a supporting staff member left the university, their replacement would retain supporting the garden in their mandate.

At one of the gardens, one of the proposed actions was to add the gardening session to the postgraduate student calendar. This was done, however there was no evidence that more people were aware of the garden or that it contributed to recruitment. On the other hand, it is likely that more people were aware of the garden because being put in the postgraduate calendar meant that it was included in weekly emails to all postgraduate students.

8.4.10 Maintain healthy relationships and social networks

It was thought that maintaining healthy relationships and networks was important. This includes relationships within the gardens, and between gardeners and external

stakeholders. Some said that being nice and staying positive was one way to do this. It was also thought to be important not to blame or judge people for not pulling their weight, and to show appreciation for the people acting in a more facilitative role as it could be a taxing but invisible job. Cliques could be prevented from forming through befriending new members, and participants thought it to be important to deal with interpersonal conflict. This included interpersonal conflict between gardeners, but also conflict or tensions between gardeners and staff, as described in Chapter 6 and 7.

8.4.11 Connect

It was also thought to be useful to connect to the students' union's current presidents' priorities, other student societies, the National Union of Students' Student Eats Network, and other Student Eats gardens. Connecting to the current priorities of the students' union's president was thought to reduce the 'cultural barrier' by linking the garden to something high on the agenda for the students' union. An example given by a staff member from the National Union of Students was to demonstrate the mental health benefits of gardening in order to connect to the student unions' mental health agenda. Connecting with other student societies was also a way to spread the word about the garden and open it up to new members. For example, one garden invited different societies to come visit their garden sessions¹⁹. Others suggested that there could be themed gardening sessions by collaborating with other societies.

Connecting to the Student Eats network was thought to put gardeners in contact with resources to improve their gardening knowledge and skills. Furthermore, connecting to other

¹⁹ This, however, did not happen as one of the 'actions' taken through this study.

Student Eats gardens through the network was thought to help gardeners feel they were part of something bigger and made them feel better about the work they were doing.

8.4.12 Create vibrancy and focus on value

I borrowed the phrase 'focus on value' from Wenger et al. (2002) to capture sentiments expressed by the research participants. The meaning of 'value' is contextdependent. What is valuable in one garden may not be in another. Furthermore, actions that are valuable at one point may become less valuable at another point, as the social and organisational dynamics in the garden change. Some of the suggestions about how gardeners can focus on value included ensuring gardening activities were not derailed by committee meetings, making sure recruitment remained a secondary rather than primary purpose, continuing with tasks that need doing even if other people do not show up to gardening sessions, and focusing on activities that were fun and/or useful. It was thought these actions would improve maintenance, and although this was not said explicitly, this suggestion to focus on value seemed to address the 'existential crisis' caused by an overemphasis on recruitment. Focusing on value was thought to help maintain balance between activities that people want to do and activities that need to get done. One students' union staff member said:

The struggle we have is that we get task-obsessed sometimes. So not the whole clique thing, but we forget to befriend. So, people who have been there and done it can be like, let's get stuck in that, let's go! Let's do this today, this, this, and this. And then five new people show up and... you just kind of assume you know what's going on. You don't get to actually know them. And then they do it once and they don't come back. Because, you know, it wasn't a purposely trying to exclude people. But they were so focused on something else and forgotten to take a minute...

Becoming task-obsessed and forgetting to befriend is as problematic as focusing only on the social aspects of being part of the garden group and forgetting to garden. Focusing on activities that bring value to the group can help prevent falling into such either/or traps.

One of the ways to create vibrancy was to have passionate people in the group that believe in the garden. Passionate people were thought to make others care about the garden and break down the 'cultural barrier.' They were also thought to be more likely to take action that creates a better sense of continuity and legacy for the gardens. Another way to create a feeling of vibrancy, according to the gardeners, was to have an impressive harvest in the autumn to inspire new recruits and generate interest from those who might not have heard of the garden. Overall, the focus on creating vibrancy and focusing on value was thought to make the garden more fun and create a positive energy that decreased negative emotional responses.

8.4.13 Appeal to extrinsic motivation & resource adequately

It was also thought that appealing to students' extrinsic motivation would be useful through, for example, giving away free plants and refreshments, or feeding participation in the gardens into volunteering award schemes so students could receive credit for their work. Some also said that paying for membership at the beginning of the academic year would encourage people to continue to take part. Others, however, did not think this was the case, and that it would only discriminate against those who could not afford to pay. It was also felt that the gardens should be funded and resourced adequately, and some thought paid student roles would be useful.

8.5 Problems without solutions

There were a number of barriers to participation (see section 6.2) that were not addressed by any of the suggestions. This is to be expected because, like transience, some of these barriers are not problems that need solving. For example, students trying out gardening and finding it is not for them ('experimenting'), is natural and to be expected. There were other problems that were not addressed by any solutions because they were minor issues. For example, while some gardeners thought that the cost of society fees or buying seeds could be prohibitive for potential recruits, in the three gardens studied, all prospective gardeners were welcomed in regardless of whether they were officially members of the society or could contribute to purchasing seeds or other materials.

However, there were also some solution-less problems that are more problematic and need to be unpacked further. First of all, prospective gardeners' expectations and misconceptions (see section 6.2.1) about the space were not addressed by any of the suggestions. I suspect this is because it is assumed that prospective gardeners coming to the garden and taking part would be the best way to address this. For example, when a new gardener comes into the space with a vision that does not match the group's capacity to act, the best way to recalibrate the vision is to try to put it into practice. Likewise, short-term thinking and lack of insight into transience is difficult to address directly other than to take part, and over time insight into transience and longer-term mindset can be developed. This said, recruiting long-term participants could also be a potential solution to this. As unmet expectations and misconceptions, short-term thinking, and lack of insight into transience all can contribute to negative emotional responses, like frustration, it would also likely be useful to address negative emotional responses to these as a downstream solution (see Table 26 for strategies to address this). Students being unwilling to commit was another solution-less problem. To some extent, students not wanting to commit is a decision that should be accepted as it may appropriate for students to be able to maintain a balanced, healthy schedule. As such, I would be reluctant to provide strategies that directly deal with this. Instead, I would prefer to focus on increasing initiative, motivation or responsibility, and hope that a greater willingness to commit would be a by-product. However, a similar solution-less problem was gardeners' lack of initiative. It is not entirely unexpected that solutions addressing 'being unwilling to commit' and a 'lack of initiative' were not highlighted because research participants often used terms like confidence, initiative, responsibility, ownership, and motivation to capture the same thing: the intangible quality that makes someone exercise their agency for the good of the garden. Because of the fact that, oftentimes, these terms were being conflated, it is unsurprising some of them were missed when solutions were being suggested.

Untangling how research participants understand each of these terms and how they correspond to how they are defined would be a challenging and interesting route of inquiry. However, in the pragmatist tradition, I would suggest it is sufficient to understand that the solutions increase participants' exercising their agency for the good of the garden, regardless of whether it is to do with increasing confidence, initiative, responsibility, ownership, or motivation. I will elaborate on this further in the following section.

8.6 Increasing participation through 'action competence'

In this section, I will make the argument that there is a mutually reinforcing relationship between the development of 'action competence' (Mogensen & Schnack, 2010; Hedefalk et al., 2014) and participation in student-led food gardens (see Figure 34). That is to say, the more students participate, the more they develop 'action competence,' which makes them more likely to become a more active participant in their garden. But conversely, if students do not participate, then they do not develop the action competence that drives them to exercise their agency, making them even less likely to participate.





To make this argument, I would first like to suggest that this 'intangible quality' that makes participants exercise their agency for the good of the garden are the pieces that make up 'action competence,' as it is called in sustainability education. As highlighted in Chapter 2, 'action competence' tends to defy definition. However, when scholars do describe it, they use terms like motivation and desire (Morgensen & Schnack, 2010), ownership and commitment (Fien & Skoien, 2002), confidence and responsibility (Almers, 2013), and so on. In the previous section I said that I believe when the research participants used these terms – confidence, initiative, responsibility, ownership, or motivation – they were trying to grasp at the same idea: the intangible quality that makes someone exercise their agency for the good of the garden. I believe this intangible quality to be action competence because it has to do with "ability, motivation and desire to play an active role" in a sustainability-focused initiative (Morgensen & Schnack, 2010, p. 61).

If we take this to be true, the next step is then, to demonstrate that action competence increases participation in student-led food gardens (the upper arrow in Figure 34). As said in Chapter 2, action competence is associated with the intrinsic tradition of sustainability education, and as such is process- rather than outcome-orientated. As such, action competence certainly inspires action of some kind – whether or not it is in student-led food gardens. However, I suggest that it is a safe assumption that if someone has the "ability, motivation and desire to play an active role" in a sustainability-focused initiative, they would be more likely to take part in a student-led food garden than someone who does not have such an ability, motivation or desire. That is not to say that developing action competence will automatically make a person participate in a student led food garden. Rather, I suggest that the development of action competence in a pool of (potential) gardeners would increase participation in student-led food gardens in general.

Finally, I argue that students can develop action competence in student-led food gardens (the lower arrow in Figure 34). I will do this by outlining how many of the learning environments that are thought to be conducive to building action competence are present in student-led food gardens. First, gardeners worked with "power relations and conflicting interests" through collaborating with other students, and university and students' union staff (Breiting et al., 2005, p. 25). It is through negotiating the challenges associated with working with power relations and conflicting interests, that action competence can be developed (Breiting et al., 2005, p. 25). Through prolonged engagement in the gardens, students can come to consider the "relations between the past, the present and the future," another instance in which action competence can be developed, in order to acquire an understanding of how students' transience impacts student-led food gardens (Breiting et al., 2005, p. 19). In some cases, gardeners had "the opportunity to appreciate and confront diversity," such as through engaging with people with different politics, like in the exchange in section 6.2.2 where a student describes how the garden put vegans and non-vegans into conversation (Breiting et al., 2005, p. 23). This too, is thought to be a situation which can foster action competence. As such, a logical conclusion of this is that the student-led food gardens provide environments in which action competence could be developed.

If we accept all of these propositions to be true, and the relationship between action competence and participation in student-led food gardens is mutually reinforcing, then why is

there a problem with participation? It is, likely in part, because there are many other variables affecting participation, and perhaps development of action competence is not a strong enough driver of continued participation to make a difference. It is also likely that action competence is not developed though peripheral or passive participation because those who engaged in activities that are conducive to building action competence tended to be the more active gardeners. For example, long-term mentors and student leaders were most often the ones managing power relations and conflicting interests, activities conducive to building action competence.

Therefore, to activate this amplifying feedback between participation and action competence, it may be necessary to foster more active participation from passive gardeners. It may also be helpful to create 'boundary interactions' in which gardeners are exposed to opportunities to confront diversity to develop action competence. A boundary interaction is where a negotiation of meaning take place through the juxtaposition of an individual's experience against the socially defined competences of the community (Wenger, 2002). In these boundary interactions, learning happens through "negotiation of meaning as individual experiences and socially defined competences intertwine, or fail to intertwine" (Bendt et al., 2013, p. 19). Wenger et al. (2002) have suggested some strategies to create such boundary interactions. One of these is to "open a dialogue between inside and outside perspectives" (Wenger et al., 2002, p. 51). In addition to building action competence, these interactions can also contribute to problem solving within the community. This is because insiders "appreciate the issues at the heart of the domain, the knowledge that is important to share, the challenges their field faces, and the latent potential in emerging ideas and techniques," whereas outsiders can "help members see the possibilities" for their initiative and how to address challenges it might be facing (Wenger et al., 2002, p. 54-55).

Another strategy to build action competence is through pairing action and reflection (Breiting et al., 2005). In co-curricular activities like student-led food gardens, the focus tends to be on action. It would be untrue to say no reflection takes part, and it certainly did take place in gardening sessions I attended, especially through conversations between gardeners. However, reflection typically only happened incidentally and could be built in more strategically if sustainability education is the principle outcome of these initiatives as suggested in Chapter 5. It is possible that formal opportunities to reflect on their participation could be valuable for those taking part in student-led food gardens. Embedding opportunities like this into already existing programmes could be a strategy to do this. At Keele University, students were able to log their volunteering hours to receive volunteering awards (KeeleSU, n.d.). The volunteer hour log could potentially be accompanied by a written or oral reflection in order for students to get the most learning out of their volunteering. A different, but also optional, scheme run at Keele was a skills portfolio programme accredited by the Institute for Leadership and Management (Keele University, n.d.). Students undertaking the programme are required to read articles on, for example, critical thinking, time management, public speaking, leadership, and so on. They are also required to write between 2400 and 9000 words reflecting on six pre-determined personal and professional development skills, such as techniques for getting organised and dealing with stress. Such a portfolio could be an excellent place to embed reflective exercises reflecting on sustainability-related co-curricular activity. Similar programmes exist at other higher education institutions in the United Kingdom and could be leveraged in this way as well.

8.7 Retaining social-ecological memory

In this section, I will draw on the concepts of 'social-ecological memory' to make sense of how memory was being retained within the student-led food gardens and how it could be better retained. Social-ecological memory is a relatively new concept that has built on prior theorisation about social memory (Coser, 1992; Gongaware, 2003) and social-ecological systems (Folke et al, 2003; Berkes et al., 2003). Social memory (sometimes called collective memory or cultural memory) is thought to store "experiences of living pasts and influencing behaviours of societies and groups" (Barthel et al., 2010, p. 255). The concept of 'organisational memory' (as brought up in Chapter 6) has typically been theorised through an approach to knowledge akin to Freire's (2000) 'banking model,' in which organisational memory is thought of as a "storage bin" rather than thinking of memory within an organisation as being something that is socially-constructed (Rowlinson et al., 2010, p. 69). However, in recent years the approach to memory in organisational studies' approach has been shifting towards more of a social constructivist view (Rowlinson et al., 2010). As such, one might think of this social constructivist interpretation of organisational memory as the storage of experiences of living pasts that influences behaviours of an organisation. Socialecological memory is a bit broader, in that it is made up of a set of "collectively shared mental maps for dealing with a complex world" that carry social and "ecological experiences and revises them over time and between people" (Barthel et al., 2010, p. 256). Social memory, organisational memory, and social-ecological memory are all means of carrying forward experiences and mental maps and have relevance to the study of student-led food gardens, but social-ecological memory is the only conceptualisation that encompasses the ecological system in full recognition of its complexity. Furthermore, student-led food gardens are extremely amorphous organisations that are more akin to social networks than the somewhat more formal organisations that the concept of organisational memory is typically written about (e.g., Olivera, 2000; Ibert, 2004). As such, social-ecological memory is the most useful concept to use in this instance.

Social-ecological memory can contribute to local resilience by retaining knowledge about management of social and environmental ecosystem services that can be drawn on in times of crisis (Barthel et al., 2010). It has been found that the main ways that socialecological memory is retained in urban gardens is through rituals/habits, oral communications, rules-in-use/metaphors (e.g., organic practices, phrases/sayings), physical forms/artefacts, and engagement from/embeddness in external institutions or physical forms (e.g., artefacts, laws, social networks) (Barthel et al., 2010). These were all present in studentled food gardens.

As highlighted in section 8.4.8 (on documenting and planning), in-person communication (whether it be through rituals/habits, oral communications, or rules-inuse/metaphors) are effective ways to retain memory within a student-led food garden. An example of a ritual that took place through which memory was retained was how, every week before the regular gardening session in one of the gardens, one of the self-directed volunteers would post a picture on social media of what had taken place the previous week:

I'm hoping [...] just capturing these photos will make a difference in encouraging people. I know it's not the same as [...] the actual experience you have, but I'm hoping... that will have an impact on future cohorts coming in.

Posting the photos not only reminded participants to come to the garden, but produced a log of what had been done in the garden, week by week. However, it was unclear if the log of photos had been used in a way that made use of these memories.

However, given the high levels of student transience in student-led food gardens, it often happened that rituals, oral communication, and rules-in-use/metaphors were not retained or passed on because of high turnover, seasonal breaks in participation, and periods of no participation at all. Over shorter periods and between generations of students that overlap, these strategies may be crucial for retaining memory. However, over longer periods, I found that physical forms/artefacts, embeddedness in external institutions, and objects beyond the gardens had a more important role to play in keeping retaining social-ecological memory. For example, at one garden there was little or no overlap between participants, which meant that rituals, oral communication, and rules-in-use/metaphors were ineffective for retaining social-ecological memory. However, physical forms/artefacts, like physical garden space itself, and external sources of support, like the students' union, played an important role in retaining social-ecological memory.

It was thought that artefacts like written records, sowing calendars, or Gantt charts were important for addressing challenges associated with lack of knowledge/skills transfer, especially when they were retained in a way that would be accessible to future generations of students. This was because rituals, oral communication, and rules-in-use/metaphors could not be used in cases when generations of gardeners did not overlap temporally. However, the physical features of the gardens themselves seemed to play a more important role in practice. Raised beds, greenhouses, and other physical features served as a reminder of what the gardens had been, even when they weren't being actively managed. Observing the soil told gardeners about what has been there previously. For example, in one of the gardens both miner's lettuce and nasturtiums returned to the beds year on year, regardless of what had been planted there since. Latent disease present in the soil manifested when the crop rotation was not maintained. At Keele, when the garden was created, it was even chosen to be placed in the historical walled garden of the campus estate from the late 1700s, serving as social-ecological memory manifest.

As highlighted in Chapter 6, long-term stakeholders, although they may be 'outside' the initiative, have a vital role in retaining memory within the gardens. Most directly, university staff with gardening experience shared their knowledge and skills with inexperienced gardeners. University and students' union staff also supported students with

the retention of organisational memory by adding gardening sessions to university calendars, into internal communications, and on university webpages. Staff from the National Union of Students contributed to the development of social-ecological memory by supporting student gardeners with guidance about food growing and structure to support record-keeping, such as templates of sowing calendars. However, long-term stakeholders could help students even more by ensuring that gardens' plans and documentation are passed on to future generations of students rather than leaving it entirely in the students' hands. As such, in times of high turnover, physical forms/artefacts and external sources of memory played a more important role in contributing to social ecological memory than did rituals, oral communication, and 'rules-in-use'/metaphors.

8.8 Summary of advice for different stakeholders

Different stakeholders have different roles to play in managing the impacts of transience within student-led food gardens. All the suggestions from earlier in this chapter, including those summarised from Chapters 6 and 7, have been separated out into advice for students, advice for university and students' union staff, and advice to the staff at the National Union of Students and presented in Table 27. Sustainability science and action research are intended to have practical impacts. This advice, summarised and disaggregated by stakeholder, is intended to contribute practically.

The roles actors within universities and students' unions take in the governance of student-led food gardens can be expected to vary between institutions, depending on staff competencies, interests, and institutional roles and arrangements. For example, some universities' sustainability staff are located within the estates, grounds, or operations teams, and other institutions have sustainability staff located with academic faculties, or as a unique unit within the institution. Furthermore, some institutions operate their student-facing

sustainability work out of the students' union. As such, advice to students' union and university staff have been lumped together because to disaggregate them would make the recommendations less widely applicable.

In summary, advice to students is tailored towards how to run the core activities and recruitment in the garden, with a focus on keeping a long-term perspective in mind. The advice for university and students' union staff focuses on supporting student gardeners in a way that students' autonomy over the garden and how it is run is retained. Their support should focus particularly on students' blind spots, like having a long-term view of the garden, ensuring knowledge handover, embedding the garden in the university's activities and infrastructure. In all of this, university and students' union staff should become and stay attuned to the power dynamics between themselves and students, and develop strategies to compensate for them. Advice for the staff from the National Union of Students focuses on providing support to students through providing gardening-specific resources, providing inspiration, facilitating and mediating staff-student relationships, and helping student gardeners see they are part of something bigger. I also suggest that all stakeholders should strive to learn, through reflection and from each other, and treat learning as a process rather than as an outcome.

Some of the advice in this chapter and Table 27 is very specific to the context of the United Kingdom. As highlighted earlier, not all countries have such active or well-funded students' unions at university or national levels. However, it is clear that the National Union of Students has and can continue to fulfil a unique role in linking student groups from different universities to provide guidance and inspiration, and help students see they are part of a bigger movement. As external stakeholders with insight into both the student and staff perspective, hey also help facilitate healthy student-staff relationships, and build bridges across different parts of the university. What this means is that in countries where nation- or

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Maintain an appropriate and well-managed space	 Grow low-maintenance crops that are either early or late cropping (as long as it doesn't prevent gardeners from being creative or dampen interest) 	 Provide physical infrastructure required for gardening (e.g., water taps), and infrastructure to reduce the maintenance needs of the space (e.g., raised beds), provide sheltered space for gardening in poor weather, and create social spaces (e.g., seating) If setting up a garden, consider its location with regards to connectivity with the rest of the campus Ensure the space is widely accessible, including out of business hours Pay attention to the physical state of the garden as an indicator of the overall status of the gardening group 	 Support student gardeners with information about low-maintenance growing that is congruent with the academic calendar
Run events and activities	 Organise regular year-round gardening sessions on a set day and time on a weekly basis Create variety in the garden's activities, with balance between 'fun' activities (e.g., socials, workshops) and work that just needs to get done (e.g., recruitment, gardening tasks that some might find boring) Create opportunities for structured volunteering and more free-form modes of taking part 	 Fund visits, short courses, or other events or activities, as appropriate 	 Continue to run conferences Provide information to students about external events and activities they can take part in Fund visits, short courses, or other events or activities, as appropriate

Table 27. Advice to different stakeholders of student-led food gardens for how to address the impacts of transience in student-led food gardens.

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
are supported	 Seek out support from: University and students' union staff The National Union of Students Knowledgeable gardeners Your peers For help with: What to grow and how to grow it Retaining the garden's historical knowledge and pass it forward to future generations of students Embedding the garden in the university's activities and infrastructure 	 Ensure there is an appropriate balance between staff and student involvement such that students feel supported but that students' autonomy over the garden and how it is run is retained Reassure students when they face challenges and become frustrated or disappointed 'Kick-start' the garden if it falls into a period of low participation or inactivity Be responsive and timely in interactions with students, but acknowledge they are volunteers and may still be learning to extend the same level of professionalism Students' unions should offer support with raising awareness of the garden, recruiting new participants, running the garden as a student society and support knowledge handover Staff with gardening experience can offer to share their knowledge/skills Grounds or operations staff can mow the lawn within or around the garden and provide compost, as relevant 	 Reassure students when they face challenges and become frustrated or disappointed Support students in building and maintaining productive relationships with university and students' union staff

Table 27 (continued). Advice to different stakeholders of student-led food gardens for how to address the impacts of transience in student-led food gardens.

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Nurture interest and creativity	 Provide leadership, mentorship, and facilitate engagement in the space, even if you are not sure if it is your role to do so Reassure others when they face challenges and become frustrated or disappointed Do not overburden yourself or others, but do encourage taking on a bigger role and coming back again Do not tell other people what to do; instead, let them ask Be flexible, pay close attention, and sense into the needs of the group so you can create an atmosphere where people can engage in ways that suit them Communicate in person where possible Ask other students what their visions for the space are Listen to others and help them 	 Devolve as much decision-making power to students as possible Ensure that averting risk stifles students' creativity and self-determination as little as possible 	Provide case studies of impressive or innovative initiatives
Increase visibility	 Advertise the garden through the student newspaper, social media, leafletting, emailing, blogging, and creating videos, for example Use social media to remind people to come to the garden on a weekly basis through creating online events or messages Create lively conversations on social media with many people contributing Hand out or walk around with food/plants Spread the word about the garden in social conversations 	 Put the garden on the university campus map Create signage directing people to the garden and/or an interpretive sign for the gardening space 	

Table 27 (continued). Advice to different stakeholders of student-led food gardens for how to address the impacts of transience in student-led food gardens.
	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Create organisational infrastructure	 Establish a student society with committee roles, and have elections for these roles Create individual responsibilities for tasks if you, collectively, feel this is appropriate to your group Long-term mentors and student leaders should engage less-active participants in decision-making as much as possible 	 Provide gardeners with the framework to set up a student society and reminders to have elections 	 Provide guidance to students and Students' Union for the sorts of roles and tasks that are appropriate to a food growing society (in contrast to other student societies)
Recruit	 Recruit as many people as possible through increasing the visibility of the garden, acknowledging that many people who are targeted will never show up Ask friends to come to the garden Strive to recruit diverse participants Strive to involve participants that will stay involved for longer periods (such as postgraduate students, staff, first-year students, or campus residents) Strive to recruit people who are knowledgeable about gardening 		

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Document and plan	 Focus on retaining knowledge within the garden by: Keeping records on what has been planted where each year Creating to-do lists and/or how-to guides (could be Gantt charts or visual calendars) Keeping records in an accessible and low-maintenance way so they can be used in handovers between generations of gardeners and catch the eye of outsiders Plan for the future by visioning what the garden should look like in future and creating legacy strategies for the next generation of students Construct and communicate the story of your garden – where has it come from and where is it going to? 	 Help students transfer plans and documentation on to future generations of students Prompt students to document and plan at appropriate intervals 	 Provide guidance (such as gardening information and resources) and structure (such as templates) to support record- keeping

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Embed within wider organisational infrastructure	 Affiliate the garden with a students' union through creating a student society Take advantage of opportunities provided by an affiliation with the students' union, such as funding or advertising Embed the gardens into students' union volunteering schemes, if relevant Seek to add gardening sessions to university calendars, into internal communications, and on university webpages 	 Add gardening sessions to university calendars, into internal communications, and on university webpages Academic staff can offer to embed the garden in their formal curricula, where appropriate Grounds or operations staff can embed the garden within the schedules to mow lawns and provide compost, if relevant Embed the responsibility for the garden into university or students' union's staff member's role description Embed opportunities for reflection about taking part in the garden in leadership accreditations, volunteering awards, or other such schemes, as relevant 	
Maintain healthy relationships & social networks	 Create a lively and cohesive garden group, and remain open to new recruits to prevent being seen as clique-y Be nice to and befriend newcomers Stay positive and reserve judgement and blame if others do not 'pull their weight' Address interpersonal conflict Engage with staff at the university, students' union, and National Union of students in a professional and timely manner, and be patient with them as helping the garden is likely only one of many tasks on their plates 	 Help mediate interpersonal conflicts between gardeners, as appropriate Pay attention to power asymmetries between staff and students, and develop strategies to compensate for them Pay attention to the quality of relationships between staff and students to identify changes, especially changes for the worse 	 Pay attention to the quality of relationships between staff and students to identify changes, especially changes for the worse Play a facilitative role in student-staff conversations to create a mutual sense of trust and understanding, and mediate as necessary in the event of friction or conflict

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Connect	 Connect the garden to current priorities of the students' union or university Invite other student societies to the garden or run joint activities Long-term mentors and student leaders should strive to act as a bridge between the less active gardeners and university, students' union, and National Union of Students staff 		• Use the Student Eats network to connect gardens from different universities
Create vibrancy and focus on value	 Find passion in your work in the garden and support other in finding theirs Plan for an impressive harvest in the autumn Pay attention to the balance between an emphasis on recruitment compared to the 'core business' of the garden as an indicator of whether the focus needs to shift Focus on activities that bring the most value to the garden and try not to get side- tracked by activities that might not bring as much value (such as too many committee meetings, or tasks that do not contribute to the vision for or fun in the garden) Ensure long-term mentors and student leaders get to take part in gardening activities, and not only administrative activities 		

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Appeal to extrinsic motivation	• Give away plants or serve refreshments at gardening sessions	 Create opportunities for taking part in the garden to contribute to volunteering award schemes, if relevant 	
Secure funding/resourcing		 Consider paid part-time student roles for coordinating the garden and volunteering activities in the garden Ensure time is allotted for supporting students, preferably from both more traditionally student-facing roles (i.e., students' union or sustainability staff) and estates, grounds, or operations staff 	

	Advice to students	Advice to university & students' union staff	Advice to staff at the National Union of Students
Other key messages	 When trying to improve participation and reduce the negative impacts of problematic participation, consider which of the issues in Figure 26 are the most relevant in your garden, and use Table 26 to find appropriate solutions to address those issues Be patient with uncertainty, and do not let it prevent you from taking part in the garden Do not worry if your actions do not have an impact right now. They might have an impact in future that you cannot perceive, or it might take ten tries for an action to have impact. Do not stop on your ninth try! Be open to learning because 'you don't know what you don't know' Think of learning as a process rather than an outcome – it's not what you get out of it, but how you get there Reflect, and be open minded with people who might not think the same way as you do because you never know what you might learn from that interaction – even if it is uncomfortable! Understand that there will be different levels of participation in the garden, and it is okay that some people are more active than others 	 When trying to improve participation and reduce the negative impacts of problematic participation, consider which of the issues in Figure 26 are the most relevant in your garden, and use Table 26 to find appropriate solutions to address those issues Work closely with other staff supporting the garden, whether they be in the students' union, an academic, in the estates, grounds, or operations team, at the National Union of Students, or elsewhere Ensure students are actively involved in decision-making about the garden space Be open to learning because 'you don't know what you don't know' Reflect, and be open minded with people who might not think the same way as you do because you never know what you might learn from that interaction – even if it is uncomfortable! Encourage students to think of learning as a process rather than an outcome 	 Be open to learning because 'you don't know what you don't know' Reflect, and be open minded with people who might not think the same way as you do because you never know what you might learn from that interaction – even if it is uncomfortable! Encourage students to think of learning as a process rather than an outcome

region-wide students' union and advocacy groups do not exist or are weak, other strategies for accessing guidance and inspiration, helping students see they are part of a bigger movement, and developing and mediating student-staff relationships may be needed, as least in the short-term. In the long term, student-facing sustainability organisations have an important role to play in linking between local pockets of activity, and these therefore should be strengthened.

8.9 Conclusion

In summary, this chapter has provided suggestions for how to address the impacts of transience and build resilience into student-led food gardens based on the analyses from Chapters 6 and 7, the suggestions made by the research participants, and the actions actually taken in the gardens. It is important for student-led food gardens to maintain an appropriate and well-managed space in which to run events and activities, nurture interest and creativity, create vibrancy, and take actions that bring value to the space. Efforts should be made to increase the visibility of the gardens and recruit widely. At times, this may be through appealing to potential participants' extrinsic motivation. Students should be supported by relevant longer-term stakeholders, which were, in this case, staff from the university, students' union, and National Union of students. Gardeners should be supported in creating organisational infrastructure, like having an elected committee of students. The gardens themselves should also be embedded in the wider organisational infrastructure, like campus maps and calendars. Gardeners should strive to document what has happened in the gardens and what is planned to happen next, and long-term stakeholders should support handingover documentation and plans between generations of gardeners.

Healthy relationships and social networks should be maintained within the gardens and between gardeners and other stakeholders, such as university maintenance teams.

Connecting with other gardens and organisations can also help gardeners feel like they are a part of something bigger when participating in their garden. Finally, it was important for the gardens to be adequately funded and resourced, especially resourcing for staff time to support the gardens.

Learning was one of the main ambitions of the gardens and the primary outcome of university community gardens, as highlighted in earlier chapters. However, learning and retention of social-ecological memory also play an important role in managing the impacts of transience, particularly though the development of 'action competence', and the use of physical forms/artefacts and external engagement and embeddedness as sources social ecological memory.

At the end of the chapter, I offer practical advice tailored to different stakeholders of student-led food gardens. Advice to students focuses on how to run the core activities and recruitment in the garden, with a focus on keeping a long-term perspective in mind, while the advice for university and students' union staff focuses on supporting student gardeners in a way that students' autonomy over the garden and how it is run is retained. The advice for the staff of the National Union of Students focused more on supporting the gardeners from a distance with gardening resources, providing inspiration, managing staff-student relationships, and helping gardeners see the big picture. The next chapter will also be futureorientated, but will trade the prescriptive orientation for a more reflective one, considering this study's methodology and potential future research directions.

Methodological reflections and future directions

Chapter 9

...a single cause does not lead to a specific affect and yet in research and evaluation we tend to behave as if it did.

Willem van der Eyken, 1996



Mushrooms Spring 2014

9.1 Introduction

This thesis has caused me to reflect considerably upon the methodology I used, which is where I will begin this chapter. The further lines of inquiry generated in this study will be presented in the second half.

9.2 Methodological reflections

I have already reflected on methodology to some extent in Chapter 7. I reflected on the usability of the resilience principles, and made some suggestions about how to improve the principles' fit with social science literature and intrinsic educational tradition. I also outlined how applying the resilience principles might be done with greater humility, and suggested that the synergies between the ideas of communities of practice and learning for resilience could be drawn out further in future. This section will focus on exploring some further methodological issues.

In writing these methodological reflections, I have attempted to be as candid and honest as possible. Authenticity is an important quality criterion for action research, and part of being authentic is being transparent. Attempting to obscure mistakes or hide behind academic language in order to bolster the credibility of research findings does not serve to deepen understanding or contribute to positive societal outcomes. Therefore, I urge the reader to recognise my mistakes, errors of judgement, and oversights, and use them to learn from and contextualise. As a researcher from Cook's (1998, p. 106) article on mess in action research said: "it's more systematic to admit that you're bumbling than to dress up your bumbling as something else. That's just plain dishonest." Mess in research allows for "getting below rhetoric" through creating "space for contesting interpretations" and methodological plurality (Cook, 2009 p. 288, 289). Scholars in the field of geography echo this, calling for

"open discussion of mistakes, failures, and setbacks [as they] should be part of our collective learning process" (Harrowell et al., 2018, p. 236). In contrast to more conventional views of how research should be presented, I contend that the systematic cataloguing of mess and failure is a mark of rigour. This section (and parts of Chapter 4) are intended to catalogue mess, mistakes, and even failure, to this end.

In this section, I will begin by explaining how keeping research diary enhanced the rigour of this study. This will be followed by a reflection on how my unique positionality as a student-staff member provided me with a unique opportunity for inquiry into the both the worlds of students and staff. The next section will unpack some of the challenges of conducting action research with transient participants, and following that I will reflect on the weaknesses and strengths of combining action research with a single embedded case study approach. I then discuss how I believe my naiveté affected my study, the discomfort an interviewee had with focusing on 'problems' in the first phase of the research, and missing stakeholder involvement. Before going on to discuss potential future research directions, I reflect on conducting causal loop mapping and applying the resilience principles late in the research process.

9.2.1 The changing research focus – and the value of a research diary

Keeping a research diary was a crucial part of my research process. This was also complemented by several 'proposals' I wrote at various stages in my research, for which the main purpose was to clarify my thinking about my study as a whole. I will now use a passage from an analytical memo to unpack how my research diary served my doctoral research. In an analytic memo I wrote while working on early drafts of my methodology, I reflected:

Part of what has struck me in this process is the fallibility of my own memory. My proposals are actually a really interesting way for me to see the progress in my

thinking. And what's really interesting, I think, is that before looking [...] at my research proposals I had a narrative in my mind of how my ideas developed which was much more linear than the reality. Upon reflection on these proposals, I discovered there were two main problems underpinning the initial framings of my research. In my mind I thought that I had arrived wanting to study the environmental impacts of community gardening, and through a participatory process I had realised that measuring environmental impacts would have very little relevance for student gardeners on the ground, and that they told me that primarily struggled with participation and their own transience. What actually happened, evidenced by my proposals, was much messier. I actually engaged with the 'transience' phenomenon much earlier than I remembered doing so. A proposal I wrote in October 2013 already stated a research aim as to: "investigate if and how community gardens tended to by transient populations, like university students, are affected by the mobility of their participants". [...] But in January of 2014, I had departed somewhat from these ideas in favour of developing indicators to measure the 'social and environmental outcomes' of student food growing projects as I wanted to "evaluate, in collaboration with community gardens affiliated with Student Eats, their social and environmental sustainability." I think the reality here is that I was struggling with understanding what makes research 'legitimate,' and had been highly influenced by Guitart et al.'s (2012) paper that noted most papers about community gardening came from the social sciences, and that while many purport that community gardens contribute to environmental sustainability, few actually demonstrate this. At the time I started writing about wanting to develop indicators for sustainability I was also beginning my own quantitative systematic review of university community gardens (Laycock [Pedersen] & Robinson, 201[8]). And I was also, I believe, feeling the aftershock of my Master's

degree in Environmental Studies and Sustainability Science at Lund University, particularly from a module about 'Sustainability Science.' I very much agreed with the main tenets of sustainability science – a discipline aiming to be transdisciplinary, to break down the divides between "scientists from different disciplines and nonacademic stakeholders from business, government, and the civil society in order to address sustainability challenges and develop solution options" (Lang et al., 2012, p. 26). At the time I didn't feel this, but I increasingly do now, the language of 'Sustainability Science' operates much more comfortably within the natural sciences than the social sciences. Even the term itself – Sustainability Science – seems to imply that it is a Science (e.g., geoscience, chemical sciences), rather than a field of study (e.g., gender studies). To me, tapping into the language of 'Science' reads as a strategy to legitimise the field ('It is a real science! It is rigorous!'), which actually undermines its transdisciplinary ambitions through subjugation of some of forms of social sciences and the humanities.

But coming back to the topic at hand, the first main problem with my research framing was that I was attempting impose a quantitative (and environmental scienceinfluenced) approach on a case in which messy, complex social phenomena were at stake.

As is evident from this passage, my research diary and cataloguing of progress served several purposes. On a mundane level, I collected factual information, like my changing research aims and rationale. Collecting this meant that I was more accurately able to chart development of my thinking. I was then able to use this to construct my methodology chapter in a more accurate way than I would have been able to otherwise. This factual information also served my reflective practice. When revisiting this information after several years, it challenged my present perceptions, thereby prompting me to think critically about what

happened, my understandings of the issues at play, how they have changed, and what I have learned. As Engin (2011, p. 303) put it, "the opportunity to re-read and interact with my thoughts was also a strong mediator in understanding my role of researcher and the research process."

Research diaries are thought to be especially useful for novice researchers, enabling us to scaffold our own understandings through using a research diary as "repository for thoughts and reflections" and "as a written account of [our] research [journeys]" (Engin, 2011, p. 303). As can be seen towards the end of the above passage, I reflected on some of my (perhaps unconscious) assumptions about what 'science' is and how this might have affected my initial assumptions that influenced my choice of how to frame my research. In doing this, I have been able to be more transparent (Clark, 2009) about factors that may have influenced my analysis and framing. While not all of these factors were unpacked in the above passage, I revisited my research diary and catalogue of documents whilst writing the section about my methodology to enable me to be as accurate and transparent as possible.

9.2.2 Gaining privileged access to the separate worlds of students and staff

In Chapter 4, I discussed my boundary spanning (Williams, 2002) role as a studentstaff member and how this gave me privileged access to the different worlds occupied by students and other stakeholders from the universities, students' unions and the National Union of Students. I would like to note that this boundary spanning role is a bit more complex than a by-the-book practitioner-researcher role because I was a practitioner in two senses: a student practitioner and a sustainability professional practitioner. My focus here is on the boundary spanning between these two practitioner roles. As a student, I took courses, was a member of the students' union, took part in student activities, and was unpaid. As a staff member, I was also employed, interacted with other university staff in a professional capacity,

and built my schedule around a conventional work-week. Many doctoral students have teaching responsibilities, but it is much less common for them to have responsibilities with administration, student support services, or the campus grounds/built environment, in the way that I did. I believe this dual role was pivotal for providing insight into the interface between these two worlds that student-led food gardens occupy. I also believe that these crossover roles could be beneficial for research into other student-focused campus initiatives. Partnerships between student advocacy organisations, like the National Union of Students and universities would well-suited to conceptualise and/or fund such initiatives.

These dual roles, and other complex configurations of positionalities that such doctoral researchers may find themselves in, require a high degree of reflexivity to bring unconscious assumptions to light and allow for inquiry into potential ethical quandaries. As such, the methodologies employed by these boundary spanning researchers are also crucial. Reflection on my positionality in my research diary helped me to find ethical ways to exploit my boundary spanning role, as well as highlight my biases and assumptions. More traditional research approaches that focus on distancing the researcher from the study could actually prove to be more problematic for researchers with these complex roles because of unproblematised positionality that could result in bias or abuse of power (England, 1994). Even in quantitative inquiry, reflexivity "can be an effective, ongoing means of critically reviewing work, process and researcher development" (Walker et al., 2013, p. 38). Some strategies that might be useful for this are self-interviews (Keightley et al., 2012), examining one's positionality (Coghlan & Brannick, 2019), and writing reflexive 'asides' within one's thesis (Forbes, 2008).

9.2.3 Action research with and about transient partners

As highlighted in Chapter 2, there have been calls for action-orientated research in the field of sustainability, a research approach that requires prolonged engagement. However, the world is globalising, and people are moving and travelling at rates that are unprecedented in human history. For example, career change is more common than it used to be and episodic volunteering is on the rise (MacDuff, 2005). This means that understanding how to use action research when working with transient partners or participants is of critical importance. This study has provided some insights into the challenges of doing action research with transient participants and how these might be navigated.

I found several key challenges in doing action research with transient participants. They were (1) a superficial understanding of issues by volunteers as a result of short term engagement, resulting in difficulties identifying or theorising the issue to research; (2) difficulties ensuring the implementation of 'actions' or 'interventions' given those involved in the 'planning' phase had often left by the 'action' phase; (3) a lack of capacity to engage in research given the ongoing challenges to maintain an adequate level of participation; and (4) that participants reported that the introduction of the researcher changed the group dynamics, prompted critical reflections that had not previously taken place, and improved continuity, thereby preventing a 'naturalistic' inquiry into transience as a phenomenon.

9.2.3.1 Transient participants lack insight into their own transience

Part of what makes this study unique is that it is not only conducting action research with transient participants, but it is actually researching the phenomenon of transience itself. Most student gardeners had superficial understandings of the issues faced by student-led food gardens as a result of short term and irregular engagement (as seen in Chapters 6 and 7). This resulted in difficulties collaboratively identifying, theorising, and evaluating the issue to research. This was particularly evident in the photovoice component of the study. I would characterise the use of this method as relatively unsuccessful in terms of how the participants engaged in answering the questions posed. Most of the responses were off-topic. For example, participants took submitted photos and captions that described things they enjoyed about the garden or particular features of the garden (see Figure 35). As such, these did not provide insight into the issue of transience or participation. The discussion about the photos did provide useful reflections, however the most useful parts of the conversation were disconnected from the photos. The challenge of participants providing off-topic photos has been mentioned in the literature about photovoice (e.g., Gaboardi et al., 2018), however most articles giving an overview of photovoice and challenges associated with it do not mention this (e.g., Catalani & Minkler, 2010; Powers & Freedman, 2012; Povee et al., 2014).



I absolutely LOVE having the opportunity to be a part of something as special and as colorful as gardening at Keele is. Being an international student from the United States and having always loved the outdoors, I fell in complete adoration of the individuals who made up the community as well as the ability to tend to the Earth just as I did back home. xo

Figure 35. One of the off-topic photovoice submissions.

A mantra of many of those who engage in participatory research is that people are experts on their own lived experiences (e.g., Einerson, 1998; Sampson & Wills, 2013). This was the thinking that underpinned my understanding of participatory research, and as a result I turned to the research participants to help me better understand transience and how it might be managed. However, in doing so, I found that the participants who were the most transient struggled to understand how their own transience impacted the garden. On the other hand, participants that provided nuanced insights were typically ones that would be considered the least transient. While I consider it to be true that people are experts of their own lived experiences, from my research I came to see that people are not always experts on social milieu they operate within. I reflected on this in my research diary:

In terms of doing action research with transient participants, I think it's a real challenge, because the people who end up being 'key informants' are the less transient participants. So it's actually the more permanent participants who have the most voice. But similarly, they end up being the ones to say the most interesting things. For example, [one of the long-term mentors] certainly isn't transient, but [they have] so many interesting observations. But at the same time, [they have] obviously spent a long time thinking about the transience of participants. Whereas the genuinely transient participants haven't reflected on it, on their role, and on the project in anywhere near the same depth. So with each person I interview, they say more or less the same things.

As such, when conducting participatory research with transient participants, gathering perspectives from less transient stakeholders may be necessary. Transient stakeholders are able to provide insights into their lived reality (that is, psychological phenomena). However, equally important are the perspectives of less transient stakeholders because they are better able to provide insight into social and organisational dynamics at play (that is, sociological phenomena) that transient stakeholders may not be able to perceive. As such, researchers may benefit from considering which perspectives will best help them answer their research

questions, depending on the aims of the research study and whether the focus leans further towards the psychological or the sociological (or otherwise).

9.2.3.2 Difficulty providing 'value' to research participants

One of the main features of participatory action research is a focus on reciprocity between the researcher and participants (Maiter et al., 2008). As such, I intended for my research to deliver benefits to the student-led food gardens and their participants. As was outlined in Chapter 8, the benefits of the 'actions' taken were mixed, and it was hard to tell if the 'actions' actually had a meaningful impact. However, because students were only involved in the gardens for short periods, it was difficult to deliver 'value' for them because their engagement was so short. For example, some students' first engagement in their university's garden was through my research, therefore it was hard for me to find ways to 'give back.' Students were incentivised to commit time to the research because they wanted to help me and there were usually snacks provided, however seeing reduction of the negative impacts of transience was not possible for most students.

This said, a long-term mentor reflected that they appreciated engaging in this project because it gave them the time to reflect on the challenges faced by their garden and inspired action that wouldn't have taken place otherwise:

So it's been kind of useful to have this engagement. Just full stop. To have some kind of sense of outside perspective being brought in. Whatever that perspective is. It's just helped us to think about our own actions and what actions might be best to address the problems we're facing. Rather than, which has kind of been the norm so far. Which is that we are acutely aware of the problem... but not... really... ever really doing anything about it. [laughs] It's just, kind of, making that part of the experience of the garden. We all struggle in summer. 'Well, don't you know the agricultural calendar is the inverse of the academic calendar!' And, you know, I just find myself saying that,

for, like the millionth time. And, it's been like, well I guess we'd better do something about that then. And I have done things off the back of this research that I wouldn't have otherwise done.

As such, the study was reportedly beneficial to some participants, at least to some extent. Given that this stakeholder had been involved for a longer period of time (the full length of data collection during this study) they were able to identify some more emancipatory outcomes (*helped us to think about our own actions*) and some more pragmatic actions (*I have done things off the back of this research that I wouldn't have otherwise done*). However, given that most of the research participants were much more transient, many did not see or experience these outcomes.

However, for many there was a sense of doing something for the greater good, and therefore the need to somehow benefit from the research directly didn't matter. One gardener spoke about another gardener who would take part in their garden and harvest (i.e., take home produce) from another without any guilt, under the belief that what goes around comes around: as long as you put in the work somewhere, you can reap the rewards from somewhere else. This willingness to accept ambiguity in the reciprocity of giving and getting from the garden they took part in mirrored the willingness to accept ambiguity in the reciprocity in the research process. This is an especially useful way to think about this for research with transient participants because the benefits of research can come with a delay. As one student observed, *sustainability is about passing to future generations. As students, we are, in generations, coming and leaving, and coming and leaving* (Self-directed volunteer). Like with the sustainability challenges we face, the benefits of today's actions may not benefit those who take them; the benefits may only come to future generations. This sort of thinking may be useful for action researchers when thinking about reciprocity in their research.

Our present realities are shaped by those who created the foundations of them, and future realities will depend on the foundations we lay for our descendants, whether it be our global ecological reality or the reality of a student-led food garden. This idea of intergenerational justice is important for participatory research with transient participants, because today's participants may not benefit from the research, though people in future may. As such, the evaluation of such research should take into account not only the benefits of actual participants who took part, but also potential future beneficiaries.

9.2.3.3 Difficulties implementing 'actions'

A more pragmatic challenge was ensuring a smooth transition from the 'action planning' phase to the 'taking action' phase as most of the participants that helped plan actions had left the project by the time they were expected to take the actions. This is in spite of the fact that the execution of actions were planned to be executed during term-time.

After my realisation that I had been too prescriptive in determining the direction of my research in the early stages, I was particularly keen to make sure I was genuinely participatory in the later stages of the research. I interpreted this as meaning that the research participants would take actions with my support so that there was ownership by the participants, as is often advocated in action research (Löfman et al., 2004). However, students' short-term engagement, combined with a lack of insight into the implications of their own transience, meant that it was difficult to ensure actions were taken and to produce meaningful outcomes. Furthermore, there was a lack of urgency to take the proposed actions by the more transient participants. This meant that many of the planned actions were not implemented. However, I still did not take any actions myself because I felt it was important for people participating in the research to have ownership over the actions that were taken and not feel like they were imposed by an outsider. In hindsight, I also believe that this was due to my unconscious discomfort operating as an 'activist' within the garden, in tandem with

being a 'researcher.' In hindsight, I can now see that, especially at Keele, I was in a much better position to take meaningful actions to address challenges associated with transience than many other gardeners were. Having taken actions would have also informed my reflective process, potentially contributing to a higher level of understanding.

Another issue in doing action research with students in an informal setting was that participation was highly varied and irregular. Most of the suggested interventions would have benefited from a 'little and often' approach, however students tended to concentrate their activity into bursts of engagement, followed by periods of absence and disengagement.

Therefore, I would also advocate that researchers should embrace their agency in their practitioner or activist role, if they have one. The researcher taking action in partner with other project stakeholders need not undermine other stakeholders' ownership if it is done with their blessing and best interests at heart. In working with transient participants, actions taken by a researcher may even be the most effective type of action that can be taken given that they may be the actor with the most capacity to follow through the process from planning to taking action. This is common practice within educational action research, when the researcher inquires into their own teaching practice.

9.2.3.4 Lacking capacity to engage in research

Gardeners also lacked capacity to engage in research given the already ongoing challenge of maintaining an adequate level of participation. In my research diary I reflected that:

There needs to be a kind of soft-touch approach to doing research working with students involved in these sorts of projects because they already struggle for participation and are time-strapped as is. So engaging them in a group activity that is not explicitly gardening-focused (or an informal social event) risks putting potential

of.

This worry was not unfounded. Because, as I wrote in my research diary:

At Keele we actually lost one person who was a regular because I tried to organize a workshop one week, which flopped (because no one really came), and so re-ran it again the next week. The next week was better attended, and worked, but the person never came after that second week of workshops because it wasn't what he was there for. He just wanted to garden, I think, and he probably was coming to take part because it wasn't an academic thing. And my workshop made a fun, relaxed activity an exercise in critical thinking and... well, work.

In some cases, it was possible to work around the lack of capacity to engage in research by offering some of my own time to help with gardening tasks and by using go-along interviews (Kusenbach, 2018) so that participants' time was not taken away from their core tasks. Furthermore, I aimed to 'give back' to participants through my research, though, as highlighted above, the results of this were mixed. As such, the first of the main strategies that can help address the challenge of participants not having time to engage in action research are to plan research activities that are in synergy or, at least, do not compete with existing activities. The second is to ensure that the action research undertaken provides value to those taking part. This may be through the outcome(s) of the action research, but at the very least, can be through the provision of opportunities to critically reflect.

9.2.3.5 The researcher as 'intervention'

Am I actually gathering data on a truly student-led initiative if I am facilitating interventions? And the answer is probably at least in part, no. (My research diary) Participants reported that my being involved in the gardens changed the group dynamics, prompted critical reflections that had not previously taken place, and improved

continuity. All of these are, of course, positive outcomes. However, I was concerned this had prevented a 'naturalistic' inquiry into transience as a phenomenon. That is to say, I was involved in the gardens in order to better understand how transience impacts them, and by in getting involved I disrupted the impacts of transience, making it difficult to study them *in situ*.

The impact I had at Keele's garden was most profound because I tended the garden over the summer and had historical knowledge of the space which gardeners would not have had without me. Furthermore, my longer-term engagement imbued me with a higher degree of authority, even though this was something I resisted and attempted to subvert. At the garden at Warwick, the main impact of my visits appeared to be to stimulate periodic reflection that spurred on action. However, given I was much less involved at the garden at Sheffield (due to the low participation levels present), I never met the same person twice. Therefore, at Sheffield, my involvement did not result in any noticeable changes.

Studying three sites meant I was able to triangulate my findings from these different situations to understand the extent to which my presence prevented a 'naturalistic' inquiry of the impacts of transience. Ultimately, this concern appeared to be less problematic than it seemed on the surface, as the findings from the different gardens were not substantially different in spite of my engagement. I reflected on the reasons why in my research diary:

I think my interventions are artificial – however, I think they mimic natural processes that happen in gardens. Outsiders do come in and start up projects. How is this any different that a highly engaged student coming in and running a series of workshops over the course of a year? How is it different that a member of academic staff getting involved and making suggestions at various intervals? Or a skilled and knowledgeable community member? The nature of these gardens is that they have transient participants... people like me being one of them!

My involvement did impact the gardens, and in fact, making change in a real-world context is the intention of an action research inquiry. However, I would suggest that my involvement did not prevent me from exploring transience as a phenomenon because my engagement in the gardens were within the realm of what was normal for a student-led food garden. As highlighted in the passage from my research diary, my involvement may have been longer than many of the other gardeners', but it was not outside of what could be considered normal for a student-led food garden in a university.

In action research, first-person inquiry is thought to be important for enhancing reflexivity using an 'upstream' approach, enhancing reflexivity with regard to researchers' "basic assumptions, desires, intentions and philosophy of life," as well as through a 'downstream' approach, enhancing reflexivity with regard to researchers' "behaviours, ways of relating and their action in the world" (Coghlan, 2019, p. 7). However, action research in the field of sustainability (e.g., Bradbury, 2005; Leclerc, 2009; Hallstedt, 2017) has historically tended not to engage in first-person inquiry to the extent that has been seen in other disciplines. I used a research diary to attempt a more holistic inquiry by engaging in first-person inquiry, in addition to second- and third-person inquiry which are more common in action research in the field of sustainability.

In the action research tradition, within each type of inquiry (first-person, secondperson, third-person) there should be both action and reflection in order to engage in all four forms of knowing (as outlined in Chapter 4). Therefore, I advocate that action research in the field of sustainability should engage more in first-person inquiry to complement second- and third-person inquiry in order to attend to the researcher's own "thinking, valuing, way of learning, and behaving" to produce more authentic research (Coghlan, 2019, p. 191).

9.2.4 Using an embedded case study approach while conducting action research

Although not always used in conjunction, action research and case study methodologies are thought to marry well because they both involve a detailed analysis of a particular situation, group of people, or individual (Corcoran et al., 2004). However, the use of multiple, distinct sites within an action research project (as in this study) is much less common, although has been done previously (e.g., Pereira & Vallance, 2006).

I chose to include three gardens in this study because of the expected instability of student initiatives. This decision was affirmed by the fact that the garden at Sheffield ended up being an unable to participate in an action research cycle. However, action research is a context-specific methodology. It cannot be expected that different groups in different contexts and locations would undergo the same sort of process and evolution. Furthermore, if research is intended to be participatory, as in this study, the general coherence in the research agenda between sites may be difficult to maintain if participants at each site have different ambitions. Indeed, all three gardens did undergo their own journey during the research. At Keele and Warwick, different actions were implemented, and no action was undertaken at Sheffield at all.

My engagement at the three gardens was also uneven which affected the data. My engagement at Sheffield and Warwick was considerably less than at Keele by virtue of the fact that I was based at Keele. This limited the passing insights I gleaned by participating regularly in their gardening sessions, which may have resulted in an over-representation of Keele in my results and analysis. I, however, reflected about this in my research diary in order to consider the extent to which my conclusions were Keele-specific and to what extent they were relevant for the other two gardens. Furthermore, I attempted to buffer potential bias by

providing initial results back to the participants (Appendix E) to receive feedback and check the accuracy of my interpretations.

In contrast, a single, holistic case study at only one site (Yin, 2003) would have allowed for deeper immersion in the inner reality of the garden that I was not otherwise able to because my energy was spread across three sites. However, it would have been difficult to elicit more information from the pool of interviewees. There was a clear case of theoretical saturation (Bryman, 2016) due to the fact that most gardeners were involved in the gardens in a time-limited capacity and therefore had limited insights into the question of transience. Data collection would have therefore needed to focus on other methods, such as participant observation, or accessing new data sources within the projects, such as Student Eats' social media.

Since findings of case studies "will be more robust if there are more mediating variables" and more sites provide more variables, the findings from several sites are likely to be more robust than a single-site action research case study (Pereira & Vallance, 2006). This can allow for a broader scope for analytic generalisation. My interactions with gardens beyond Keele also spurred me to critically reflect about the context specificity of the problems and suggested solutions through the juxtaposition of the different gardens. For example, the role a doctoral student played within the project may not have been as apparent without the juxtaposition between the garden at Sheffield (where there were not any) and gardens at Keele and Warwick (where there was). As such, using a single, holistic case study would have required a greater commitment to reflexivity because the lack of interaction with other gardens would mean fewer opportunities for comparison.

In summary, it is difficult to say which would have been a preferable approach in hindsight: focusing on a single site or multiple sites. Having multiple sites meant that it was challenging to create a coherent research project because the different sites were each on

their own journey. Furthermore, it also meant that my high level of exposure to the garden at Keele compared to Warwick and Sheffield may have led me to over-generalise the Keele experience to other sites. However, having multiple sites meant that there was a buffer in case of drop out, lowering the risk of having inadequate data for my doctoral work. It also meant to there were more mediating variables at play, and more points of comparison which meant I was able to provide a more robust analysis.

9.2.5 Implications of early career researcher naiveté

Doctoral study is just that: a form of study, studentship, or apprenticeship. All research can be thought of as a process of learning, however, doctoral research in particular is deliberately designed to *train* a student to be an independent researcher. As such, a doctoral student can be expected to approach research with a certain degree of inexperience and naiveté. Having completed my study, I certainly feel this was present, especially in the early stages of my research. This manifested through how I articulated the problem of transience in my research journey, practiced journaling, and constructed the aims of my research.

When I articulated the problem of transience, I often used the term 'transience' explicitly in interviews and struggled to make clear what I meant by it. This resulted in difficulties with some research participants understanding what I was asking them (however, this was further compounded by the fact that many students struggled to understand how their own transience in the university community impacted their garden).

Furthermore, as a relatively new action researcher, I have been less practiced in the process of journalling and archiving my experience. I mainly focused on reflecting upon experience and exploring my own understandings of the issues in question. In hindsight, I would have made a more consistent practice of journalling with a focus on reflection on the

data and as "an analytic tool where the data can be examined and analysed" (Coghlan & Brannick, 2014, p. 34). I believe this would have resulted in better clarity in my understandings of the themes in the data at an earlier stage through forcing me to engage with the data more analytically rather than descriptively in the analysis process.

I would also have engaged in a more participatory approach to constructing the aims of the research from the start. Prior to my doctoral research, I had little experience engaging with people in a professional capacity, as research required me to do. Many of my ideas of what a good research question was and what I wanted to examine in my research were fairly rigid. This was exacerbated by the tradition of first reviewing the literature and applying for ethical approval before engaging research participants rather than co-constructing the research agenda. Engaging with staff from the National Union of Students and student gardeners at an earlier stage would have helped mould my aims into a more contextuallyappropriate research agenda.

Another way my inexperience impacted the elegance (or lack thereof) of my methodology was in the ethical review process. This thesis was my first time undergoing a process of ethical review. I learned a number of things in the process. I learned that often for action research studies, a multi-phase approach to applying for ethical approval is preferable. One of the crucial challenges relating to ethical review of participatory research is that often methods are not decided until first engagement with the community of interest. However, first engagement with the community of interest usually requires ethical approval. For this reason, Khanlou and Peter (2005) suggest that researchers may need to seek ethical approval several times during a project, as the methodology and focus of the project evolves. However, when I applied for ethical clearance, I attempted to apply for my entire study in spite of the fact that how the later stages of the research were to manifest were dependent on how the earlier stages played out. I produced a single information sheet and consent form for

participants to sign for activities taking place years after participants consented. The form was long and contained a lot of information, which raised a pragmatic issue, which was that someone who might have only taken part once had to wade through pages of information that was not necessarily relevant to their participation in the study given that most participants only took part in one activity. However, this also raised ethical issues in terms of whether the participants could reasonably be expected to consent to something so far in the future. I intuitively managed this by reiterating information from the information sheet and consent form to participants and sought out verbal consent when they were taking part in an activity a considerable time after they had reviewed the information sheet and signed the consent form. I also amended my application twice (approval letters can be found in Appendix D) and created new information sheets and consent forms for participants of the fishbowl discussion and photovoice component.

Since undergoing the ethical review process at Keele my first time, there have been changes to simplify the administrative process, and inclusion of a special question about whether the research is 'action research'. This is a good step forward. A useful next step might be to offer some resources with tips and considerations for researchers planning to conduct participatory and/or action-orientated research. Based on my experiences, some useful guidance would include to consider whether it would be more appropriate to break studies into several ethical review applications, and the length of the information sheet and consent form(s). Many of these issues may seem intuitive to more experienced social researchers, however special guidance about navigating ethical review in introductory action research texts and for postgraduate supervision of action research would be valuable. For ethical review boards, it may also be worth considering a different application strategy for participatory and action-orientated research projects. Ethical review panels should also consider that "the potential burden and cost of multiple reviews could be minimized through

the development of guidelines and procedures for participatory action research that facilitate an expedited review process for each cycle of the research" (Khanlou & Peter, 2005, p. 2337).

9.2.6 Problem-focus discomfort

Action research is a solutions-focused approach to research. Its aim is to make positive change. However, in order to create solutions, a problem first needs to be identified and understood. Therefore, the first stage of action research is a reconnaissance to explore an issue to fix or improve. When presenting the initial findings of this stage and the action planning (see Appendices E and F) to a staff member at the National Union of Students, they expressed considerable concern about the focus on problems. The interview began somewhat uncomfortably as the staff member found the framing to be inappropriate. They were concerned the participation in the garden at Sheffield had dropped off dramatically after my research began, and speculated that it may have been caused by focusing too much on negative things during the 'constructing' phase.

Our conversation from there mainly centred around the documents (in Appendices E and F) I had handed the interviewee. They asked many clarifying questions, and as the interview went on, I clarified and elaborated on the text. As we moved into the section about solutions, the interviewee changed their tone and seemed less threatened by the document I produced. They then began to open up, asking if they could take notes, and offering some suggestions of how the research question might be reframed.

Eventually, we came to a common understanding of how the framing of the research ought to be, however, this encounter made me worried that my framing of the research question had negatively impacted the people who had participated in the study or the gardens they were involved with. Therefore, I asked one of the gardeners who I felt would be candid with me if they thought the framing had been problematic. They said:

I think it's fine [to talk about problems], and I share your perspective that... all research, especially participatory action research, should have some degree of looking for problems. Or should have a critical approach embedded within it. Otherwise what's the point? As far as how that's played out... you know, for someone like me or [a student leader] who've kind of been working on more on the inside, on the organisational part... those problems are ones that we may not have talked about that readily. Or... people just don't see, and it doesn't just come up because why would you talk to everyone about the problems of running the space, when we always want to say how good it is. So those moments when people have been invited in who are more on kind of the periphery, or new to the project. To talk about those problems in a systematic way... I think it's really useful. Because people are like, 'maybe it doesn't just materialise here.' This is an important thing for people to understand. And I think, actually, people who have... those people who participated in those workshops. Or who are otherwise getting involved and want to see the project succeed... they, uh, that actually deepened their appreciation of what it is. And gave them a, kind of, an impetus to do more. So, I think... I don't think it's like, 'oh we started dwelling on problems...' and then everyone kind of got really down and left. Not at all. I mean, people left, but it wasn't for that reason. It was reasons of leaving the country or whatever. Yeah, leaving [the university]. I think, again, that helped. And to do it in a kind of... I stressed, I felt it was done in this systematic way, where we identify problems, and then the broader horizon is kind of moving towards... kind of a better system. Rather than being like, 'hey guys, let's just talk about our problems openendedly.' It wasn't that. That might have been... a problematic way to do it. But that wasn't what we did. So... it was couched in a framework that people could understand

was.

This said, this quote represents the perspective of one individual, and therefore others may have felt differently about the focus on problems in the early stages of the study. My own experiences, however, echoed what this participant said. In addition to providing the opportunity to reflect as mentioned in the above quote, I also sensed that some participants valued the opportunity to vent about their frustrations. When recruiting participants or running gardening sessions, lead gardeners often needed to create a fun, friendly environment, in spite of the fact that they may be frustrated, annoyed, or exhausted. Interviews, in particular, seemed to provide a space where they could let out some of their negative feelings. This said, the workshops did seem to frustrate a small handful of gardeners as the workshops distracted from the main gardening tasks. However, this frustration was not to do with the focus on problems, but the research in general.

Regardless of how the research framing impacted the gardens, the incident raised an important question, which is, to what extent is it constructive to reflect on problems and when does it become unhelpful? I reflected on this in my research diary:

I guess what's good about the 'dwelling on the problem', so to speak, is that people are very quick to come up with solutions and how to fix the problem, as they see it. And I think slowing down and sitting with the problem for as long as we do, did benefit the research quite a bit. And I wonder if there's a bit of resistance from people to dwell on the problems because it is demotivating and depressing and they worry that sitting with that will cause even lower participation. Which might be true... but by sitting with the problem I feel like there is a much better understanding of transience and whatnot. And there is a danger of jumping the gun and suggesting solutions that aren't really related to the root of a problem.

Indeed, an unwillingness to talk openly about 'negatives' may result in a lack of understanding of problems and their causes. However, the concerns raised by the interviewee from the National Union of Students about overemphasising negative aspects of student-led food gardens are genuine as no one wants to climb aboard a sinking ship. The question we must then ask is, how can we talk about these 'negative' things in ways that are helpful²⁰? Perhaps one way is to find ways to destigmatise failure, as failure has considerable stigma attached to it (Shepherd & Haynie, 2011). Such stigma may cause a fear of failure, which has been associated with surface learning and "reduced understanding" (Boyle et al., 2007, p. 302). Therefore, it is possible that creating environments where people can talk about 'problems' and 'focus on the negatives' without fear can contribute to learning.

9.2.7 Missing stakeholder involvement

When designing my study, I included participants of student-led food gardens, staff members of the National Union of Students involved in the Student Eats scheme, and staff from the organisations providing funding for Student Eats. The latter were not possible to interview, so I focused mainly on participants of student-led food gardens, staff members of the National Union of Students involved in the Student Eats scheme. I did not consider including university or students' union staff as the gardens were chosen because they were student-led. I therefore assumed that contributions that university or students' union staff would be able to offer would not be particularly helpful to understand how to manage the impacts of students' transience. In two of the gardens, the relationship between students and staff was not a subject of focus, likely because it was positive and/or unremarkable. However, in one of the gardens, the relationship between staff and students was tension-ridden.

²⁰ In this thesis, I have deliberately chosen to use the term 'problem,' both in order to be accurate – some of the things I have written about are problems – but also because I believe it is necessary to talk about problems openly in order to destigmatize failure.
Interviews with the staff at each of the universities could have provided more insight into what made the relationships what they were, and particularly, could have provided further insight into the roots of these tensions. This is a considerable limitation of the study, because although staff involvement was very limited, the results indicated that the relationship between staff, especially the operational estates team, and students was important. It would therefore be useful to understand the 'other side.' Did staff perceive friction in their relationships with students? If so, how?

However, as a university staff member myself, I was able to reflect from this perspective in my research diary. Staff from students' unions also took part in the fishbowl discussion at the Student Eats conference. These participants were not affiliated with Keele, Warwick, or Sheffield and therefore were not able to offer contributions related to the specific dynamics at those institutions, however, they were able to contribute more generally from the students' union perspective. As such, students' union and university staff perspectives were represented, but were not as fully integrated as they could have been.

9.2.8 On the *ex-post* causal-loop mapping and use of the resilience principles

When I began my doctoral research, I did not intend use resilience as a theoretical framing or use causal loop mapping to analyse the data. The decision to use a causal loop diagram came at the end of the first round of data collection, and the first iterations of data analysis and even early drafts of this thesis were written before I brought in the resilience principles. When I was collecting the data, I therefore did not actively seek out evidence of alignment or misalignment with the resilience principles, nor interrogate the causal relationships between different variables to the extent I might have done had I known I was going to create a causal loop diagram from the beginning. It is therefore possible that I missed opportunities for probing and interrogating the connections between the different variables.

As a result, I chose to reduce the number of connections depicted in the causal loop diagram in order to illustrate only the most robust and important connections so as not to overstate my contribution to knowledge. It is also possible that the assessment of the vulnerability in Chapter 7 could have been more complete had I employed the resilience principles from the beginning. This being said, there were still many reflections in my analytical memos and my research diary that were rooted in resilience thinking in spite of having brought the resilience principles in at a later stage of my thesis. I attribute this to the fact that systems thinking and resilience was a key component of my undergraduate and taught postgraduate studies. I also revisted the raw data (as described in Chapter 4) to see if there was anything else that could be picked up when a resilience principle perspective was applied. Therefore, in spite of this *ex-post* use of the resilience principles, I am comfortable stating that my assessment of the vulnerability of student-led food garden using Biggs et al.'s (2012, 2015c) resilience principles is as complete as could be expected for a complex social-ecological system.

9.3 Future research directions

Elsewhere in this study, I have suggested some future research directions to take under consideration. These are to:

- Explore the extent to which the typologies from Figure 14 hold up under scrutiny with a larger sample of cases and participants,
- Explore the extent to which community gardens provide environmental benefits,
- Experiment with Biggs et al.'s (2012, 2015c) resilience principles in different settings and for sustaining different social-ecological systems and constructs (e.g., organisations, policy environments, governance regimes, shared mental maps, women's rights) to see if it is possible to identify axes of diversity, connectivity, and feedbacks that are common for particular systems, and

• Untangle how student gardeners understand terms like confidence, initiative, responsibility, ownership, or motivation and explore how they correspond to definitions used in academic literature.

In this section I will present some further lines of inquiry that were prompted by this study that should be considered in future research. These are the long-term implications of attitude and behaviour change initiatives in tertiary education; what and how students are learning in student-led initiatives; critical perspectives on pedagogy in urban agriculture; failure in sustainability; intersectional interrogations about student-led food gardens, studentstaff relationships in university sustainability initiatives; the ways in which resilience might be at odds with other aims; and transience in sustainability organising. I will begin by explaining why research exploring the long-term implications of attitude and behaviour change initiatives is needed.

9.3.1 Long-term implications of attitude and behaviour change initiatives in tertiary education

Judging by research from the National Union of Students' sustainability team, there are already high levels of pro-sustainability attitudes amongst students (NUS, 2018). However, they have also shown that most students want to know more about sustainability and students want their universities to "actively incorporate and promote" sustainable development (NUS, 2018, p. 2). The extent to which attitude and behaviour change resulting from taking part in student-led food gardens stay with participants beyond their tertiary education, however, has not been demonstrated empirically, even if logical theoretical conclusions have been drawn using the habit discontinuity hypothesis (Verplanken et al., 2008). At a theoretical level, the habit discontinuity hypothesis is a compelling argument for the National Union of Students' sustainability behaviour change initiative. However, there is little peer-reviewed literature based on empirical assessments of the extent to which the habit discontinuity hypothesis holds up in the context of pro-sustainability behaviour change interventions in higher education.

There is some evidence of university gardening schemes having positive effects on sustainability attitudes and behaviours. Several of the articles about food growing university community gardens reviewed in Chapter 5 demonstrated that taking part in these gardens enabled participants to develop healthy, ethical and eco-friendly attitudes and behaviours towards food. For example, Wharton and Harmon (2009) said that community supported agriculture membership "allows members to deepen their relationship and involvement with food" by decreasing "the number of shopping trips to more conventional venues such as the supermarket" (p. 114). Members of community supported agriculture may also change their food behaviours (such as choosing to eat together as a family or experimenting "with the art of composting") because of the lack of pre-sliced, -washed and -packaged foods provided to them (Wharton and Harmon, 2009, p. 114). In a number of the articles, the change in attitudes and behaviour related to sustainable food was as a consequence of other outcomes. For example, several of the reviewed articles (Falk et al., 2005; Parr and Van Horn, 2006; Wharton and Harmon, 2009; Kobayashi et al., 2010; Datta, 2016) said the sustainable university community gardens they wrote about provided educational opportunities about sustainable food systems, and such education may lead to changes in sustainable food attitudes and behaviours. However, many articles only discussed (rather than demonstrated) the benefits associated with engagement, attitudes, and behaviours for sustainability, indicating that there is a need for more evidence to back these claims.

There is also evidence that volunteering in a university community garden can improve self-efficacy (the belief in one's capabilities to take action on a given problem) (Pierce & Seals, 2006; Hoffman et al., 2007; Teig et al., 2009; Evans et al., 2012). Given that a lack of self-efficacy is a barrier to changing sustainability-related behaviours (such as

recycling), it may be possible that this is also the same for ethical and eco-friendly food behaviours. This is echoed by Delind (2006) who suggests that local initiatives that can connect to people emotionally, sensually and expressively, like community gardens, "may best promote long-term attitudinal and behavioural change in relation to food" (Turner, 2011, p. 513). Improved self-efficacy through food growing projects could increase self-efficacy in other areas of a person's life, increasing their ability to make wider, positive changes for themselves, their communities, and the environment.

A variety of studies have been conducted explicitly about how community gardens impact attitude and/or behaviour change with regard to the environment and participants' food choices (Somerset & Markwell, 2009; Turner, 2011). The results are mixed, and the reason may be that different participants are affected in different ways (Middlemiss, 2011). Middlemiss (2011, p. 265) found that "those who are new to sustainability and who are actively involved in cohesive groups, which are specifically targeting their lifestyles, are more likely to report substantial changes" in sustainable behaviours than those with histories of engagement in sustainable lifestyles or behaviours. Therefore, the capacity of growing projects to contribute to enhancing pro-environmental behaviour is contingent on (1) recruiting participants not already active in sustainability or environmentalism, (2) the cohesiveness of the group, and (3) the beliefs, behaviours, and attitudes of the group.

The implications of this for student-led food gardens are that the transience of their participants may both enhance *and* limit the capacity of these initiatives to actually increase engagement in sustainable lifestyles or behaviours. Because student-led food gardens experience such high turnover, a need for ongoing recruitment is one of their central features. This could result in engaging participants without a history of sustainability or environmental engagement who would be better candidates for attitude or behaviour

change. However, the high turnover could also impact the cohesiveness of the group, providing fewer opportunities for social learning.

Furthermore, there are two other mitigating factors influencing pro-environmental behaviours that have yet to be considered. Firstly, while students may develop sustainable habits while at university, the change from student life to working life may actually be another "moment of change" which could "undo" all the habits developed at university. Therefore, it is equally as important to study how partaking in initiatives like student-led food gardens impact behaviours beyond the participants' time at university to understand the longevity of the initiatives' impacts. Verplanken and Roy (2016) have made some steps in testing the habit discontinuity hypothesis, but so far there is only evidence of short-term (in this case, eight weeks) changes caused by behaviour change interventions at life course transitions (Verplanken & Roy, 2016). Thompson et al. (2011) have conducted a limited number of interviews which explored the potential for university-based sustainability initiatives to contribute to behaviour change. They found that while there are already many existing behaviour change initiatives in universities throughout the United Kingdom that have considerable potential for behaviour change, given that students are a captive audience, there is little evidence of the impact these actually have. This therefore is an area requiring significant further research. Longitudinal studies spanning pre-university to post-university life would be particularly appropriate for understanding the impact of student-led food gardens beyond the university, as has been done in other studies on life course transitions (e.g., Elder, 1998).

Secondly, the impact of pro-environmental behaviour change initiatives needs to be reflected on in relation to infrastructural factors/barriers (in the case of sustainable food habits, whether there are growing spaces, food waste and composting facilities, good access to markets and shops with sustainable food options, etc.). There are also ethical

considerations that instrumental attempts to change students' attitudes and behaviours are somewhat paternalistic in assuming that students are not already conscious of their behaviour, and in reality may actually prevented by infrastructural barriers (e.g., lack of composting facilities, adequate public transport, etc).

Furthermore, there is a risk that the need to demonstrate impact of Student Eats, and other similar projects, to funders or prospective funders brings bias into how research assessing attitude/behaviour change interventions. In particular, it is particularly desirable for the National Union of Students' sustainability team to be able to say that their initiatives increased students' pro-sustainability attitudes and behaviours. However, it would be useful to also inquire into what structural barriers exist that prevent students from behaving more sustainably. What are the university structures that are affecting student behaviour? For example, how well are the student residences' kitchens equipped for cooking and storage of fresh food? How many people share a kitchen? What kinds of fresh food are available to students on campus, and when are they available? What kind of food diversity is available, particularly in terms of plant-based proteins? What kind of messaging are students getting from on-campus catering in terms of meat consumption, portion sizes, availability of fresh foods, and mealtime culture?

9.3.2 What and how are students learning?

In Chapter 5, it was shown that the dominating discussed and demonstrated benefits of sustainable university community gardens were educational. Indeed, one of the aims of Student Eats was "to give students (and other volunteers) the chance to learn about the ethical and environmental impact of their food choices" (NUS, n.d.c). Much of the National Union of Students' sustainability work with educational components tends to take quite an instrumental view of learning, as illustrated in the above section. This reflects the pragmatic

reality of the need to work on sustainability education "with and within the marketised reality" of higher education (Bessant & Robinson, 2019, p. 561). However, based on this study it appears that attitude and behaviour change do not seem to be the overriding outcomes for students taking part in these gardens given that many students involved in them appeared to be 'sustainability converts' already. Indeed, one of the barriers to participation that was highlighted by many students was that students involved in the gardens had difficulty relating to the wider student body because they already had such strong sustainability sensibilities. This meant that the sustainability sensibilities espoused by some existing gardeners that others found to be alienating, in conjunction with the wider student population's apathy or indifference towards gardening/sustainability issues, was preventing the students who would be candidates for learning about ethical and environmental food choices from taking part in the gardens. If student-led food gardens are intended to help students learn about the ethical and environmental impact of their food choices, but most students involved in these initiatives are already sympathetic to sustainability issues, a crucial question must then be raised: Are student-led food gardens preaching to the converted?

I contend that, while most of the students involved in the gardens studied could be considered 'sustainability converts,' there was still much for the 'converts' to learn by taking part in these spaces. Most undergraduate students are in a liminal state, where they are notquite-children, yet not-quite-adults either (Holdsworth, 2010). The learning these students are doing at this stage therefore might have more to do with their "negotiation of the transition to adulthood" (Holdsworth, 2010, p. 431) than, perhaps, the ethical and environmental impact of their food choices or how to grow food. This might include negotiating their place in the sustainability movement, understanding what realistic tasks for part-time volunteers are, and learning about the types of tasks voluntary (and, indeed, professional) organisations need to undertake to be successful. This sort of learning is much

more akin to the learning advocated in the intrinsic tradition of sustainability education, rather than the instrumental tradition that National Union of Students' initiatives tend to lean into. For example, if students have unrealistic expectations and misconceptions of what a student-led food garden can do, taking part in one can help them better understand what a voluntary, self-organising initiative can offer. Another example of such an opportunity in disguise was the conflict and mistrust between short-term and long-term stakeholders. One student said that it could be intimidating to approach people who are in supposed positions of *power*, like university staff. Indeed, the power relations between staff and students, whether actual or perceived, complicated trust building. However, Holdsworth and Quinn (2012, p. 393) have suggested that "the most valuable form of volunteering" is that which "provides learning opportunities which enable volunteers to deconstruct and resist power structures and inequalities." Students interacting with university staff were faced with such opportunities. However, this would be worth examining further to understand the extent to which students grappling with and resisting power structures within the university actually contributes learning, and if so, what they learn. Therefore, while some barriers or issues could prevent students from engaging in student-led food gardens, students who continue to take part may learn important lessons and skills to take with them into their post-university lives.

Based on all of this, it is clear that further exploration is needed to assess more definitively what students learn from student-led food gardens and other student-led sustainability initiatives, and what conditions facilitate different learning. For example, in what ways does volunteering affect students' negotiation of their identities at this point in their lives? How is learning through volunteering impacted by students' liminality? What type of learning is favoured in this life stage? Does it enable higher levels of reflexivity? What theories can help to explain this type of learning? Are students always learning 'good things'?

Or are students learning things that disempower them and prevent them from being effective change agents?

For the National Union of Students, it may be advisable to evaluate sustainability education programmes not only in terms of the extent to which it facilitates changes in attitude and behaviour, but also in terms of generation of key sustainability competencies (Wiek et al., 2011) and more intrinsic learning (Breiting et al., 2005) to ensure monitoring of all the impacts of their initiatives. Furthermore, knowing that Student Eats is made up mostly of 'sustainability converts' means that future support offered by the National Union of Students can focus more on facilitating intrinsic learning since the more instrumental attitude and behaviour change outcomes would likely be minimal. This said, other sustainability initiatives run out of the National Union of Students, such as Green Impact or Student Switch-Off may be engaging more 'mainstream' students upon which attitude and behaviour interventions may have more impact.

It may also be useful for the National Union of Students to have a two-pronged approach to student sustainability activities, with a prong for sustainability-newcomers and another prong for sustainability 'converts'. Attitude and behaviour change aims and corresponding evaluations would be more useful for initiatives that are geared towards sustainability-newcomers. However, for initiatives, like Student Eats, that draw in more sustainability-converts than sustainability-newcomers, it may be more useful to assess both sustainability and action competence development (Wiek et al., 2011; Breiting et al., 2005). If sustainability-converts already have pro-sustainability attitudes and behaviours, then it would be a case of preaching to the converted. However, these initiatives still provide important development opportunities for students that are sustainability-converts though the development of sustainability competencies (Wiek et al., 2011).

9.3.3 Pedagogy in the critical research agenda for urban agriculture

In 2014, Tornaghi published an article in which she called for a critical geography of urban agriculture. In her article, she proposed four areas of research on urban agriculture that critical geographers should explore: "cultural and political meanings of urban agricultural initiatives in different historical conjunctures and urban contexts in the Global North," "exposure of the socio-environmental injustice and exclusionary dynamics in place within urban agricultural initiatives," "proposing alternative models for a critical envisioning of postcapitalist, de-growth inspired urban living," and "re-politicizing the role of [urban agriculture] in the urban structure" (Tornaghi, 2014, p. 561-563). I suggest that there may be another prong to this critical take on urban agriculture, inspired by this study. Learning in community gardens including, but not limited to, those at universities is an important outcome. But many treat education and 'awareness raising' as unproblematic ways to address sustainability problems. However, to critical educators, education is always a political process. In critical sustainability education, it is generally understood that it is usually more than a knowledge deficit that causes unsustainable behaviour (Fink, 2011), and this is supported by empirical work from psychology (i.e., Schultz et al., 2007). As such, I suggest that another critical perspective on urban agriculture, in addition to those outlined by Tornaghi (2014), is needed. This critical perspective should be through the lens of critical pedagogy. Some of the questions that this research could address are: In what ways does and can urban agriculture act as a form of popular education? What sorts of urban agriculture contribute to building what sorts of sustainability competencies? What kind of horticultural or agricultural knowledge/skills are being developed in different types of urban agriculture, and how might these increase (or reduce) food system resilience? To what extent does dysfunctionality of urban agriculture initiatiatives correlate with development of sustainability competencies?

9.3.4 Failure

As highlighted in section 9.2.6, focusing on problems can be uncomfortable. However, understanding our problems, failures, and mis-steps is important to avoid repeating them. However, failure of community gardens is under-explored in both academic and popular literature. Only positive cases were discussed in the literature about sustainable university community gardens reviewed in Chapter 5, and likewise, in Guitart et al.'s (2012, p. 368) literature review of urban community gardens, only five in eighty-seven papers described negative outcomes (those that did were all related to "insecure land tenure and gardens being demolished for future development.") There were also some mixed and neutral outcomes, but the overwhelming majority of articles described positive outcomes (Guitart et al., 2012). Furthermore, Corcoran et al. (2004) have found that many case studies published about sustainability in higher education are written as success stories (or "make-your-casestudies") and have a "'policing function' that identifies good [cases] from bad" (Corcoran et al., 2004, p. 18). Together, this seems to indicate that sustainability in higher education has a rather one-sided picture of sustainability initiatives. This may even be true of the wider field of sustainability science, where literature on 'seeds of the good Anthropocene' (Bennett et al., 2016), pockets of activity representing what a sustainable future could look like, are being widely celebrated, while the rotting or dormant seeds are hard to find.

It is important to study failure because when we only study positive examples, we can misunderstand what is genuinely making them successful (Denrell, 2003). Commonalities between successful cases can be assumed to contribute to their success when in reality they may be coincidental, and equally feature in examples of failure. Only through exploration of both positive and negative examples can we see what factors contribute to success.

A similar problem is publication bias against papers with negative outcomes. This is a well-documented, however unresolved, phenomenon in medical (Easterbrook et al., 1991)

and social sciences (Franco, et al., 2014). Some of the reasons this may have taken place are "reviewer or editor preferences" and authors' anticipation of "the rejection of negative outcomes" (Easterbrook et al., 1991; Lee et al., 2013, p. 10). To reviewers, editors and authors, positive examples may be perceived as more useful to know about so they can be replicated. Furthermore, authors may prefer to write about their successes rather than their failures for promotional purposes or because they are more likely to write about projects they have positive feelings about (Gill, 2009). Even if researchers are interested in studying examples of failure, they may struggle to find information about failed projects because projects may have collapsed before they came to fruition making it difficult to identify individuals involved.

There is also a time-lag associated with publishing negative or statistically insignificant findings because researchers may prioritise their positive findings (Ioannidis, 1998). Even though results may eventually be published, delays can also force researchers and practitioners to rely on positive findings for up-to-date information.

Scholars of various disciplines are taking steps to understand (Helmig et al., 2014), make space for (Harrowell et al., 2018) define (Halberstam, 2011) and typologise failure (Van Rooij, 2015). However, there is very much a need to understand what failure is in the context of student-led food gardens, and sustainability initiatives more broadly – or if it is even allowed to exist in the minds of those leading them. Indeed, a staff member at the National Union of Students was uncomfortable with even entertaining the idea of any of the Student Eats gardens failing.

So, what is a "failed" community garden? How do temporality and positionality affect understandings of failure in community gardens? Most projects tend to have successful elements from at least one perspective – is this perhaps why community gardens are so frequently framed in the literature as being successful, or at least having mixed outcomes?

And more pragmatically, if we can agree that there is such a thing as a 'failed' community garden, what can be done to prevent them from failing? Or is preventing failure even desirable? This last question is one Halberstam (2011, p. 185) has grappled with in his book, *The Queer Art of Failure*, challenging the idea that failure is something to be avoided, by subverting the idea of a "productive" failure, or "failing again, failing better." This is a useful conceptualisation of failure for researchers, because "good" or "bad" dichotomies are not necessarily helpful in generating nuanced understandings. However, it is unclear how these concepts might be used in the field of sustainability, where to fail might mean the end of humanity as we know it. As such, there is a considerable need to explore the meaning of failure for sustainability initiatives, 'failed' sustainability initiatives (like student-led food gardens), and destigmatise failure in research and the sustainability community.

9.3.5 Intersectional interrogations of student-led food gardens

There is also a need for intersectional interrogations of student-led food gardens, and indeed, student volunteering in general. Brewis and Holdsworth (2011) found that student volunteers with disabilities, from ethnic minorities, and from outside of the UK were more likely to be less satisfied with their volunteering experiences, and we need to inquire as to why.

In the case of the student-led food gardens, there were several students who took part but left university mid-way through their degrees after expressing quite high levels of anxiety and hopelessness in my interviews and conversations with them. This was not a core part of my study and I did not have ethical approval for this line of inquiry, and therefore I did not ask why they chose to leave university. However, I believe there is an important avenue of inquiry here. Students reported feeling overburdened, overwhelmed and burned out, and some stopped taking part in the garden because of it. Some students, however, continued to

be involved in spite of personal and academic struggles, and I worry that their involvement in voluntary activity could have been exacerbating potential mental ill-health.

Those who are involved in student-led food gardens advocate for social sustainability, of which mental health is an important part. There is a recognition that activism can take a toll on the mental health of activists (Vaccaro & Mena, 2011). However, the relationship between mental health and sustainability activism is not yet well-interrogated (Fritze et al., 2008), and the relationship between young person's mental health and sustainability, even less so (Ojala, 2012). However, these are clearly some important lines of inquiry to enable students engaging in these projects to maintain their own health, as well as in terms of actually delivering the aims of the gardens. For example, what is the quality of student learning from taking part in student-led food gardens if students are grinding themselves into the ground in the process?

Another important dimension that I did not dwell much on in this thesis is the engagment of interational and exchange students in student-led food gardens. Why are there such high proportions of international and exchange students in these initiatives? Is it because self-selecting students who choose to travel abroad for their education are more likely to be highly engaged? Or is it to do with place-making and meeting locals? Or is it just to do with making the most of their time abroad? Furthermore, what implications does their engagement have on the projects? Do they usually bring with them knowledge about foodgrowing from their home countries as in Mundel and Chapman's (2010) study? Or is it typical for international students to be unfamiliar with the practice of food growing before studying abroad? Furthermore, staff from the National Union of Students also said that international students may want to grow foods from their home countries in their gardens, whereas one student at Keele I spoke with this about said they would prefer to grow traditionally English

plants because they already knew what is grown in their own country and wanted to learn about what people grow and eat in England.

It would also be useful to explore ways in which gender roles and stereoypes might be disrupted or reinforced in student-led food gardens, particularly given that it has been found that gender roles can be reinforced in community gardens (Parry et al., 2005). An interviewee from the National Union of Students' said:

I think from memory, the men were more interested in the more physical setting-up tasks, like sawing planks of wood, and digging holes and mud and stuff like that. And the women's responses were more interested in the planting and the stuff that came next. Which is really, like gender stereotyped.

A better understanding of the practices that help disrupt unhelpful norms would be useful, particularly given the ambitions of sustainability education to contribute to an equitable future.

9.3.6 Student-staff relationships in university sustainability initiatives and living labs

One notable gap in my data collection is the limited engagement with non-student stakeholders at the univeristies and students' unions. Given the important role of these stakeholders and the fact that sometimes their interactions with students are reportly so fraught, inquiry into how these groups can better collaborate and create shared understandings would be useful. This does not need to be limited to research about studentled food gardens, and could be a line of inquiry that would be extremely relevant to institutions attempting to adopt the living lab model on their campuses. For example, how can different stakeholder groups (academic staff, professional services staff, students' union staff, local community) with different ambitions for how university sustainability initiatives like living labs should be used (teaching, research, enhancing student experience, delivering

services to the community, etc.) cooperate in a mutually beneficial way? It would also be interesting to examine how the living lab agenda could be used, or perhaps is already being used, as a boundary object (Star & Griesemer, 1989) between different stakeholders.

9.3.7 Resilience at odds with other aims

Further inquiry with a more critical stance on resilience would be a worthwhile undertaking, focusing on the extent to which enhancing the resilience of a system might be at odds with other aims. There has already been some initial exploration of how sustainability and resilience might be in tension (Elmqvist, 2017). For example, a key contributor to resilience is redundancy, whereas efficiency is often an important contributor to sustainability (for example, energy efficiency). It may also be useful to explore if there are tensions between learning and resilience. For example, can an organisation that lacks resilience, like a student-led food garden, provide a productive learning environment by creating space to confront and wrestle with complex problems? In other words, is a resilient student-led food garden less likely to contribute to the development of sustainability competencies?

9.3.8 Understanding transience

Surprisingly little has been written about transience in sustainability science, particularly given that the field has only existed for several decades, a time where people have become more mobile than they have ever been. Furthermore, much of today's sustainability work relies on grassroots and volunteer efforts, and *ad hoc*, one-time, and seasonal volunteering is on the rise (MacDuff, 2005). Therefore, understanding transience and how it affects and can be managed by sustainability organisations is absolutely vital if we want these organisations to be effective in leveraging change for a sustainable future.

Further inquiry about what could be learned from temporary organisations (Ibert, 2004; Bakker, 2010; Bakker et al., 2011; Brookes et al., 2017) for the management of

organisations with transient participants would also be worthwhile. This may include further examination of the similarities between the roles of what I have termed 'long-term stakeholders,' and Bakker et al. (2011) calls 'parent organisations,' or looking into how risk is approached, avoided, or exploited in student-led food gardens and other organisations with transient members.

9.4 Conclusion

The purpose of this chapter was twofold: to reflect on this study's methodological issues and the future research directions it has prompted. Some aspects of the methodological discussion demonstrated the strengths of the approaches adopted in this study, such as the use of a research diary. Others shed light on the weaknesses with the ambition to be honest and transparent, for example, how I had only limited input from university and students' union staff. However, most were more complicated contemplations without conclusive recommendations or outcomes. For example, involving three different gardens in an action research study proved to be challenging and prevented the in-depth inquiry that could have been conducted with just one garden. The future lines of inquiry inspired by the study included exploring both the pragmatic and emancipatory roles of student sustainability initiatives in higher education, critical perspectives on pedagogy in urban agriculture, failure in sustainability, intersectional interrogations about student-led food gardens, the use of the resilience principles in social science-focused sustainability research, and transience in sustainability organising. Engagement with these methodological reflections and exploration of these lines of inquiry will contribute to more rigorous, relevant, and learningful action-orientated research in sustainability in higher education and beyond.



Chapter 10



Carrots ready to harvest November 2013

10.1 Introduction

This study is the first comprehensive, empirical examination of how university students' transience impacts student-led initiatives. It contributes to emerging areas of sustainability scholarship inquiring into the internal realities of self-organising sustainable food organisations (Keafsey & Hasanov, forthcoming), and the use of university campuses as living laboratories for developing sustainability practices and learning. In this final chapter, I will briefly summarise the main findings from each chapter in order to answer the research questions. In doing so, I will outline the original contributions this thesis makes to both research and practice.

10.2 The benefits of sustainable university community gardens

The first step I undertook in my inquiry into student-led food gardens was to assess the benefits of student-led food gardens in universities in order to understand how they could contribute to addressing some of the sustainability challenges outlined in the literature review in Chapter 2. I also compared this to the literature about urban community gardens to understand how sustainable university community gardens differ from urban community gardens, and the extent to which the benefits they provide are unique. I did this through conducting a quantitative systematic literature review, drawing on a protocol from Guitart et al.'s (2012) literature review of urban community gardens in order to enable comparison. Given that very little was written about student-led food gardens, I broadened the scope of the review to 'sustainable university community gardens'. There were only twenty-two articles that met the inclusion criteria. These articles gave an indication of the general trends in the literature, but the limited number of articles meeting the criteria also demonstrated that this body of literature is currently under-explored. The literature demonstrated a clear geographical bias mirroring the geographical bias in Guitart et al.'s review, with most articles coming out of the United States and focusing mainly on gardens in the United States. The portfolio of benefits provided by sustainable university community gardens and urban community gardens were very similar, with the most prominent being social, educational, economic, and health benefits, as well as benefits associated with accessing fresh food. However, it was clear that educational benefits were more evident in sustainable university community gardens compared to urban community gardens. There was also evidence that taking part in sustainable university community gardens can contribute to building sustainability competencies (Wiek et al., 2011; Hedefalk et al., 2014).

Beyond the literature review, there was also evidence from the empirical components of the study that learning was an important benefit of taking part in student-led food gardens. However, it appeared that the sort of learning that took place was more akin to the intrinsic rather than the instrumental educational tradition. For example, participants were negotiating their place in the sustainability movement and learning about how to make change in organisations, rather than learning about sustainable food in order to foster prosustainability attitude and behaviour change. Overall, there were considerable benefits provided by student-led food gardens, making them worthy of support.

10.3 Existing understandings of the impacts of students' transience

Crucially, it was also found that there was little more than a passing mention of the impacts of students' transience in the literature on sustainable university community gardens, in spite of the fact that participation from students was mentioned in nearly all the articles reviewed. Based on the near-to-non-existent mention of the impacts of students' transience in the literature, and the gardens' potential educational benefits, especially to do with the development of sustainability competencies, I concluded that an exploration of how to understand and manage university student-led food gardens was indeed worth pursuing. I therefore went on to explore the causes and effects of short-term irregular, and low participation.

10.4 Problematic participation: causes, effects, and feedbacks

The second set of research questions I addressed focused on understanding the impacts of students' transience through examining the dynamics of problematic participation, including its causes, effects, and any feedbacks between them. Students' transience was found to have a paradoxical role in that it caused a form of problematic participation (shortterm participation), while indirectly increasing participation overall by refreshing the pool of new recruits on an annual basis. There were many other factors causing problematic participation, which I categorised into barriers to 'getting through the garden gate' and barriers to continued engagement. There were many of each type of barrier, although there were none that stood out as key to address. Rather, it seemed that students' transience was the dominating factor, but this was not something that could or should be changed: students do and should leave university. Naturally, problematic participation led to a range of problems. These included: a constant need to recruit, a fast-changing and unpredictable organisational structure, friction between long-term and short term stakeholders, discontinuity and difficulty ensuring longevity, gardeners not seeing the results of their work, short-term thinking and a lack of insight into transience, difficulty retaining knowledge and skills, lack of funding, gardeners feeling overburdened, overwhelmed, and just burned out, inadequate maintenance, reduced social interaction, and negative emotional responses. A key finding that emerged out of the causal loop mapping was that short-term stakeholders experienced time in the gardens linearly, while long-term stakeholders experienced it

cyclically. These dual conceptions of time likely resulted in university staff members engaging in taxing and repetitive interactions with students, which they may have found frustrating. Students and staff also had different expectations on what appropriate timeframes were for tasks, which also caused frustration in university and students' union staff. As such, these differing conceptions of time were, in some cases, a source of friction between university staff and students. This analysis also indicated that student-led food gardens, as organisations with transient participants, might actually have an organisational form unique from temporary and permanent organisations.

From the causal loop mapping I found three accelerating feedback loops in the interaction between problematic participation, its causes and effects. These were buffered only by an annual refreshment of the pool of potential new recruits, illustrating the vulnerability of the gardens that many of the research participants expressed. Because transience was the main factor driving the vulnerability of the gardens, there were no other factors that stood out as key to address to reduce the effects of problematic participation. As such, it was the cumulative effects of all the barriers to participation and effects of problematic participation that were causing the vulnerability rather than one particular factor that could be leveraged. Because of this, I made some initial recommendations that both upstream solutions (targeting causes of problematic participation) and downstream solutions (effects of problematic participation) should be implemented, and which causes and effects of problematic participation solutions should focus on addressed should be decided on a garden-to-garden basis. Having mapped the relationship between transience, problematic participation, and its causes and effects, I then went on to explore other ways in which student-led food gardens might be vulnerable to students' transience.

10.5 Student-led food gardens' vulnerabilities in the face of students' transience

From developing an understanding of how students' transience impacts the participation dynamics of student-led food gardens, I found student-led food gardens appeared to be vulnerable to the transience of their own participants. Chapter 7 went further with this analysis of the vulnerability of student-led food gardens to students' transience by exploring the extent to which the gardens aligned or misaligned with Biggs et al.'s (2012, 2015c) resilience principles. Through this analysis I make a contribution to social science resilience scholarship empirically and theoretically as, to date, it is underrepresented in comparison to the natural sciences.

In Chapter 7, I explored the diversity and redundancy of the stakeholders involved and found that student-led food gardens were, naturally, lacking long-term involvement. In the garden where the relationships between short-term and long-term stakeholders were characterized by mistrust and conflict, the garden was thought to be particularly vulnerable. Gardens that lacked choice in ways that prospective gardeners could take part and ran a limited variety of different activities struggled to maintain participation. Furthermore, when there was an unsuitable balance between the 'fun' gardening activities and the ones that felt more like work for long-term mentors and student leaders, they struggled to see the value in taking part. In one garden, there was a high level of top-down control over the use of the space, preventing gardeners from exercising their autonomy over the space. There also appeared to be limited time allocated for university staff, especially those involved in managing the grounds, to engage with or support gardeners. Support was also needed from a variety of actors, although in most gardens the support from the students' union and university was limited. Staff from the National Union of Students, however, can and did play a crucial facilitative role in supporting staff-student conversations.

The *connectivity* between gardeners changed rapidly, and a fine line between cliqueyness and cohesiveness needed to be balanced. Throughout the study, there was evidence of this balance both being maintained and falling off-kilter. All three of the gardens studied were poorly connected physically to the rest of the campus, which was thought to increase their vulnerability to transience by limiting incidental engagement. In one garden, the connections between university/students' union staff and student gardeners were not perceived to be mutual, and there was a notable amount of distrust characterising the relationships. Power asymmetries were not attended to or compensated for. In other gardens, the relationships between staff and students were not particularly strong, but overall were positive. However, support to help students retain knowledge within the gardens was limited.

Staff from the National Union of Students played an important role in building the resilience of the gardens through playing a facilitative and mediating role between the universities, students' unions and student gardeners. They also helped to connect gardens at different universities through visits and conferences, which helped gardeners feel they were a part of something bigger and inspired them.

Slow variables and feedbacks that I identified as being in need of monitoring were students' indifference/apathy to food growing and sustainability, the quality of the relationships between long-term and short-term stakeholders, the physical state of the garden, and the balance between an emphasis on recruitment compared to the 'core business' of the garden. Long-term stakeholders, like university/students' union staff or longterm mentors, were in a better position to monitor these and intervene as needed than those who had shorter-term engagement. However, there was little evidence of this sort of monitoring.

Short-term participants tended to be less tolerant to uncertainty and ambiguity, although it was found that it could be developed over time through participation in the

gardens. There was considerable evidence of learning taking place, although the learning seemed to not be the attitude/behaviour change or instrumental learning that was expected by the National Union of students, or perhaps even the students themselves. Furthermore, the retention of knowledge and skills within the gardens was poor which limited the learnings' contribution to resilience. However, the learning that came from participating in the student-led food gardens seemed to be more about students negotiating their transition to adulthood, how to navigate change in complex systems, and the development of the "ability, motivation and desire to play an active role" in a sustainability-focused initiative (Morgensen & Schnack, 2010, p. 61).

10.6 Applying the resilience principles beyond ecosystem services

This study marks a first attempt to apply the resilience principles to a context beyond sustaining ecosystem services, therefore, I also reflected on their transferability in Chapter 7. Overall, I found the principles useful, however because each of the principles had the potential to contain so many variables it was challenging to offer an overarching assessment of the resilience of the student-led food gardens with the depth that a qualitative case study approach requires. I also found it difficult to disaggregate the properties of student-led food gardens as systems ('the governed') and their governance ('the governors') given the focus of the study was weighted more heavily on the social component of the gardens. That is to say, that which needed governance and those who governed were often one and the same. I also found that too much diversity (i.e., in participants' worldviews) negatively affected connectivity and learning, thereby reducing resilience, indicating that it may not always be desirable to increase diversity. I also suggested that the application of the complex adaptive systems lens should be done with more ontological and epistemological humility to create opportunities for complex adaptive systems skeptics to engage with the principles. Finally, I suggested that intrinsic orientations towards learning should be encouraged in the resilience

principles, in addition to the extrinsic orientation already present. Rooted in these reflections, I suggested new phrasings and ways to organise the resilience principles for the application to social-ecological systems where the social component, rather than ecosystem services, is in focus.

10.7 Strategies to address the impacts of students' transience and build resilience

Having explored how transience impacts student-led food gardens and the vulnerabilities it causes, the last part of the thesis was future-orientated. In the first chapter of this part, I explored how to address the impacts of students' transience and build resilience into their initiatives such that they continue to persist. Drawing on the causal loop mapping, analysis of student-led food gardens through the lens of the resilience principles, and research participants' perspectives, I created a list of strategies and a matrix of advice for actions different stakeholders. These are the more prescriptive and practical contributions of this thesis.

I suggested that an appropriate and well-managed garden space should be maintained. This means being located in an accessible, central location, that the site is wellmaintained, and it is aesthetically appealing. Physical infrastructure, such as raised beds, greenhouses and benches, should be used to prevent the gardens from becoming overgrown during periods of low participation, as indoor space for gardening in poor weather, and to facilitate a more social experience. Appropriate crops should also be grown to reflect the student calendar. Activities and events should be run weekly on a set day and time. Activities and events should be diverse and varied to appeal to a diverse student population, including gardening-orientated and social activities.

Support should be sought out by gardeners, and should be provided by university and students' unions' staff, National Union of Students' staff, peers, and more experienced and 366

knowledgeable gardeners. Support should be offered in a way that capitalises on the supportgiver's competencies and their role in relation to the garden. That is to say, students' union staff would likely be able to support gardeners in a different capacity than staff working with campus operations. Support should also be offered in a way that enhances, or at least does not infringe upon gardeners' capacities to self-determine within their garden.

Creativity and interest should also be nurtured through high-quality leadership and facilitation (from all stakeholders). This should foster a sense of ownership and motivation through creating opportunities for gardeners to self-determine. Care should be given to ensure participants do not feel overburdened, but still feel encouraged to take on a bigger role and come back again. Participant input should be sought out by asking others what they want to do and finding out what participants want for the garden and from taking part. University or students' union staff overseeing the space should ensure opportunities for students to make changes, be creative, and experiment with the garden space as much as possible.

The visibility of the garden should be increased through advertising and making the actual garden more visible on the campus, such as though putting the garden on the campus map or creating signage. Organisational infrastructure should also be created. This includes establishing a society with committee roles, having elections for these roles, and assigning individual responsibility for tasks. It may also be effective to have a large number of roles within a society in order to distribute tasks more widely and yield a sense of responsibility form participants.

Recruitment should also be in focus, both in terms of who to recruit and how to recruit them. Potential participants that would be able to participate for a longer time and/or over the summer would be more strategic to recruit, such as first-year students, postgraduates, staff members, and campus residents. Recruitment can be done through

advertising, but reaching out through one's social network is typically a more efficient route. Diversity and redundancy should be built into participation, which is why recruitment in so important.

Documenting the history of and planning for the future of gardens ought to be done by keeping records to demonstrate impact, catalogue useful resources, and provide future gardeners with vital information for future strategic planning. The National Union of Students should continue to offer support for this through providing templates and guidance. These records should be used as part of a 'handover' between generations of gardeners. Planning for the future could be done through creating and regularly updating a shared vision; having to-do lists and guidance documents; and provision of general gardening advice. Documenting and planning were thought to contribute to building a narrative or a story of the gardens, helping participants to see that they are a part of something bigger.

To build the resilience of the student-led food gardens, they should be embedded within existing organisational infrastructure, such as student volunteering schemes, in university calendars/internal communications/webpages, the estates maintenance teams' schedule, the formal curriculum, staff members' role descriptions, and by becoming a student society. This can account for participation drop-offs because staff can 'kick-start' the group by advertising and recruiting new members. It also could create more incidental or serendipitous encounters with the initiative that might not have existed otherwise.

Connecting up with the students' union's current priorities, other student societies, the National Union of Students' Student Eats Network, and other Student Eats gardens was also thought to be helpful to break down the 'cultural barrier' and spread the word about the garden. Connecting to the National Union of Students' Student Eats network, and other Student Eats gardens was also thought to put gardeners in contact with resources to improve

gardeners' gardening knowledge and skills and help gardeners feel they were part of something bigger.

Although a less concrete strategy, focusing on value and creating vibrancy was equally as important. Focusing on value meant that gardening activities were not derailed by committee meetings, ensuring recruitment remained a secondary rather than primary purpose, continuing with tasks that need doing even if other people do not show up to gardening sessions, and focusing on activities that were fun and/or useful. Passionate people and having impressive harvests were thought to create vibrancy, resulting in a positive atmosphere people would return to. Appealing to extrinsic motivations to get and keep potential recruits involved may also be a useful strategy. Gardens should be funded and resourced adequately (including staff time). Some also thought paid student roles would be useful.

Learning played an important role in managing the impacts of transience. In particular, I found that there was a mutually reinforcing relationship between participation and the development of action competence. To further amplify this relationship and increase both participation and the development of action competence, I suggested that stakeholders of student-led food gardens should create opportunities for boundary interactions and embed opportunities for reflection-on-action in, for example, volunteering awards schemes or skills portfolio programmes.

I also found that while some strategies (such as running events and activities), had the potential to contribute to social-ecological memory over the short-term, over the long-term the most effective actions retain social-ecological memory were documenting activities in the garden and embedding the garden within wider organisational infrastructure, because these were strategies that did not require overlap between generations of gardeners. Ultimately, there is no silver bullet to address the impacts of transience and build resilience in student-

led food gardens. Rather, attending to as many of these strategies as possible creates space for prospective gardeners to engage in different ways, and to handle unpredictable or unexpected challenges and issues as they arise.

10.8 Methodological reflections

In Chapter 9, I offered some reflections on the methodology in order to be as candid, honest, authentic, and transparent as possible. I also hope that suggestions borne out of my methodological reflections will be help other action researchers navigate the confusing landscape of scholar-activism.

I found that keeping a research diary enhanced the rigour of my study, as did my unique positionality as a student-staff member. Rooted in this, I made the recommendation that this cross-over role would be beneficial for research into other student-focused campus initiatives. However, I found that conducting action research with transient participants came with considerable challenges. These were (1) a superficial understanding of issues by volunteers as a result of short term engagement, resulting in difficulties identifying or theorising the issue to research; (2) difficulties ensuring the implementation of 'actions' or 'interventions' given those involved in the 'planning' phase had often left by the 'action' phase; (3) a lack of capacity to engage in research given the ongoing challenges to maintain an adequate level of participation; and (4) that participants reported that the introduction of the researcher changed the group dynamics, prompted critical reflections that had not previously taken place, and improved continuity, thereby preventing a 'naturalistic' inquiry into transience as a phenomenon. Rooted in these reflections, I argued that, when conducting research (especially action research) with transient participants or partners, gathering perspectives from less transient stakeholders is a vital complement to perspectives from transient participants. Also, when conducting research with transient participants, it is worth

considering reciprocity not only on an individual basis, but also in a wider sense. That is, a researcher might not be able to 'give back' to the individual who took part in their research, but the researcher can give back to the community the individual is part of. Dealing with a lack of participant capacity to engage in research can be done by harmonising the data collection activities with activities participants need to do anyway, such as through go-along interviews. Researchers can also offer their time or efforts to return the time the research participant sacrificed for the researcher. I also suggest that first-person inquiry should be used to complement second- and third-person inquiry in order to attend to the researcher's own "thinking, valuing, way of learning, and behaving" to produce more authentic research (Coghlan, 2019, p. 191). Furthermore, researchers should also embrace their agency in their practitioner or activist role, as long as it is done with partners' blessings and best interests at heart, as they may be the actor with the most capacity to follow through the process from planning to taking action.

I also found that using an embedded case study approach within an action research study was both a strength and a weakness. I was able to develop more complete insights by capitalising on my different positionalities within the gardens. However, action research is a methodology that needs to be applied in context-specific ways. This meant that there was some inconsistency between the process between each of the three gardens that underwent more in-depth inquiry. This made the development of a coherent narrative for this thesis challenging.

I also reflected about an interaction with a staff member at the National Union of Students who expressed considerable concern about the focus on problems in my study and how I handled the feedback. Rooted in this, I said that 'failure' has stigma attached to it that would be good to break down. This was brought up as a future research direction later in Chapter 9.

Finally, I noted two limitations of this study. One is that I did not engage with or interview university or students' union staff in the three gardens that were the main focus of this study, although as a university staff member myself, I was able to reflect from this perspective in my research diary. The staff-student relationships in student-led sustainability initiatives in higher education would be an interesting future route for inquiry. The second limitation was the *ex-ante* causal-loop diagramming and use of the resilience principles. Because I had not planned to use causal loop diagramming or the resilience principles in the initial data collection (or during data collection at all, for the resilience principles), it meant that it is possible that I missed opportunities for probing and interrogating the connections between the different codes, meaning the causal loop diagram may be incomplete. Furthermore, it is also possible that the assessment of the vulnerability of student-led food gardens was not as complete as it could have been.

10.9 Future research directions

At the end of this thesis, I offered some potential research directions that were inspired by this study. I suggested that long-term implications of attitude and behaviour change initiatives in tertiary education, what and how students are learning in student-led initiatives, and critical perspectives on pedagogy in urban agriculture are needed. I also suggested that understandings of failure in the context of sustainability are currently limited, possibly due to stigma, and that this should be rectified. Given some gardeners' expressions of high levels of distress in my interviews and conversations, and the high proportion of international and exchange students who took part in the gardens, intersectional inquiry of student-led food gardens would be useful. These might consider mental health, (inter)national student status, ethnicity, (dis)ability, gender, or other axes of difference. Student-staff relationships in university sustainability initiatives/living labs would be worth further inquiry to explore how to better maximise benefits to all stakholders. Further research applying Biggs et al.'s (2012, 2015c) resilience principles in different contexts should be undertaken to explore the extent of their applicability, and the extent to which enhancing the resilience of a system might be at odds with other aims (such as sustainability or learning). Finally, surprisingly little has been written about transience in sustainability science, although transience is very much a defining feature of our time. Further inquiry about the what could be learned from the scholarship about temporary organisations might help us to better understand how living in a more transient world affects student organising for sustainability.

10.10 Conclusion

Having studied impacts of transience in university student-led food gardens and how to manage them, it is clear that student-led food gardens operate in a perpetually vulnerable state. Students' transience impacts student-led food gardens in paradoxical manner, in that it both increases and decreases problematic participation. In some cases, there were power asymmetries between students and staff, underrepresentation of long-term stakeholders (like postgraduate students and university/students' union staff), slow variables were not monitored, and a lack of tolerance to ambiguity and uncertainty, all contributing to the vulnerability of the student-led food gardens studied. In spite of this vulnerability, there is evidence that the gardens provide a range of benefits, especially opportunities for learning, including learning that contributes to building sustainability competencies.

In addition to these findings, this study has practical outcomes. The action research process had participants deliberating and reflecting on how to address students' transience, and this not only contributed to answering the study's research questions, but also providing opportunities for learning for gardeners and staff at the National Union of Students alike. In the gardens at Keele and Warwick, gardeners took actions to address the impacts of
transience through this study although some of these were more impactful and long-lasting than others. There were also initial indications that the main learning outcomes of studentled food gardens were not 'converting' mainstream students into pro-sustainability advocates, but rather, helping pro-sustainability advocates develop their competencies to be better sustainability advocates. Knowing this will be useful to capture the full impacts of these projects in future evaluation of Student Eats and other pro-sustainability initiatives in tertiary education. The results of this action research study culminated in a practical set of recommendations for different stakeholders of the gardens.

However, there are outside pressures that may affect the extent to which these recommendations are taken up within universities, students' unions, student-led food gardens, and the National Union of Students. The higher education sector in England is increasingly subject to neoliberal agenda which is shaping how sustainability activity is manifesting in universities (Bessant, 2017). There do appear to be opportunities to harness this agenda for pro-sustainability activity. For example, the increasing importance of studentsas-consumers can be leveraged for pro-sustainability aims by drawing out links between student-driven agendas (e.g., employability, internationalisation) and sustainability (Bessant & Robinson, 2015). One such opportunity that could be exploited is drawing out the mental health benefits of gardening, given the current focus on improving student mental health (Pereira et al., 2019). This opportunity could be harnessed by the National Union of Students when seeking out funding for Student Eats, or by supporters of student-led food gardens from universities or students' unions when advocating on their behalf. This said, it is not clear what the net impact student-led food gardens have on the mental health of their participants, as has been problematised in this study.

However, there are also considerable pressures from the neoliberal agenda that may impact or already be impacting student-led food gardens negatively. The pressure to be more

374

efficient with resources (including staff time) would likely limit staff support for such initiatives in both a quantitative and qualitative sense. The drive for efficiency may result in less staff time, quantitatively, allocated for such initiatives. This may be especially so for grounds staff, who are not typically expected to act in student-facing roles. Furthermore, the pressures of staff needing to do more with less may also negatively affect the quality of support they can offer to students. This might include staff lacking patience, reducing their capacities to guide student engagement and learning effectively. Moreover, the most important benefits of student-led food gardens are ones that are notoriously hard to measure. With the rise of audit culture associated with neoliberalism, it may be difficult to provide a compelling rationale for the continued support of such initiatives if their outcomes are not captured by league tables or other metrics. As such, it is not only the dynamics within student-led food gardens that have the capacity to undermine them, given that the activities taking place in the higher education sector at large also play into maintaining their longevity.

University student-led food gardens are a niche phenomenon, even within the world of sustainability in higher education. However, studying them has provided insights with much wider implications. I found that thinking about long-term and short-term stakeholders in organisations as having dual conceptions of time can help us to understand why friction emerges and how to address it. This study was also the first, to the best of my knowledge, to apply Biggs et al.'s (2012, 2015c) resilience principles to sustain something other than ecosystem services – that is, student-led food gardens. In doing so, I found that (1) the principles were indeed useful beyond their original intended context, and that (2) the stratification into 'the governed' and 'the governors' was less useful for application to socialecological systems where the social system is in stronger focus. This is why I proposed a rephrasing of the principles for these sorts of contexts. Furthermore, I suggest that for socialecological memory to be retained within social-ecological systems where 'generations' of

375

people do not overlap, physical forms/artefacts, and engagement from/embeddness in external institutions or physical forms were the most effective strategies to use.

Using action research with transient stakeholders also provided some useful recommendations for future attempts to employ the methodology in the field of sustainability: action research with transient stakeholders should seek out less transient stakeholders to provide complementary perspectives; reciprocity can be thought of at a community-level rather than an individual level; and researcher agency and first-person inquiry should be embraced. I also suggested that sustainability scholars need to become more comfortable working with and reporting on failure so as not to create a one-sided picture of their research topic.

As the number and types of sustainability initiatives in universities increase, and universities work increasingly with sustainability and sustainable food through their campuses and education, the growing number and types of sustainability-focused initiatives, like student-led food gardens, can provide sites of learning and experimentation outside of the formal curriculum. These initiatives are meaningful contributors to sustainability education in the higher education sector. While vulnerable to their own participants' transience, students, university staff, students' union staff, and student sustainability organisations, like the National Union of Students sustainability team, all have the potential to contribute to ensuring the longevity of these sites of learning.

376

References

@KeeleStdentEats (n.d.). Student Eats. Retrieved on August 13, 2019, from https://twitter.com/KeeleStdentEats

- Abson, D. J., Fischer, J., Leventon, J., Newig, J., Schomerus, T., Vilsmaier, U., ... & Lang, D. J. (2017). Leverage points for sustainability transformation. *Ambio: A Journal of the Human Environment*, 46(1), 30-39. doi: 10.1007/s13280-016-0800-y
- Adger, W. N. (2000). Social and ecological resilience: are they related?. *Progress in Human Geography*, 24(3), 347-364. doi: 10.1191/030913200701540465
- Adger, W. N., Hughes, T. P., Folke, C., Carpenter, S. R., & Rockström, J. (2005). Social-ecological resilience to coastal disasters. *Science*, 309(5737), 1036-1039. doi: 10.1126/science.1112122
- Adrangi, M. (2013). From Campus to Community: Challenges and Opportunities of University Based Activism. *Canadian Dimension*, 47(1), 33-35.
- Ahmed, S. (2017). Living a Feminist Life. USA: Duke University.
- Almers, E. (2013). Pathways to action competence for sustainability—Six themes. *The Journal of Environmental Education*, 44(2), 116-127. doi: 10.1080/00958964.2012.719939
- Apel, H. (2004). *The Future Workshop*. Deutsches Institut für Erwachsenenbildung. Retrieved on October 30, 2014, from http://www.die-bonn.de/esprid/dokumente/doc-2004/apel04_02.pdf
- Apul, D. S., & Philpott, S. M. (2011). Use of outdoor living spaces and Fink's taxonomy of significant learning in sustainability engineering education. *Journal of Professional Issues in Engineering Education and Practice*, 137(2), 69-77.
- Anderies, J. M., Janssen, M.A., & Ostrom, E. (2004). A framework to analyze the robustness of social-ecological systems from an institutional perspective. *Ecology and Society*, 9(1), 18.
- Anderies, J., Walker, B., & Kinzig, A. (2006). Fifteen weddings and a funeral: case studies and resilience-based management. *Ecology and Society*, 11(1), 21.
- Argyris, C. (1977). Double loop learning in organizations. Harvard Business Review, 55(5), 115-125.
- Argyris, C. & Schön, D. (1974). *Theory in practice: Increasing professional effectiveness*. San Francisco, USA: Jossey-Bass.
- Arora-Jonsson, S. (2016). Does resilience have a culture? Ecocultures and the politics of knowledge production. *Ecological Economics*, 121, 98-107. doi: 10.1016/j.ecolecon.2015.11.020
- Babich, R., & Smith, S. (2010). "Cradle to Grave": An Analysis of Sustainable Food Systems in a University Setting. Journal of Culinary Science & Technology, 8(4), 180-190. doi: 10.1080/15428052.2010.535747
- Bacon, C. M., Mulvaney, D., Ball, T. B., Melanie DuPuis, E., Gliessman, S. R., Lipschutz, R. D., & Shakouri, A.
 (2011). The creation of an integrated sustainability curriculum and student praxis projects. *International Journal of Sustainability in Higher Education*, 12(2), 193-208. doi: 10.1108/1467637111118237
- Bakker, R. M. (2010). Taking stock of temporary organizational forms: A systematic review and research agenda. *International Journal of Management Reviews*, 12(4), 466-486. doi: 10.1111/j.1468-2370.2010.00281.x
- Bakker, R. M., Cambré, B., Korlaar, L., & Raab, J. (2011). Managing the project learning paradox: A set-theoretic approach toward project knowledge transfer. *International Journal of Project Management*, 29(5), 494-503. doi: 10.1016/j.ijproman.2010.06.002
- Baldwin, S., & Chung, K. (2007). Sustainable disposal of edible food byproducts at university research farms. International Journal of Sustainability in Higher Education, 8(1), 69-85. doi: 10.1108/14676370710717607

- Barth, M., Godemann, J., Rieckmann, M., & Stoltenberg, U. (2007). Developing key competencies for sustainable development in higher education. *International Journal of Sustainability in Higher Education*, 8(4), 416-430. doi: 10.1108/14676370710823582
- Barthel, S., Folke, C., & Colding, J. (2010). Social–ecological memory in urban gardens—Retaining the capacity for management of ecosystem services. *Global Environmental Change*, 20(2), 255-265. doi: 10.1016/j.gloenvcha.2010.01.001
- Barton, S. S., Pineo, R. S., & Carter, L. (2010). Implementing Sustainable Practices on Campus through Student Independent Study. *HortTechnology*, 20(3), 495-498. doi: 10.21273/HORTTECH.20.3.495
- Barlett, P. F. (2011). Campus sustainable food projects: critique and engagement. *American Anthropologist*, 113(1), 101-115. doi: 10.1111/j.1548-1433.2010.01309.x
- BBC (2019). *Tomorrow's Food*. Retrieved on August 12, 2019, from https://www.bbc.co.uk/programmes/b06r3jh5
- Belfrage, C., & Hauf, F. (2017). The gentle art of retroduction: Critical realism, cultural political economy and critical grounded theory. *Organization Studies*, 38(2), 251-271. doi: 10.1177/0170840616663239
- Bendell, K., & Sylvestre, J. (2017). How different approaches to taking pictures influences participation in a photovoice project. *Action Research*, 15(4), 357-372. doi: 10.1177/1476750316653812
- Bendt, P., Barthel, S., & Colding, J. (2013). Civic greening and environmental learning in public-access community gardens in Berlin. *Landscape and Urban Planning*, 109(1), 18-30. doi: 10.1016/j.landurbplan.2012.10.003
- Bennett, E. M., Solan, M., Biggs, R., McPhearson, T., Norström, A. V., Olsson, P., ... & Carpenter, S. R. (2016).
 Bright spots: seeds of a good Anthropocene. *Frontiers in Ecology and the Environment*, 14(8), 441-448.
 doi: 10.1002/fee.1309
- Bergvall-Kareborn, B., & Stahlbrost, A. (2009). Living Lab: an open and citizen-centric approach for innovation. International Journal of Innovation and Regional Development, 1(4), 356-370. doi: 10.1504/IJIRD.2009.022727
- Berkes, F., Colding, J., & Folke, C. (2003). *Navigating social-ecological systems: Building resilience for complexity and change*. Cambridge, UK: Cambridge University.
- Berkes, F., & Ross, H. (2013). Community resilience: toward an integrated approach. *Society & Natural Resources*, 26(1), 5-20. doi: 10.1080/08941920.2012.736605
- Bessant, S. (2017). Exploring the interface of marketisation and education for sustainable development in English higher education (Doctoral thesis, Keele University, Keele, UK).
- Bessant, S. E., & Robinson, Z. P. (2019). Rating and rewarding higher education for sustainable development research within the marketised higher education context: experiences from English universities. *Environmental Education Research*, 25(4), 1-18. doi: 10.1080/13504622.2018.1542488
- Bhaskar, R. (1986). Scientific realism and human emancipation. Oxon, UK: Routledge.
- Bhaskar, R. (2008). A Realist Theory of Science (3rd ed.). Oxon, UK: Routledge.
- Biernbaum, J. A., Ngouajio, M., & Thorp, L. (2006). Development of a year-round student organic farm and organic farming curriculum at Michigan State University. *HortTechnology*, 16(3), 432-436. doi: 10.21273/HORTTECH.16.3.0432
- Biggs, R., Schlüter, M., Biggs, D., Bohensky, E. L., BurnSilver, S., Cundill, G., ... & Leitch, A. M. (2012). Toward principles for enhancing the resilience of ecosystem services. *Annual Review of Environment and Resources*, 37, 421-448. doi: 10.1146/annurev-environ-051211-123836

- Biggs, R., Gordon, L. J., Raudsepp-Hearne, C., Schlüter, M., & Walker, B. (2015a). Principle 3 Manage slow variables and feedbacks. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 105-141). Cambridge, UK: Cambridge University.
- Biggs, R., Schlüter, M., & Schoon, M. L. (2015b). An introduction to the resilience approach and principles to sustain ecosystem services in social-ecological systems. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 1-31). Cambridge, UK: Cambridge University.
- Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.) (2015c). *Principles for building resilience: Sustaining ecosystem services in social-ecological systems*. Cambridge, UK: Cambridge University.
- Bohensky, E. L., Evans, L. S., Anderies, J. M., Biggs, D., & Fabricius, C. (2015). Principle 3 Manage slow variables and feedbacks. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 142-173). Cambridge, UK: Cambridge University.
- Boser, S. (2006). Ethics and power in community-campus partnerships for research. *Action Research*, 4(1), 9–21. doi: 10.1177/1476750306060538
- Blok, V., Gremmen, B., & Wesselink, R. (2016). Dealing with the wicked problem of sustainability: The role of individual virtuous competence. *Business and Professional Ethics Journal*. doi: 10.5840/bpej201621737
- Bloom, B. S., Engelhart, M. D., Furst, E. J., Hill, W. H., & Krathwohl, D. R. (1956). *Taxonomy of educational objectives: The classification of educational goals*. Ann Arbor, USA: David McKay. Retrieved on August 3, 2019, from https://www.uky.edu/~rsand1/china2018/texts/Bloom%20et%20al%20-Taxonomy%20of%20Educational%20Objectives.pdf
- Bochner, A. P. (2000). Criteria against ourselves. *Qualitative Inquiry*, 6(2), 266-272. doi: 10.1177/107780040000600209
- Boyle, A., Maguire, S., Martin, A., Milsom, C., Nash, R., Rawlinson, S., ... & Conchie, S. (2007). Fieldwork is good: The student perception and the affective domain. *Journal of Geography in Higher Education*, 31(2), 299-317. doi: 10.1080/03098260601063628
- Bradbury, H. (2005). Learning with the Natural Step: Action Research to Promote Conversations for Sustainable Development. In Reason, P., & Bradbury, H. (Eds.). *Handbook of action research: Concise paperback edition* (pp. 236-242). London, UK: Sage.
- Brand, F., & Jax, K. (2007). Focusing the meaning(s) of resilience: resilience as a descriptive concept and a boundary object. *Ecology and Society*, 12(1), 23.
- Brandt, P., Ernst, A., Gralla, F., Luederitz, C., Lang, D. J., Newig, J., ... & Von Wehrden, H. (2013). A review of transdisciplinary research in sustainability science. *Ecological Economics*, 92, 1-15. doi: 10.1016/j.ecolecon.2013.04.008
- Breiting, S., Mayer, M., & Mogensen, F. (2005). *Quality Criteria for ESD-Schools; Guidelines to enhance the quality of Education for Sustainable Development*. Vienna, Austria: Austrian Federal Ministry of Education, Science and Culture.
- Breiting, S., & Mogensen, F. (1999). Action competence and environmental education. *Cambridge Journal of Education*, 29(3), 349-353. doi: 10.1080/0305764990290305
- Brewis, G. (2014). *A social history of student volunteering: Britain and beyond, 1880-1980*. New York, USA: Palgrave Macmillan.
- Brewis, G., & Holdsworth, C. (2011). University support for student volunteering in England: historical development and contemporary value. *Journal of Academic Ethics*, 9(2), 165-176. doi: 10.1007/s10805-011-9129-0

- Broman, G. I., & Robèrt, K. H. (2017). A framework for strategic sustainable development. *Journal of Cleaner Production*, 140, 17-31. doi: 10.1016/j.jclepro.2015.10.121
- Brookes, N., Sage, D., Dainty, A., Locatelli, G., & Whyte, J. (2017). An island of constancy in a sea of change: Rethinking project temporalities with long-term megaprojects. *International Journal of Project Management*, 35(7), 1213-1224. doi: 10.1016/j.ijproman.2017.05.007
- Brundiers, K., Wiek, A., & Redman, C. L. (2010). Real-world learning opportunities in sustainability: from classroom into the real world. *International Journal of Sustainability in Higher Education*, 11(4), 308-324. doi: 10.1108/14676371011077540
- Bruun Jensen, B., & Schnack, K. (1997). The action competence approach in environmental education. Environmental Education Research, 3(2), 163-178. doi: 10.1080/1350462970030205
- Brydon-Miller, M. (2009). Covenantal ethics and action research: Exploring a common foundation for social research. In Donna M. Mertens & Pauline E. Ginsberg (Eds.), *The Handbook of Social Research Ethics* (pp. 243-258). Thousand Oaks: USA, Sage.
- Bryman, A., Becker, S., & Sempik, J. (2008). Quality criteria for quantitative, qualitative and mixed methods research: A view from social policy. *International Journal of Social Research Methodology*, 11(4), 261-276. doi: 10.1080/13645570701401644
- Bryman, A. (2012). Social Research Methods (4th ed.). Oxford, UK: Oxford University.
- Bryman, A. (2016). Social Research Methods (5th ed.). Oxford, UK: Oxford University.
- Burnard, K., & Bhamra, R. (2011). Organisational resilience: development of a conceptual framework for organisational responses. *International Journal of Production Research*, 49(18), 5581-5599. doi: 10.1080/00207543.2011.563827
- Butler, C. (2018). Five steps to organisational resilience: Being adaptive and flexible during both normal operations and times of disruption. *Journal of Business Continuity & Emergency Planning*, 12(2), 103-112.
- Byers, R. D. (1999). Reaching out: A university botanical garden builds long-distance relationships. *HortTechnology*, 9(4), 573-576. doi: 10.21273/HORTTECH.9.4.573
- Cambridge Green Challenge, The (n.d.). *Living Laboratory Annual Report 2012-2013*. Retrieved on July 12, 2018, from https://www.environment.admin.cam.ac.uk/files/living_laboratory_annual_report_2012-2013.pdf
- Cannon-Bowers, J. A., Salas, E., & Converse, S. (1993). Shared Mental Models in Expert Team Decision Making. In Castellan, J. N. (Ed.), *Individual and Group Decision Making: Current Issues* (pp. 221-246). Hillsdale, USA: Lawrence Erlbaum Associates.
- Capaldi, C. A., Dopko, R. L., & Zelenski, J. M. (2014). The relationship between nature connectedness and happiness: a meta-analysis. *Frontiers in Psychology*, 5, 976. doi: 10.3389/fpsyg.2014.00976
- Carpenter, S. R., Arrow, K. J., Barrett, S., Biggs, R., Brock, W. A., Crépin, A. S., ... & Li, C. Z. (2012). General resilience to cope with extreme events. *Sustainability*, 4(12), 3248-3259. doi: 10.3390/su4123248
- Catalani, C., & Minkler, M. (2010). Photovoice: A review of the literature in health and public health. *Health Education & Behavior*, 37(3), 424-451. doi: 10.1177/1090198109342084
- Chapin, F. S., Trainor, S. F., Huntington, O., Lovecraft, A. L., Zavaleta, E., Natcher, D. C., ... & Fresco, N. (2008). Increasing wildfire in Alaska's boreal forest: pathways to potential solutions of a wicked problem. *AIBS Bulletin*, 58(6), 531-540. doi: 10.1641/B580609
- Chapin, F. S., Kofinas, G. P., & Folke, C. (2009). *Principles of Ecosystem Stewardship: Resilience-Based Natural Resource Management in a Changing World*. New York, USA: Springer.

- Cilliers, P., Biggs, H. C., Blignaut, S., Choles, A. G., Hofmeyr, J. H. S., Jewitt, G. P., & Roux, D. J. (2013). Complexity, modelling, and natural resource management. *Ecology and Society*, 18(3). doi: 10.5751/ES-05382-180301
- Clark, W. C. (2007). Sustainability Science: A room of its own. *Proceedings of the National Academy of the Sciences*, 104(6), 1737-1738. doi: 10.1073/pnas.0611291104
- Clark, W. C., & Dickson, N. M. (2003). Sustainability science: the emerging research program. *Proceedings of the National Academy of Sciences*, 100(14), 8059-8061. doi: 10.1073/pnas.1231333100
- Clarke, K. A. (2009). Uses of a research diary: learning reflectively, developing understanding and establishing transparency. *Nurse Researcher*, 17(1), 68-76. doi: 10.7748/nr2009.10.17.1.68.c7342
- Coghlan, D. and Brannick, T. (2005). *Doing Action Research in Your Own Organization* (2nd ed.). London, UK: Sage.
- Coghlan, D., & Brannick, T. (2014). Doing Action Research in Your Own Organization (4th ed.). London, UK: Sage.
- Coghlan, D. (2019). Doing Action Research in Your Own Organization (5th ed.). London, UK: Sage.
- Cohen, N. (2010). Designing the sustainable foodshed: A cross-disciplinary undergraduate environmental studies course. *Innovative Higher Education*, 35(1), 51-60. doi: 10.1007/s10755-009-9126-z
- Colding, J. & Barthel, S. (2017). The Role of University Campuses in Reconnecting Humans to the Biosphere. *Sustainability*, 9(12), 2349. doi: 10.3390/su9122349
- Colding, J., & Barthel, S. (2019). Exploring the social-ecological systems discourse 20 years later. *Ecology and Society* 24(1), 2. doi: 10.5751/ES-10598-240102
- Cook, K., & Quigley, C. F. (2013). Connecting to our community: Utilizing photovoice as a pedagogical tool to connect college students to science. *International Journal of Environmental and Science Education*, 8(2), 339-357. doi: 10.12973/ijese.2013.205a
- Cook, T. (1998). The importance of mess in action research. *Educational Action Research*, 6(1), 93-109. doi: 10.1080/09650799800200047
- Cook, T. (2009). The purpose of mess in action research: building rigour though a messy turn. *Educational Action Research*, 17(2), 277-291. doi: 10.1080/09650790902914241
- Coombs, P. (1973). Non-formal Education for Rural Development: Strengthening learning opportunities for Children and Youth. New York, USA: ICED.
- Corcoran, P. B., Walker, K. E., & Wals, A. E. (2004). Case studies, make-your-case studies, and case stories: a critique of case-study methodology in sustainability in higher education. *Environmental Education Research*, 10(1), 7-21. doi: 10.1080/1350462032000173670
- Coser, L. A. (1992). The revival of the sociology of culture: The case of collective memory. *Sociological Forum*, (7)2, 365-373.
- Costanza, R., d'Arge, R., De Groot, R., Farber, S., Grasso, M., Hannon, B., ... & Raskin, R. G. (1997). The value of the world's ecosystem services and natural capital. *Nature*, 387(6630), 253.
- Cote, M., & Nightingale, A. J. (2012). Resilience thinking meets social theory: situating social change in socioecological systems (SES) research. *Progress in Human Geography*, 36(4), 475-489. doi: 10.1177/0309132511425708
- Cresswell, J. W. (2014). *Research design: qualitative, quantitative, and mixed methods approaches*. Los Angeles, USA: Sage.
- Cretney, R. (2014). Resilience for whom? Emerging critical geographies of socio-ecological resilience. *Geography Compass*, 8(9), 627-640. doi: 10.1111/gec3.12154

- Crichton, S., & Childs, E. (2005). Clipping and coding audio files: A research method to enable participant voice. International Journal of Qualitative Methods, 4(3), 40-49. doi: 10.1177/160940690500400303
- Crotty, M. (1998). *The Foundations of Social Research: Meaning and Perspective in the Research Process*. London, UK: Sage.
- Cundill, G., Leitch, A. M., Armitage, D., & Peterson, G. D. (2015). Principle 5 Encourage learning. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 174-200). Cambridge, UK: Cambridge University.
- Dakos, V., Quinlan, A. E., Jacopo, A. B., Bennett, E. M., Bodin, Ö., & Burnsilver, S. (2015). Principle 2 Manage connectivity. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 80-104). Cambridge, UK: Cambridge University.
- Danielsen, F., Burgess, N. D., & Balmford, A. (2005). Monitoring matters: examining the potential of locally-based approaches. *Biodiversity & Conservation*, 14(11), 2507-2542. doi: 10.1007/s10531-005-8375-0
- Darwen, J., & Grace Rannard, A. (2011). Student volunteering in England: a critical moment. *Education + Training*, 53(2/3), 177-190. doi: 10.1108/00400911111115717
- Datta, R. (2016). Community garden: A bridging program between formal and informal learning. *Cogent Education*, 3(1), 1177154. doi: 10.1080/2331186X.2016.1177154
- de Haan, G. (2006). The BLK '21' programme in Germany: a 'Gestaltungskompetenz'-based model for Education for Sustainable Development. *Environmental Education Research*, 12(1), 19-32. doi: 10.1080/13504620500526362
- De Medeiros, J. F., Ribeiro, J. L. D., & Cortimiglia, M. N. (2014). Success factors for environmentally sustainable product innovation: a systematic literature review. *Journal of Cleaner Production*, 65, 76-86. doi: 10.1016/j.jclepro.2013.08.035
- Delate, K. (2006). Incorporating organic and agroecological approaches into the university curricula: The Iowa State University graduate program in sustainable agriculture. *HortTechnology*, 16(3), 445-448.
- Delind, L. B. (2006). Of bodies, place, and culture: Re-situating local food. *Journal of Agricultural and Environmental Ethics*, 19(2), 121-146. doi: 10.1007/s10806-005-1803-z
- Denrell, J. (2003). Vicarious learning, undersampling of failure, and the myths of management. *Organization Science*, 14(3), 227-243.
- Design Trust for Public Space (n.d). *Data Collection Toolkit, Farming Concrete BARN*. Retreived May 1, 2015 from http://farmingconcrete.org/barn/data-collection-toolkit/
- Dewey, J. (1897). My Pedagogic Creed. Journal of the National Education Association, 18(9), 291–295.
- Dick, B. (2014). Case Study. In Coghlan, D. & Brydon-Miller, M. (Eds.) *The Sage Encyclopedia of Action Research*, Volume 1 (pp. 86-89). London, UK: Sage.
- Doyle, R., & Krasny, M. (2003). Participatory rural appraisal as an approach to environmental education in urban community gardens. *Environmental Education Research*, 9(1), 91-115. doi: 10.1080/13504620303464
- Draper, C., & Freedman, D. (2010). Review and analysis of the benefits, purposes, and motivations associated with community gardening in the United States. *Journal of Community Practice*, 18(4), 458-492. doi: 10.1080/10705422.2010.519682
- Easterbrook, P. J., Gopalan, R., Berlin, J. A., & Matthews, D. R. (1991). Publication bias in clinical research. *The Lancet*, 337(8746), 867-872. doi: 10.1016/0140-6736(91)90201-Y

- Economic and Social Research Council (ESRC) (2015). *ESRC Framework for Research Ethics*. Retrieved on August 12, 2019, from https://esrc.ukri.org/files/funding/guidance-for-applicants/esrc-framework-for-research-ethics-2015/
- Einerson, M. J. (1998). "Do Ya Wanna Dance?": Collaborating with and Empowering Preadolescent Girls in Feminist Interpretive Research. Frontiers: A Journal of Women Studies, 19(3), 42-57.
- Eizenberg, E. (2012). The changing meaning of community space: Two models of NGO management of community gardens in New York City. *International Journal of Urban and Regional Research*, 36(1), 106-120. doi: 10.1111/j.1468-2427.2011.01065.x
- Egmose, J. (2015). Action Research for Sustainability: Social Imagination Between Citizens and Scientists. Surrey, UK: Ashgate.
- Elder, G. H. (1998). The life course as developmental theory. *Child Development*, 69(1), 1-12. doi: 10.1111/j.1467-8624.1998.tb06128.x
- Elliott, J. (1991). Action Research for Educational Change. Buckingham, USA: Open University.
- Elmqvist, T. (2017). Development: Sustainability and resilience differ. *Nature*, 546(7658), 352.
- Enfors-Kautsky, E., Järnberg, L., Quinlan, A, & Ryan, P. (2018). *Wayfinder: a resilience guide for navigating towards sustainable futures*. GRAID programme, Stockholm Resilience Center. Retrieved on October 3, 2018, from www.wayfinder.earth
- Engin, M. (2011). Research diary: A tool for scaffolding. *International Journal of Qualitative Methods*, 10(3), 296-306. doi: 10.1177/160940691101000308
- England, K. V. (1994). Getting personal: Reflexivity, positionality, and feminist research. *The Professional Geographer*, 46(1), 80-89. doi: 10.1111/j.0033-0124.1994.00080.x
- English Association of Universities and Colleges (EAUC) (2018). *Living Labs Opportunities, Benefits and Challenges of Different Models Globally*. English Association of Universities and Colleges. Retrieved on November 7, 2018, from http://www.eauc.org.uk/living_labs_opportunities_benefits_and_challeng
- Evans, A., Ranjit, N., Rutledge, R., Medina, J., Jennings, R., Smiley, A., ... & Hoelscher, D. (2012). Exposure to multiple components of a garden-based intervention for middle school students increases fruit and vegetable consumption. *Health Promotion Practice*, 13(5), 608-616. doi: 10.1177/1524839910390357
- Evers, A., & Hodgson, N. L. (2011). Food choices and local food access among Perth's community gardeners. Local Environment, 16(6), 585-602. doi: 10.1080/13549839.2011.575354
- Extinction Rebellion (2019). Extinction Rebellion. Retrieved on August 14, 2019, from https://rebellion.earth/
- Fairclough, N., Jessop, B., & Sayer, A. (2002). Critical realism and semiosis. *Alethia*, 5(1), 2-10. doi: 10.1558/aleth.v5i1.2
- Falk, C. L., Pao, P., & Cramer, C. S. (2005). Teaching diversified organic crop production using the community supported agriculture farming system model. *Journal of Natural Resources and Life Sciences Education*, 34, 8-12.
- Faulkner, S. L., & Trotter, S. P. (2017). Data Saturation. In Donsbach, W. (Ed.) *The International Encyclopedia of Communication Research Methods*. doi: 10.1002/9781118901731.iecrm0060
- Feldman, A. (2003). Validity and quality in self-study. *Educational Researcher*, 32(3), 26-28. doi: 10.3102/0013189X032003026
- Feldman, A. (2007). Validity and quality in action research. *Educational Action Research*, 15(1), 21-32. doi: 10.1558/aleth.v5i1.2

- Ferguson, J. J., Lamb, E., & Swisher, M. (2006). Developing an interdisciplinary organic and sustainable agriculture curriculum at the University of Florida. *HortTechnology*, 16(3), 436-438.
- Fien, J., & Skoien, P. (2002). "I'm Learning... How You Go about Stirring Things Up—in a Consultative Manner": Social capital and action competence in two community catchment groups. *Local Environment*, 7(3), 269-282. doi: 10.1080/1354983022000001642
- Fink, H. S. (2011). Promoting behavioral change towards lower energy consumption in the building sector. Innovation: The European Journal of Social Science Research, 24(1-2), 7-26. doi: 10.1080/13511610.2011.586494
- Finlay, L. (2002). Negotiating the swamp: the opportunity and challenge of reflexivity in research practice. *Qualitative Research*, 2(2), 209-230. doi: 10.1177/146879410200200205
- Fischman, M. W. (2000). Informed consent. In B. D.Sales & S.Folkman (Eds.), *Ethics in research with human participants* (pp. 35–48). Washington, DC: American Psychological Association.
- Fischer, L. R., & Schaffer, K. B. (1993). Older volunteers: A guide to research and practice. California, USA: Sage.
- Fleetwood, S. (2014). Bhaskar and critical realism. In Adler, P., Du Gay, P., Morgan, G. and Reed, M., (Eds.),
 Oxford Handbook of Sociology, Social Theory and Organisation Studies: Contemporary Currents (pp. 182-219). Oxford, UK: Oxford University. Retrieved on August 12, 2019, from http://eprints.uwe.ac.uk/26526
- Flood, R. L., & Romm, N. R. (1996). Diversity management: Triple loop learning. Chichester, UK: Wiley.
- Flynn, K. C. (2001). Urban agriculture in Mwanza, Tanzania. *Africa*, 71(4), 666-691. doi: 10.3366/afr.2001.71.4.666
- Foley, J. A., Ramankutty, N., Brauman, K. A., Cassidy, E. S., Gerber, J. S., Johnston, M., ... & Balzer, C. (2011). Solutions for a cultivated planet. *Nature*, 478(7369), 337. doi: 10.1038/nature10452
- Folke, C. (2006). Resilience: The emergence of a perspective for social–ecological systems analyses. *Global Environmental Change*, 16(3), 253-267. doi: 10.1016/j.gloenvcha.2006.04.002
- Folke, C. (2016). Resilience (Republished). Ecology and Society, 21(4), 44. doi: 10.5751/ES-09088-210444
- Folke, C., Biggs, R., Norström, A. V., Reyers, B., & Rockström, J. (2016). Social-ecological resilience and biospherebased sustainability science. *Ecology and Society*, 21(3). doi: 10.5751/ES-08748-210341
- Folke, C., Carpenter, S., Elmqvist, T., Gunderson, L., Holling, C. S., & Walker, B. (2002). Resilience and sustainable development: building adaptive capacity in a world of transformations. *AMBIO: A Journal of the Human Environment*, 31(5), 437-440. doi: 10.1579/0044-7447-31.5.437
- Folke, C., Colding, J., & Berkes, F. (2003). Synthesis: building resilience and adaptive capacity in social-ecological systems. In Berkes, F., Colding, J., & Folke, C. (Eds.), *Navigating social-ecological systems: Building resilience for complexity and change* (pp. 352-387). Cambridge, UK: Cambridge University.
- Forbes, J. (2008). Reflexivity in professional doctoral research. *Reflective Practice*, 9(4), 449-460. doi: 10.1080/14623940802431523
- Frame, B. (2008). 'Wicked', 'messy', and 'clumsy': long-term frameworks for sustainability. *Environment and Planning C: Government and Policy*, 26(6), 1113-1128. doi: 10.1068/c0790s
- Franco, A., Malhotra, N., & Simonovits, G. (2014). Publication bias in the social sciences: Unlocking the file drawer. *Science*, 345(6203), 1502-1505. doi: 10.1126/science.1255484
- Freire, P. (2000). Pedagogy of the oppressed. New York, USA: Continuum.
- French, W. L., & Bell, C. H. (1999). Organization development: Behavioral science interventions for organization improvement. Upper Saddle River, USA: Prentice-Hall.

- Fritze, J. G., Blashki, G. A., Burke, S., & Wiseman, J. (2008). Hope, despair and transformation: climate change and the promotion of mental health and wellbeing. *International Journal of Mental Health Systems*, 2(1), 13. doi: 10.1186/1752-4458-2-13
- Gaboardi, M., Zuccalà, G., Lenzi, M., Ferrari, S., & Santinello, M. (2018). Changing the method of working with homeless people: a photovoice project in Italy. *Journal of Social Distress and the Homeless*, 27(1), 53-63. doi: 10.1080/10530789.2018.1446407
- Galt, R. E., Parr, D., Kim, J. V. S., Beckett, J., Lickter, M., & Ballard, H. (2013). Transformative food systems education in a land-grant college of agriculture: the importance of learner-centered inquiries. *Agriculture and Human Values*, 30(1), 129-142. doi: 10.1007/s10460-012-9384-8
- Gardiner, M. M., Prajzner, S. P., Burkman, C. E., Albro, S., & Grewal, P. S. (2014). Vacant land conversion to community gardens: influences on generalist arthropod predators and biocontrol services in urban greenspaces. *Urban Ecosystems*, 17(1), 101-122. doi: 10.1007/s11252-013-0303-6
- Garg, A. X., Hackam, D., & Tonelli, M. (2008). Systematic review and meta-analysis: when one study is just not enough. *Clinical Journal of the American Society of Nephrology*, 3(1), 253-260. doi: 10.2215/CJN.01430307
- Garrison, K., & Munday, N. K. (2012). Toward Authentic Dialogue: Origins of The Fishbowl Method and Implications for Writing Center Work. *Praxis: A Writing Center Journal*, 9(1), 2-7.
- Geertz, C. (1973). Thick description: Toward an interpretive theory of culture. *The Interpretation of Cultures: Selected Essays* (pp. 3-30), New York USA: Basic Books.
- Giddings, B., Hopwood, B., & O'brien, G. (2002). Environment, economy and society: fitting them together into sustainable development. *Sustainable Development*, 10(4), 187-196. doi: 10.1002/sd.199
- Gill, R. (2009). Breaking the silence: The hidden injuries of neo-liberal academia. In Flood, R., & Gill, R. (Eds.) Secrecy and silence in the research process: Feminist Reflections. London, UK: Routledge.

Global Alliance for Monitoring Learning (GAML) (2017). *Measurement strategy for* SDG Target 4.7 - Proposal by GAML Task Force 4.7. Retrieved on July 12, 2018, from http://uis.unesco.org/sites/default/files/documents/gaml4-measurement-strategy-sdg-target4.7.pdf

- Gongaware, T. B. (2003). Collective memories and collective identities: maintaining unity in Native American educational social movements. *Journal of Contemporary Ethnography*, 32(5), 483-520. doi: 10.1177/0891241603255674
- Greenwood, D. J., & Levin, M. (2008). Introduction to action research: Social research for social change (2nd ed.). California, USA: Sage.
- Guitart, D. A., Byrne, J. A., & Pickering, C. M. (2015). Greener growing: assessing the influence of gardening practices on the ecological viability of community gardens in South East Queensland, Australia. *Journal of Environmental Planning and Management*, 58(2), 189-212. doi: 10.1080/09640568.2013.850404
- Guitart, D., Pickering, C., and Byrne, J. (2012). Past results and future directions in urban community gardens research. *Urban Forestry & Urban Greening*, 11, 364-373. doi: 10.1016/j.ufug.2012.06.007
- Guitart, D. A., Pickering, C. M., & Byrne, J. A. (2014). Color me healthy: Food diversity in school community gardens in two rapidly urbanising Australian cities. *Health & Place*, 26, 110-117. doi: 10.1016/j.healthplace.2013.12.014
- Gunderson, L. H. (2000). Ecological resilience-in theory and application. *Annual Review of Ecology and Systematics*, 31, 425-439. doi: 10.1146/annurev.ecolsys.31.1.425
- Gunderson, L. (2010). Ecological and human community resilience in response to natural disasters. *Ecology and Society*, 15(2), 19.

- Gunderson, L. H., & Holling, C. S. (2002). *Panarchy: understanding transformations in systems of humans and nature*. Washington, DC: Island Press.
- Habermas, J. (1984). *The Theory of Communicative Action: Reason and the Rationalization of Society*. Cambridge, UK: Polity.
- Habermas, J. (1987). *The Theory of Communicative Action: The Critique of Functionalist Reason*. Oxford, UK: Blackwell.
- Haggar, P., Whitmarsh, L., & Skippon, S. M. (2019). Habit discontinuity and student travel mode choice. *Transportation Research Part F: Traffic Psychology and Behaviour*, 64, 1-13. doi: 10.1016/j.trf.2019.04.022
- Halberstam, J. (2011). The Queer Art of Failure. Durham, USA: Duke University.
- Hallstedt, S. I. (2017). Sustainability criteria and sustainability compliance index for decision support in product development. *Journal of Cleaner Production*, 140, 251-266. doi: 10.1016/j.jclepro.2015.06.068
- Hamann, R., Giamporcaro, S., Johnston, D., & Yachkaschi, S. (2011). The role of business and cross-sector collaboration in addressing the 'wicked problem' of food insecurity. *Development Southern Africa*, 28(4), 579-594. doi: 10.1080/0376835X.2011.605581
- Hamari, J., Sjöklint, M., & Ukkonen, A. (2016). The sharing economy: Why people participate in collaborative consumption. *Journal of the Association for Information Science and Technology*, 67(9), 2047-2059. doi: 10.1002/asi.23552
- Hamilton, S. L. (1999). The roles of the University of Tennessee gardens in a public horticulture teaching program. *HortTechnology*, 9(4), 552-556. doi: 10.21273/HORTTECH.9.4.552
- Harrowell, E., Davies, T., & Disney, T. (2018). Making space for failure in geographic research. *The Professional Geographer*, 70(2), 230-238. doi: 10.1080/00330124.2017.1347799
- Haynes, C., & Trexler, C. J. (2003). The perceptions and needs of volunteers at a university-affiliated public garden. *HortTechnology*, 13(3), 552-556. doi: 10.21273/HORTTECH.13.3.0552
- Hector, D. C., Christensen, C. B., & Petrie, J. (2014). Sustainability and sustainable development: Philosophical distinctions and practical implications. *Environmental Values*, 23(1), 7-28. doi: 10.3197/096327114X13851122268963
- Hedefalk, M., Almqvist, J., & Lidar, M. (2014). Teaching for action competence. *Sage Open*, 4(3), 2158244014543785. doi: 10.1177/2158244014543785
- Heikkinen, H. L., Huttunen, R., & Syrjälä, L. (2007). Action research as narrative: five principles for validation. *Educational Action Research*, 15(1), 5-19. doi: 10.1080/09650790601150709
- Helmig, B., Ingerfurth, S., & Pinz, A. (2014). Success and failure of nonprofit organizations: Theoretical foundations, empirical evidence, and future research. *Voluntas: International Journal of Voluntary and Nonprofit Organizations*, 25(6), 1509-1538. doi: 10.1007/s11266-013-9402-5
- Heron, J., & Reason, P. (1997). A participatory inquiry paradigm. *Qualitative Inquiry*, 3(3), 274-294. doi:10.1177/107780049700300302
- Heron, J., & Reason, P. (2008). Extending epistemology within a co-operative inquiry. In P. Reason & Bradbury, H.
 (Eds.) *The Sage Handbook of Action Research: Participative Inquiry and Practice* (pp. 366-380). London, UK: Sage.
- Higher Education Funding Council for England (HEFCE) (2001). *HEFCE Higher Education Active Community Fund Guidance and Allocations*. 01/65, London, November.

- Hiles, D. R. (2008). Transparency. In Given, L. M. (Ed.) *The SAGE Encyclopedia of Qualitative Research Methods* (pp. 890-892). Thousand Oaks, USA: Sage.
- Hill, A. (2013). The place of experience and the experience of place: Intersections between sustainability education and outdoor learning. *Australian Journal of Environmental Education*, 29(1), 18-32. doi: 10.1017/aee.2013.13
- Hilsen, A. I. (2006). And they shall be known by their deeds: Ethics and politics in action research. Action Research, 4(1), 23-36. doi: 10.1177/1476750306060539
- Hirsch Hadorn, G. H., Bradley, D., Pohl, C., Rist, S., & Wiesmann, U. (2006). Implications of transdisciplinarity for sustainability research. *Ecological Economics*, 60(1), 119-128. doi: 10.1016/j.ecolecon.2005.12.002
- Hjorth, P., & Bagheri, A. (2006). Navigating towards sustainable development: A system dynamics approach. *Futures*, 38(1), 74-92. doi: 10.1016/j.futures.2005.04.005
- Hockenberry Meyer, M., Hokanson, S., Galatowitsch, S., & Luby, J. (2010). Public gardens: Fulfilling the university's research mission. *HortTechnology*, 20(3), 522-527. doi: 10.21273/HORTTECH.20.3.522
- Hoffman, A. J., Morales Knight, L. F., & Wallach, J. (2007). Gardening activities, education, and self-esteem: Learning outside the classroom. *Urban Education*, 42(5), 403-411. doi: 10.1177/0042085907304909
- Holdsworth, C. (2010). Why volunteer? Understanding motivations for student volunteering. *British Journal of Educational Studies*, 58(4), 421-437. doi: 10.1080/00071005.2010.527666
- Holdsworth, C., & Quinn, J. (2010). Student volunteering in English higher education. *Studies in Higher Education*, 35(1), 113-127. doi: 10.1080/03075070903019856
- Holdsworth, C., & Quinn, J. (2012). The epistemological challenge of higher education student volunteering: "reproductive" or "deconstructive" volunteering?. *Antipode*, 44(2), 386-405. doi: 10.1111/j.1467-8330.2011.00844.x
- Holland, L. (2004). Diversity and connections in community gardens: a contribution to local sustainability. *Local Environment*, 9(3), 285-305. doi: 10.1080/1354983042000219388
- Holling, C. S. (1973). Resilience and stability of ecological systems. *Annual Review of Ecology and Systematics*, 4(1), 1-23. doi: 10.1146/annurev.es.04.110173.000245
- Howlett, S. (2010). Developing volunteer management as a profession. *Voluntary Sector Review*, 1(3), 355-360. doi: 10.1332/204080510X538338
- Hughes, T. P., Huang, H. U. I., & Young, M. A. (2013). The wicked problem of China's disappearing coral reefs. *Conservation Biology*, 27(2), 261-269. doi: 10.1111/j.1523-1739.2012.01957.x
- Hustinx, L., Vanhove, T., Declercq, A., Hermans, K., & Lammertyn, F. (2005). Bifurcated commitment, priorities, and social contagion: the dynamics and correlates of volunteering within a university student population. *British Journal of Sociology of Education*, 26(4), 523-538. doi: 10.1080/01425690500200111
- Hyde, M. K., Dunn, J., Bax, C., & Chambers, S. K. (2016). Episodic volunteering and retention: An integrated theoretical approach. *Nonprofit and Voluntary Sector Quarterly*, 45(1), 45-63. doi: 10.1177/0899764014558934
- Ioannidis, J. P. (1998). Effect of the statistical significance of results on the time to completion and publication of randomized efficacy trials. *Jama*, 279(4), 281-286. doi: 10.1001/jama.279.4.281
- Ibert, O. (2004). Projects and firms as discordant complements: organisational learning in the Munich software ecology. *Research Policy*, 33(10), 1529-1546. doi: 10.1016/j.respol.2004.08.010
- Intergovernmental Panel on Climate Change (IPCC) (2019). *Climate Change and Land: An IPCC Special Report on climate change, desertification, land degradation, sustainable land management, food security, and*

greenhouse gas fluxes in terrestrial ecosystems. Retrieved on August 13, 2019, from https://www.ipcc.ch/site/assets/uploads/2019/08/4.-SPM_Approved_Microsite_FINAL.pdf

- Ison, R. L. (2008). Systems thinking and practice for action research. In: Reason, P. W. and Bradbury, H. (Eds.). *The Sage Handbook of Action Research Participative Inquiry and Practice* (2nd Ed.) (pp. 139-158). London, UK: Sage.
- Jacobsen, K., Niewolny, K., Schroeder-Moreno, M., Van Horn, M., Harmon, A. H., Chen Fanslow, Y., ... & Parr, D. (2012). Sustainable agriculture undergraduate degree programs: A land-grant university mission. *Journal* of Agriculture, Food Stuffs, and Community Development, 2(3), 13-26. doi: 10.5304/jafscd.2012.023.004
- Jerneck, A., Olsson, L., Ness, B., Anderberg, S., Baier, M., Clark, E., ... & Persson, J. (2011). Structuring sustainability science. *Sustainability Science*, 6(1), 69-82. doi: 10.1007/s11625-010-0117-x
- Johansson, A. W., & Lindhult, E. (2008). Emancipation or workability? Critical versus pragmatic scientific orientation in action research. *Action Research*, 6(1), 95-115. doi: 10.1177/1476750307083713
- Johnston, L., Collins, B. L., Boyle, A., & Womack, H. D. (2012). Looking at sustainability through a different LENS. *Sustainability: The Journal of Record*, 5(4), 244-247. doi: 10.1089/sus.2012.9943
- Jucker, R. (2011). ESD between systemic change and bureaucratic obfuscation: Some reflections on environmental education and education for sustainable development in Switzerland. *Journal of Education for Sustainable Development*, 5(1), 39-60. doi: 10.1177/097340821000500109
- Karaan, A. S. M., & Mohamed, N. (1998). The performance and support of food gardens in some townships of the Cape Metropolitan Area: An evaluation of Abalimi Bezekhaya. *Development Southern Africa*, 15(1), 67-83. doi: 10.1080/03768359808439996
- Kasa, S. (2008). Globalizing Unsustainable Food Consumption: Trade Policies, Producer Lobbies, Consumer Preferences, and Beef Consumption in Northeast Asia. *Globalizations*, 5(2), 151-163. doi: 10.1080/14747730802057480
- Kates, R. W., Clark, W. C., Corell, R., Hall, J. M., Jaeger, C. C., Lowe, I., ... & Faucheux, S. (2001). Sustainability Science. *Science*, 292(5517), 641-642. doi: 10.1126/science.1059386
- Keck, M., & Sakdapolrak, P. (2013). What is social resilience? Lessons learned and ways forward. *Erdkunde*, 67(1), 5-19. doi: 10.3112/erdkunde.2012.01.02
- Keele University (n.d.). The KUSP. Retrieved on August 8, 2019, from https://www.keele.ac.uk/kusp/
- KeeleSU (n.d.). KeeleSU Volunteering Awards. Retrieved on August 8, 2019, from https://keelesu.com/activities/volunteering/awards/
- Keessen, A. M., Hamer, J. M., Van Rijswick, H. F., & Wiering, M. A. (2013). The concept of resilience from a normative perspective: examples from Dutch adaptation strategies. *Ecology & Society*, 18(2), 45. doi: 10.5751/ES-05526-180245
- Keightley, E., Pickering, M., & Allett, N. (2012). The self-interview: A new method in social science research. International Journal of Social Research Methodology, 15(6), 507-521. doi: 10.1080/13645579.2011.632155
- Kemmis, S. (2008). Critical theory and participatory action research. In P. Reason & Bradbury, H. (Eds.) *The Sage Handbook of Action Research: Participative Inquiry and Practice* (pp. 121-138). London, UK: Sage.
- Kesebir, S., & Kesebir, P. (2017). A growing disconnection from nature is evident in cultural products. *Perspectives on Psychological Science*, 12(2), 258-269. doi: 10.1177/1745691616662473
- Khanlou, N., & Peter, E. (2005). Participatory action research: considerations for ethical review. *Social Science & Medicine*, 60(10), 2333-2340. doi: 10.1016/j.socscimed.2004.10.004

- Kingsley, J. Y., Townsend, M., & Henderson-Wilson, C. (2009). Cultivating health and wellbeing: members' perceptions of the health benefits of a Port Melbourne community garden. *Leisure Studies*, 28(2), 207-219. doi: 10.1080/02614360902769894
- Kneafsey, M. & Hasanov, M. (forthcoming). Community Self-Organisation, Sustainability, and Resilience in Food Systems [special issue]. *Sustainability*.
- Kneen, B. (1999). Farmageddon: Food and the culture of biotechnology. Gabriola Island, Canada: New Society.
- Kobayashi, A. (2003). GPC ten years on: Is self-reflexivity enough?. *Gender, Place and Culture*, 10(4), 345-349. doi: 10.1080/0966369032000153313
- Kobayashi, K. D., Radovich, T. J., & Moreno, B. E. (2010). A tropical perspective on environmental sustainability in horticultural education. *HortTechnology*, 20(3), 503-508. doi: 10.21273/HORTTECH.20.3.503
- Kolb, D. (1984). *Experiential learning as the science of learning and development*. Englewood Cliffs, USA: Prentice Hall.
- Komiyama, H., & Takeuchi, K. (2006). Sustainability science: building a new discipline. *Sustainability Science*, 1(1), 1-6. doi: 10.1007/s11625-006-0007-4
- Kotschy, K., Biggs, R., Daw, T. M., Folke, C., West, P. C. (2015). Principle 1 Maintain diversity and redundancy. In:
 Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem* Services in Social-Ecological Systems (p. 50-79). Cambridge, UK: Cambridge University Press.
- Koutsouris, A. (2010, June 28-July 1). The riddle of knowledge forms and the 'paradox' of participation. In *Innovation and Sustainable Development in Agriculture and Food* Conference Proceedings, Montpellier, France (pp. 1-10).
- Krasny, M. E., & Tidball, K. G. (2009). Community gardens as contexts for science, stewardship, and civic action learning. *Cities and the Environment (CATE)*, 2(1), 8.
- Kusenbach, M. (2016). The go-along method. In Schwanhäußer, A. (Ed.) *Sensing the City: A Companion to Urban Anthropology* (pp. 154-158). Basel, Switzerland: Birkhauser.
- Kusenbach, M. (2018). Go-alongs. In Flick, U. (Ed.) *The SAGE Handbook of Qualitative Data Collection* (pp. 344-361). London, UK: Sage.
- Kyburz-Graber, R. (2004). Does case-study methodology lack rigour? The need for quality criteria for sound casestudy research, as illustrated by a recent case in secondary and higher education. *Environmental Education Research*, 10(1), 53-65. doi: 10.1080/1350462032000173706
- Lankenau, G. R. (2018). Fostering connectedness to nature in higher education. *Environmental Education Research*, 24(2), 230-244. doi: 10.1080/13504622.2016.1225674
- Lang, D. J., Wiek, A., Bergmann, M., Stauffacher, M., Martens, P., Moll, P., ... & Thomas, C. J. (2012). Transdisciplinary research in sustainability science: practice, principles, and challenges. *Sustainability Science*, 7(1), 25-43. doi: 10.1007/s11625-011-0149-x
- Lave, J., Wenger, E., & Wenger, E. (1991). *Situated learning: Legitimate peripheral participation*. Cambridge, UK: Cambridge University.
- Laycock, R. (2013). The tip of the iceberg lettuce: what direct and indirect factors enable knowledge and skill sharing in community gardens? (Master Thesis, Lund University, Sweden).
- Laycock Pedersen, R., & Robinson, Z. (2018). Reviewing University Community Gardens for Sustainability: taking stock, comparisons with urban community gardens and mapping research opportunities. *Local Environment*, 23(6), 652-671. doi: 10.1080/13549839.2018.1463210

- Leclerc, G., Bah, A., Barbier, B., Boutinot, L., Botta, A., Daré, W. S., ... & Toure, I. (2009). Managing tricky decentralised competencies: case study of a participatory modelling experiment on land use in the Lake Guiers area in Northern Senegal. *Sustainability Science*, 4(2), 243. doi: 10.1007/s11625-009-0083-3
- Lee, C. J., Sugimoto, C. R., Zhang, G., & Cronin, B. (2013). Bias in peer review. *Journal of the American Society for Information Science and Technology*, 64(1), 2-17. doi: 10.1002/asi.22784
- Leggatt-Cook, C. (2011). An uncertain balance: Negotiating theory, politics and love in academic writing. *Feminism & Psychology*, 21(3), 393-410. doi: 10.1177/0959353510370029
- Leigh Star, S. (2010). This is not a boundary object: Reflections on the origin of a concept. *Science, Technology, & Human Values*, 35(5), 601-617. doi: 10.1177/0162243910377624
- Leitch, A.M., Cundill, G., Schultz, L., Meek, C.L. (2015). Principle 6 Broaden participation. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 201-225). Cambridge, UK: Cambridge University.
- Lewin, K. (1946). Action Research and Minority Problems. *Journal of Social Issues*, 2(4), 34–46. doi: 10.1111/j.1540-4560.1946.tb02295.x
- Lewis, A. J., & Affolter, J. M. (1999). The state botanical garden of Georgia: A living laboratory for student education. *HortTechnology*, 9(4), 570-572. doi: 10.21273/HORTTECH.9.4.570
- Liu, S. C., & Lin, H. S. (2014). Undergraduate students' ideas about nature and human–nature relationships: an empirical analysis of environmental worldviews. *Environmental Education Research*, 20(3), 412-429. doi: 10.1080/13504622.2013.816266
- Löfman, P., Pelkonen, M., & Pietilä, A. M. (2004). Ethical issues in participatory action research. *Scandinavian Journal of Caring Sciences*, 18(3), 333-340. doi: 10.1111/j.1471-6712.2004.00277.x
- Longhurst, J., Bellingham, L., Cotton, D., Isaac, V., Kemp, S., Martin, S., ... & Tilbury, D. (2014). *Education for sustainable development: Guidance for UK higher education providers*. Gloucester, UK: QAA. Retrieved on November 6, 2018, from http://eprints.uwe.ac.uk/23353
- Lorenz, C. (2012). If you're so smart, why are you under surveillance? Universities, neoliberalism, and new public management. *Critical Inquiry*, 38(3), 599-629. doi: 10.1086/664553
- Louv, R. (2008). *Last child in the woods: Saving our children from nature-deficit disorder*. Chapel Hill, NC: Algonquin Books.
- Lovell, S. T. (2010). Multifunctional urban agriculture for sustainable land use planning in the United States. *Sustainability*, 2(8), 2499-2522. doi: 10.3390/su2082499
- Lovell, S. T., & Taylor, J. R. (2013). Supplying urban ecosystem services through multifunctional green infrastructure in the United States. *Landscape Ecology*, 28(8), 1447-1463. doi: 10.1007/s10980-013-9912-y
- Lund, M., & Orth, G. (2010). From the White House to Our House: The Story of an Honors College Vegetable Garden. *Honors in Practice*, 6, 177–188.
- MacDuff, N. (2005). Societal changes and the rise of the episodic volunteer. *Emerging Areas of Volunteering*, 1(2), 49-61.
- MacKinnon, D., & Derickson, K. D. (2013). From resilience to resourcefulness: A critique of resilience policy and activism. *Progress in Human Geography*, 37(2), 253-270. doi: 10.1177/0309132512454775
- Maiter, S., Simich, L., Jacobson, N., & Wise, J. (2008). Reciprocity: An ethic for community-based participatory action research. *Action Research*, 6(3), 305-325. doi: 10.1177/1476750307083720

- Markhart, A. H. (2006). Organic educational opportunities at the University of Minnesota: The role of a studentrun organic farm. *HortTechnology*, 16(3), 443-445. doi: 10.21273/HORTTECH.16.3.0443
- Marktanner, M., & Noiset, L. P. (2013). Food price crisis, poverty, and inequality. *The Developing Economies*, 51(3), 303-320. doi: 10.1111/deve.12020
- Marshall, N., & Marshall, P. (2007). Conceptualizing and operationalizing social resilience within commercial fisheries in northern Australia. *Ecology and Society*, 12(1).
- Martínez-Alier, J., Pascual, U., Vivien, F. D., & Zaccai, E. (2010). Sustainable de-growth: Mapping the context, criticisms and future prospects of an emergent paradigm. *Ecological Economics*, 69(9), 1741-1747. doi: 10.1016/j.ecolecon.2010.04.017
- Masters, J. (1995). The history of action research. *Action Research Electronic Reader*. Retrieved on August 6, 2019, from http://www.behs.cchs.usyd.edu.au/arow/Reader/rmasters.htm
- Max-Neef, M. A. (2005). Foundations of transdisciplinarity. *Ecological Economics*, 53(1), 5-16. doi: 10.1016/j.ecolecon.2005.01.014
- McDowell, G. R. (2003). Engaged universities: Lessons from the land-grant universities and extension. *The Annals of the American Academy of Political and Social Science*, 585(1), 31-50. doi: 10.1177/0002716202238565
- McEvoy, P., & Richards, D. (2006). A critical realist rationale for using a combination of quantitative and qualitative methods. *Journal of Research in Nursing*, 11(1), 66-78. doi: 10.1177/1744987106060192
- McKinne, K. L., & Halfacre, A. C. (2008). "Growing" a campus native species garden: sustaining volunteer-driven sustainability. *International Journal of Sustainability in Higher Education*, 9(2), 147-156. doi: 10.1108/14676370810856297
- McNiff, J. (2013). Action Research: Principles and practice (3rd ed.). Abingdon, UK: Routledge.
- McNiff, J., & Whitehead, J. (2001). Doing and writing action research. London, UK: Sage.
- McNiff, J., and Whitehead, J. (2011). All you need to know about action research. London, UK: Sage.
- McTaggart, R. (1994). Participatory Action Research: issues in theory and practice. *Educational Action Research*, 2(3), 313-337. doi: 10.1080/0965079940020302
- Meadows, D. (1999). Leverage points: Places to Intervene in a System. The Sustainability Institute: Vermont, USA.
- Meadows, D. (2001). Dancing with Systems. *The Systems Thinker*, 13(2).
- Meerow, S., Newell, J. P., & Stults, M. (2016). Defining urban resilience: A review. *Landscape and Urban Planning*, 147, 38-49. doi: 10.1016/j.landurbplan.2015.11.011
- Meyerson, D.; Weick K. E.; Kramer, R. M. (1996). Swift trust and temporary groups. In Kramer, R.M., Tyler, T.R., (Eds.), *Trust in organizations: Frontiers of theory and research* (pp.166-195). Thousand Oaks, USA: Sage.
- Middlemiss, L. (2011). The effects of community-based action for sustainability on participants' lifestyles. *Local Environment*, 16(3), 265-280. doi: 10.1080/13549839.2011.566850
- Millennium Ecosystem Assessment (MEA). (2005). *Ecosystems and human well-being: Synthesis*. Retrieved on November 5, 2018, from http://www.millenniumassessment.org/documents/document.356.aspx.pdf
- Miller, D., & Harkins, C. (2010). Corporate strategy, corporate capture: food and alcohol industry lobbying and public health. *Critical Social Policy*, 30(4), 564-589. doi: 10.1177/0261018310376805
- Mingers, J. (2015). Helping business schools engage with real problems: The contribution of critical realism and systems thinking. *European Journal of Operational Research*, 242(1), 316-331. doi: 10.1016/j.ejor.2014.10.058

- Missimer, M., Robèrt, K. H., Broman, G., & Sverdrup, H. (2010). Exploring the possibility of a systematic and generic approach to social sustainability. *Journal of Cleaner Production*, 18(10-11), 1107-1112. doi: 10.1016/j.jclepro.2010.02.024
- Mitchell, K. M. (2017). Academic voice: On feminism, presence, and objectivity in writing. *Nursing Inquiry*, 24(4), e12200. doi: 10.1111/nin.12200
- Mogensen, F., & Schnack, K. (2010). The action competence approach and the 'new'discourses of education for sustainable development, competence and quality criteria. *Environmental Education Research*, 16(1), 59-74. doi: 10.1080/13504620903504032
- Mok, H. F., Williamson, V. G., Grove, J. R., Burry, K., Barker, S. F., & Hamilton, A. J. (2014). Strawberry fields forever? Urban agriculture in developed countries: a review. *Agronomy for Sustainable Development*, 34(1), 21-43. doi: 10.1007/s13593-013-0156-7
- Moon, K., & Blackman, D. (2014). A guide to understanding social science research for natural scientists. *Conservation Biology*, 28(5), 1167-1177. doi: 10.1111/cobi.12326
- Mooney, L. A., & Edwards, B. (2001). Experiential learning in sociology: Service learning and other communitybased learning initiatives. *Teaching Sociology*, 181-194. doi: 10.2307/1318716
- Moyer, J. M., Sinclair, A. J., & Diduck, A. P. (2014). Learning for sustainability among faith-based organizations in Kenya. *Environmental Management*, 54(2), 360-372. doi: 10.1007/s00267-014-0289-8
- Mundel, E., & Chapman, G. E. (2010). A decolonizing approach to health promotion in Canada: the case of the Urban Aboriginal Community Kitchen Garden Project. *Health Promotion International*, 25(2), 166-173. doi: 10.1093/heapro/daq016
- Munro, M. (2018). Principles for embodied learning approaches. *South African Theatre Journal*, 31(1), 5-14. doi: 10.1080/10137548.2017.1404435
- Mycock, A. E. (2011). 'Local Food' Systems in County Durham: The capacities of community initiatives and local food businesses to build a more resilient local food system (Master's thesis, Durham University, Durham, UK).
- Nastar, M., Boda, C., & Olsson, L. (2018). A critical realist inquiry in conducting interdisciplinary research: an analysis of LUCID examples. *Ecology and Society*, 23(3). doi: 10.5751/ES-10218-230341
- National Commission for the Protection of Human Subjects of Biomedical and Behavioural Research, The (1978). *The Belmont Report: Ethical Principles and Guidelines for the Protection of Human Subjects of Research*. Retrieved on August 12, 2019, from https://www.hhs.gov/ohrp/regulations-and-policy/belmontreport/access-other-reports-by-the-national-commission/index.html
- National Resource Council. 1987. *Agricultural Biotechnology: Strategies for National Competitiveness*. Washington, USA: National Academies.
- National Resource Council. 1995. *Colleges of Agriculture at the Land Grant Universities: A Profile*. Washington, USA: National Academies.
- National Union of Students (NUS) (2018). *Sustainability Skills Survey* | 2017-18 Research into students' experiences of teaching and learning on sustainable development. Retrieved on October 24, 2018, from https://sustainability.nus.org.uk/resources/sustainability-skills-2017-18
- National Union of Students (NUS) (n.d.a). *Green Impact*. Retrieved on August 9, 2019, from https://sustainability.nus.org.uk/green-impact/
- National Union of Students (NUS) (n.d.b). *Student Eats Participating Institutions*. Retrieved on February 15, 2019, from https://sustainability.nus.org.uk/student-eats/grow-your-veg/participating-institutions

- National Union of Students (NUS) (n.d.c). *Student Eats Evaluation Baseline survey | Student summary report*. Retrieved on October 24, 2018, from https://sustainability.unioncloud.org/student-eats/research
- National Union of Students (NUS) (n.d.d). *Students for Trees*. Retrieved on August 6, 2019, from https://sustainability.nus.org.uk/students-for-trees/about
- National Union of Students (NUS) (n.d.e). *Students Green Fund*. Retrieved on August 6, 2019, from https://www.studentsgreenfund.org.uk/
- National Union of Students (NUS) (n.d.f). *Student Switch Off.* Retrieved on August 6, 2019, from http://studentswitchoff.org/
- Nell, W., Wessels, B., Mokoka, J., & Machedi, S. (2000). A creative multidisciplinary approach towards the development of food gardening. *Development Southern Africa*, 17(5), 807-819.
- Nielsen, K. A., & Nielsen, B. S. (2006). *Methodologies in action research. Action and Interaction Research-beyond practice and theory*. Maastricht, The Netherlands: Shaker.
- Nikulina, V., Simon, D., Ny, H., & Baumann, H. (2019). Context-adapted urban planning for rapid transitioning of personal mobility towards sustainability: A systematic literature review. *Sustainability*, 11(4), 1007. doi: 10.3390/su11041007
- Nisbet, E. K., Zelenski, J. M., & Murphy, S. A. (2008). The nature relatedness scale: Linking individuals' connection with nature to environmental concern and behavior. *Environment and Behavior*, 41, 715–740. doi: 10.1177/0013916508318748

Norman, W., & MacDonald, C. (2004). Getting to the bottom of "triple bottom line". *Business Ethics Quarterly*, 14(2), 243-262. doi: https://doi.org/10.5840/beq200414211

- Obrist, B., Pfeiffer, C., & Henley, R. (2010). Multi-layered social resilience: A new approach in mitigation research. *Progress in Development Studies*, 10(4), 283-293. doi: 10.1177/146499340901000402
- Ojala, M. (2007). *Hope and worry: Exploring young people's values, emotions, and behaviour regarding global environmental problems* (Doctoral thesis, Örebro University, Örebro, Sweden).
- Ojala, M. (2012). Regulating Worry, Promoting Hope: How Do Children, Adolescents, and Young Adults Cope with Climate Change?. *International Journal of Environmental and Science Education*, 7(4), 537-561.
- Olivera, F. (2000). Memory systems in organizations: an empirical investigation of mechanisms for knowledge collection, storage and access. *Journal of Management Studies*, 37(6), 811-832. doi: 10.1111/1467-6486.00205
- Olsen, S., Amundsen, D., Varga, B., Minch, D., & Anderson, D. (1999). The Utah botanical gardens: an educational resource for the university and the community. *HortTechnology*, 9(4), 562-565. doi: 10.21273/HORTTECH.9.4.562
- Olson, M. M., & Raffanti, M. A. (2006). Leverage points, paradigms, and grounded action: Intervening in educational systems. *World Futures*, 62(7), 533-541. doi: 10.1080/02604020600912897
- Olsson, L., Jerneck, A., Thoren, H., Persson, J., & O'Byrne, D. (2015). Why resilience is unappealing to social science: Theoretical and empirical investigations of the scientific use of resilience. *Science Advances*, 1(4), e1400217. doi: 10.1126/sciadv.1400217
- Onwuegbuzie, A. J., & Collins, K. M. (2007). A typology of mixed methods sampling designs in social science research. *The Qualitative Report*, 12(2), 281-316.
- Orsini, F., Kahane, R., Nono-Womdim, R., & Gianquinto, G. (2013). Urban agriculture in the developing world: a review. Agronomy for Sustainable Development, 33(4), 695-720. doi: 10.1007/s13593-013-0143-z

- Ostrom, E. (2007). A diagnostic approach for going beyond panaceas. *Proceedings of the National Academy of Sciences*, 104(39), 15181-15187. doi: 10.1073/pnas.0702288104
- Parker, K. A. (1996). Pragmatism and Environmental Thought. In Light, A. & Katz, E. (Eds.). *Environmental Pragmatism* (pp. 21-37). London, UK: Routledge.
- Parr, D. M., & Van Horn, M. (2006). Development of organic and sustainable agricultural education at the University of California, Davis: A closer look at practice and theory. *HortTechnology*, 16(3), 426-431. doi: 10.21273/HORTTECH.16.3.0426
- Parry, D. C., Glover, T. D., & Shinew, K. J. (2005). 'Mary, mary quite contrary, how does your garden grow?': examining gender roles and relations in community gardens. *Leisure Studies*, 24(2), 177-192. doi: 10.1080/0261436052000308820
- Partelow, S., Schlüter, A., von Wehrden, H., Jänig, M., & Senff, P. (2018). A sustainability agenda for tropical marine science. *Conservation Letters*, 11(1), e12351. doi: 10.1111/conl.12351
- Pereira, M. D., & Vallance, R. (2006). Multiple site action research case studies: Practical and theoretical benefits and challenges. *Issues in Educational Research*, 16(1), 67-79.
- Pemberton, S. (2017). *The importance of super-diverse places in shaping residential mobility patterns. A Report to the Leverhulme Trust.* Keele, UK: Keele University.
- Pereira, S., Reay, K., Bottell, J., Walker, L., Dzikiti, C, Platt, C., & Goodrham, C. (2019). University Student Mental Health Survey 2018: A large scale study into the prevalence of student mental illness within UK universities. *Dig In. The Insight Network*. Retrieved on November 18, 2019, from https://uploadsssl.webflow.com/561110743bc7e45e78292140/5c7d4b5d314d163fecdc3706_Mental%20Health%20Rep ort%202018.pdf
- Petticrew, M., & Roberts, H. (2006). *Systematic reviews in the social sciences: a practical guide*. Malden, Australia: Blackwell.
- Phillips, J. (2011). Building on Tradition--Tribal Colleges Can Lead the Way to Food Sovereignty. *Tribal College Journal of American Indian Higher Education*, 22(3), 15-20.
- Pickering, C., & Byrne, J. (2014). The benefits of publishing systematic quantitative literature reviews for PhD candidates and other early-career researchers. *Higher Education Research & Development*, 33(3), 534-548. doi: 10.1080/07294360.2013.841651
- Pierce, C. A., & Seals, L. M. (2006). The importance of community gardening for homeless women: A pilot study. *Journal of Therapeutic Horticulture*, 17, 20-26.
- Popa, F., Guillermin, M., & Dedeurwaerdere, T. (2015). A pragmatist approach to transdisciplinarity in sustainability research: From complex systems theory to reflexive science. *Futures*, 65, 45-56. doi: 10.1016/j.futures.2014.02.002
- Povee, K., Bishop, B. J., & Roberts, L. D. (2014). The use of photovoice with people with intellectual disabilities: Reflections, challenges and opportunities. *Disability & Society*, 29(6), 893-907. doi: 10.1080/09687599.2013.874331
- Powers, M. C., & Freedman, D. A. (2012). Applying a social justice framework to photovoice research on environmental issues: A comprehensive literature review. *Critical Social Work*, 13(2), 80-100. doi: 10.22329/csw.v13i2.5867
- Pudup, M. B. (2008). It takes a garden: Cultivating citizen-subjects in organized garden projects. *Geoforum*, 39(3), 1228-1240. doi: 10.1016/j.geoforum.2007.06.012
- Pyle, R. M. (2007). Pulling the Plug. *Orion Magazine*. Retrieved on August 20, 2019, from https://orionmagazine.org/article/pulling-the-plug/

- Quinlan, A. E., Berbés-Blázquez, M., Haider, L. J., & Peterson, G. D. (2016). Measuring and assessing resilience: broadening understanding through multiple disciplinary perspectives. *Journal of Applied Ecology*, 53(3), 677-687. doi: 10.1111/1365-2664.12550
- Ramaley, J. A. (2014). The changing role of higher education: Learning to deal with wicked problems. *Journal of Higher Education Outreach and Engagement*, 18(3), 7-22.
- Reason, P., and Bradbury, H. (Eds.) (2001). *Handbook of Action Research: Participative Inquiry and Practice*. London, UK: Sage.
- Reece, J. (2014). Journaling. In Coghlan, D., & Brydon-Miller, M. (Eds.) *The Sage Encyclopedia of Action Research* (online). Retrieved on August 12, 2019, from http://methods.sagepub.com/reference/encyclopedia-of-action-research/n185.xml
- Reed, B. (2007). Shifting from 'sustainability' to regeneration. *Building Research & Information*, 35(6), 674-680. doi: 10.1080/09613210701475753
- Reed, M. S., Evely, A. C., Cundill, G., Fazey, I., Glass, J., Laing, A., ... & Stringer, L. C. (2010). What is social learning?. *Ecology and Society*, 15(4), r1.
- Robles, A. G., Hirvikoski, T., Schuurman, D., & Stokes, L. (Eds.) (2015). *Introducing ENoLL and its living lab community*. Retrieved on August 11, 2019, from https://issuu.com/enoll/docs/enoll-print
- Robèrt, K. H., Schmidt-Bleek, B., De Larderel, J. A., Basile, G., Jansen, J. L., Kuehr, R., ... & Wackernagel, M. (2002). Strategic sustainable development—selection, design and synergies of applied tools. *Journal of Cleaner Production*, 10(3), 197-214. doi: 10.1016/S0959-6526(01)00061-0
- Robinson, J. (2004). Squaring the circle? Some thoughts on the idea of sustainable development. *Ecological Economics*, 48(4), 369-384. doi: 10.1016/j.ecolecon.2003.10.017
- Robinson-O'Brien, R., Story, M., & Heim, S. (2009). Impact of garden-based youth nutrition intervention programs: a review. *Journal of the American Dietetic Association*, 109(2), 273-280. doi: 10.1016/j.jada.2008.10.051
- Rocchi, B., Romano, D., & Hamza, R. (2013). Agriculture reform and food crisis in Syria: Impacts on poverty and inequality. *Food Policy*, 43, 190-203. doi: 10.1016/j.foodpol.2013.09.009
- Rochester, C. A. (2010). *Gateway to Work: The role of Volunteer Centres in supporting the link between volunteering and employability*. London, UK: Institute for Volunteering Research.
- Rogers, K., Luton, R., Biggs, H., Biggs, R. O., Blignaut, S., Choles, A., ... & Tangwe, P. (2013). Fostering complexity thinking in action research for change in social–ecological systems. *Ecology and Society*, 18(2). doi: 10.5751/ES-05330-180231
- Rojas, A., Richer, L., & Wagner, J. (2007). University of British Columbia food system project: Towards sustainable and secure campus food systems. *EcoHealth*, 4(1), 86-94. doi: 10.1007/s10393-006-0081-1
- Rosol, M. (2010). Public participation in post-Fordist urban green space governance: The case of community gardens in Berlin. *International Journal of Urban and Regional Research*, 34(3), 548-563. doi: 10.1111/j.1468-2427.2010.00968.x
- Roubanis, J. L., & Landis, W. (2007). Community gardening project: Meredith College students explore sustainability, organics. *Journal of Family and Consumer Sciences*, 99(3), 55.
- Ross, N. J. (2005). Bringing you fresh food from local farms and our garden: A college class designs a program to meet peer and institutional needs. *Journal of Nutrition Education and Behavior*, 37(2), 102-103. doi: 10.1016/S1499-4046(06)60026-2
- Rowlinson, M., Booth, C., Clark, P., Delahaye, A., & Procter, S. (2010). Social remembering and organizational memory. *Organization Studies*, 31(1), 69-87. doi: 10.1177/0170840609347056

Saldaña, J. (2013). The coding manual for qualitative researchers. London, UK: Sage.

- Saldaña, J. (2018). The coding manual for qualitative researchers (2nd ed.). London, UK: Sage.
- Saldivar-Tanaka, L., & Krasny, M. E. (2004). Culturing community development, neighborhood open space, and civic agriculture: The case of Latino community gardens in New York City. *Agriculture and Human Values*, 21(4), 399-412. doi: 10.1007/s10460-003-1248-9
- Sampson, D., & Wills, C. (2013, September 14-15). Culturally appropriate food: Researching cultural aspects of food sovereignty. Paper presented at *Food Sovereignty: A Critical Dialogue* (pp. 1-14). New Haven, USA: Journal of Peasant Studies.
- Savage, M. (2005). Working-class identities in the 1960s: Revisiting the affluent worker study. *Sociology*, 39(5), 929-946. doi: 10.1177/0038038505058373
- Schäfer, M., Jaeger-Erben, M., & Bamberg, S. (2012). Life events as windows of opportunity for changing towards sustainable consumption patterns?. *Journal of Consumer Policy*, 35(1), 65-84. doi: 10.1007/s10603-011-9181-6
- Schlüter, M., Biggs, R., Schoon, M. L., Robards, M. D., & Anderies, J. M. (2015). Reflections on building resilience: Interactions among principles and implications for governance. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 251–282). Cambridge, UK: Cambridge University.
- Schoon, M.L., Robards, M. D., Meek, C. L., & Galaz, V. (2015). Principle 7 Promote polycentric governance systems. In: Biggs, R., Schlüter, M., & Schoon, M. L. (Eds.), *Principles for Building Resilience: Sustaining Ecosystem Services in Social-Ecological Systems* (pp. 226–250). Cambridge, UK: Cambridge University.
- Schroeder, M. S., Creamer, N. G., Linker, H. M., Mueller, J. P., & Rzewnicki, P. (2006). Interdisciplinary and multilevel approach to organic and sustainable agriculture education at North Carolina State University. *HortTechnology*, 16(3), 418-426.
- Schultz, P. W., Nolan, J. M., Cialdini, R. B., Goldstein, N. J., & Griskevicius, V. (2007). The constructive, destructive, and reconstructive power of social norms. *Psychological Science*, 18(5), 429-434. doi: 10.1111/j.1467-9280.2007.01917.x
- Senge, P. M. (2006). *The Fifth Discipline: The Art and Practice of the Learning Organisation*. London, UK: Random House.
- Seville, E., Brunsdon, D., Dantas, A., Le Masurier, J., Wilkinson, S., & Vargo, J. (2008). Organisational resilience: Researching the reality of New Zealand organisations. *Journal of Business Continuity & Emergency Planning*, 2(3), 258-266.
- Shan, H., & Walter, P. (2015). Growing everyday multiculturalism: Practice-based learning of Chinese immigrants through community gardens in Canada. *Adult Education Quarterly*, 65(1), 19-34. doi: 10.1177/0741713614549231
- Sharp, L. (2002). Green campuses: the road from little victories to systemic transformation. *International Journal of Sustainability in Higher Education*, 3(2), 128-145. doi: 10.1108/14676370210422357
- Shephard, K. (2008). Higher education for sustainability: seeking affective learning outcomes. *International Journal of Sustainability in Higher Education*, 9(1), 87-98. doi: 10.1108/14676370810842201
- Shephard, K., Rieckmann, M., & Barth, M. (2019). Seeking sustainability competence and capability in the ESD and HESD literature: An international philosophical hermeneutic analysis. *Environmental Education Research*, 25(4), 532-547. doi: 10.1080/13504622.2018.1490947
- Shepherd, D. A., & Haynie, J. M. (2011). Venture failure, stigma, and impression management: A self-verification, self-determination view. *Strategic Entrepreneurship Journal*, 5(2), 178-197. doi: 10.1002/sej.113

- Silva, P. (2013). *Outcomes Monitoring within Civic Ecology Practices in the NYC Region*. Civic Ecology Lab, Cornell University. Retrieved from https://civeco.files.wordpress.com/2013/09/silva-2013.pdf
- Silva, P., & Krasny, M. E. (2016). Parsing participation: models of engagement for outcomes monitoring in urban stewardship. *Local Environment*, 21(2), 157-165. doi: 10.1080/13549839.2014.929094
- Silver, H., Scott, A., & Kazepov, Y. (2010). Participation in urban contention and deliberation. *International Journal of Urban and Regional Research*, 34(3), 453-477. doi: 10.1111/j.1468-2427.2010.00963.x
- Simmie, J., & Martin, R. (2010). The economic resilience of regions: towards an evolutionary approach. *Cambridge Journal of Regions, Economy and Society*, 3(1), 27-43. doi: 10.1093/cjres/rsp029
- Sipos, Y., Battisti, B., & Grimm, K. (2008). Achieving transformative sustainability learning: engaging head, hands and heart. *International Journal of Sustainability in Higher Education*, 9(1), 68-86. doi: 10.1108/14676370810842193
- Sissay, L. (2016). Gold from the Stone: New and Selected Poems. Edinburgh, UK: Canongate.
- Snyder, B. R. (1970). The Hidden Curriculum. Cambridge, MA: MIT.
- Smidt, S. (2014). Introducing Freire: A Guide for Students, Teachers and Practitioners. Abingdon, UK: Routledge.
- Somerset, S., Harris, N., Wenham, K., & Rowe, F. (2010). A campus-based community garden to enhance economic, social and health promotion opportunities for African immigrants. In R. Russell-Bennet & S. Rundle-Thiele (Eds.), 2010 International Nonprofit & Social Marketing Conference (INSM) Conference Proceedings (pp. 15-16). Brisbane, Australia: Queensland University of Technology. Retrieved from https://www.aasm.org.au/wp-content/uploads/2017/04/AASM2010_ISNM_Conference_Proceedings.pdf
- Somerset, S., & Markwell, K. (2009). Impact of a school-based food garden on attitudes and identification skills regarding vegetables and fruit: a 12-month intervention trial. *Public Health Nutrition*, 12(2), 214-221. doi: 10.1017/S1368980008003327
- Spencer, P., Perkins, P. E., & Erickson, J. D. (2018). Re-establishing justice as a pillar of ecological economics through feminist perspectives. *Ecological Economics*, 152, 191-198. doi: 10.1016/j.ecolecon.2018.05.022
- Spierings, B., Van Liempt, I., & Maliepaard, E. (2018). Ownership and Membership: Practices and Experiences of Neighbourhood Residents in the Wijsgeren Community Garden in Amsterdam. *Tijdschrift voor* economische en sociale geografie, 109(5), 677-684. doi: 10.1111/tesg.12337
- Spronck, W. E. E. C., & Compernolle, T. H. L. (1997). Systems theory and family therapy: From a critique on systems theory to a theory on system change. *Contemporary Family Therapy*, 19(2), 147-175. doi: 10.1023/A:1026143317909
- Squirrell, G., Connor, H. and Peterman, B. (2009). *Student Volunteering: Background, Policy and Context*. Bristol, UK: National Co-ordinating Centre for Public Engagement.
- Staeheli, L. A., Mitchell, D., & Gibson, K. (2002). Conflicting rights to the city in New York's community gardens. *GeoJournal*, 58(2-3), 197-205. doi: 10.1023/B:GEJO.0000010839.59734.01
- Stam, H. J. (1996). Theory and Practice. In Tolman, C. W., Cherry, F., van Hezewijk, R., & Lubek, I. (Eds.). *Problems of theoretical psychology* (p. 24-32). North York, Canada: Captus Press.
- Star, S. L., & Griesemer, J. R. (1989). Institutional ecology, translations' and boundary objects: Amateurs and professionals in Berkeley's Museum of Vertebrate Zoology, 1907-39. *Social Studies of Science*, 19(3), 387-420. doi: 10.1177/030631289019003001
- Starnes, B. J., & Wymer Jr, W. W. (2001). Conceptual foundations and practical guidelines for retaining volunteers who serve in local nonprofit organizations: Part II. *Journal of Nonprofit & Public Sector Marketing*, 9(1-2), 97-118. doi: 10.1300/J054v09n01_06

- Steffen, W., Richardson, K., Rockström, J., Cornell, S. E., Fetzer, I., Bennett, E. M., ... & Folke, C. (2015). Planetary boundaries: Guiding human development on a changing planet. *Science*, 347(6223), 1259855. doi: 10.1126/science.1259855
- Stephens, M., Steil, A., Gray, M., Hird, A., Lepper, S., Moydell, E., ... & Lyons, R. E. (2006). Endowment strategies for the University of Delaware Botanic Garden through case study analysis. *HortTechnology*, 16(4), 570-578. doi: 10.21273/HORTTECH.16.4.0570
- Stephenson, A., Vargo, J., & Seville, E. (2010). Measuring and comparing organisational resilience in Auckland. *The Australian Journal of Emergency Management*, 25(2), 27-32.
- Sterling, S. (2010). Learning for resilience, or the resilient learner? Towards a necessary reconciliation in a paradigm of sustainable education. *Environmental Education Research*, 16(5-6), 511-528. doi: 10.1080/13504622.2010.505427
- Stimart, D. P. (1999). Development and integration of an instructional garden in education at the University of Wisconsin-Madison. *HortTechnology*, 9(4), 557-561. doi: 10.21273/HORTTECH.9.4.557
- Stirling, A. (2007). A general framework for analysing diversity in science, technology and society. *Journal of the Royal Society Interface*, 4(15), 707-719. doi: 10.1098/rsif.2007.0213
- Stocker, L., & Barnett, K. (1998). The significance and praxis of community-based sustainability projects: Community gardens in western Australia. *Local Environment*, 3(2), 179-189. doi: 10.1080/13549839808725556
- Stockholm Resilience Centre (n.d.). *About our publications*. Retrieved on August 6, 2019, from https://www.stockholmresilience.org/research/our-publications.html
- Stringer, E. T. (2007). Action Research (3rd ed.). California, USA: Sage.
- Stringer, L., Dougill, A., Fraser, E., Hubacek, K., Prell, C., & Reed, M. (2006). Unpacking "participation" in the adaptive management of social–ecological systems: a critical review. *Ecology and Society*, 11(2), 39.
- Student Eats (n.d.). *Student Eats*. Retrieved on August 13, 2019, from https://students-eat-keele.wixsite.com/home
- Student Volunteering England (2004). *The Student Response to the Russell Commission*. London, UK: Student Volunteering England.
- Sustainable Development Solutions Network (2018). *SDGacademy*. Retrieved on October 3, 2018, from https://sdgacademy.org/
- Taylor, J. R., & Taylor Lovell, S. (2014). Urban home food gardens in the Global North: research traditions and future directions. *Agriculture and Human Values*, 31(2), 285-305. doi: 10.1007/s10460-013-9475-1
- Teig, E., Amulya, J., Bardwell, L., Buchenau, M., Marshall, J. A., & Litt, J. S. (2009). Collective efficacy in Denver, Colorado: Strengthening neighborhoods and health through community gardens. *Health & Place*, 15(4), 1115-1122. doi: 10.1016/j.healthplace.2009.06.003
- Terstappen, V., Hanson, L., & McLaughlin, D. (2013). Gender, health, labor, and inequities: a review of the fair and alternative trade literature. *Agriculture and Human Values*, 30(1), 21-39. doi: 10.1007/s10460-012-9377-7
- Thompson, S., Michaelson, J., Abdallah, S., Johnson, V., Morris, D., Riley, K., & Simms, A. (2011). 'Moments of Change' as opportunities for influencing behaviour: A report to the Department for Environment, Food, and Rural Affairs. the new economics foundation (nef). London, UK: DEFRA.

- Tidball, K. G., & Krasny, M. E. (2007). From risk to resilience: What role for community greening and civic ecology in cities. In Wals, A. (Ed.), *Social Learning Towards a More Sustainable World* (pp.149-164). The Netherlands: Wageningen Academic.
- Tooth, R., & Renshaw, P. (2009). Reflections on pedagogy and place: A journey into learning for sustainability through environmental narrative and deep attentive reflection. *Australian Journal of Environmental Education*, 25, 95-104. doi: 10.1017/S0814062600000434
- Tornaghi, C. (2014). Critical geography of urban agriculture. *Progress in Human Geography*, 38(4), 551-567. doi: 10.1177/0309132513512542
- Tracy, S. J. (2010). Qualitative quality: Eight "big-tent" criteria for excellent qualitative research. *Qualitative Inquiry*, 16(10), 837-851. doi: 10.1177/1077800410383121
- Transient [Def 1b]. In *Merriam Webster Online*, Retrieved September 25, 2018, from https://www.merriamwebster.com/dictionary/transient
- Turner, B. (2011). Embodied connections: sustainability, food systems and community gardens. *Local Environment*, 16(6), 509-522. doi: 10.1080/13549839.2011.569537
- United Kingdom Research Integrity Office (UKRIO) (2019). Research involving human participants, human material or personal data. Retrieved on August 12, 2019, from https://ukrio.org/publications/code-of-practice-for-research/3-0-standards-for-organisations-and-researchers/3-7-research-involving-human-participants-human-material-or-personal-data/
- United Nations (UN) (1987). Report of the World Commission on Environment and Development: Our Common Future. Retrieved on November 5, 2018, from http://www.un-documents.net/our-common-future.pdf
- United Nations (UN) (2019). *The Sustainable Development Goals Report 2019*. Retrieved on August 12, 2019, from https://unstats.un.org/sdgs/report/2019/The-Sustainable-Development-Goals-Report-2019.pdf
- United Nations Conference on Environment and Development (UNCED) (1992). Agenda 21. Retrieved on November 6, 2018, from https://sustainabledevelopment.un.org/outcomedocuments/agenda21
- United Nations Educational, Scientific and Cultural Organization (UNESCO) (2014). UNESCO Roadmap for Implementing the Global Action Programme on Education for Sustainable Development. France: UNESCO.
- University of British Columbia (UBC) (n.d.). University of British Columbia Farm. University of British Columbia Sustainability. Retrieved on November 7, 2018, from https://sustain.ubc.ca/campus-initiatives/food/ubc-farm
- University of Warwick (2015). *About: Interactive Campus Map*. Retrieved on August 8, 2019, from https://warwick.ac.uk/about/visiting/maps/interactive
- University of Warwick (2019a). *About: People*. Retrieved on August 6, 2019, from https://warwick.ac.uk/about/profile/people/
- University of Warwick (2019b). Accommodation: Student Accommodation. Retrieved on August 6, 2019, from https://warwick.ac.uk/services/accommodation/studentaccommodation
- Vaccaro, A., & Mena, J. A. (2011). It's not burnout, it's more: Queer college activists of color and mental health. *Journal of Gay & Lesbian Mental Health*, 15(4), 339-367. doi: 10.1080/19359705.2011.600656
- Valley, W., & Wittman, H. (2018). Beyond feeding the city: The multifunctionality of urban farming in Vancouver, BC. *City, Culture and Society*. doi: 10.1016/j.ccs.2018.03.004
- VanDerZanden, A. M., & Cook, T. (1999). A multifunctional horticulture teaching garden at Oregon State University. *Horttechnology*, 9(4), 549-551.

- van der Eyken, V. (1996) *Making the Essential Difference: research and evaluation*. Speech made at the Portage Association International Conference, Winchester, England, September 1996.
- van Leeuwen, E., Nijkamp, P., & de Noronha Vaz, T. (2010). The multifunctional use of urban greenspace. International Journal of Agricultural Sustainability, 8(1-2), 20-25. doi: 10.3763/ijas.2009.0466
- van Niekerk, E., & J. G. Barnard (2011). Health and Lifestyle Practices among Female Students in a South African University Setting. *College Student Journal*, 45(3), 649–666.
- Van Rijsoort, J., & Jinfeng, Z. (2005). Participatory resource monitoring as a means for promoting social change in Yunnan, China. *Biodiversity & Conservation*, 14(11), 2543-2573. doi: 10.1007/s10531-005-8377-y
- Van Rooij, A. (2015). Sisyphus in business: Success, failure and the different types of failure. *Business History*, 57(2), 203-223. doi: https://doi.org/10.1080/00076791.2014.909808
- Vare, P., & Scott, W. (2008). Education for Sustainable Development: two sides and an edge. DEA Thinkpiece. Retrieved on August 3, 2019, from https://www.academia.edu/21256086/Education_for_Sustainable_Development_two_sides_and_an_ed ge
- Vaughn, L. M., Jacquez, F., Zhao, J., & Lang, M. (2011). Partnering with students to explore the health needs of an ethnically diverse, low-resource school: An innovative large group assessment approach. *Family & Community Health*, 34(1), 72-84. doi: 10.1097/FCH.0b013e3181fded12
- Vaughn, L. M., & Lohmueller, M. (1998). Using the Group Level Assessment in a support group setting. *Organization Development Journal*, 16(1), 99-105.
- Veland, S., Scoville-Simonds, M., Gram-Hanssen, I., Schorre, A. K., El Khoury, A., Nordbø, M. J., ... & Bjørkan, M. (2018). Narrative matters for sustainability: the transformative role of storytelling in realizing 1.5 C futures. *Current Opinion in Environmental Sustainability*, 31, 41-47. doi: 10.1016/j.cosust.2017.12.005
- Verplanken, B., & Roy, D. (2016). Empowering interventions to promote sustainable lifestyles: Testing the habit discontinuity hypothesis in a field experiment. *Journal of Environmental Psychology*, 45, 127-134. doi: 10.1016/j.jenvp.2015.11.008
- Verplanken, B., Walker, I., Davis, A., & Jurasek, M. (2008). Context change and travel mode choice: Combining the habit discontinuity and self-activation hypotheses. *Journal of Environmental Psychology*, 28(2), 121-127. doi: 10.1016/j.jenvp.2007.10.005
- Wade, I. (1987). Community food production in cities of the developing nations. *Food and Nutrition Bulletin*, 9(2), 1-7.
- Wagner, L. K., & Fones, S. W. (1999). Enhancing science education experiences through garden explorations: An inquiry-based learning opportunity at the South Carolina Botanical Garden. *HortTechnology*, 9(4), 566-569. doi: 10.21273/HORTTECH.9.4.566
- Wahlström, Mattias, Piotr Kocyba, Michiel De Vydt and Joost de Moor (Eds.) (2019). Protest for a future: Composition, mobilization and motives of the participants in Fridays For Future climate protests on 15 March, 2019 in 13 European cities. Retrieved on August 14, 2019, from http://eprints.keele.ac.uk/6571/7/20190709_Protest%20for%20a%20future_GCS%20Descriptive%20Rep ort.pdf
- Walker, B., Gunderson, L., Kinzig, A., Folke, C., Carpenter, S., & Schultz, L. (2006). A handful of heuristics and some propositions for understanding resilience in social-ecological systems. *Ecology and Society*, 11(1), 13.
- Walker, B., Holling, C. S., Carpenter, S., & Kinzig, A. (2004). Resilience, adaptability and transformability in social– ecological systems. *Ecology and Society*, 9(2), 5.

- Walker, B. H. & Salt, D. (2006). *Resilience Thinking: Sustaining Ecosystems and People in a Changing World*. Washington, DC: Island Press.
- Walker, S., Read, S., & Priest, H. (2013). Use of reflexivity in a mixed-methods study. *Nurse Researcher*, 20(3). doi: 10.7748/nr2013.01.20.3.38.c9496
- Wals, A. E. (2012). Shaping the Education of Tomorrow: 2012 Full-length Report on the UN Decade of Education for Sustainable Development. Paris, France: UNESCO. Retrieved on November 6, 2018, from https://library.wur.nl/WebQuery/wurpubs/fulltext/246667
- Walter, P. (2013). Theorising community gardens as pedagogical sites in the food movement. *Environmental Education Research*, 19(4), 521-539. doi: 10.1080/13504622.2012.709824
- Weaver, T. F., & Diamantides, J. (1993). The Future of the Land Grant University System. Agricultural and Resource Economics Review, 22(2), 130-136. doi: 10.1017/S1068280500004718
- Wenger, E., McDermott, R. A., & Snyder, W. (2002). *Cultivating communities of practice: A guide to managing knowledge*. Boston, USA: Harvard Business Press.
- Wharton, C., & Harmon, A. (2009). University engagement through local food enterprise: community-supported agriculture on campus. *Journal of Hunger & Environmental Nutrition*, 4(2), 112-128. doi: 10.1080/19320240902915235
- Wiek, A., Withycombe, L., & Redman, C. L. (2011). Key competencies in sustainability: a reference framework for academic program development. *Sustainability Science*, 6(2), 203-218. doi: 10.1007/s11625-011-0132-6
- Williams, P. (2002). The competent boundary spanner. *Public Administration*, 80(1), 103-124. doi: 10.1111/1467-9299.00296
- Wills, J., Chinemana, F., & Rudolph, M. (2009). Growing or connecting? An urban food garden in Johannesburg. *Health Promotion International*, 25(1), 33-41. doi: 10.1093/heapro/dap042
- Winter, J., & Cotton, D. (2012). Making the hidden curriculum visible: sustainability literacy in higher education. *Environmental Education Research*, 18(6), 783-796. doi: 10.1080/13504622.2012.670207
- Winter, R., & Munn-Giddings, C. (2001). A handbook for action research in health and social care. Oxon, UK: Routledge.
- Wittmayer, J. M., & Schäpke, N. (2014). Action, research and participation: roles of researchers in sustainability transitions. *Sustainability Science*, 9(4), 483-496. doi: 10.1007/s11625-014-0258-4
- Witzling, L., Wander, M., & Phillips, E. (2011). Testing and educating on urban soil lead: A case of Chicago community gardens. *Journal of Agriculture, Food Systems, and Community Development*, 1(2), 167-185. doi: 10.5304/jafscd.2010.012.015
- Wolff, P., Medin, D. L., & Pankratz, C. (1999). Evolution and devolution of folkbiological knowledge. *Cognition*, 73(2), 177-204. doi: 10.1016/S0010-0277(99)00051-7
- World Bank (n.d.). World Development Indicators. Retrieved on September 25, 2017, from https://data.worldbank.org/indicator
- Yadav, P., Duckworth, K., & Grewal, P. S. (2012). Habitat structure influences below ground biocontrol services: A comparison between urban gardens and vacant lots. *Landscape and Urban Planning*, 104(2), 238-244. doi: 10.1016/j.landurbplan.2011.10.018
- Yap, C. (2018). *The Politics of Self-Organisation and the Social Production of Space in Urban Community Gardens* (Doctoral dissertation, Coventry University, Coventry, UK).
- Yin, R.K., (1984). Case Study Research: Design and Methods. Beverly Hills, USA: Sage.

- Yin, R. K. (2003). Case study research: Design and methods (3rd ed.). California, USA: Sage.
- Yin, R. K. (2008). How to do better case studies. Bickman, L., & Rog, D. J. (Eds.). *The SAGE handbook of applied social research methods* (2nd ed.) (pp. 254-282). Thousand Oaks, USA: Sage
- Yin, R. K. (2014). Case study research: Design and methods. (5th ed.). Thousand Oaks, USA: Sage.
- Zasada, I. (2011). Multifunctional peri-urban agriculture—A review of societal demands and the provision of goods and services by farming. *Land Use Policy*, 28(4), 639-648. doi: 10.1016/j.landusepol.2011.01.008
- Zen, I. S. (2017). Exploring the living learning laboratory: An approach to strengthen campus sustainability initiatives by using sustainability science approach. *International Journal of Sustainability in Higher Education*, 18(6), 939-955. doi: 10.1108/IJSHE-09-2015-0154
- Zohar, D. 1997. *Rewiring the corporate brain: Using the new science to rethink how we structure and lead organizations*. San Francisco, USA: Berrett-Koehler.
- Zuber-Skerritt, O. & Perry, C. (2002). Action research within organizations and university thesis writing. *The Learning Organization*, 9(4), 171-179. doi: 10.1108/09696470210428895

Appendix A: Wiek et al.'s (2011) key competencies in sustainability

Table 28. Overview of core concepts and methods/methodologies as well as exemplary sources of the five key competencies in sustainability (Wiek et al., 2011, p. 213).

Competence	Concepts	Methods/methodologies	Sources
Systems- thinking competence	Variables/indicators, sub-systems, structures, functions	Qualitative and quantitative modeling	Porter and Córdoba 2009; Crofton 2000; Sterling 1996
	Feedback loops, complex cause-effect chains, cascading effects, inertia, tipping points, legacy,	Institutional, decision, governance, social systems analysis Systems multi-methodologies (e.g., "thick" description methodology)	
	resilience, adaptation, structuration		
	Across/multiple scales: local to global		
	Across/multiple/coupled domains: society, environment, economy, technology	Participatory systems approaches, including participatory modeling	
	People and social systems: values, preferences, needs, perceptions, (collective) actions, decisions, power, tactics, politics, institutions		
Anticipatory competence	Concepts of time including temporal phases (past, present, future), terms (short, long), states, continuity (dynamics, paths), non-linearity	Scenario methodology	Major et al. 2001; Withycombe and Wiek 2010; de Haan, 2006; Grunwald 2007
		Forecasting from statistical and simulation models	
	Concept of uncertainty and epistemic status including	Backcasting and envisioning methods	
	possibility, probability, desirability of future	Anticipatory multi-methodologies	
	developments (predictions, scenarios, visions) Concepts of inertia, path dependency, non- interventions	Participatory anticipatory approaches, including Delphi and future workshop	
	Concepts of consistency and plausibility of future developments		
	Concepts of risk, intergenerational equity, precaution		
Normative	(Un-)sustainability of current or future states	Multi-criteria assessment methods	Gibson 2006; Sterling 1996; Grunwald 2004
competence	Sustainability principles, goals, targets, thresholds (tipping points)	(including Life-Cycle Assessment, Multi-Attribute Utility Theory, etc.)	
	Concepts of justice, fairness, responsibility, safety,	Risk analysis	
	happiness, etc.	Sustainability efficiency analysis	
	Concept of risk, harm, damage		
	Concept of reinforcing gains ("win-win") and tradeoffs	Envisioning methods (e.g., backcasting)	
	Ethical concepts	Participatory normative methods, including negotiation methods and consensus conference	
Strategic competence	Intentionality	Methods to design governance	Bammer 2005; de Haan, 2006 Grunwald 2007
	Transitions and transformation	arrangements, policies, institutions	
	Strategies, action programs, (systemic) intervention, transformative governance	Planning methodologies Decision support methodologies	
	Success factors, viability, feasibility, effectiveness, efficiency	Transition management methodology	
	Adaptation and mitigation	Methods to support behavioral change	
	Obstacles (resistance, reluctance, path dependency, habits) and synergies	Organizational (change) management	
	Instrumentalization and alliances		
	Social learning	Methods to support learning and	
	Social movements	reflexivity	
Interpersonal competence	Functions, types, and dynamics of collaboration (within and beyond academia)		Crofton 2000; Kearins and Springett 2003; de Haan
	Strengths, weaknesses, success, and failure in teams		
	Concepts of leadership	Teamwork methods	
	Limits of cooperation and empathy		
	Concepts of solidarity and ethnocentrism		

Appendix B: Student Eats funding

The National Union of Students' sustainability team was awarded £315 000 through the Big Lottery's Local Food scheme. A portion of this funding was subsequently distributed to universities, colleges, and their students' unions. The requirement for the funding was that the project must have been student-led or at least have had substantial student engagement given that the funding was provided through the National Union of Students. Approximately £10 000 was given to each of the eighteen sites funded.

Some institutions already had gardens with considerable existing infrastructure, while others were setting up a garden from scratch. As such, each garden spent funding by mixing-and-matching different items according to their needs. For example, all sites received funding for improving access, paths, and/or raised beds, however, only six sites received funding to set up greenhouses. Most of the gardens' expenditures were expected to be spent on capital costs. The items that the National Union of Students sustainability team expected to fund are in Table 29.

Annrox

	Items	
Capital costs	Improving access/paths/raised beds	£1500
	Polytunnels including fitting	£3500
	Greenhouses including delivery/base	£1400
	Cold frames including delivery	£950
	Staging (shelving)	£1000
	Seeds, plants, fruit trees	£750
	Compost - one load, including delivery	£100
	Growing trays and herb pots	£75
	Canes, string, labels, pens, books	£40
	Water installation - mains	£1,250
	Water installation - rainwater harvesting including 2 x butts, hose, watering cans	£150
	Composting infrastructure	£150
	Boundaries - protective fencing and gates	£800
	Biodiversity enhancements - hedging, bird boxes, ponds	£350
	Bee hive - two at each, including protective equipment	£600
Revenue costs	Fixed branded interpretation boards at each site, including delivery and installation	
	Recruitment of volunteers - posters, adverts in student newspaper, windowsill taster growing kits for freshers	
	Box scheme set up - leaflets, bags, boxes	
	On campus and off-campus outreach events (cooking, school visits, etc.)	
	Volunteer insurance (extension of current policy to cover new volunteers)	

Table 29. Items funded by Student Eats' Local Food funding.

Appendix C: Reflective diary prompts

In my reflective diary, I would use the following 'Learning Window' from Coghlan and Brannick (2005, p. 43) to structure how certain I felt about what I thought I knew and to figure out what ideas I might want to explore further or test out.

What I <i>know</i> I know	What I <i>think</i> I know	
•	•	
What I <i>know</i> I don't know	What I <i>don't know</i> I don't know	
•		

I would also use the following reflective prompts I wrote for myself based on a journaling exercise created by Coghlan and Brannick (2005).

Context

What stage of the research am I at? What are the recent events that have taken place? Have we reached any milestones? Has there been any conflict? What are the challenges I've been struggling? Have there been any unexpected outcomes?

Reflecting on Recent Happenings

How did recent activity make me feel? Why? What judgements did I make? Would I have done anything differently? Why? What is within my control, and what is out of my control? How did I deal with unpredicted outcomes? Did this work? Can I see any trends or patterns emerging?

Moving Forward

What am I expecting to happen next? What steps am I taking next? Can I foresee any challenges or conflict? If so, what? And how can I prepare myself to deal with these? Can I test any of the conclusions I've come up with in my reflections? How?
Appendix D: Ethical approval letters



RESEARCH AND ENTERPRISE SERVICES

Ref: ERP1202

26th August 2014

Rebecca Laycock Room 1.25 William Smith Building

Dear Rebecca, [insert name]

Re: Cultivating social and environmental sustainability in community gardens: an action research approach to developing indicators

Thank you for submitting your revised application for review. I am pleased to inform you that your application has been approved by the Ethics Review Panel.

The following documents have been reviewed and approved by the panel as follows:

Document	Version	Date
Summary Proposal	2	01.08.14
Letter of invitation for case study participants	2	01.08.14
Letter of invitation for administrative participants	2	01.08.14
Information Sheet for case study participants	2	01.08.14
Information Sheet for administrative participants	2	01.08.14
Consent Form for case study participants	2	01.08.14
Consent Form for administrative participants	2	01.08.14
Consent Form for case study participants for the use of quotes	2	01.08.14
Consent Form for administrative participants for the use of quotes	2	01.08.14
Consent Form for case study participants use of media	2	01.08.14
Consent Form for administrative participants for use of media	2	01.08.14
Passive Consent Form Template	2	01.08.14
Interview Topic Guide	1	16.06.14
Future Workshop/Participatory Rural Appraisal Outline	1	16.06.14
Group Level Assessment Topic Guide	1	16.06.14
Conference Outline	2	01.08.14
Structured Ethical Reflection	1	16.06.14

If the fieldwork goes beyond the date stated in your application, you must notify the Ethical Review Panel via the ERP administrator at <u>uso.erps@keele.ac.uk</u> stating ERP1 in the subject line of the e-mail.

If there are any other amendments to your study you must submit an 'application to amend study' form to the ERP administrator stating ERP1 in the subject line of the e-mail. This form is available via http://www.keele.ac.uk/researchsupport/researchethics/

Research and Enterprise Services, Keele University, Staffordshire, ST5 5BG, UK Telephone: + 44 (0)1782 734466 Fax: + 44 (0)1782 733740 If you have any queries, please do not hesitate to contact me via the ERP administrator on research.erps@keele.ac.uk_stating ERP1 in the subject line of the e-mail.

Yours sincerely

PP & Bennerman

Dr Jackie Waterfield Chair – Ethical Review Panel

СС **RI Manager** Supervisor



Ref: ERP1202

19th May 2016

Rebecca Laycock William Smith Building, 1.15 Keele University

Dear Rebecca,

Re: Cultivating sustainable participation in Student-led Food Growing Initiatives with transient participants using action research

Thank you for submitting your application to amend study. I am pleased to inform you that your application has been approved by the Ethical Review Panel. The following documents have been reviewed and approved by the Panel as follows:-

Document	Version	Date
Summary Proposal	3	04-05-2016
Invitation Letter – Case Study Participants	3	04-05-2016
Information Sheet – Case Study Participants	3	04-05-2016
Consent Form	3	04-05-2016
Consent Form (for the use of video, photo and	3	04-05-2016
audio recordings)		
Photovoice Information Sheet	1	04-05-2016

If the fieldwork goes beyond the date stated in your application **31**st **January 2017**, you must notify the Ethical Review Panel via the ERP administrator at <u>research.erps@keele.ac.uk</u> stating **ERP1** in the subject line of the e-mail.

If there are any other amendments to your study you must submit an 'application to amend study' form to the ERP administrator stating **ERP1** in the subject line of the e-mail. This form is available via http://www.keele.ac.uk/researchsupport/researchethics/

Directorate of Engagement & Partnerships T: +44(0)1782 734467

Keele University, Staffordshire ST5 5BG, UK www.keele.ac.uk +44 (0)1782 732000



RESEARCH AND ENTERPRISE SERVICES

If you have any queries, please do not hesitate to contact me via the ERP administrator on <u>uso.erps@keele.ac.uk</u>_stating ERP1 in the subject line of the e-mail.

Yours sincerely,

PP

Dr Jackie Waterfield Chair – Ethical Review Panel

CC RI Manager Supervisor

> Research and Enterprise Services, Keele University, Staffordshire, ST5 5BG, UK Telephone: + 44 (0)1782 734466 Fax: + 44 (0)1782 733740



Ref: ERP1202

1st February 2017

Rebecca Laycock William Smith Building 1.15 Keele University

Dear Rebecca,

Re: Cultivating social and environmental sustainability in community gardens: an action research approach to developing indicators

Thank you for submitting your third application to amend study. I am pleased to inform you that your application has been approved by the Ethical Review Panel. The following documents have been reviewed and approved by the Panel as follows:-

Document	Version	Date
Information Sheet and Consent form for Discussion Workshop	1	24-01-2017

If the fieldwork goes beyond the date stated in your application, **28th February 2017**, or there are any other amendments to your study you must submit an 'application to amend study' form to the ERP administrator at <u>research.governance@keele.ac.uk</u> stating **ERP1** in the subject line of the e-mail. This form is available via http://www.keele.ac.uk/researchsupport/researchethics/

If you have any queries, please do not hesitate to contact me via the ERP administrator on <u>research.governance@keele.ac.uk</u> stating **ERP1** in the subject line of the e-mail.

Yours sincerely,

Benneman

Dr Jackie Waterfield Chair – Ethical Review Panel

CC Supervisor RI Manager

> Directorate of Engagement & Partnerships T: +44(0)1782 734467

Keele University, Staffordshire ST5 5BG, UK www.keele.ac.uk +44 (0)1782 732000

Appendix E. Summary of findings from the 'constructing' phase



A Summary of Findings from the 'Diagnosing' Phase from an Action Research project exploring Student-led Food Growing Initiatives

> Rebecca Laycock William Smith Building, 1.15 Keele University Staffordshire ST5 5BG email: r.laycock@keele.ac.uk telephone: +44 (0) 1782 734993

> > January 19, 2016

1

Background

In October 2014 I began fieldwork as part of my PhD project about Student-led Food Growing Initiatives (SFGIs), looking at three case studies that had been supported by the NUS' Student Eats food growing initiative. My PhD uses Action Research – an approach that combines research with taking 'actions' to address a problem. The process begins with a 'diagnosing' phase to find a problem to address, followed by planning an action to address the problem, then taking the planned action, and finally evaluating the action taken.

Diagnosing	Planning Action	Taking Action	Evaluating
•Oct 2014 - Dec 2015	•Jan-Feb 2016	•Feb-Nov 2016	•Nov 2016-Feb 2017

The purpose of this document is to communicate the findings from the 'diagnosing' phase by reporting:

- 1. What the problem is that we want to address,
- 2. The factors that influence the problem.
- 3. The challenges the problem causes, and
- 4. What can be done to address the problem.

How to Read This Report

It's recommended that you interpret this report by first looking at the flow diagram on page 3. Then if you don't understand something, go to the section of the report that elaborates on the part you don't understand.

This diagram will be able to help us make some decisions about how to move forward with this Action Research project. However, I don't want to have missed out anything important or to have misinterpreted what was said by the people who were involved in providing this data. For this reason, **I'm asking that you look over this report, including the diagram, and let me know if you think they are an accurate representation of the issues, and if there are any changes that need to be made.** For more information on how to do this, please jump to page 7 under the heading of 'Next Steps & Suggestions'.

1. The Problem We Want To Address: Participation

There were a number of challenges identified during the 'diagnosing' phase. These included: a lack of funding, friction and disagreements among gardeners, a lack of transport, and a shortage of gardening space. However, the challenges that were unanimously agreed to be the biggest were participation, student transience and continuity of the SFGIs themselves.

There were two main participation issues that were identified:

Low participation, which is when there are few people involved or the people involved can only take part for a limited time. While there may be lots of time spent on garden activities on a single day, the initiative may suffer from low participation if this happens irregularly. I think the garden is great at bringing people together, I just don't think there are enough people brought together to start with

Irregular participation is the other issue identified, and is when involvement in the garden is inconsistent. For example, regular participation might be a gardener spending an hour in the garden on a weekly basis over the course of the year, and irregular participation might be a gardener spending two hours at the garden five occasions over the course of the year.

There are also a dual set of tasks involved in SFGIs that gardeners can participate in. One of these tasks is the actual gardening, including activities as diverse as creating sowing calendars, weeding, composting, and harvesting produce. The other task is to maintain the project in an administrative sense, including equally diverse activities like maintaining a society status and structure within the Students' Union, recruiting participants, applying for funding, and liaising with University and Students' Union staff. Both of these sets of tasks are essential for SFGIs to sustain themselves.

2



The dynamics of low/irregular participation in Student-led Food Growing Initiatives (SFGIs)

2. Factors that Contribute to Low/Irregular Participation

There are a variety of factors that contribute to low and/or irregular participation, and the transience of the gardeners in SFGIs can also exacerbate some of these factors. There were two ways in which students were found to be transient: 1) they were absent for periods of time during the year as a result of the academic calendar (1A) (ie. summer holidays, Easter holidays, winter break), and 2) every three years there is an entirely new group of volunteers because students typically take three years to graduate (1B) from their course.

Some of the factors that contribute to low and/or irregular participation were found to be:

The initiative doesn't meet participants' expectations (2A). Some people who get involved with SFGIs have a romanticised notion of what community gardens are or can be, and once they get involved the gardens don't live up to their expectations. This may be in part because of the 'messiness' of the initiatives and the gardeners' expectation that the project should be well-managed, should be producing more food, or better contributing to a pro-environmental agenda.

Gardeners are busy or lack time (2B). When gardeners are busy or lack time there can be low and/or irregular participation. A number of research participants mentioned their reluctance to commit to a formal role in the committee or society they were involved with because of this.

Not enough ownership by gardeners (2C). Some participants mentioned the need for gardeners to have a more personal connection to the garden to make them feel accountable and encourage them to keep coming back. The difficulties developing ownership of the initiative may be compounded by the fact the students typically only stay at university for three years and therefore know their participation is time-bounded.

However, there were also issues of some gardeners claiming more ownership that it was perceived they were due. Common examples were participants who claimed committee roles without doing gardening tasks and participants acting like an authority on the project without actually having invested time or efforts in the groundwork of the project.

Too closed-off to newcomers (2D). If the group is too closed off, newcomers may not feel welcome to participate. One way this can happen is by a group being too close-knit that it becomes alienating for others. An example of how this alienation can occur is having meetings or social events at someone's home. However, another possible reason for the garden being closed off is because the existing gardeners feel that bringing in more people is a threat to their ownership and authority in the garden, and that newcomers may direct the initiative in an unwanted direction.

People don't feel qualified to participate (2E). Some participants felt that food growing was too unfamiliar and far out of students' comfort zones, which is why student gardeners may lack the confidence or not feel gualified to participate. This may also be due to or compounded by the 'messiness' inherent in the projects. and the trial-and-error approach to gardening and project management.

Participants mentioned that these challenges are mitigated when there is someone more experienced involved in the initiative to take leadership (either someone who has experience from other projects, or someone who has already gained experience through the initiative). It can also be mitigated by running hands-on workshops which can decision makina. feel more inviting to people who aren't as confident.

Participants also reported that some gardeners had a lack of initiative or lacked the ability to make decisions independently. This may be caused by a lack of confidence, but also could be because students' inexperience with volunteer-run projects - and they therefore struggle to understand the level of initiative and independent decision-making needed for such projects to run effectively.

Gardeners feeling overwhelmed or overburdened by the work (2F) was very common. Those that were regularly involved in the initiatives tended to get burdened with more responsibilities than they'd like, and also tended to be people with a sense of stick-to-itiveness. Therefore, it was common for these participants to be highly involved for a year and then drop out. In other words, burnout accelerates the already high turnover rate.

4

enough to keep people in, but not so cohesive that it leaves people out.

It needs to be cohesive

What annovs me most about the garden is] a lack of independent

It seems like, in my experience, they get fed up after a year, because it's too much responsibility. And then they never come again, after they're no longer on the committee. Even though they did it all the time before. Sort of, like, overload. It goes both ways. You can get too much, like, involvement, just out of, like necessity, and aet just not involvement because people won't come back.

Because those that are more involved in the initiative know that overburdening newcomers can lead to dropout, they can be reluctant to delegate responsibilities. Inadvertently, this can also reduce the ability of newcomers to develop a sense of ownership over the initiative, another factor which inhibits participation.

Challenges involving staff and other potential long-term participants (2G). Staff were challenging to involve in the SFGIs because it was difficult to accommodate their work schedules (for example, meetings couldn't be arranged after 5pm because it would be dark out). Another challenge identified was that undergraduate students and potential long-term participants, such as postgraduate students, don't usually mix so it can be hard to involve them.

People aren't familiar with the initiative (2H). There is a lack of awareness about the SFGIs, but this is also compounded by the fact that the gardens themselves can be hard to find and there is limited advertising.

In the middle of winter it's hard to appreciate [what the initiative achieves] sometimes. Gardeners find it hard to appreciate the good things about the initiative (21). Because those involved with the initiative are constantly dealing with urgent problems, it can be hard for them to see the positive aspects of the project.

3. Issues Caused by Low/Irregular Participation

The issues found to be caused by low and/or irregular participation were:

Uncertainty about what will happen to the

initiative (3A). Low or irregular participation can lead gardeners to worry about the future of their SFGI. This response is notable in people who have been involved with the initiative for longer, while newcomers tend to not experience these worries or may not even be aware that their fellow gardeners are having them.

Trust issues between short-term and long-term stakeholders (3B). Another issue caused by low or irregular participation is the

development of trust issues between long-term (ie. university staff) and short-term stakeholders (ie. undergraduate students). Because students are involved in the initiative for a limited period of time, it can be difficult for university staff and the initiative's other long term stakeholders (ie. postgraduate students, local residents) to begin to trust them. If the long-term stakeholders feel past students haven't been trustworthy, they may be reluctant to trust new students. This problem is reinforced by stereotypes about students, as one gardener pointed out, saying that university staff think students are lazy.

The initiative becoming less social (3C). Natural outcomes of having lower and more irregular participation are fewer opportunities for the initiative to facilitate socialization – a widely reported benefit of being part of the initiatives.

The garden being poorly maintained (3D). When there are participation issues, jobs in the garden don't get done. This results in the garden

can mean they don't get as much out of the experience of growing something.

...every session I'm wondering how many people are going to turn up. Did I make a good enough effort remind people that there is a session on today? Is the weather going to stop people from turning up? And then, also, the more general thing that has to do with that is who's going to run it over the summer when the students that are going home are gone? Who's going to run it when the most amount of people... and also who's going to run it next year? Is it going to dissolve?

At the fair I had a lot of people

coming to me, being like, 'Oh

they're really interested then

true that it's hard to find it

because it's so hidden.

yeah, I found your garden' and if

there's this link and they can find it. But if they're not, and they

don't know anything about it, it's

...just going up the chain of command is a pretty difficult skill, and it's quite intimidating to approach people who are in supposed positions of power [ie. university staff]. And with, like, new students coming in who have no idea who these people are, like, that in itself is hours and hours of work of trying to figure out of who do I go to talk to? How do I make a proposal? It's basically the issue of consistency again. There's no consistency, so it'll always be difficult when there's no consistency. Because when, I quess, the previous year's group of students have made, like, a good relationship with the people in the chain of command, then it makes it that much more difficult to re-establish that new relationship with a new bunch of people.

5

becoming overgrown and also being harder to maintain because the gardeners need to play catch up with the duties. It also means that there is wasted potential, which can be disappointing for the people involved and

People missing out (3E). When participation is low or irregular gardeners don't get to experience the benefits of the initiative to their full. For example, they may miss out on the enjoyment of seeing the garden at its peak during the summer. One participant also spoke about how the short lifecycle of their participation in the project influenced their choice to not grow crops that take several years to reach maturity because by the time the crops were mature they would be gone. In this way, the short term nature of gardeners involvement with the project may cause future 'generations' of gardeners to miss out on some benefits.

Existential crisis (3F). The SFGIs struggle so much with

iust trying to stay afloat (ie. maintaining the garden.

I guess, sometimes I'm here in July and August and I'm just like, this is just so beautiful. Heart-achingly beautiful. [...] And I think, everyone's missing out on this! Everyone's missing out on... I can take photos and put photos up and stuff, but people are missing out on it. But they don't know. They don't actually know. Like, their frame of reference in smaller so they don't know what they are missing. Literally. And they come back and they see it and it's vaguely... it's alright. So I think it more affects me, you know. It's disappointing that people can't enjoy it more.

How do you define this project? You know, like, where is it going? What's the goal and stuff? I don't know.

recruiting and retaining participants) that it actually becomes their main focus. However, if the main aim of the initiative is to make sure it doesn't fold (ie. the initiative is successful if it continues to exist), it leads to questions about the value and the meaning of the SFGIs.

Lack of knowledge transfer (3G). Gardening is a knowledge intensive exercise, both in terms of transferable gardening skills and knowledge and in terms of site-specific knowledge (ie. the history of the crop rotation, knowing which areas flood in the spring). Low and irregular participation means that key pieces of knowledge can be lost over the 'generations' of gardeners. General knowledge can be recouped by reading gardening books or with skilled newcomers joining, however some site-specific knowledge cannot be recouped and may result in problems down the line (for example, growing potatoes in the same spot over several seasons means they are more likely to get blight).

Negative emotional responses that may lead to further dropout (3H). It is apparent that one of the main outcomes of low or irregular participation are negative emotional responses. These took a variety of forms and included feeling worried, tired, sad, lonely, frustrated, disappointed, demotivated, and annoyed. Further dropout and more participation issues can be caused by the development of these negative emotional responses by participants. I get quite annoyed with the [others]. I don't say it to them, but inside I feel annoyed because we're supposed to be in it together, working as a team, and then they're always making excuses every [week] when they can't come. And I feel like I'm the only one who does all the work for the whole [initiative].

I think it leads to a spiral of reducing participation each week, because if someone comes and it's only them and another person, then they're not going to come back again. Because it's like, they want to be part of a community garden and if it's not a community, then it's just gardening. [...] So I don't know why they'd go to a random place to just do gardening. They want to be part of something that will end up being part of the project.

4. Strategies to Address the Participation Problem

The strategies identified to address the participation problem were:

Maintaining a well-managed garden retaining an appearance of permanence. This can be done by growing early and late crops so gardeners that are away for the summer can eat the produce they grow, and acquiring more growing space to extend the scale of the project and the number of participants that can be involved.

Having regular meetings, events, and workshops (particularly catering to those newcomers who feel underqualified); meeting in a central location to lead people to the garden.

Recruiting new participants. People with gardening/project management experience can help the garden function more smoothly, and long-term participants (such as University or Students' Union staff, postgraduates, and local residents) help to maintain continuity over time. Embedding the initiative into a staff member's formal role was also suggested.

Developing a cohesive 'core' group with a sense of comradery without being becoming too closed-off through being personable and managing tensions and frictions within the group.

Fostering a sense of ownership of the garden in newcomers. This could be done by delegating responsibilities, assigning tasks, encouraging newcomers to take on a committee role, or even through a 'plant parenting scheme' where gardeners have a plant that belongs to them. An important part of this is making sure

6

newcomers know what tasks need to be done in the garden – both the immediate tasks, but also ones that will crop up in the future.

Having a vision and building in a legacy into the project can help to give the project a better sense of continuity and a feeling that the gardeners are participating in something 'bigger' (giving relief to the 'existential crisis'). A vision can be set by outlining goals of the project, and continuity can be maintained by 'handing-over' information from the present committee to the incoming committee verbally or through a handbook. However, such handovers may benefit from a sense of ritual or use of creative methods to make them less instrumental and more fun, such as through developing a handover tradition (ie. past committee take the future committee out for dinner) or creating videos annually about what has been done in the

Because sustainability is about passing to future generations. As students, we are, in generations, coming and leaving and coming and leaving. So we have to have an action plan for the future to look at.

garden each year. Monitoring progress (ie. measuring how much food is produced or amount of volunteer hours) could also be useful so that gardeners can be motivated by seeing their efforts contributing to a larger cause.

Enhancing communication can both raise awareness and increase the visibility of the initiative, and help people to better understand the benefits of being involved. Communications include:

- Regular emails and updates on social media
- Using advertising platforms available to student societies (like TV screens around campus)
- Handing out promotional material, such as produce or tote bags
- Creating or improving signage, including creating a mural in a prominent location, creating a new
 interpretive sign, adding the garden to existing interpretive signage, and getting footprint stickers to
 lead from a central location to the garden site.

Communications can also help dispel some of the misconceptions and romanticized ideas about what being part of a SFGI is like. For example, choosing to focus on the *'radical potential'* of the space, the social benefits, and the feel-good benefits (like feeling more relaxed and developing a sense of pride in the project), rather than reducing food miles or self-sufficiency.

Next Steps & Suggestions

If you are involved in one of the case study gardens or a part of the NUS sustainability team, you are invited to offer your feedback on these initial interpretations of the findings. If you have not participated yet and want to offer feedback, please contact me so that I can give you the information you need to know before taking part in the project. If you have already signed a consent form, you are welcome to send your feedback in immediately by emailing me at r.laycock@keele.ac.uk. Feedback will be welcomed until February 14th.

I will be running further workshops using the findings from this report and the feedback I receive. There will be one workshop at each case study garden. The focus of these workshops will be to come up with an action plan to meaningfully address the challenges outlined in this report. These actions will be delivered over the course of the following 9-10 months.

My suggestion that I would like to pose for your consideration is that it might make sense to **address both the transience issue and the participation issue** with the action(s) we choose to take. The reason for this is because it is easy for actions that address issues associated with transience (such as recruiting long-term participants) to be placed on the backburner when there are more urgent concerns nudging us at the present (such as recruiting participants in general), particularly since we might not reap the benefits while we are involved in the garden. I also welcome your feedback on this suggestion.

7

Appendix F: Action plans

XXXXX's Actions to Increase Participation

This is a list of actions to be taken by XXXXX to increase participation in their project. These actions form a part of a PhD research project about participation and transience in university food growing initiatives. The research has four main stages: **constructing** the problem, **planning action(s)** to address the problem, **taking action**, and **evaluating** the actions taken. <u>This link</u> will take you to the findings from the constructing stage. This google doc summarises the 'Planning Action' phase. First you will find a flow diagram mapping out the dynamics affecting participation in the project (which, based on feedback from gardeners involved in the research, has been adapted from the version in the 'Constructing' stage's findings report). This is followed by a table which lists the proposed actions. Some actions have a person responsible for seeing them through, however others do not. If you would like to sign yourself up to undertake any of the actions, just add your name under the 'Person Responsible' column, and set yourself a deadline to complete the said action. You don't need to have participated previously in the research to take part in these actions.

I (Bekki Laycock, PhD researcher) will be in touch on a regular basis to find out how the actions are progressing. If you have any questions or comments, you can also feel free to get in touch with me via email at r.laycock@keele.ac.uk.

Please feel free to add or adapt comments, details of the action, and the issue addressed by the action.

Figure 1. The dynamics of low/irregular participation in Student-led Food Growing Initiatives (SFGIs).



Table of Actions to Increase Participation

Action	Details of Action	Issue addressed by action	Date for action to be completed by	Person responsible
Create a recipe collection		•		
Introduce benefits in an attractive way (food, gardening)		 People don't care about or don't understand the initiative (2J) 		
Updated summary of things done in the allotment and brief description of how it works		 People do not feel qualified to participate (2E) 		
Hand out food made out of produce in public areas	Hand out food with a leaftlet including information about the food and how to get involved with the allotment	 People aren't familiar with the initiative (2H) 		
Put fliers in different accommodations to spread awareness about the allotment and its events		 People aren't familiar with the initiative (2H) People don't care about or don't understand the initiative (2J) 		
Create a growing plan	Should have two parts: one saying what needs to be done, and the other reporting what has been done after each workshop	 Gardeners feel overburdened by the work (2F) Not enough ownership by gardeners (2C) People do not feel qualified to participate (2E) 	Check in on a monthly basis	XXXXX1 and XXXXX2

Make tote bags		 People aren't familiar with the initiative (2H) 		
Contact other societies to get involved	For example, the writing society to do an allotment themed session, eco therapy	 People aren't familiar with the initiative (2H) People don't care about or don't understand the initiative (2J) 		
Create packets/bouquets of dried herbs to give away		• People aren't familiar with the initiative (2H)		
Have a bonfire night in Week 10		 People aren't familiar with the initiative (2H) Gardeners find it hard to appreciate the good things about the initiative (2I) Too closed off to newcomers (2D) 		XXXXX1
Organise a sowing workshop		 People do not feel qualified to participate (2E) 	1st week of March	ХХХХХЗ
Hosting interdisciplinary projects at the allotment	Such as an art installation or poetry reading	 People aren't familiar with the initiative (2H) Too closed off to newcomers (2D) People don't care about or don't understand the initiative (2J) 		
Add to the blog	Put photos, poems, the growing plan/manual on the blog	 People aren't familiar with the initiative (2H) People do not feel qualified to participate (2E) People don't care about or don't understand the initiative (2J) 		

Bring a new person to the garden		 People aren't familiar with the initiative (2H) People do not feel qualified to participate (2E) People don't care about or don't understand the initiative (2J) 	By next term	Everyone
Post items of interest on the facebook page		 Gardeners feel overburdened by the work (2F) Gardeners find it hard to appreciate the good things about the initiative People aren't familiar with the initiative (2H) People don't care about or don't understand the initiative (2J) 	By next term	Everyone
Embed the garden in internal communications		 People aren't familiar with the initiative (2H) People don't care about or don't understand the initiative (2J) 	By April	ХХХХХЗ
Embed the garden in campus map		 People aren't familiar with the initiative (2H) 	By April	ХХХХХЗ
Embed in the Food GRP		 People aren't familiar with the initiative (2H) 	By April	ХХХХХЗ
Update online pages/resources	Especially 'Why should I care?'	 People aren't familiar with the initiative (2H) People don't care about or understand the initiative (2J) 		

Challenge VC		•	
'Low-level' advertising	Consistent, free, non-invasive advertising, such as posts on Facebook	 People aren't familiar with the initiative (2H) People don't care about or understand the initiative (2J) 	

Actions to Increase Participation in YYYYY's Initiative

This is a list of actions to be taken by YYYYY to increase participation in their project. These actions form a part of a PhD research project about participation and transience in university food growing initiatives. The research has four main stages: **constructing** the problem, **planning action(s)** to address the problem, **taking action**, and **evaluating** the actions taken. <u>This link</u> will take you to the findings from the constructing stage. This google doc summarises the 'Planning Action' phase. First you will find a flow diagram mapping out the dynamics affecting participation in the project (which, based on feedback from gardeners involved in the research, has been adapted from the version in the 'Constructing' stage's findings report). This is followed by a table which lists the proposed actions. Some actions have a person responsible for seeing them through, however others do not. If you would like to sign yourself up to undertake any of the actions, just add your name under the 'Person Responsible' column, and set yourself a deadline to complete the said action. You don't need to have participated previously in the research to take part in these actions.

I (Bekki Laycock, PhD researcher) will be in touch on a regular basis to find out how the actions are progressing. If you have any questions or comments, you can also feel free to get in touch with me via email at r.laycock@keele.ac.uk.

Please feel free to add or adapt comments, details of the action, and the issue addressed by the action.

Figure 1. The dynamics of low/irregular participation in Student-led Food Growing Initiatives (SFGIs).



Table of Actions to Increase Participation

Action	Details of Action	Issue addressed by action	Date for action to be completed by	Person responsible
Create a Twitter account		 People aren't familiar with the initiative (2H) People don't care about or don't understand the initiative (2J) 		ΥΥΥΥΥ1
Add Student Eats gardening sessions to Researcher's calendar and other university-wide calendars		 People aren't familiar with the initiative (2H) 		ΥΥΥΥΥ1
Create a blog for Student Eats		 People aren't familiar with the initiative (2H) People do not feel qualified to participate (2E) People don't care about or don't understand the initiative (2J) 		ΥΥΥΥΥ2, ΥΥΥΥΥ1
Create plant protection		 Poorly maintained garden (3D) 		ΥΥΥΥΥ1
Set up elections for a new committee		 Not enough ownership by gardeners (2C) 	Friday, February 26th	ΥΥΥΥΥ3

Create leaflets for vegbox		 People are the initiative People do r qualified to (2E) 	not feel	Thursday, February 25th	ҮҮҮҮҮ4, ҮҮҮҮҮ5
Create member bios for the website		 the initiative Too closed newcomers 	off to ownership by		ΥΥΥΥΥ6
Create Facebook events for gardening sessions		 People aren the initiative Too closed newcomers People do n qualified to p (2E) 	off to	Every week	ΥΥΥΥΥ
Organise weekly socials		the initiativeGardeners f	a't familiar with (2H) ind it hard to he good things	Every week	ΥΥΥΥΥ7
Create better signage for the garden		People aren the initiative	i't familiar with (2H)		ΥΥΥΥΥ6
Ask Adam to create a film about Student Eats	The film could be about communicating the larger vision of the garden	the initiative	t care about or stand the	1st week of March	ΥΥΥΥΥ6

Create a blog post about Student Eats for Keele's main blog		 People aren't familiar with the initiative (2H) People don't care about or don't understand the initiative (2J) 	
Have an impressive crop ready in september to inspire people	This means planting in June/July	 Poorly maintained garden Jun (3D) 	ne/July
Keele University Film Society host film screenings about growing food		 People aren't familiar with the initiative (2H) People do not feel qualified to participate (2E) People don't care about or don't understand the initiative (2J) 	
Arrange a visit to the Staffordshire University allotment at the beginning of the semester to inspire people		 People do not feel qualified to participate (2E) People don't care about or don't understand the initiative (2J) Gardeners find it hard to appreciate the good things about the initiative (2I) 	
Everyone invites a society to come to the garden on a different week	Creative writing society, photography society, media, psychology, etc	 People aren't familiar with the initiative (2H) People don't care about or don't understand the initiative (2J) 	
Create a better media presence		 People aren't familiar with the initiative (2H) People don't care about or 	YYYYY8

		don't understand the initiative (2J)	
Organise a workshop to bring students from the Staffs Uni growing project to Keele		•	
Create an annotated map of the Walled Garden	Detailing what is where, including a crop rotation	 People aren't familiar with the initiative (2H) 	YYYY8
Aesthetically improve the social space in the Walled Garden	Freecyle more 'furniture'	 Poorly maintained garden (3D) 	YYYYY5 (YYYYY8)
Network with experienced growers	For example, the keen locals at Staffs	 People do not feel qualified to participate (2E) 	
Have a set schedule of tasks through the winter to maintain participation	ie. tool maintenance, bird table maintenance, grafting, pot plants, trips, visitors	 People do not feel qualified to participate (2E) 	
Making sure newcomers are aware they can borrow appropriate clothing/footwear	Posting on Facebook/blog	 People do not feel qualified to participate (2E) 	
Everyone takes turns to post stuff on Facebook		 People aren't familiar with the initiative (2H) Not enough ownership by gardeners (2C) 	
Create an interpretive sign above the garden		 People aren't familiar with the initiative (2H) 	YYYY8
Improve signage to get inside the Walled Garden		 People aren't familiar with the initiative (2H) 	YYYY8

Set up a 'plant parenting scheme'		•	Not enough ownership by gardeners (2C)	
Link up to mental health initiatives		•	Gardeners find it hard to appreciate the good things about the initiative (2I) Gardeners feel overburdened by work (2F)	ҮҮҮҮҮ8
Offer cookery classes with people buying the vegbox		•	People aren't familiar with the initiative (2H)	YYYYY8
Teach people how to replant veggies from the vegbox	Using leaflets for workshops at the garden	•	People aren't familiar with the initiative (2H) People do not feel qualified to participate (2E)	
Give away potted plants		•	People aren't familiar with the initiative (2H)	

Appendix G: Photovoice information sheet



Study Title: Cultivating sustainable participation in Student-led Food Growing Initiatives with transient participants using action research **Rebecca Laycock, r.laycock@keele.ac.uk**

What does Photovoice involve?

In this Photovoice project, you will be asked to take photos considering the following questions:

- What does the gardening session look like in the [garden/allotment/initiative] this week?
- Is the participation (the quality/the amount) better than usual this week? Is it worse?
- What helps improve participation in the [garden/allotment/initiative]?
- What hampers participation in the [garden/allotment/initiative]?
- What do I want to tell other people about participation in the [garden/allotment/initiative]?

Between May 31st and November 30th 2016, you can send photos to r.laycock@keele.ac.uk. Please accompany it with the date it was taken and a short caption to make it clear what the photo is of and how it addresses one or more of the question(s) above. You can send in as many or a few photographs as you'd like – however you may be asked to choose a maximum of five to display.

The photos will be displayed at [Keele University's Sustainability Hub/Warwick Students' Union/Sheffield Students' Union/ other appropriate location] for a photo exhibition and celebration. You will be invited to come early to take part in a discussion for research purposes, and then after the discussion, other guests can arrive and hot drinks and nibbles will be served. Further information will be provided closer to the date.

To participate, you must have agreed to take part in the research project called *Cultivating sustainable participation in Student-led Food Growing Initiatives with transient participants using action research* undertaken by Rebecca Laycock from Keele University. If you have not yet been given an information sheet or signed a consent form but would like to take part, get in touch by emailing r.laycock@keele.ac.uk.

Some rules and tips for taking photos!

No experience needed! Owning a camera or having experience taking photographs is not necessary – all you need is a camera or a phone with a camera.

Photo quality is not necessarily important. Photovoice is not about the quality of your photographs. It is about taking pictures that mean something to you as a participant in a student-led food growing initiative.

Ask permission to take someone's photo. Always ask permission before you take someone's picture! If they say no, explain briefly what you are doing and why you want to take their picture. Your explanation can simply be: "I'm working on a photography project about participation in student-led food growing projects, would you mind being in one of the photos?" If they still say no, take a picture of something or someone else instead.

Permission isn't necessary when... In a public place like a park, you can take someone's photo without permission if they are far away and can't be recognized in the picture. But keep in mind that universities are small communities and people could be recognised more easily than in other locations (or example, someone could be recognised by their clothing or by the location they are standing in the [garden/allotment], even if their face is not visible).

Be respectful. If certain people don't their photo taken, respect their feelings.

Write captions from your heart. Write your captions as if you were talking to someone else about your photos. Speak from your heart. If you'd prefer, you can dictate your captions to someone else to write down for you.

Send your photos in as soon as possible. The sooner you send in your photos, the less likely you'll forget about them! The same goes for the captions – write them as soon as you can so you don't forget why you took the photo.

Some things to think about when creating good photographs:

- Try different angles
- Try different points of view
- Keep the sun to your back, or to the side
- Is your subject in the center of the photo?
- Does your subject fill the photo?
- Keep your finger away from the lens
- Don't cover the flash (if you are using it!)
- Stand about three to eight feet away from your subject

• To prevent blurry pictures, hold your elbows close to your sides, and hold your breath when you press the shutter (button).

Appendix H. Fishbowl discussion guidelines

The following guidelines were explained to participants verbally, with the use of a powerpoint presentation.

Introduction

Consent forms

If you want to participate in the discussion...

• You must read & fill out the consent form

If you only want to listen...

• You don't need to fill in the consent form, but you do need to wear a sticker

Please remember: if you don't fill in the consent form, you can't speak during the discussion!

How the Fishbowl works

The fishbowl isn't meant to offer concrete solutions or closure on a topic – Its purpose is an **open-ended exploration** of an issue.

Inner fishbowl rules

- There should always be one seat empty
- The focus of the discussion should be on the question(s) presented
- Strive to be an *active* listener
- You can leave the inner circle at any time by moving to the outer circle or 'tapping out'
- Encourage others to share their opinions

Outer fishbowl rules

- No one should speak when seated in the outer circle
- Strive to be an *active* listener
- You can join the inner circle at any time by occupying the spare seat or 'tapping' someone out

The fishbowl will end with a short debrief in one large circle.

Discussion

How can we maintain sustainable participation given...

- The typical undergraduate degree is only three years, after which most students leave
- Most student leave for the summer (and spring/winter) holidays
- The ebbs and flows of academic workloads, course work and exams

Remember, it's not just about increasing participation... it's about maintaining participation sustainably over the long-term.

- The typical undergraduate degree is only three years, after which most students leave
- Most student leave for the summer (and spring/winter) holidays
- The ebbs and flows of academic workloads, course work and exams

Debrief

As individuals and as a group, what are our biases limiting us from understanding the dynamics of participation?

What did you learn from this discussion (if anything)?

Will you be trying out anything new in your food-growing project based on this discussion?

Appendix I: Visualisations of the causal loop mapping using

TheBrain software

TheBrain software was used to map the causal relationships between the different factors affecting and affected by problematic forms of participation. Note that these figures are only for illustrating how the software was used to map relationships. The causal loop diagram was further iterated and simplified using PowerPoint software to create the simplified versions in Chapter 6.

Busy, lacking time, competing commitments Cliqueyness ·Control by and dependence on key people Hippies Old fashioned People don't know and don't care People don't understand Cultural barrier Long term participants Difficulty recruiting people Interested in the idea of it but not the actual practice Don't like gardening Benefits not clear until afterwards **Delayed Rewards** Don't see what they will get out of it Expectations and misconceptions Having to pay Inaccessible space Infrastuctural problems Lack of cohesion ·Perceptions of being undergualified Lack of confidence Lack of initiative Lack of motivation Issues with structure Ownership issues Lack of opportunity to self-determine Lack of responsibility Not appreciating the good things Overwhelmed, overburdened, burned out People forgetting to come People haven't heard of the project Poor weather Transience Unpredictability Unwillingness to commit Participation problems

Figure 36a. An early iteration of the mapped causal relationships between the different factors affecting and affected by problematic forms of participation in student-led food gardens.



Figure 36b. An early iteration of the mapped causal relationships between the different factors affecting and affected by problematic forms of participation in student-led food gardens.